ABSTRACT

"Learning on the Run" refers to the educational experiences of the primary school children travelling along the agricultural show 'circuits' in coastal and western Queensland. This thesis examines those educational experiences by drawing on the voices of the show children, their parents, their home tutors and their teachers from the Brisbane School of Distance Education, which from 1989 to 1999 implemented a specialised program of Traveller education for these children (in 2000 a separate school was established for them).

The thesis focusses on the interplay among marginalisation, resistance and transformation in the spaces of the show people's itinerancy. It deploys Michel de Certeau's (1984, 1986) concept of 'tactics of consumption' and Mikhail Bakhtin's (1986a) notions of 'outsidedness' and 'creative understanding' to interrogate the show people's engagement with their absence of place, the construction of their otherness and forms of seemingly unproblematic knowledge about their schooling. Data gathering techniques included semi-structured interviews with forty-two people between 1992 and 2000 in seven sites in Queensland – Mackay, Bundaberg (over two years), Emerald, Brisbane, Rockhampton and Yeppoon – and document collection.

The thesis's major finding is that the show people's resistance and transformation of their marginalising experiences have enabled them to initiate and implement a significant counternarrative to the traditional narrative (and associated stereotypes) attending their itinerancy. This counternarrative has underpinned a fundamental change in their schooling provision, from a structure that worked to marginalise and disempower them to a specialised form of Traveller education. This change contributes crucially to understanding and theorising the spaces of itinerancy, and highlights the broader significance of the Queensland show people's "learning on the run".



Learning on the Run: Traveller Education for Itinerant Show Children in Coastal and Western Queensland

Patrick Alan Danaher

Thesis submitted in fulfilment of
the requirements for the degree of
Doctor of Philosophy,
School of Education and Innovation,
Faculty of Education and Creative Arts,
Central Queensland University,
March 2001

DEDICATION

This thesis is dedicated with love and affection to the memory of

MAURICE DANAHER

(28 September 1930 - 5 February 1988)

"Yet in my lineaments they trace

Some features of my father's face."

Lord Byron, "Parisina"

"—Adieu, dit le renard. Voicimon secret. Il est très simple: on ne voit bien qu'avec le cœur. L'essential est invisible pour les yeux. ... C'est le temps que tu as perdu pour ta rose quifait ta rose si importante. ... Les hommes ont oublié cette vérité Mais tu ne dois ges l'oublier. Tu deviens responsable pour toujours de ce que tu as apprivoisd. Tu es responsable de ta rose. ... Antoine de Saint-Exupéry, Le Petit Prince, pp. 72-74

"When he [José Arcadio Buendia] became an expert in the use and manipulation of his instruments, he conceived a notion of space that allowed him to navigate across unknown seas, to visit uninhabited territories, and to establish relations with splendid beings without having to leave his study."

Gabriel Garcia Márques, *One Hundred Years of Solitude*, **p. 4**

CONTENTS

				P	PAGE
Abstract					1
Dedication					111
Contents					iv
List of Figure	es				ix
Acknowledge	ements				X
Declaration					xvi
Selected Publ	ications	and Presentations Related	to This Work		xvii
Chapter One:	Tl	ne Problem and The Questic	ons		1
1.1	The prol	olem		2	
1.2	The sign	nificance of the problem		4	
1.3	The rese	arch questions		8	
1.4	The out	line of the thesis		11	
1.5	A person	nal note		13	
Chapter Two:	T1	ne Review of the Literature			15
2.1	Overvie	w of the chapter		16	
2.2	Australi	an shows		17	
2.3	Travelle	er education		35	
	2.3.1	The dearth of research	37		
	2.3.2	The prevalence of negative			
		stereotyping	40		
	2.3.3	Some exceptional studies	55		

İV

2.4	Review of	f the chapter			72	
Chapter Thre	e:	The Conceptual Framework				76
3.1	Overview	of the chapter			77	
3.2	De Certea	u's concept of 'tactics of consump	otion'		80	
	3.2.1	Thepractice & everyday life (19	84)	81		
		"Consumerproduction"	85			
		"The tactics &practice"	91			
	3.2.2	Heterologies: Discourse on the o	ther			
		(1986)		101		
	3.2.3	Applications and critiques of de				
		Certeau's work		108		
		'Strategies' and 'tactics'	114			
		'Place' and 'space'	125			
		Marginalisation and resistance	132			
		Change and transformation	143			
3.3	Bakhtin's	notions of 'outsidedness' and				
	'creative	inderstanding'			150	
	3.3.2	Bakhtin and de Certeau		152		
	3.3.2	'Outsidedness' and 'creative				
		understanding'		160		
3.4	Synthesis	of the conceptual framework			168	
3.5	Review of	f the chapter			172	
Chapter Four	: The	e Research Design				176
4.1	Overview	of the chapter			177	
4.2	Marginali	sation, resistance and transformati	on and	l researchin	g	
	Traveller	education			179	

4.0	·			100	
4.3		consumption' and researching Traveller		182	
4.4	'Outsidedn	Outsidedness' and 'creative understanding' and researching			
	Traveller e	ducation		190	
4.5	The ethical	l and political dimensions of the study		197	
4.6	The research	ch design of the study		207	
4.7	The data g	athering techniques of the study		209	
	4.7.1	Semi-structured interviews	209		
	4.7.2	Document collection	212		
4.8	The delim	itations and limitations of the study		213	
4.9	Review of	the chapter		215	
Chapter Five:	Res	earch Question One: The Show P	eople's		
	Exp	eriences of Marginalisation			218
5.1	Overview	of the chapter		219	
5.2	Strategies	of marginalisation		222	
5.3	Absence of	f place		228	
5.4	Constructi	ions of otherness		238	
5.5	Forms of u	unproblematic knowledge		251	
	5.5.1	Sending show children to local school	S		
		on the show circuits	257		
	5.5.2	Sending show children to boarding			
		schools	264		
	5.5.3	Teaching show children correspondence	ee		
		lessons on the show circuits	265		
	5.5.4	Show people coming off the circuits			
		and sending their children to local			
		schools	269		

	5.5.5	Sending show children to live with			
		relatives and attend local schools	271		
	5.5.6	Not sending show children to school			
		at all	273		
5.6	Review	of the chapter		277	
Chapter Six:	R	Research Question Two: The Show	People's		
	P	ractices of Resistance			280
6.1	Overvi	ew of the chapter		281	
6.2	Tactics	of resistance		283	
6.3	Resistii	ng absence of place: "Where is home to yo	ou?"	289	
6.4	Resisting constructions of otherness: 'Showies', 'locals'				
	and 'm	ugs'		297	
	6.4.1	Giving 'showies' a positive valence	301		
	6.4.2	Disrupting the 'showie'-'non-showie	,		
		dichotomy	307		
	6.4.3	Giving 'mugs' a negative valence	311		
6.5	Resisti	ng forms of unproblematic knowledge: Th	he Brisbane		
	School	of Distance Education program		315	
	6.5.1	Establishing the program	3 16		
	6.5.2	Consuming the program	325		
6.6	Review	of the chapter		336	
Chapter Seven:		Research Question Three: The Show	w People's	}	
	T	ransformation of Their Marginalis	ing		
	E	Experiences and Resistant Practices			338
7.1	Overvi	ew of the chapter		339	

7.2	Outsidedness, creative understanding and transformation		
7.3	Transforming absence of place: A place of their own		
7.4	Transforming constructions of otherness: Valuing difference		
7.5	7.5 Transforming forms of unproblematic knowledge: A separate		
	show school	362	
7.6	Review of the chapter	370	
Chapter Eigh	t Some Answers to the Questions		373
8.1	The problem addressed and the research questions answered	374	
8.2	The personal note revisited	378	
8.3	Possible directions for future research	381	
References			386
Appendix A	Semi-StructuredInterview Schedule		407

LIST OF FIGURES

		PACE
3.1	The conceptual framework of the study	169
4.1	Phases in the study's research design	208
4.2	Interviews conducted for the study	211

ACKNOWLEDGEMENTS

Many individuals and groups have provided invaluable assistance in the writing of this thesis. The relationship between research student and supervisor is a curiously intimate one, requiring the supervisor to challenge and encourage in equal measure. Dr Leonie Rowan fulfilled the manifold responsibilities of associate, and more recently principal, supervisor with commendable grace, patience and good humour. She applied her enviable knowledge of poststructuralist theory, particularly the work of Michel de Certeau and Mikhail Bakhtin, to assist my elaboration of the study's conceptual framework, and she tolerated with a smile my often naive questions. She also insisted on an appropriate integration of theory, method and data analysis, and she facilitated my efforts to attain that goal. In short, the thesis would not have been completed without Dr Rowan's support.

Professor Leo Bartlett, who was my principal supervisor for most of the study, took time from his extremely busy schedule to read my tentative drafts, to focus my thinking and to suggest potentially fruitful ways of proceeding. His questions and criticisms have greatly enhanced my understanding of educational research. This is only a small part of his enormous contribution to establishing and nourishing a research culture in the then Faculty of Education at Central Queensland University, in his former role as Foundation Professor and Dean in that faculty. Professor Bartlett's support was also instrumental in my being awarded a bursary of six months by Central Queensland University.

Four well-disposed spirits – Ms Phyllida Coombes, Dr Geoffrey Danaher, Ms Lucy Jarzabkowski and Dr Máirín Kenny – undertook the tedious but greatly appreciated task of painstakingly reading late drafts of the text. Their efforts contributed immeasurably to the improvement of that text.

Staff members of the Central Queensland University Library provided prompt and courteous service. Ms Beverley Corness, former liaison librarian with the then Faculty of Education, and Ms Mary Bevis, current holder of that position, were interested and encouraging. Ms Cathy Dennis, inter-library loans officer, coped well with my requests for abstruse items.

I am grateful to the organisers of and the participants in the annual symposia of the Central Queensland University Postgraduate Student Association. I acknowledge also the searching questions **posed** by audience members at other conferences where I have discussed the study. These events gave me an opportunity to sharpen my thinking and to respond to the questions of interested outsiders.

I was greatly assisted in writing this thesis by virtue of its forming part of a larger collaborative research project with my fellow members of the former Professional Growth Research and Teaching Group in the then Faculty of Education at Central Queensland University. My thanks go to Dr Geoffrey Danaher, Dr Paul Duncum, Mr Peter Hallinan, Mr Ian Kindt, Ms Patricia Moran, Dr Beverley Moriarty, Dr Ken Purnell, Mr Colin Rose, Mr Robert Thompson, Ms Christine Woodrow and Mr Doug Wyer. Of these, Messrs. Hallinan, Kindt, Rose, Thompson and Wyer and Drs Danaher and Moriarty constitute, with myself, the past and present core of the Australian Traveller education research project at Central Queensland University.

Taped interviews recorded **as** part of the aforementioned broader project and on which I drew for this study were expertly transcribed by Dr Geoffrey Danaher, Ms Bonita Frank and Ms Pam Gale.

Several individuals discussed the study or the broader project to which it relates with me and provided information and sometimes advice, in person and more often in correspondence. In this regard, I am indebted to Associate Professor Chris Bigum, Dr Ron Blaber, Mr John Bowman, Dr Ian Buchanan, Ms Helen Currie, Professor Roger Dale, Dr Caroline Dyer, Professor Terry Evans, Mr Keith Harry, Ms Pat Holmes, Mr Tom Jenkin, Dr Elizabeth Jordan, Professor Stephen Kemmis, Dr Máirín Kenny, Ms Cathy Kiddle, Mr Ludo Knaepkens, Mr Ken Lee, Mr Robert T. Pullin, Mr Nello Raciti, Ms Ursula Scholten, Ms Tracy Smith, Mr Mark St Leon, Mr Emile Steffann, Professor Gidado Tahir, Dr Abdurrahman Umar, Mr Ruud van de Rakt, Mr Wim van de Steeg, Associate Professor Bill Warren and Mr Matthew Weinstein. Ms Maryanne Lynch provided invaluable editorial advice on a related project that incidentally enhanced the presentation of this study.

Associate Professor Robert Baker provided the impetus for the study, by musing aloud one day, "Iwonder where and when the show children go to school". His receipt (with Mr Peter Hallinan) of a University Research Grant (awarded through the Research Centre for Open and Distance Learning at Central Queensland University) provided the initial funding for the research, which was supplemented by funding by the former Faculty of Education and a second University Research Grant awarded through the Research Centre for Open and Distance Learning. Associate Professor Baker also performed the

role of supervisor of the thesis until his move to become Director of the Centre for Education at Southern Cross University.

I am very grateful to my friends and colleagues in the Faculty of Education and Creative Arts at Central Queensland University. They have ensured that 'going to work' is a fulfilling and enriching experience. I trust that it will not be thought amiss if I name some individuals whose assistance and friendship are particularly important to me:

- past and present members of the Faculty of Education and Creative Arts general office for enormous efficiency and professionalism leavened by good humour, particularly my current administrative assistant, Ms Sandi Weedon
- my three fellow Gooleys former Associate Professor Robert Baker, Ms
 Bobby Harreveld and Ms Pat Moran for great food and even better company
- Dr Trevor Gale, for his intellectual companionship and his commitment to socialjustice in theory and practice
- former Associate Professor Michael Garbutcheon Singh, for introducing me to the thought of Mikhail Bakhtin and amazing me with his industry and productivity
- my four fellow 'lounge bar lizards' Ms Lynda Kennedy, Ms Jackie
 Newdick, the late Ms Angie Padfield and Dr Leonie Rowan for
 participating with me in essential distractions from the daily grind.

I thank all the members of my family – Phyllida and Cedric Coombes, Michael and Haruko Danaher, Geoffrey Danaher, Christine Danaher and Reanna Urquhart, Jemima Coombes and Charlie Danaher – for their manifold interest, encouragement and support. My brother Dr Geoffrey Danaher was particularly helpful in suggesting some theoretical links between Michel de Certeau and Mikhail Bakhtin, and I benefited from reading his doctoral thesis (Danaher, 1995). My Venezuelan host family – Emilio A., Princess and Poockie Anteliz – provided a 'home away from home' that rekindled my energies to complete the project. My greatest regret is that the dedicatee of the thesis, and both my grandmothers, are unable to read the finished product.

My largest debt is to two groups of people. Firstly, the participants on the coastal and western Queensland show circuits – children, parents and home tutors – responded politely and tolerantly to the inquisitive researchers from Central Queensland University, whose annual incursion into one of the local show weeks they took in their stride. Mr Dick Dargin, current executive officer of the Showmen's Guild of Australasia, and Mr Peter A. Johnstone, former holder of that position, have been encouraging of the research project and facilitated my contacts with other show people.

Secondly, I am very grateful for the ongoing support and the many kindnesses of Mr Bob Rasmussen, principal of the Brisbane School of Distance Education, and for the invaluable assistance of the teachers involved in the show children's education program, especially Ms Cathie Fullerton, Ms Cathy O'Connor and Mr Duncan Robertson.

Specifically, I acknowledge with thanks the special assistance of the following individuals who granted me interviews or who otherwise contributed to the study: Mr Mark Adsett, Ms Robyn Allan, Ms Sarah Butler, Ms Jessamy Bell, Ms Sandra Bell, Ms Tammy Bell, Ms Kylie Boon, Mr Johnny Castle, Ms Priscilla Castle, Mr Luke Cheyne, Mr Greg Davenport, Ms Raelene Davis, Ms Nadine Eckert, Mr Josh Evans, Ms Cathie Fullerton, Ms Sarah Gill, Ms Jessica Gilmore, Ms Zoe Gilmore, Mr Christopher Hennessy, Ms Jessica Hennessy, Mr Luke Hennessy, Mr Peter A. Johnstone, Ms Jan Marshall, Mr Jesse McDonald, Ms Claire McPherson, Ms Robyn McPherson, Ms Kathleen Miller, Ms Julie Miller, Mr Liam Miller, Ms Cathy O'Connor, Mr Brod Pavier, Ms Nanette Pavier, Mr Damien Phillips, Ms Tyrie Phillips, Ms Briony Pink, Ms Kylie Pink, Mr Bob Rasmussen, Mr Duncan Robertson, Ms Kym Silver and Ms Jodie Wardle.

Both the groups identified above – show people and educators – were crucial to the success of the research project, and they contributed immeasurably to my growing understanding of "learning on the run".

DECLARATION

I declare that the main text of this thesis is entirely my own work and that such work has not been previously submitted **as** a requirement for the award of a degree at Central Queensland University or any other institution of higher education.

(Patrick Danaher)

30 March 2001

SELECTED PUBLICATIONS AND PRESENTATIONS RELATED TO THIS WORK

Anteliz, E. A. & Danaher, P. A. (2000). *Carritos chocones* and dodgem cars: A preliminary comparison between Venezuelan and Australian fairgrounds. *Journal of Nomadic Studies*, **3**, **12-17**.

Coombes, P. N., Danaher, G. R. & Danaher, P. A. (2000, July). Educational technologies and lifelong learning: The place of the habitus. *ODLAA Occasional Papers*, **9-15.**

Danaher, P. A. (1994, October). Show day all year round: How children on the Queensland show circuit are educated. *Social Alternatives*, *13*(3-4), 27-31.

Danaher, P. A. (1995, September). Show children's perceptions of their schooling experiences. *Unicorn*, 21(3), 43-50.

Danaher, P. A. (1996). Experiences regarding ODL for occupational travellers in Australia. In Report: European seminar open and distance learning for the children of occupational travellers Blankenberge November

15th and 16th 1996 (pp. **34-59**). Brussels, Belgium: European Federation for the Education of the Children of the Occupational Travellers.

Danaher, P. A. (1997, June). Showing the way. *Community Quarterly*, 43, 33-38.

Danaher, P. A. (1997, August 5). Educating Australian show children: A national and international research project. Invited keynote paper presented at the 7th national conference of the Showmen's Guild of Australasia, Mercure Hotel, Brisbane.

Danaher, P. A. (1997, October-December). Researching the education of Australian show children. *The Outdoor Showman*, 406, 37.

Danaher, P. A. (1997, November). The flexible delivery of education: Lessons from Queensland travelling show children. In M. J. Garbutcheon Singh, R. E. Harreveld & N. Hunt (Eds.), *Virtual flexibility: Adult literacy and new technologies in remote communities* (pp. 40-48). Rockhampton, Qld: Central Queensland University.

Danaher, P. A. (1998, January). From structural-functionalism to traveller education: Personal and public meanings for distance education research. DERUN: Distance Education Research Update Newsletter, 3, 2.

Danaher, P. A. (1998, May). Distance education for space travellers: Educational provision for occupational itinerants/Éducation h distance pour les voyageurs de l'escpace: Disposition pédagogique pour les itinkrants professionnels. Communiqué, 13(3-4), 5-7/6-8.

Danaher, P. A. (Ed.)(1998). Beyond the ferris wheel: Educating Queensland show children (Studies in open and distance learning number 1). Rockhampton, Qld: Central Queensland University Press.

Danaher, P. A. (1998). Consumption and dialogue in educating Queensland show children. In W. O'Brien, D. Grasby, M. Brigg & L. Hungerford (Eds.), *Diverse dialogues: Postgraduate contributions from Central Queensland University* (pp. 20-30). Rockhampton, Qld: Central Queensland University Press and Central Queensland University Student Association.

Danaher, P. A. (1998). Past, present and future in theorising Australian Traveller education. In B. Cranston (Ed.), *National Open & Distance Education Student Network Inc (NODESNet) 7thannual conference 6-9 July 1998 Rockhampton campus Central Queensland University: Original and revised papers From the 1998 NODES Net annual conference* (pp. 35-47). Rockhampton, Qld: Central Queensland University and National Open & Distance Education Student Network.

Danaher, P. A. (1999). Learning under ferris wheels and big tops: Comparing the education of Australian show and circus people. *Journal of Nomadic Studies*, 2, 24-3 1.

Danaher, P. A. (2000). What's in a name?: The 'marginalisation' of itinerant people. *Journal of Nomadic Studies*, 3, 67-71.

Danaher, P. A. (Ed.) (2000). *Mapping international diversity in researching Traveller and nomadic education*, special issue of the *International Journal of Educational Research*, 33(3).

Danaher, P. A. (2000). Guest editor's introduction. In P. A. Danaher (Ed.), Mapping international diversity in researching Traveller and nomadic education, special issue of the International Journal of Educational Research, 33(3), 221-230.

Danaher, P. A. (2001, in press). The researcher as occupational Traveller: From strategic essentialism to creative understanding. *Journal of Nomadic Studies*, 4.

Danaher, P. A., Coombes, P. N., Danaher, G. R. & Anteliz, E. A. (2000, December). Life imprisonment or learning for life: Re-opening the social control—social capital debate about lifelong learning. *Re-Open, 1*(1), 1-14 (http://www.edca.cqu.edu.au/lit/reopen/current.html).

Danaher, P. A. & Danaher, G. R. (2000). 'Power/Knowledge' and the educational experiences and expectations of Australian show people. In P. A. Danaher (Ed.), *Mapping international diversity in researching Traveller and nomadic education*, special issue of the *International Journal & Educational Research*, 33(3), 309-3 18.

Danaher, P. A., Hallinan, P. M., Kindt, I. D., Moriarty, B. J., Rose, C. G., Thompson, R. & Wyer, D. W. (1995). *School on a show day* (thirteen minute videotape). Rockhampton, Qld: Educational Media Section, Central Queensland University.

Danaher, P. A., Pullin, R. T. & Wyer, D. W. (1997). Show people's expectations of Traveller education: Issues in international collaboration in researching open and distance learning. In 18th ICDE world conference The new learning environment: A global perspective June 2-6, 1997 Hosted by

The Pennsylvania State University (CD-ROM) (G1DANAH) (11 pp.). State College, PA: Pennsylvania State University.

Danaher, P. A., Pullin, R. T. & Wyer, D. W. (1997). Technology on the move: Educational lessons from British and Australian travelling show children. In J. Osborne, D. Roberts & J. Walker (Ms.), Open, flexible and distance learning: Education and training in the 21st century: Selected papers from the 13th biennial forum of the Open and Distance Learning Association of Australia (In association with the Australian Association of Distance Education Schools) (pp. 94–98). Launceston, Tas: University of Tasmania.

Danaher, P. A., Rose, C. G. & Hallinan, P. M. (1994). Lessons from the carnival. *Epistolodidaktiku: The European Journal & Distance Education*, 1, 91-110.

Danaher, P. A. & Wyer, D. W. (1997). Itinerant education as border pedagogy: The globalisation and localisation of show culture. In L. O. Rowan, V. L. Bartlett & T. D. Evans (Eds.), *Shifting borders: Globalisation, localisation and open and distance education* (pp. 99-111). Geelong, Vic: Deakin University Press.

Danaher, P. A., Wyer, D. W. & Bartlett, V. L. (1998, February). Theorising open learning for researching home school and itinerant settings. *OpenLearning*, 13(1),9-17.

Danaher, P. A., Wyer, D. W., Rowan, L. O. & Hallinan, P. M. (1994). The role of course development and design in an itinerant schooling program: The perceptions of staff members of the School of Distance Education in

Brisbane, Queensland. In *Distance education: Windows on the future -- conference proceedings* (pp. 70-75). Wellington, New Zealand: New Zealand Correspondence School.

Moriarty, B. J., Danaher, P. A. & Hallinan, P. M. (1996). Language on the show circuit: A broadening appreciation of critical literacies. In 1996 ERAIAARE joint conference papers, Singapore (7 pp.). Australian Association for Research in Education, http://www.swin.edu.au/aare/con/96/MORIB96.145.

Moriarty, B. J., Hallinan, P. M., Danaher, G. R. & Danaher, P. A. (2000). All I know is what I learned from my colleagues: Reflections on research from Australian Traveller education. *Queensland Journal & Educational Research*, 16(1),56-75.

Rose, C. G., Moriarty, B. J. & Danaher, P. A. (1995). Margins within margins?: Voices speaking through a study of the provision of an educational program for the children of one Australian show circuit. In L. O. Rowan & J. McNamee (Eds.), *The voices of a margin: Speaking for yourself* (pp. 160-167). Rockhampton, Qld: Central Queensland University Press.

Thompson, R. & Danaher, P. A. (1994, winter). Fairy floss, ferris wheels and going to school: Regional development and itinerant education in Central Queensland. *Central Queensland Journal of Regional Development*, 3(1),37-40.

Wyer, D. W., Danaher, P. A., Kindt, I. D. & Moriarty, B. J. (1997). Interactions with Queensland show children: Enhancing knowledge of educational contexts. *Queensland Journal of Educational Research*, 13(2), 28-40.

CHAPTER ONE

THE PROBLEM AND THE QUESTIONS

"Questionsare invented, like anything else. If you aren't allowed to invent your questions, with elements from all over the place, from never mind where, it people 'pose' them to you, you haven't much to say. The art of constructing a problem is very important; you invent a problem, a problem–position, before finding a solution."

Deleuze, 1987; cited in Buchanan, 1996c, p. 99

"Hecame to the point. **He** remarked that in his job, the problem isn't really finding the answers. It's finding the questions. 'We need the man **who** can find the key questions.'"

Lynn & Jay, 1986, p. 14

1.1 The problem

This thesis is concerned with marginalisation, and with the possibilities of resistance and transformation of that marginalisation, in the lives of the children of itinerant show families whose travels take them through coastal and western Queensland. More specifically, the problem with which the thesis engages is the ways in which educational provision for these children has been complicit with that marginalisation as well as being the site of alternative understandings about how Travellers can and should be educated. (Throughout the thesis, I use the British term 'Traveller education' – including the capital 'T' to reflect respect for the self-asserted ethnicity of many itinerant groups — to denote educational provision for occupational Travellers such as show children, and I use 'itinerancy' to refer to the set of conditions involved in occupational travelling.)

The statement of this problem establishes the parameters for this thesis. The focus is clearly on the show children and their families, as a specialised group whose educational needs depart from the 'norm' of children living in permanent residences and attending their local schools. The show children's home tutors, and the teachers from the Brisbane School of Distance Education, are important, but here their roles are analysed from the standpoint of how they contribute to the show children's educational experiences.

The statement of the problem also signals my interest in the recipients' consumption of an educational policy **as** opposed to the designers' development of such a policy. **As** I explain at length in Chapter Three, the conceptual framework guiding this study has been selected for its utility in explaining

how the show people used their consumption of the education program designed for their children to subvert their marginalised status, and in the process to capitalise on educational providers' outsidedness that promotes greater mutual and creative understanding of the other group's circumstances. This approach reflects my continuing interest in the actions of minority groups as exercises of social agency (Giddens, 1984) in educational contexts.

In view of the problem to be addressed in this thesis, the following concepts have been assigned prominence in the study:

- marginalisation
- resistance
- transformation.

These concepts underpin the research questions of the study and provide the framework for the analysis of data in the substantive part of the thesis. The concepts are linked logically and sequentially. The show people's marginalised status (and its educational consequences) provide both a backdrop and an impetus for their resistance of that status, which in turn creates the possibilities of counternarratives, or 'alternative stories', about the show people's lives and educational opportunities. The intention is to emphasise marginalisation, resistance and transformation as shifting and fluid markers of the dynamic relations between show people and educational providers, not as fixed receptacles for ossified interactions between 'us' and 'them'.

Martinez (1994) captured something of the spirit of this intention when she commented:

My growing conviction is that our work on the context of teaching should lead us to recognise the importance not of the factors, aspects or components we identify, but of what goes on at the human level as real people interact with these factors in particular sites. The action is not in the named parts of our conceptual maps, but in the arrows that join them. (p. 138)

The aim of this thesis is to analyse "the action" that takes part in "the arrows" joining the show people and the educational providers in coastal and western Queensland, particularly in reference to the key concepts of marginalisation, resistance and transformation.

1.2 The significance of the problem

The 1990s in Australia, as in many other developed countries, witnessed a proliferation of educational programs designed to address particular manifestations of 'special needs'. These 'special needs' are generally involved in some way with particular constructions of marginalised identities: women (see for example Rowan, Bartlett & Danaher, 1996); Indigenous Australians; residents in rural communities; students with physical and intellectual disabilities; the illiterate; and the itinerant.

Of these groups, there is no doubt that itinerants have a considerably lower profile. Yet one of the most predictable questions likely to be exchanged when strangers meet is: "Where do you live?" – or possibly its variant: "Where do you come from?" (Winning, 1990, p. 246). Establishing a new

acquaintance's physical location is "a pervasive and effective mechanism for regulating social relations" (Danaher, 1993, p. 71).

Permanent residence is also a means for constructing one incarnation of 'theother' – the shiftless wanderer whose address at 'no fixed abode' renders her or him at best unreliable and improvident, and at worst likely to be found in a court of law on some kind of property charge. (Danaher, 1993, p. 71)

Within the field of open and distance learning, Traveller education is further marginalised. By far the most common view sees clients of distance education programs as residing permanently in one of two sites: either on farms and properties, or else in towns and cities. The two groups are assumed to be prevented from enjoying the benefits of face-to-face teaching and learning – the first group by 'the tyranny of distance', the second by nonnegotiable work commitments or by being 'tied to the home' for other reasons. In both cases 'home' is a fixed element in the equation. The notion that 'home' – or at least the physical manifestations of 'home' – can move on a regular but sometimes unpredictable basis is rarely acknowledged.

The relative positionings in distance education of people whose homes are fixed and those who take their homes with them were neatly summarised in the following extract from a review of "Distance education around the world".

Itinerant wanderers delivering information by word **d** mouth were perhaps the world's first distance educators bringing information **from** afar to eager recipients encountered during their travels. This centuries

old practice was irrevocably changed by the invention of writing and later, print. Print was first put to use in distance education with development of correspondence courses created by universities during the middle 1800s to disseminate learning beyond the walls of existing institutions. Technology of information transfer has now outstripped the ability of educators and trainers to develop programs to serve the exponential increase in learners who wish to be educated outside traditwnalplace-bound learning sites. (Brown & Brown, 1994, p. 5)

Here "itinerant wanderers" are positioned as belonging to the 'prehistoric age', before the advent of writing and printing. This view of itinerancy provides a stark contrast to the technological developments that enable learners "to be educated outside traditional place-bound learning sites". By contrast, this thesis reveals that the itinerant show children in fact receive an education that is precisely non-place-bound in the way described by Brown and Brown (1994).

This study is significant, therefore, in seeking to direct attention to representatives of a generally undervalued lifestyle that has nevertheless a large number of manifestations: homeless youth, Gypsies, seasonal fruit pickers, fisherpeople, miners, circus people and participants in Australia's several agricultural show circuits. These people and their forebears have been speaking for centuries, but their voices have typically not been heard by 'the mainstream' or 'the centre' —largely on account of their residential patterns, around which so much that is connected to identity and education is organised, deviating from 'the norm' of permanent residence.

Here I am using 'speaking' and 'voice', not literally in the sense of verbal interactions, but rather from a highly politicised perspective, which emphasises the analytical necessity of identifying what is said by whom, to whom, in which contexts, thereby drawing attention to the fact that, while some may speak and their voices are interpreted as commands that will be obeyed immediately, others may speak and their voices are regarded as being safely ignored. It is in this sense that this thesis is conceived as a counternarrative — as helping to record and disseminate an alternative 'story' about the purposes and effects of the show children's educational experiences.

In addition, there are at least three 'practical' ways that the problem outlined above is significant. Firstly, policy makers and planners of the increasing number and range of specialised educational programs need information about the perceived effectiveness of such programs in addressing the clients' needs, and also an understanding of how the results of policy *can* be consumed. This is so particularly in the context of the simultaneous trends during the 1990s and early 2000s of greater local control over decision making and increased requirements for accountability in the expenditure of public funds.

Secondly, the area of activity that **has** been assigned the broad label 'open and distance learning' has been promoted in some quarters as a possible panacea for all lunds of educational and socio-cultural 'deficiencies' in the existing education system. This study provides an opportunity to assess the extent to which such optimism is justified, by investigating one specific initiative in open and distance learning from the viewpoints of its clients.

Thirdly, and perhaps most importantly, educational programs encapsulate many of the trends, pressures, values and behaviours evident in the society at large. In particular, identities — whether marginalised or otherwise — are created and contested within constructed learning situations, with the potential to render students empowered and literate (in its widest sense), or alternatively disempowered and less likely to participate fully in the post-industrial age. It is hoped that more far-reaching conclusions about the responses to and the consequences of educational programs in Australia will be made possible **as** a result of the investigation outlined below.

1.3 The research questions

As I indicated in the first section of this chapter, this study is concerned with the operation of a specialised education program designed for the show circuits of coastal and western Queensland. The program arose in 1989 as a direct consequence of the lobbying of the then Queensland Department of Education by members of the Showmen's Guild of Australasia. These members were motivated by a strong desire to improve the educational opportunities available to their children, while at the same time to maintain their children's contact with the rich traditions of show life. The Guild members were successful in attaining their objective of pressing for the establishment of an education program specifically designed to meet the identified learning needs of their children. In 2000, the program was superseded by the establishment of a separate school for the show children.

This situation suggests two important characteristics of the show people. The first characteristic is their conviction that they constitute a special group, different from 'mainstream' Australian society. They have distinctive mobility and residential patterns; they use terms such as 'showies' and 'mugs' as conscious ways to valorise and strengthen their own identities; they cherish their links with previous generations of show people, both in Australia and in the United Kingdom. The show people see themselves, and they believe that other people see them, as belonging to a minority group.

The second characteristic of the show people with which this thesis is concerned is their political 'know-how'. They use their heightened self consciousness of what they believe makes them different from other Australians to articulate a set of objectives, ranging from the education of their children to greater access to water and electricity at particular showgrounds. They formulate and enact a series of tactics to achieve those objectives, from seeking meetings with government ministers to reinforcing the show people's considerable contributions to the local economy. They pride themselves on their persistence in overcoming obstacles that they believe would defeat less well-organised and less cohesive minority groups.

These two characteristics of the show people – their 'difference' and their political 'know-how' or agency – are taken up in the study's three research questions. The first research question asks: "How do the show people experience marginalisation?". This question seeks to identify those aspects of the show people's identity constructions and educational experiences that reflect their sense of being both distinctive and disadvantaged in comparison with 'mainstream' Australians.

The second research question asks: "How do the show people resist their marginalised status?". This question derives from an assumption that, rather than passively accepting their disadvantage, show people exercise agency to ameliorate and if possible to change the bases of that situation.

Another dimension of the interplay between the show people's 'difference' and their agency is the extent to which their position *vis-a-vis* other groups undergoes positive change. Pursuing this theme, the third research question asks: "How do the show people transform their marginalising experiences and resistant practices?". While accepting that such a transformation is neither inevitable nor necessarily permanent, the question examines the possibilities of thinking otherwise about the show people's identities and educational experiences.

These three research questions, then, are linked and integrated in the ways that they seek to reveal different aspects of the show people's 'difference' and their agency. The first question examines how the show people's marginalisation originates from their 'difference' from 'mainstream' society. The second question investigates how the show people exercise their agency to resist that marginalisation. The third question discusses the show people's capacity to channel that agency to transform the marginalisation of their 'difference' to an understanding and a valuing of that 'difference'. Or to express the political importance of each of the three data analysis chapters slightly differently: each helps to tell the story of the show people's resistance and transformation, first by explaining the need for change, second by highlighting the potential for resistance, and third by discussing transformative moments associated with the circulation and endorsement of new ways

of thinking about show people and their education. The questions are therefore all filtered through reference to educational issues, but of necessity they also engage with a wider range of issues that help to demonstrate the concerns identified in the questions.

1.4 The outline of the thesis

This thesis is arranged in eight chapters. This chapter has stated the problem to be investigated and has outlined the significance of the problem, and has also articulated and explained the research questions to be answered. Chapter Two reviews selected literature about two key issues. Accounts of Australian shows are discussed. Several Australian and European studies in Traveller education are interrogated. In both cases, gaps in, and limitations of, the existing literature are identified to create spaces for this study's contribution to knowledge.

Chapter Three elaborates the conceptual framework of the study, beginning with an analysis of the connections between itinerancy and marginalisation from the perspective of Traveller education. Michel de Certeau's (1984) concept of 'tactics of consumption' is examined in detail, as are Mikhail Bakhtin's (1986a) notions of 'outsidedness' and 'creative understanding'. These ideas in combination constitute a conceptual approach that assists in understanding and theorising the interplay among marginalisation, resistance and transformation in the lives of the show people.

Chapter Four discusses the research design of the study. That design derives from, and builds on, the links between researching Traveller education and marginalisation, resistance and transformation on the one hand and de Certolian 'tactics of consumption' and Bakhtinian 'outsidedness' and 'creative understanding' on the other. These elements have implications for the study's ethical and political dimensions, which also articulate well with the conceptual framework outlined in Chapter Three. Specific data gathering techniques are outlined, **as** are the study's delimitations and limitations.

The next three chapters constitute the **data** analysis of the study. Chapter Five addresses the first research question, by examining the show people's experiences of marginalisation. The second research question is considered in Chapter Six, which analyses the show people's resistance of those marginalising experiences. Chapter Seven provides information relevant to the third research question, which examines how the show people's tactics of consumption, and their capitalising on outsidedness and creative understanding, contribute to transforming their marginalising experiences and resistant practices, as well as to presenting a counternarrative about what Traveller education *can* and should be.

Chapter Eight draws the study to a close by synthesising the data analysis chapters' answers to the three research questions outlined at the outset. It concludes by revisiting the personal note outlined below, **as** well **as** suggesting some possible directions for future research projects in Traveller education.

1.5 A personal note

In keeping with the practice of eschewing authorial anonymity favoured by the approaches to research ethics and politics adopted in this study, it is appropriate to insert here a brief personal note about the researcher. My research interests have a bearing on this study in at least two important respects.

Firstly, my substantive position as Foundation Lecturer, and more recently Senior Lecturer, in Open and Distance Learning in the Faculty of Education and Creative Arts at Central Queensland University constitutes the framework for my interest in Traveller education. I consider that open and distance learning has much of value to learn—conceptually, methodologically and empirically—from the lives of people whose physical spaces change constantly, yet who aspire to continuity of learning in two senses: continuation of the education program regardless of physical location; and ensuring the continuation of a rich cultural tradition that is intimately connected with itinerancy. These shifting and fluid understandings of 'place' can significantly enhance current efforts within open and distance learning to theorise 'mixed mode' and 'flexible' approaches to educational provision.

Secondly, my previous research projects have concentrated on several types of 'marginalised groups'. These have included Indigenous Australians, women students, student teachers, beginning teachers, academic staff members in recently proclaimed universities and proponents of unfashionable disciplinary paradigms. Despite the diversity of these groups and the varied rates of intensity with which I have studied them, the projects have given me a developing understanding of people whose identities are seen by significant

others **as** inferior or less valued in particular ways – or else **as** unacceptably challenging to the *status quo*. This understanding should assist my examination of the 'marginalised group' under review in this study. Furthermore, my long standing interest in 'marginalised groups' constitutes the academic background against which my use of postmodernist theory in this work takes place. That is to say, like Griffiths (1995, p. 224), I am "interestedin postmodernism—but only in so far as it is relevant to my wider values". For me, those "wider values" are intimately connected with the links and relations among social groups with differential power and status being reflected in the implementation of particular education programs.

In combination, these two points – myconviction of Traveller education's potential contribution to theorising open and distance learning, and my focus on members of 'marginalised' communities – have had a significant impact on the way that I have designed and conducted the study reported in this thesis. In Chapter Four, I outline how I have sought to carry out Traveller education research by applying a more fluid and less dichotomised approach to the 'researcher'—'researched' relationship. In Chapter Eight, I revisit this personal note by reflecting anew on the "situated,partial and interested" (Kenway & Willis with Blackmore and Rennie, 1998, p. xii) aspects of my developing subjectivity as a researcher. In this thesis, my interest in delineating my own 'learning on the run' operates in parallel with my identification of the show people's educational experiences.

CHAPTER TWO

THE REVIEW OF THE LITERATURE

"Itinerancyis also a problem **cf** international importance and not one peculiar to Australia. Mobility and distance are synonymous with the educational disadvantage **cf** children."

Lally, 1993, p. 202

"I would make the same argument for all children—the necessity for parents and teachers to work together to let the children have the chance to learn from both and then move onto be truly themselves. ...I argue it particularly for Gypsy and Traveller children because for them the tensions are greater, the hostility they can face is more intense, the stereotyping is more negative, history weighs more heavily on them and they have been too long at the margins."

Kiddle, 1999, p. 156

2.1 Overview of the chapter

The purpose of this chapter is to present a detailed and critical review of selected sections of the literature relevant to the research problem outlined in Chapter One. The aim is to identify what has already been written about the context in which the problem is situated, and to discern the strengths and limitations of those writings. This will assist the substantive section of this study in contributing results that are original and of interest to other researchers.

The chapter is divided into the following major sections:

- Australian shows
- Travellereducation.

In each section, the intention is to review the existing literature critically, identify its strengths and limitations, and indicate how this thesis can contribute to addressing those limitations. Thus I seek to establish a point of departure for my own research, which as I noted in Chapter One is conceived as a counternarrative to traditional ways of understanding — as demonstrated in the existing literature — Australian shows and Traveller education.

In particular, I established in the previous chapter my focus on three key organising concepts throughout the thesis: marginalisation, resistance and transformation. Those same concepts constitute the framework informing this chapter's interrogation of the existing literature. My goal is to present this study **as** a counternarrative to the stereotypical and marginalising images of itinerant people, and particularly of the Australian show people, contained in a

large proportion of existing studies of Australian shows and of Australian and European Traveller education. Similarly, I seek to position the study as akin to, and contributing to, the still small but steadily growing literature that emphasises the resistant and transformative potential of itinerant people and their education.

2.2 Australian shows

It is appropriate to acknowledge at the outset of this account of the literature pertaining to Australian shows that lack of space precludes my inclusion here of the detailed study of the literature relating to British fairs that I undertook in an earlier draft of the thesis. Suffice to say that many striking parallels between the lives and educational experiences of Australian show people and British fairground people were revealed, with the themes of marginalisation, resistance and transformation being very strongly in evidence.

Similarly, the thesis has not interrogated the literature relating to North American carnivals. Partly this is because Australian 'showies' generally portray themselves as being significantly different from their North American counterparts. Frank Foster, one of Bill Morgan's (1995) informants, identified a crucial element of that difference: "But in the history & American showman [sic] their business is basically, and mainly, run by tent men, even today they still have their tent shows" (p. 136).

A practical difficulty with interrogating the North American literature is a difference in terminology. There the term 'migrant' is used to refer to occupational Travellers, such as Mexican fruit pickers working in the southern United States of America, as well as being used in the more conventional sense to denote people who travel from one country to another, without necessarily being itinerant. This difference in terminology prompted a British researcher in Traveller education to assert that there is no North American equivalent of the European literature on occupational Travellers (Pullin, personal communication, 1996) – although that year marked the publication of the edited book *Children & la frontera* (Flores, 1996).

Therefore, although I am aware of the existence of both the Showmen's Guild of Great Britain and the Outdoor Amusement Business Association (founded in 1964) in the United States of America, and of the latter's affiliation with several Showmen's Associations in individual states (http://www.oaba.org/index.htm, 1996), I acknowledge that a limitation of this thesis is that an intensive examination of the British and North American literatures on fairs and carnivals respectively has been postponed until an appropriate future occasion.

Within the parameters set by that limitation, then, the literature about Australian agricultural shows can be characterised as scattered and fragmented – much as Australian show people are often depicted by the 'mainstream'

literature.' This characterisation creates a point of entry and of departure for my study, which aims in part to synthesise and extend certain key elements of the literature about shows and show people. The most substantial Australian literature emanates from the Showmen's Guild of Australasia. The Victorian Showmen's Guild's quarterly magazine, *The Outdoor Showman*, includes articles of interest to travelling show people in Australia and New Zealand, such as reports on major shows, references to legislation pertaining to equipment registration and advertisements for rides (like the ferris wheel) and 'joints' (like the laughing clowns). A regular feature is devoted to circuses.

Several articles in *The Outdoor Showman* reflect the 'showies' conviction that they have a specialised and valuable lifestyle – a conviction that, as the data analysis chapters of this thesis demonstrate, has considerable impact on their attitudes to and experiences of marginalisation, resistance and transformation. For example, **a** New Zealand 'showie' wrote in the 'Letters' page:

We can be perceived by others as strange. We are both divided and united. Anywhere in the world you would be greeted warmly by show families without prejudice. I read letters from old show people worldwide and the stories are similar to our own. Sad that their friends have

The same is true of studies of circuses, another group of occupational Travellers. The European literature is substantial and steadily growing; the Australian literature is relatively fragmented and the academic dimension of that literature is largely dominated by the writings of a single author, Mark St Leon (see for example Cannon with St Leon, 1997; Ramsland with St Leon, 1993; St Leon, 2000).

passed on and happy that **our** new generations are coming through to carry on with this wonderful profession. (Ashworth, 1997, p. 7)

The fact that this letter was written by a New Zealander gives special point to the assertion that "We can be perceived by others as strange". In other words, being rejected by one's fellow citizens contrasts with a sense of identification with 'showies' in other countries. This is a construction of marginalisation with which 'mainstream' Australia would not be familiar. The last sentence of the paragraph emphasises the importance of passing on the cultural traditions of "this wonderful profession" to the members of "our new generations".

Similar feelings were evident in Lew Osborne's (1997) retrospective essay "Behindthe loudspeakers, toffee apples and fairy floss". For example, he referred to:

The thoughts of a young boy that takeme back to when **as** a Showie kid going to school, inundated with questions about the show. What's the best ride? The fastest one? Will it **make me** sick? What's your dad got at the show? The questions that went on. If elt important, commanding this unbelievable respect from kids that normally wouldn't give me the time of day. (1997, p. 21)

Here the contrast is between an identification among children of a shared delight in the excitement and glamour of the show **and** an awareness that those same children "normallywouldn't give me the time of day". This sense of being marginalised on the basis of a perceived 'difference' and 'strangeness' is a major theme of Chapter Five. Similarly, Chapter Six takes up the ways in

which show people actively resist, rather than passively accept, that experience of being marginalised.

Chapter Seven elaborates the notion of transformation to both marginalisation and resistance of and by show people. Osborne (1997) identified the source of change within the show circuits that contributes to that transformation: competition among 'showies' for the money of the 'locals' attending the shows.

Showmenfor 70 plus years have lived and breathed this competition factor improving their business, knowing full well that **f** they don't the operator across the alley will. Looking for the next thrill ride from America that will turn heads **or** buying the best plush toys. What will be the most popular one this year? The monkey **or** the big wrinkleface dog to put on their games. (1997, p. 21)

This account of internal competition driving change on the showgrounds is a timely reminder that there is no such phenomenon **as** a 'typical showie', despite the linguistic convenience of using this **term**. Chapter Six analyses the ways in which the multiplicity of experiences of being **a** 'showie' helps to strengthen the show people's resistance of their marginalised identities.

Osborne's (1997) essay ended with a characteristic call to unity among 'showies':

The atmosphere, this magic that is created is unique to shows and can only get better **as** we strive to hold a place in a rapidly changing world and it's happening right now, behind the loudspeakers, toffee apples and fairy floss. (1997, p. 21)

This rhetorical flourish is characteristic of many of the show people's statements recorded in the data analysis chapters. The intention is clearly to depict Australian shows as "unique" in creating "this magic", which is evoked by such images as "the loudspeakers, toffee apples and fairy floss". The optimistic belief in progress – "this magic. ... can only get better" – contrasts with the slightly ominous reference to the need to "striveto hold a place in a rapidly changing world". This last point is particularly important. The writer has constructed "a place" as the show people's 'portion', their equitable share of the available resources. That "place" exists independently of the show people's itinerant lifestyle, which means that their "place" changes from one week to the next. These themes – multiple understandings of terms like "place", using those terms to assert a right to consuming available resources and in the process to engage and hopefully to transform the understandings of resource providers – are elaborated throughout the data analysis chapters of this thesis. The thesis thereby extends the existing literature, by providing an analytical and a conceptual dimension to terms of that kind used by and about show people.

Written in a similar vein to *The Outdoor Showman*, Bob Morgan's (1995) book *The showies: Revelations of Australian outdoor side-showmen* is a celebratory text that recorded the life stories of several 'showies' who

A major theme of the book was the distinctive contribution made by these individuals to Australia's cultural history. One chapter **dealt** with Jack Allan, who joined the show in 1928at the age of fourteen and who developed a very successful career as, among other roles, a 'spruiker' (a 'front man' charged with attracting the crowds to a particular act) and a promoter of the wrestler

"arefairly representative and reflect the history of their industry" (p. xvi).²

Chief Little Wolf. In a representative passage about Jack and his wife Dawn, Morgan (1995)wrote:

...they have the satisfaction of knowing that they were a part of those magnificent people who brought so much joy and entertainment to the multitude of Australians who were so dependent on them and their kind for decades – from the Depression days of the 1930s through the dark days of the Second World War, and into the post-war years leading up to the advent of television. (p. 112)

Due allowance must be made for the exaggeration with which any old man who has achieved a respectable prosperity tends to view the hardships of his youth, yet equally we must be cautious of dismissing as fantasy the vicissitudes which a more protected age finds hard to credit. (p. 3)

In any case, it is precisely the voices of the show people such as those presented in *The showies*, and the constructions of 'reality' that they represent, with which I am concerned in this thesis.

In presenting selected quotation from *The showies*, I am conscious of the methodological injunction of Duncan Dallas, author of *The travelling people* (1971):

This close identification between 'showies' and central events in Australian history is intended to emphasise the size of the contribution that show people continue to make to Australia's cultural life. This integral association contrasts with other passages in *The showies* in which Morgan emphasised the feelings of his selection of 'showies' that they were marginalised from 'mainstream' Australian society. This was particularly evident in their recollections of their schooling experiences. For example, Frank Foster, born in 1927 and "a member of one of Australia's oldest and best known show business families" (p. 113), had the following recollection of his school days (which, in view of its encapsulation of several themes underpinning this thesis, is worth quoting in full):

Well, believe it **or** not basically I had very little schooling at all. I was self taught, we went from school to school. This was no fault of my parents[;] it was just the way things were in those days. I went to each public school that we could **make** it to, but. • the roads were so bad in those days it could take three to four days to get to a show and town.

By the time we got into a town it might be Thursday and then you'd go and see the local school and they just couldn't be bothered with you. The teacher would tell you to sit in the class and if you couldn't pick up the work it was too bad. With the result that by the time you had to move on it almost proved a waste of everyone's time.

Then we got correspondence courses and for **a** while that was all right, but going from town to town the mail was never able to catch up. Some of those outbackplaces we would have left before the mail arrived. I remember one time the teacher **from** the correspondence school wrote

to my mum and said, "Frank's work is all right but I wish he'd stop using too much bread". Everyone used to help me, and to rub out words we used bread — there were so many different handwriting styles in my work they must have thought I was Chinese! Now at my stage of life I look back on those early days and wonder how much better off I might have been had the education system for we travelling people been a better one.

Again on reflection, in those days your education was on how to survive and what you did for a living, there was no need to be an academic, your education, believe it or not, was your craft and how you hundled things—such things as what to do when the rent is too dearfor a show! By that I mean, if you get on a showgrounds you know a good position from a bad one. You don't need a college education for that, nor do you need a college education to know how to get the pitch for an act up there on the platform. All you know you have to work for it. The average person walks around a fairground or showground and may think we get it easy but it's never that way. Mostly we live out of doors, and we have to forgo a lot of other things that other people have. (p. 123)

Several comments can be made about this memoir of one 'showie's' educational experiences. The general and crucial point underpinning those comments is that the literature on Australian shows examined in this section of the chapter has shaped the design of the thesis, not by accident, but rather by my deliberate structuring of interview questions to ensure that the issues that recur in the literature – including the three themes in Foster's recollec-

tions (cited in Morgan, 1995) identified below – are acknowledged and used to extend existing knowledge about Australian show people. (Chapter Four elaborates the study's research design and makes these links among that design, current literature and the study's conceptual framework explicit.)

Firstly, many of the show people whose voices are heard in Chapter Five recollect the same kinds of difficulties associated with travelling from school to school and dealing with teachers who were ill-equipped to engage with these temporary visitors. The sense is very strongly of incomprehension, apathy and possibly distrust or dislike on the part of the 'locals' against the 'showies'—all crucial elements of the show people's marginalisation.

Secondly, Foster's reference to receiving family assistance with his "correspondencecourses" evokes the multi-age learning and peer tutoring that are strong features of current educational provision for the Queensland show children. This reference also emphasises the continuing involvement of show parents in their children's education, either through their employment of a home tutor or through their taking that role themselves. This valuing of formal education, and the preparedness of family members to provide assistance in completing this important task, echo Foster's recollection of the fact that "Everyoneused to help me".

Thirdly, this valuing of formal education sits somewhat uneasily beside a distinction between 'book' and 'real' learning. This can be seen in Foster's assertions that "You don't need a college education" to decide "what to do when the rent is too dear for a show!" or "toknow how to get the pitch for an act up there on the platform". Foster's pride in this arcane 'situated learning' was reflected in the comments of many of my interviewees, who also

concurred with Foster that show life is difficult and "we have to forgo a lot of other things that other people have". The point to emphasise here is that Foster's constructions of his educational experiences were firmly located in the context of his working life and the cultural traditions surrounding that life. This same context and these same traditions also animate the voices of the show people heard in the data analysis chapters of this thesis, and form the framework for understanding the show people's accounts of marginalisation, resistance and transformation on the show circuits. As such, I build on and extend significantly this crucial dimension of the literature in this thesis.

Many of these themes were also exemplified in *Sideshow alley* (Broome with Jackomos, 1998), a collaboration between Australian historian Richard Broome and Indigenous Australian travelling boxer and wrestler Alick Jackomos. The authors' account of the rise and fall of 'sideshow alley' referred not to the joints and rides that that term evokes today, but instead to the 'alley of wonder' that comprised animal acts, boxing troupes (many of whose members were Indigenous Australians), 'freak' shows and illusion acts. The authors emphasised the resistant agency of the itinerant inhabitants of 'sideshow alley', arguing that their extended family provided a refuge from a marginalising broader community and a recognition of skills that were not valued outside the 'alley':

A dominant theme of the book is that Sideshow Alley was a place of power for its participants. This may seem surprising given that showpeople were viewed by the rest of society with both fear and wonder, and as outcasts. However, showpeople evolved their own culture over generations and this enabled them to forge their own self-

assured identities and a unique way of life. Those who fought in the boxing tents in Sideshow Alley were often seen as victims of low wages, a hard life and too many punches. Despite this view, they too relished their unique life, felt powerful being tent boxers, and they look back on it as a golden age in their lives. Seen in their own terms there are few victims in Sideshow Alley. (p. viii)

This thesis certainly eschews a portrayal of show people **as** "victims", and aspires to communicate their experiences of marginalisation, resistance and transformation "in their own terms". At the same time, Broome and Jackomos recognised that ambivalence was evident in other Australians' views of the inhabitants of 'sideshow alley':

Sideshow Alley was also a place of powerfor other Australians. Those who did not visit it demonised it as a powerful place of low and dangerous entertainment. Those who frequented Sideshow Alley found it a powerful source of wonder in their lives. For both groups it helped to shape their identities by gazing at difference; because human identities are formed partly by knowing, seeing and experiencing difference. We know who we are, in both a positive and a negative sense, by that which we are, as well as by that which we are not. (p. viii)

For me, ambivalence lies at the centre of these 'mixed feelings' about difference. Ambivalence can be enlisted to further the strategies of marginalisation, if the 'fear' of difference is emphasised and exploited. Alternatively, ambivalence *can* be deployed to facilitate the agency of those who are marginalised and accordingly assist them to resist and transform and thereby become more powerful. *So "a dominant theme"* of Broome with

Jackomos's historical account of 'sideshow alley' from the 1870s to the 1950s and 1960s is played out in this thesis in the context of contemporary concerns. This is another vital respect in which the thesis is conceived as contributing to and extending the store of knowledge about Australian show people.

I turn now to consider how 'showies' are depicted in a small number of Australian newspaper and magazine articles. The intention is to record some elements of popular constructions of how 'locals' are held to regard shows and 'showies'. In June 1996, when the show came to Rockhampton, a provincial city in Central Queensland, Ken Coombe, the president of the Rockhampton Agricultural Society, was interviewed for a local newspaper.

Mr Coombe said the show provided an important social and economic function.

He said the show's rural component was an essential element in the success of the show which offered the opportunity for town and country residents to exchange views and talents. . .

Mr Coombe said the Rockhampton Show incorporated a lot of tradition while remaining progressive and responsive to what the community wanted. ("Show offers the best of town and country", 1996, p. 6)

These remarks underline the symbiotic – *albeit* sometimes tense – relationship between show people and show societies. A major attraction for many people to attend the show each year is undoubtedly the delights of sideshow alley and items like fairy floss and show bags. So the show societies

depend on the 'showies' to provide these attractions, which are helpful and perhaps even necessary to entice people to view local products in the paddocks and the pavilions. Yet the 'showies' rely on the show societies for the logistical support and the organisational framework required to conduct the annual show.

Coombe's reference to "alot of tradition" echoed another sentence in the article: "Visitorspassing through the gates follow in the footsteps of a marvellous tradition as they enter the annual show to marvel at this year's latest attractions" ("Show offers the best of town and country", 1996, p. 6). This complements statements by some of the interviewees in The showies (Morgan, 1995): local people should attend the show because it is part of a significant rural tradition that celebrates rural life and strengthens the solidarity of living in a rural community. Suggestively show people themselves received no mention in this article, implying that 'showies' do not spring immediately to 'locals" minds as being an integral part of this significant rural tradition. This invisibility carries over into scholarly constructions of itinerant people, as the next section of the chapter demonstrates. This thesis is intended to counteract, as far as possible within the study's parameters, the marginalising impact of that popular and academic invisibility – an approach that accords with Terry Evans' (1998) assertion:

In many respects, show children can be seen as travelling the margins of modem Australian society, living and learning as their families eam their livelihood providing the facilities and services which keep the country shows alive. (p. xii)

The week before the publication of the interview with Ken Coombe was 'show time' in Gladstone, a port city in Central Queensland south of Rockhampton. The local newspaper marked this event by featuring an article about a travelling show family. The article was entitled incorrectly "Codeyis circus' fifth generation traveller" (1996), the misconception that shows and circuses are synonymous evidently extending to newspaper subeditors. Beginning with the statement, "CodeyMiller represents the fifth generation of afamily which has chosen to make travelling with the show a way of life" (p. 1), the article featured an interview with Richard Miller, who contrasted his own schooling experiences ("In his day the [show] children had to attend school at their various stops along the way" [p. 2]) with those of his grandchildren: "His grandchildren and the 30 or so other students who travel with the show now have a teacher who travels with them and teaches them from 9am to 3pm" (p. 1).

Although this was an accurate – *albeit* abbreviated – account of the education program implemented for the show children by the Brisbane School of Distance Education, what interests me here is the construction of this phenomenon of attending school "*from 9am to 3pm*" as a special event worthy of relatively lengthy mention in a brief article. Again we encounter the ambivalence attending many of the interactions between 'showies' and 'non-showies': on the one hand attending school on a regular basis is constructed as something that 'showies' and 'locals' now have in common, yet on the other hand this is contrasted with Richard Miller's own schooling experiences, which marked his family and him as essentially different from 'normal' Australian families. This textual treatment is part of a superficial celebration of

the exotic lifestyle of itinerant people that is no less marginalising than constructing those people as 'other' to sedentary people.

This same ambivalence was manifested in a longer article (Olszewski, 1995) about show children in *The Australasian Post*, a popular magazine. The combination of seemingly discordant images in the headline "Carnivalswith class" was followed by the subheading "Showkids don't have to learn on the road any more. ..school follows them around" (p. 2). This suggestion of a previously disadvantaged group just now 'catching up' with their 'normal' peers was carried into the beginning of the article.

It's one **Australia's most unusual classrooms**, a tiny caravan that travels backwards and forwards across Australia each year.

But to Wendy Lou Stewart it's the world's most important classroom – because it's her very own.

Wendy Lou reckons she's just another ordinary Aussie schoolkid, but to people who don't understand the way **a** life she's regarded as different.

Until recently, Wendy and her mates were considered outsiders, and denied things lots of kids take for granted, especially a good education.

Wendy Lou is a "showies" kid – her mum and dad travel the agricultural show network and make a living by running a sideshow shooting gallery.

Showies are a special group of people and some of them, the "aristocrats" of the show circuit, have been running their stalls for two or three generations.

The showies keep to themselves, many families often intermarrying, which is probably a legacy of the way they've been regarded over decades by townsfolk who have shunned them as sort-ofgypsies, not to be trusted.

If you can cross the barrier and actually get to meet the showies, you discover that behind the gruff exteriors they're warm-hearted, fascinatingfolk.

Many are successful small-business people. Some are downright wealthy. (p. 1)

These paragraphs traverse the gamut of possible constructions of Australian show people. 'Showies' are represented as barred by their itinerancy from enjoying what most Australians take for granted, "especially a good education". Furthermore, they are marginalised because of decades of being regarded by 'locals' as "diflerent", "outsiders" and "sort-ofgypsies, not to be trusted". The construction of marginalisation gives way to that of a 'minority group', by possessing "one of Australia's most unusual classrooms" and by being "a special group of people". The putative identification of 'showies' with 'mainstream Australia' comes in the revelation that "beyond the gruff exteriors they're warm-hearted, fascinating folk". Finally, the suggestion that show people are an Clite with privileges not available to other Australians is contained in the references to "the 'aris-

tocrats' of the show circuit, [who] have been running their stalls for two or three generations", and to some of the 'showies' being "downright wealthy".

The point to emphasise about these constructions is that they reveal that 'showies' do not 'fit the mould' of itinerant people constructed by people who are not, and usually never will be, itinerant. They seem to the journalist — and through him to the readers of *The Australasian Post* — to be fascinatingly shifting and fluid in seeming from one perspective to be 'exotic' and 'strange', from another to be 'just like us', from another to be shamefully disadvantaged, from another to enjoy 'the best of both worlds' — travelling and operating successful businesses.

These somewhat confused and conflicting constructions of show people for popular consumption articulate with the earlier quotations from *The Outdoor Showman* and *The showies* (Morgan, 1995). These quotations demonstrate that 'showies' themselves consume and 'make use' of such constructions, and engage directly with their 'difference' from 'locals', in order to disrupt marginalising stereotypes of show people and transform people's understandings of the multiple elements of 'showie' identity. This crucial issue is acknowledged and built upon in the thesis, which goes beyond ambivalent description to a critical analysis of the ambivalence and its relationship to the production and contestation of marginalisation. As the data analysis chapters will indicate, these 'tactics of consumption' and this focus on 'outsidedness' and 'creative understanding' provide an appropriate framework for conceptualising, not only the show people's marginalisation, but also their resistance and transformation of that marginalisation.

The principal finding of this section of the literature review, then, has been to locate these processes of marginalisation, resistance and transformation directly in the 'showies" and popular accounts of Australian shows. Furthermore, I have staked a claim for the thesis's original and substantial contribution to knowledge, as a significant extension of the existing literature on Australian shows and show people, particularly their educational experiences and opportunities. Equally importantly, the study is conceived as a counternarrative to the marginalising stereotypes still prevalent in most 'mainstream' renditions of the show people's lifestyle and culture. It is therefore designed to contribute to filling some fundamental gaps and attaching sound to some crucial silences in the established understandings of the show people and their education.

2.3 Traveller education

As with the literature on Australian shows, the literature on Traveller education has been for a long time, and largely remains, scattered and fragmentary. As I elaborate later in this section, there is a strong tradition of educational provision for, although less so of research into, Travellers in Europe; as I indicated above, there is a limited literature about American itinerant 'migrant' people's education (Flores, 1996); the Nigerian National Commission for Nomadic Education has conducted a vigorous research program into educating nomadic pastoralists and migrant fisherpeople (see for example Tahir, 1991; Tahir & Muhammad, 1998; Umar & Tahir, 2000); there are a few studies of nomadic pastoralists in Asian countries such as India (Dyer, 2000;

2000; Dyer & Choksi, 1998) and Mongolia (Robinson, 1999); and relatively little has been written about Australian Travellers.

This scattered international coverage illustrates important and substantive differences among different groups of itinerant people. I have concentrated in this study on the literature relating to occupational Travellers, rather than to other itinerant groups such as Gypsy Travellers, military personnel (who, as Duffy [1987] pointed out, exhibit significant differences from other kinds of Travellers) and New Age Travellers. However, I refer to literature about those groups if they appear to have particular implications for researching the education of occupational Travellers.

This critical review of selected literature about Traveller education is framed around the interplay among the thesis's organising concepts: marginalisation, resistance and transformation. I argue that much of the literature is posited on blinkered and unexamined assumptions that actually contribute to Travellers' ongoing marginalisation. By contrast, a small but growing number of studies, mainly from the late 1990s, directly challenge the marginalising assumptions of earlier (and in some cases continuing) literature, and focus instead on Travellers' agency and the possibilities of alternative, more enabling educational provision and research. This thesis is located unequivocally in the community of research formed by those exceptional studies.

The review is organised around the following themes:

- the dearth of research
- the prevalence of negative stereotyping

some exceptional studies.

2.3.1 The dearth of research

Two comments about the literature on Traveller education can be made immediately: not very much has been published; and what little is published — with some significant exceptions, detailed below — tends to portray — and to marginalise — occupational Travellers as unfortunate 'victims' of an 'unnatural' lifestyle. This situation was graphically summarised by Lucassen, Willems and Cottar (1998) in their review of the literature in a related field:

The student of European history who searches for Gypsies willfind them only infootnotes. Today we still know little about how they worked and lived in the past. The same holds true for itinerant groups in general. (p. 1)

With regard to the extent and scope of the literature, no single text attempting comprehensively to identify and categorise various itinerant groups was encountered, apart from the specialised publications of the European Federation for the Education of the Children of the Occupational Travellers, which I discuss below. Indeed, the literature can be characterised as focusing separately on a number of such groups, particularly defence force personnel (in Australia and North America) and Gypsies (in the United Kingdom), but also including caravan park dwellers, fruit pickers and other seasonal employees, and circus workers (see for example Ramsland with St Leon, 1993). These items tend to be isolated articles appearing in journals dealing with general social concerns; a special issue of the *Journal & Social Issues*

devoted to residential mobility (Shumaker & Stokols, 1982)was a noteworthy exception to this observation.

The absence of a substantial literature is especially surprising given that fifteen per cent of the Australian (Rahmani, 1985) and twenty per cent of the American (Shumaker & Stokols, 1982) populations were estimated as being itinerant in the early 1980s. More recently, a journal article published in 1987 (Welch, 1987) had the dramatic title "As many as 100,000 Australian school children move school each year". In the same year, Harrington (1987) referred to "530,000 migrant [itinerant] students" (p. 36) in the United States of America. Miller and Cherry (1991) reported that, in the United States of America, "Mobilityprojects remain at approximately 20%" (p. 52). Fields (1997) reported literature that indicated that Australia's population is "one of the most highly mobile in the world" (p. 45), and that between 1986 and 1991 forty-six per cent of children aged between five and nine years, and thirty-eight per cent of children aged between ten and fourteen years, changed their permanent location at least once (p. 45).

Despite these statistics, only limited research has been conducted into Traveller education since then. For example, itinerant children were conspicuous by their absence from Keith Harry's (1991) review of the distance education literature, which conceived of distance education students as living permanently in one place, at a distance from the source of instruction.

This theme of a dearth of research about Traveller education was reiterated by several commentators, although surprisingly this recognition did not seem to prompt them to remedy the perceived deficiency. One example was the comment by Blair, Marchant and Medway (1984) that "there is no

comprehensive body of literature" (p. 251) about the impact of itinerancy on children, and another was the observation of Brown and Orthner (1990) that "Despitethefact that millions of family relocations occur each year, we are only beginning to understand the effects of these moves on the youngest members of these households" (p. 380). In confirming this deficiency, these commentators often identifies particular questions that they believed should inform such research. After stating baldly, "Much remains unknown about mobility", Shumaker and Stokols (1982) argued:

We need to explore the processes that mediate people's decisions to remain in an area or to move, and how these processes relate to the health of individuals and communities. (p. 2; emphasis in original)

One conclusion of a meeting of specialists in itinerant education (United Nations Educational, Scientific and Cultural Organisation, 1989, paragraph 62(a)) was the claimed need for "a general international policy for the collection of data" about itinerant children.

This is necessary to ensure researches [sic] at national and international levels and to establish an international data bank on the conditions governing the education of nomads, for dissemination of information, for proper planning and implementation of their education projects.

According to Duffy (1987):

The complexity **a** the many issues associated with mobility, and the very uncertainty **a** its real magnitude and extent throughout the school-age population, pose difficulties in the interpretation of the available

research findings. No clear picture of the problems of mobility and methods for coping with them emerges from the literature. . . (p. 544)

On this basis, "More research into the effects Amobility on the psychosocial development of children" was warranted. Birch, Lally and Tomlinson (1986) complained:

Evidence accumulated from several decades of research. . .[into itinerancy] has not provided any firminsights into long-term consequences of itinerancy for school learning. If anything the results are either indeterminate or contradictory.

Clearly this study is seen as making a substantial contribution to redressing the imbalance resulting from this dearth of research. Its particular anticipated significance lies in its intended status as a counternarrative to traditional 'stories' about Australian show people and about itinerant people more broadly. It is in that guise that it is proposed to supplement and augment the still too few studies of Traveller resistance and transformation analysed in a later subsection of this chapter. Equally clearly, the thesis is premised on the belief that this is a valuable area of analysis — one with highly significant implications for show children in particular, but also for understandings of marginalisation generally.

2.3.2 The prevalence of negative stereotyping

According to Lucassen and his colleagues (1998):

Our knowledge has been severely restricted not only because of historical negligence, but also because of two closely connected paradigms, one which views Gypsies and other itinerant groups as criminal, marginal and poor, and another which focuses almost exclusively on their alleged common ethnic identity and origin. (p. 2)

Furthermore, they referred to "the inclination to view itinerant groups predominantly as down and out riff-raff" (p. 2).

In terms of the negative portrayals of Traveller education that this "inclination" is likely to promote, while academic educationists might very well eschew the prejudices outlined in the previous subsection in favour of an 'objective' understanding of itinerancy as a complex social phenomenon, they do not necessarily produce more 'balanced' accounts of the educational dimensions and implications of itinerancy. On the contrary, many of the assumptions that position fixed residence as 'normal' and itinerancy as 'abnormal' carry over into the literature dealing with the schooling experiences of itinerant children – and doing so helps to perpetuate the marginalisation of those children.

This construction of the 'itinerant:abnormal' homology was implicit in the definition of 'itinerant children' presented by the compilers of the *Australian thesaurus* of *education descriptors* (Lavender & Findlay, 1984):

Children who move frequently with their families from one semipermanent location to another – includes children of military personnel, construction workers, gipsies, etc. (p. 114)

Attempting to include the widely varying experiences and patterns of itinerancy in a single, brief definition such as this is not effective. In this case, for example, the significant differences between the show people's organisational strength that derives from the Showmen's Guild of Australasia and the 'extended family' basis of operating most Australian circuses are elided in the almost throwaway "etc.". The inadequacy of this definition derives from its implicit assumption that 'the norm' is permanent location, and the corollary belief that if people are unfortunate enough to have to move at least they can enjoy the benefits of moving "fromone semipermanent location to another". In other words, the location (or what de Certeau [1984] would call 'the place') is privileged, and in the process any understanding that moving is potentially pleasurable and enabling is elided.

This depiction of itinerancy as 'abnormal' characterised the majority of studies in the 1960s and 1970s – and has continued as a theme in some research in the 1990s. To be a travelling student in that period was to be the recipient of considerable academic concern, verging on pity. Swendson (1958) used a gardening metaphor, by comparing itinerant students to the processes of "transplanting vegetables, flowers, or nursery stock" (p. 332). She acknowledged that for children "The results of the transition are infinitely more important than the results of moving plants, but many of the relationships are parallel" (p. 332). She concluded by emphasising the ameliorative role that education could perform for such an itinerant child:

Best of all, like the well-adjusted plant, he [sic passim] sends out new shoots of growth. In school achievement, in social relations, and in personal development, he blossoms into fuller living and learning. When

these signs appear, the teacher knows that the transplanted child has taken root and is at home. (p. 334)

Levine (1966), an American clinic psychologist, hypothesised that "any move represents both a problem in adaptation for children, and an opportunity for the development of preventative mental health programs" (p. 62). Although he acknowledged, "While it seems reasonable to expect a relationship between the frequency of moves and academic performance, some studies do not obtain the expected [negative] result. ..." (p. 67), his overall finding was emphatic in relation to the connections between itinerancy and school adjustment: "The problem is clear, but the solution is far from obvious" (p. 68).

The response by Morris, Pestaner and Nelson (1967) to a confounding in their research of their hypotheses about the different achievement levels of itinerant and non-itinerant students was to point to what they assumed were the contaminating effects of "individual prediction and personality variables" (p. 78). The authors identified "the etiologically significant variable" in the link between itinerancy and academic achievement as likely to be "the value system and motivation of the child and his [sic passim] family; i. e., what kind of child tends to improve, or retain his status with mobility and what kind of child suffers a decrement with mobility?" (p. 78). They expressed their frustration that "'Analyses in the research literature on mobility, to date, do not contribute to this vexing problem" (p. 78). As I elaborate below, I regard these kinds of "Analyses" as misconceived and as contributing little to a genuine understanding of the educational needs and

aspirations of itinerant people. This is therefore a literature that I disavow rather than one upon which I seek to build.

Two other and extreme explications of the assumption that itinerancy is a mostly negative experience (a proposition from which I vigorously dissent in this thesis) also appeared in 1967. One was the *Plowden report* (Plowden, 1967), which designated Gypsies as "*Britain'smost educationally deprived group*". The other, which itself used the term "*extreme*" to refer to a particular group of itinerant people, was a study of Gypsy education in Britain:

...although the group of children involved is too small to justify a discussion of this length in the body of our Report, the children's educational needs are nevertheless extreme and largely unmet. Moreover the economic and social handicaps of the group from which they come arise to a large extent from the fact that successive generations of gipsy children are deprived of the education that would enable them to compete on equal terms with the rest of the community. Extreme as they are, the needs of gipsy children cannot be effectively met by measures of the kind we recommend for the more general problems of urban deprivation. (Adams & Smith, 1967,p. 595)

In the early and mid 1970s, the Swedish psychologist Joseph Schaller published several studies relating itinerancy to such issues as emotional wellbeing and school behaviour. Often his studies reported contradictory findings. For example, "Research results on the relations of residential change to mental health are inconsistent" (Schaller, 1972,p. S), and "There are children who suffer from a move, but also children who benefit from it"

(Schaller, 1975, p. 3). Schaller's response to this inconsistency conformed to that of most writers on Traveller education: he questioned the research design of the studies (1972, p. S), or he emphasised the apparently unequivocal negative associations of itinerancy: "These results show clearly that geographic mobility is related to poor school adjustment, especially peer relations" (1975, p. 3). The following statement encapsulated his overall view of itinerancy: "Itis very important to recognise this potential social problem and to try to help the child during the adjustment process after a family move" (p. 3).

Whalen and Fried (1973) extrapolated from the results of their tests of itinerant and non-itinerant senior high school students in Livermore, California that "a relationship does exist between mobility and achievement" (p. 165). They acknowledged that this relationship could be either positive or negative, so that "It is possible that the interests and attitudes of higher intelligence students are stimulated by frequent geographic relocations", and "Less capable students may find frequent moves too bewildering to cope with" (p. 165). (Long [1975] made a similar finding, hypothesising that "Interstate migration is most likely to be undertaken by well-educated persons whose children tend to do well in school" [p. 369], but he also noted that "Exceptfor children of college graduates,. ... increasing frequency of interstate migration is associated with increasing likelihood of a child's being enrolled below the modal grade" [p. 3731.) Nevertheless, their overall conclusion was in keeping with the 'problem' and 'difficulty' discourse that characterises most studies of Traveller education: "Becausewe live in such a highly mobile society, educators and counselors should be aware of the problems faced by their mobile students and be ready to help those **who** find more difficulty in adjusting to **new** school settings" (p. 165).

Commentators in the 1980s and 1990s would hesitate to use words such as "extreme", "handicaps" and "deprivation" in relation to itinerancy. These underlying preconceptions nevertheless continue to characterise the tenor of most contemporary writings on Traveller education. For example, King-Stoops (1980), in presenting her list of "Goalsfor Migrant [Itinerant] Education" (p. 16), referred to the children of seasonal fruit pickers in the United States of America, and related the statement of Carlos, one such child, that he did not want to become a lettuce picker like his father. According to King-Stoops, "Neitherdoes the school want to see Carlos end up as a lettuce picker. The school's ultimate goal for Carlos is to have him gut as much intellectual and social distance between himself and the lettucefields as he is able" (p. 16). This aim contrasts starkly with the determination of the Queensland show people to maximise their children's access to formal education without threatening the continuation of their cultural traditions. Such a contrast reflects the point that, while this kind of literature can be useful, it fails to address the broader questions with which I am concerned regarding culturally produced understandings of marginalisation and disadvantage, and the ways in which marginalising effects can be challenged without the abandonment of a particular lifestyle.

An equally marked contrast was evident in an account of Operation SAIL ("Students Assimilated Into Learning") (Panagos, Holmes, Thurman, Yard & Spaner, 1981,p. 452), an innovative orientation program for new students in a Missouri suburban school district. The program's three components included

teacher professional development, parental orientation and involvement, and student development in cognitive and affective domains (p. 453). On the one hand, the program was described very positively, and was shown to have "significantgains within the academic domains" (p. 463) (although the gains in the affective domains were found to be "minimal" [p. 463]). On the other hand, the students involved in the program, who had been selected on the basis of having migrated from the inner city areas and who therefore exhibited limited patterns of itinerancy, were constructed as "new students with educational deficits [who] can be remediated and assimilated into the mainstream of their new school district" (p. 467). In other words, the wellconceived educational program was judged by its effectiveness in remedying the students' "educational deficits". That those "deficits" were constructed as linking itinerancy (even in a limited form) to poverty, low socio-economic status and people of colour reflects many of the preconceptions of the literature on Traveller education (and confirms the synthesis cited earlier by Lucassen and his colleagues [1998] of the "two closely connected paradigm" [p. 2] that have perpetuated negative stereotypes about itinerant people for centuries).

In the same way, Blair, Marchant and Medway (1984) reported on a program that they had developed for mobile military parents to assist their children to assimilate into new schools. Generalising from the effectiveness of that program, the authors concluded, in terms that again linked itinerancy with "schoolproblem" that school personnel must work to "reduce":

By recognizing the needs of families in transition, counselors and psychologists in schools can broaden their roles to serve a needed function, namely, the provision of planned consultation and training services designed to reduce the number of children exhibiting school problems because of relocation. (p. 258)

Taking up some of the themes of the *Plowden report* (Plowden, 1967) in the United Kingdom, *Educationfor all* (Swann, 1985)included Travellers in a report devoted to the education of children from ethnic minority groups in that country, prompted by concerns by West Indian parents about their children's education (p. vii). Fairground and circus children were excluded from the section of the report dealing with Travellers because of lack of resources rather than because they did not conform to the criterion of "ethnicminority groups" (p. 748, note 11). However, *Educationfor all* included a 1983 discussion paper from the English Department of Education and Science that emphasised that "Fairground and circus children experience particular problems inmaintaining continuity of education, because their families move so frequently" (p. 759). The correlation in this construction is clear: the itinerant parents are the cause of their children's educational "problems', which the Department is thereby required to address.

Rahmani (1985) developed the concept of "turbulence" to describe the harmful effects on the children of service personnel of continually transferring from one school to another. Particular problems have been identified as including parental perceptions of inconsistencies in State based educational delivery (Duffy, 1986), and teachers' and itinerant students' preconceptions

about one another and the children's schooling experiences (Mills, 1985, 1986a, 1986b, 1986c).

All of this indicates that there is a great deal of emphasis almost exclusively on what is 'wrong' with itinerancy. Indeed, the usually unspoken assumption and starting point for these accounts is that itinerancy is inherently problematic and scandalous. This false assumption creates a scholarly 'blind spot': literature predicated on this assumption fails to address, or even to acknowledge, the issue of what is 'wrong' with a 'mainstream' education that can represent itinerancy only in negative terms of 'deviance' and 'lack'. This thesis, by contrast, seeks to highlight that 'blind spot' and to turn the spotlight onto 'mainstream' education that routinely marginalises the education of Travellers but that potentially can help to make Traveller education transformative and enabling.

In a similar vein to other studies that described education programs in terms of 'remedying' the 'educational deficit' involved in itinerancy, Harrington (1987) listed some of the elements of that 'deficit': "Migrant [itinerant] children are among the most vulnerable in America's classrooms. Theirs is a history of poverty, mobility, cultural alienation, and low expectations becoming self-fulfilling prophecies" (p. 37). Although he acknowledged that itinerant students' strengths included "resiliency, resourcefulness, and responsiveness" (p. 38), Harrington's overriding construction of itinerancy as educationally harmful was encapsulated in his pleasure at recording one fruit picking family's ambitions to graduate from high school and attend college: "Alex and his siblings will be bucking the statistics, which show that about 90 percent of migrant [itinerant] kidsfollow

their parents into the fields. But kids like Alex are the success stories that remind all educators what is possible" (p. 39). For Harrington, "success stories" and itinerancy were mutually exclusive phenomena.

A group of specialists in itinerant education at a UNESCO sponsored conference in 1989 asserted:

The education of the children of nomadic, migrant and itinerant groups has been considered a major problem needing the attention and consideration of governments and national and international organisations. It is the firm belief that, without a prompt consideration of this problem, at a time when other children of sedentary groups are advancing in science and technology, the children of mobile groups will be marginalised for lack of adequate preparation to cope with the changes of the future. This will create an impediment to them and to others. (United Nations Educational, Scientific and Cultural Organisation, 1989, paragraph 53)

Miller and Cherry (1991) presented a series of strategies designed to assist itinerant students to adjust to their new schools. In doing so, they acknowledged that itinerancy influenced those students in different ways, and that more intelligent students often found itinerancy an intellectually stimulating experience (pp. 15 and 51). Nevertheless, the dominant discourse in their publication constructed itinerancy as 'stressful' and 'difficult' and as creating educational challenges that schools must address. For example, "Frequentmoves do not reduce the stress associated with relocation" (p. 17), "Childrenwho move about are at risk socially and emotionally" (p. 51) and "The special needs of these mobile children demands schools and

parents accept the responsibility for addressing the needs" (p. 52). Once again 'blame' for the 'problem' was directed at itinerant families, rather than at educational systems, even though schools were identified as carrying the burden of ameliorating the educational side effects of this undesirable lifestyle.

Miller and Cherry drew a particularly bleak picture in relation to the educational opportunities of the children of itinerant fruit picking families: "Migratorychildrenface challenges and stresses unique to their nomadic way of life" (p. 13). The authors also noted that "Althoughthese youngsters pass from state to state, they acquire little knowledge of the places they pass through since most of the travel is done at night" (p. 14). By contrast, the Queensland show families travel mostly during the day, and many of them comment on the educational uses to which this travel is put.

Despite their complaint about the dearth of research into itinerant education, Birch, Lally and Tomlinson (1986, p. 1) felt sufficiently emboldened to hypothesise that "frequentchanges of schools will be disruptive to the social and intellectual development of children". Their own preliminary research (1986, p. 21) suggested that itinerancy "may have a cumulative negative effect on academic achievement".

Lally's (1993, pp. 201-202) summary of the study in which he had earlier been involved (Birch, Lally & Tomlinson, 1986) drew a bleak, even depressing, picture of itinerant education:

- 1. Itinerancy poses a world-wide educational problem. The research evidence is conclusive in suggesting that children in itinerant families are disadvantaged educationally, compared with stable-resident family populations. Furthermore, it is the younger child who is most adversely [affected]. . . . Hence not only is there a special problem faced by the itinerant child, it is one which has to be addressed at the earliest possible age, in the pre-school and early school years.
- 2. Parents of itinerant families may themselves have come from itinerant families and, f so, it is also likely that they may have had limited access to post-school educational provisions. . .
- 3. It is symptomatic of the attention paid to itinerant families that Australian statistics are not available to determine their number. . .
- **4.** Itinerancy is not merely afeature of the tyranny of distance such as Australia experiences. It is also afeature of the major metropolitan centres which contain most of the country's population. The caravan parks and other evidence of mobility support this contention.

Itinerancy is also a problem of international importance and not one peculiar to Australia. Mobility and distance are synonymous with the educational disadvantage of children.

The last point in Lally's list echoed an earlier finding by Smith, Husbands and Street (1969). These authors argued that "pupilmobility has significant retarding influences upon intellectual achievement among slum children", and that this relationship between mobility and achievement "is greater before third grade than afterwards" (p. 269). The authors claimed to

draw on "anextensive tradition of research pointing to the fact that mobility itself has basically disruptive properties" (p. 270), which could be considered to underscore the marginalised status of itinerant students.

Further discussion of the negative connotations of educational itinerancy was included in Binns' (1990) account of recent attempts to expand the provision of Traveller education in the United Kingdom, in a clause of the Education Reform Act of 1988. He alleged that Traveller children were at least potentially liable to greater rather than reduced marginalisation, contrary to the original intention of the policy makers, because local management of schools increased pressure to discourage school attendance by lower achieving students at schools competing for government grants. Similarly, with the introduction of the National Curriculum:

Teachers will find an extra difficulty in having to cater for pupils who have missed out on large areas of school experience. . . . A sudden influx of numbers of unschooled, unskilled pupils could completely disrupt the teacher's planned progress through the levels of the Key Stages for the class. (1990, p. 257)

Binns (1990) acknowledged that both local management of schools and the introduction of the National Curriculum also had the potential to assist the educational provision for Traveller children. The point to emphasise here is that Traveller education is portrayed as unstable and subject to the uncertainties — if not the vagaries — of broader government policy and educational change. An associated assumption by the educational system is that itinerant students are automatically disposed to be less academically successful than

their permanently resident peers — clearly a supremely unhelpful assumption if the aim is to disrupt and reduce the marginalisation of itinerant people.

A recent Australian commentator on itinerancy (Fields, 1995, 1997) echoed most of the themes identified in this subsection. For example, he claimed in one article, "Thefindings of this study are strongly indicative of mobility having adverse effects on both the social and academic achievement of young adolescents" (1995, p. 30). Similarly, in another paper he noted, "It is widely believed that children who change schools frequently are adversely affected by the experience. Research on the effects of student mobility seems to support this belief" (1997, p. 47). Fields acknowledged that "Therehave been contrary findings, however, and this has led some researchers to describe the link between mobility and school adjustment as 'inconsistent'... "inconclusive"..." (1997, p. 47). Despite this qualification, he concluded his article with the following comment about the Australian literature: "What is missing is a broad recognition of the significance of the problem [of itinerancy] as a social and educational issue. Such recognition should be the impetus for a far greater resolve to do something about it" (1997, p. 53). In many ways very little has changed in the perceptions of itinerancy from the studies in the 1960s and 1970s reported at the beginning of this subsection.

I have not intended in this subsection to suggest that itinerancy is a universally happy and educationally valuable experience. On the contrary, in the data analysis chapters many show people refer to the difficulties of occupational travelling. My intention has been to highlight constructions in the literature of itinerancy as inherently educationally disabling, creating

'deficits' and 'gaps' that formal education must work hard to address and remedy. I have sought to relate these constructions to a deeply embedded stereotype that views itinerants **as** deviating from accepted social 'norms' associated with permanent residence. As the data analysis chapters demonstrate, it is precisely this marginalising stereotype that the Queensland show people actively resist and seek to transform into a more enabling and satisfying form of educational provision. It follows that this is a section of the literature – and a major and in some ways still dominant one at that – from which I depart in this thesis; indeed, I pose a very different framework for looking at the issues of itinerancy and Traveller education.

<u>2.3.3</u> Some exceptional studies

Despite the prevalence of marginalising stereotypes about itinerant people identified above, a small but growing number of publications has departed from the pervasive view of Traveller education **as** inherently disadvantageous to participants. From the perspective of this thesis, these publications have contributed to a necessary resistance and intended transformation of the marginalisation too often associated with itinerancy and Traveller education.

The earliest example that I encountered was written by an American primary school principal (Evans, 1966), who deduced from itinerant and non-itinerant students' scores on achievement tests that itinerant students performed consistently, but not significantly, more effectively in arithmetic, reading, science and social science than their non-itinerant peers. Arguing that "mobility does not have an adverse effect upon the academic achievement of

those students who have experienced it" (p. 22), Evans's concluding words make a salutary contrast to the stereotypes perpetuated by considerably more recent studies:

If moving must be considered a "handicap," as we have traditionally thought it is, then this study shows definite ability on the part of the mobile students to adjust. The resiliency so evident in the bones of the youngster seems to manifest itself in his [sic passim] personality, and thus he is able to bounce back from the "handicap" and achieve acceptably. (p. 22)

Barrett and Noble (1973) applied a questionnaire and the Louisville Behavior Check List to one hundred and fifty-nine families in Louisville in the United States of America to establish whether mothers' fears about the impact of long distance moves on their children were realised. The researchers found that "The results of this study suggest that anxiety about negative effects of moving on the emotional adjustment of children represented by this sample were largely unfounded" (p. 187). Furthermore, emphasising social agency in a similar way to this thesis, the researchers concluded:

Within the limits of this study, it is our view that the long distance move should be laid to rest as a specific variable in children's disorders. It seems more appropriate to us for families who face a move to focus on their adaptive strategies rather than to seek out ways to avoid "stress." (p. 188)

Continuing this theme of resisting and seeking to transform the marginalising stereotype of itinerancy, while conducting research into British Traveller education, Reiss (1975,p. 2) felt

...increasing apprehension about the almost universal view expressed by teachers that Travellers and their children were suffering from acute verbal deprivation which was, in many cases, seen as an insurmountable obstacle to educational progress.

With tongue firmly in cheek, Reiss (1975,p. 3) described the discipline of education in the early 1970s:

There was more interest than ever before in the world's distinctive minority groups. . . . The ubiquitous European Gipsies and travellers, the fairground and circus showmen, and the bargees of Europe could be looked upon as another fascinating area for action and research. The days of concentration on the 'normal'child were over.

Presenting a summary of his findings, Reiss (1975, p. 8) cautioned against the automatic designation of itinerant people as "disadvantaged":

Though Travellers often reveal classic symptoms of severe social and cultural deprivation, they cannot easily be placed within the general spectrum of the disadvantaged. Their unique and fascinating case presents a very real challenge to teachers and administrators.

Continuing this cautionary tone, Lacey and Blane (1979)concluded from their meta-analysis of studies of itinerancy: The simple assumption. ...that more geographic mobility necessarily means an impairment of academic attainment can be shown to be erroneous. The direct effect of mobility is likely to be small and be itself affected by the social context and reasons for mobility. (p. 200)

Lacey and Blane (1979) warned in particular against generalising to all itinerant groups from the various studies of children of military personnel, on the grounds that to do so "ignores the complex interrelationship of social class factors and educational attainment and the confounding variables within the mobility matrix" (p. 205).

Brett (1982) discussed a study of three hundred and fifty United States families whose jobs were highly mobile, compared with three samples of non-mobile people. Her interest lay in discerning the relationship between the participants' mobility and their sense of wellbeing. Her study found that mobile people were generally similar to non-mobile people in relation to the work, self, marriage and family life, and standard of living dimensions of their wellbeing. On the other hand, "the only data that consistently separated the mobile sample from the comparison samples" (p. 460) pertained to a higher level of dissatisfaction with social relationships for both mobile adults and children. Brett's conclusion was accordingly positive about the link between mobility and wellbeing:

Few families in the transfer sample believed moving is easy. However, the data from this study show that despite their mobility, these families were as satisfied with all aspects of their lives, except social relationships, as were stable families. (p. 462)

Talung a positive view of what others have perceived **as** the drawbacks of Traveller education, Pullin (1985), himself a member of a longstanding Yorkshire fairground family, cited a British school principal whose school was attended by show children:

The Travellers are a breath of fresh air in the school. With their supportive home backgrounds, they are courteous, co-operative, hardworking and want to learn. They know who they are, what they are and where they are going. (p. 2; emphasis in original)

Pullin (1985) commented about this statement, "Thiswill certainly have an important bearing on formulating appropriate aims and methods in providing an appropriate education for these children" (p. 2). In other words, the positive attributes of itinerancy, such as the show children's social maturity and personal confidence, need to be enhanced rather than eroded by the schooling provided for them. The alternative is the situation described by one of Pullin's informants: "Because of my people's wandering about most af the year, our children don't get the education they should" (p. 1).

Pullin's (1985) set of case studies began with "a success story" that in his view "convincingly illustrates what an individual can achieve, given determination and support from people in a position to advise and assist others to achieve their educational potential and ambitions" (p. 4). This "success story" concerned a young woman who left school early but whose thirst for academic knowledge led her, against considerable odds, to return to secondary school, complete a degree in economics and begin a preservice teacher education course. Pullin's comment on this "success story" reflected

his construction of itinerancy as not inherently disabling but rather as the potential site of the exercise of individual agency:

Whether or not she returns to follow her traditional life of a travelling showman [sic] following the completion of her university studies or pursues a career outside showland is insignificant. She will have realised her personal potential and ambition and can choose to return or not return as a free agent. She is now able to make a free, rational choice, the most important gift of education. (p. 6; emphasis in original)

The then Queensland Department of Education Northern Region (1992) provided a somewhat grudging acknowledgment that itinerancy was not necessarily intrinsically disadvantageous to participants: "There are both positive and negative aspects to moving, and care should be taken to ensure that mobility is not always seen as a 'problem' (Book 1,p. 10).

Brown and Orthner (1990) studied the experiences of itinerancy of seven hundred and twenty early adolescents from five communities in the United States of America, in an effort to trace the connection between mobility and personal wellbeing (measured in terms of self-esteem, alienation, depression and life satisfaction) in this age group. They were particularly interested in the gender dimension of this connection. Contrary to much of the literature on mobility and adolescence, Brown and Orthner found that the only significantly lower measure was mobile girls' general life satisfaction, which led the researchers to hypothesise "thatfemales may require more time to make substantial adaptation to relocations and that high mobility rates may inhibit that capability" (p. 378). On the other hand, the researchers' overall conclusion conveyed a positive view of itinerancy: "The present study's

findings that substantial changes in self-worth were not found to result from moving may offer some hope to parents and others who are concerned with dramatic psychological effects following a work-induced family relocation" (p. 378).

Lee and Warren (1991) explored the sophisticated alternative conception of education developed by the Romanis or Gypsies, a group often compared with occupational Travellers such as show people, and with its own substantial literature that is outside the provenance of this thesis. This educational conception is derived from Sophistic and Socratic traditions and the ideas of Georges Sorel and quantitative instrumentalism (whereby "The Romani is quite happy to be trained. But this is always and everywhere only to the extent that this training reinforces Romani life — [p. 3 19; emphasis in original]), and is based on a crucial distinction between education and schooling. Lee and Warren's concluding note (1991, p. 322) was infectiously admiring:

Perhaps a boast expressed by the Romanis – but one with a large element of truth and one which few Gaje [non-Romani] would feel confident to make – captures the value they place on their own education system, and is a fitting note on which to end: "You could put me down anywhere in the world and I could make a living".

Máirín Kenny's **book** The,routes of resistance: Travellers and second-level schooling (1997) provided an admirable of educational provision for Irish Travellers in one school. Kenny's approach, drawing on theorists such **as** Bourdieu, Giddens and Giroux, was to examine the posited links between the Travellers' ethnicity and their practices of resistance in the educational setting that she investigated. In the process, she sought to shift the focus of

attention from "theall too commonly discussed 'Travellerproblem'" to what she termed "theprovider problem" (p. 7). In doing so, Kenny consciously eschewed "[s]ub-culture ofpoverty theory" in favour of "ethnicitytheory" as "themost power-l framework to date for analysis of issues relating to Irish Travellers" (p. 59), and, with clear parallels to the approach taken in this study, she focussed on how Irish Traveller children's identity (including their ethnic identity) animated their resistance—and potentially their transformation—of marginalising schooling practices.

In an exemplary demonstration of teacher-as-researcher and reflective practitioner, Cathy Kiddle, Coordinator of the Devon County Council Traveller Education Consortium in the United Kingdom, has drawn on her life experiences as an educator of both fairground children and Gypsy Travellers to highlight the erroneousness of 'deficit' assumptions about Travellers. In *Travelling children: A voice for themselves* (Kiddle, 1999), she traced negative educational experiences to the prevailing discriminatory stereotypes ascribed to Travellers, yet she also demonstrated the mutual benefits accruing from positive educational experiences based on reciprocal trust and understanding (a theme that she continued in Kiddle, 2000). (A similar approach was taken by Elizabeth Jordan [2000] with her research into Scottish Traveller families, and also by Ursula Scholten [2000] about the education of Dutch bargee people.) Earlier (Kiddle, 1981) she had used her account of living in a caravan to work for a touring theatre company to emphasise the fundamental link between itinerancy and educational provision – orlack thereof.

Although the education of nomadic pastoralists and fisherpeople is outside the province of this study, it is pertinent to note the work of the Nigerian National Commission for Nomadic Education. Established in 1990, the Commission has carried out an effective lobbying of the Nigerian government to enhance educational provision for those two groups. The Commission distributes a biannual magazine called *Nomadic Education* News, and in 1998 published the inaugural issue of a planned annual journal entitled Journal of Nomadic Studies. In addition, the Commission has also published monographs about nomadic education in Nigeria (Ezewu & Tahir, 1997; Tahir, 1991; Tahir & Muhammad, 1998) and more broadly in Africa (Tahir, 1997). Consciously eschewing a 'deficit' approach to conceptualising nomadism, the Commission locates nomadic education in the context of the culture, economics, history and politics of nomadic people and in the broader perspective of a developing country that continues to position itself as a postcolonial nation. A similarly socially critical approach underpins the work of Caroline Dyer (2000; Dyer & Choksi, 1998) in her research into Indian nomadic pastoralists, and also Bernadette Robinson's (1999) study of the education of Mongolian nomadic pastoralists.

Another group of itinerant people who are outside the parameters of this thesis are the Gypsy Travellers. Nevertheless it is important to note the pioneering work of the Gypsy Research Centre in Paris, which was established at about the same time as the European Federation for the Education of the Children of the Occupational Travellers, which I discuss below. Under the leadership of Jean-Pierre Liégeois, the Gypsy Research Centre has lobbied the European Commission to help to improve the life chances of European Gypsy Travellers, including the enhancement of educational

provision, partly through its regular publication *Interface* and partly through monographs that it has published or sponsored (Advisory Council for the Education of Romany and Other Travellers, 1993; Liégeois, 1998). The dominant discourse underpinning the Centre's activities has been the previous and ongoing discrimination against European Gypsy Travellers and the denial of their equal human rights.

The resistant and transformative potential of studies such as those promoted by the Gypsy Research Centre, and indeed such as this thesis aspires to be, was realised in a recent publication by the Leeds Travellers Education Service (Saunders, Clarke, Kendall, Lee, Lee & Matthews, 2000). Entitled *Gypsies and Travellers in their own words: Words and pictures of travelling* life, the **book** featured lively, amusing and often moving accounts by Gypsy Travellers from the Leeds area of England of their lives and their educational experiences. The movement from being 'written out of' official and academic discourse to producing their own text demonstrates the power of constantly circulating counternarratives to the traditional 'stories' about itinerant people.

A particularly significant contribution to the literature on Gypsy Travellers, with strong resonances with this thesis, was McVeigh's (1997) effort to engage in what he called "[t]heorising sedentarism", whereby "the roots of anti-nomadism" (p. 7) could be laid bare. McVeigh defined sedentarism "as that system of ideas and practices which serves to normalise and reproduce sedentary modes of existence and pathologise and repress nomadic modes of existence" (p. 9). This definition sets in sharp focus, and helps to explain, the various negative assumptions and stereotypes underpin-

ning many conventional studies of Traveller education. McVeigh's analysis has the virtue of a historical grounding that demonstrates that these assumptions and stereotypes have a very lengthy provenance, having been evident in different ways in the transition to agriculture, the fall of the Roman Empire and the move to industrialisation. The contemporary marginalisation of Australian show people elaborated in Chapter Five thus has a much older heritage than the beginning of British colonisation in Australia in 1788.

McVeigh's (1997) analysis is helpful also in strengthening and putting in a broader historical and sociological perspective the two logical extremes of 'unproblematic othering' and 'unproblematic celebration' posited below. Following McVeigh, the major point to emphasise here is that both these extremes are equally destructive and antithetical to accurate representations and understandings of itinerant people. According to McVeigh:

It is wrong to use notions which reproduce the dichotomy between 'good' Travellers (ethnic, exotic, romantic, free) and bad travellers (non-ethnic, dispossessed and debased sedentaries, subcultures of poverty). Infact, the suggested dichotomy between the construction of the romanticised 'RaggleTaggle Gypsy' and the pathologised 'itinerant' is a false one. Both simultaneously inform Contemporary ideas about, and the treatment of, all nomadic peoples. (p. 15; emphasis in original)

Furthermore, McVeigh (1997) argued that, as well as both these "romanticised" and "pathologised" constructions of itinerant people (which parallel respectively the 'unproblematic celebration' and 'unproblematic othering' identified at the end of this section of the chapter) underpinning "contemporaryideas about, and the treatment of" those people, those same

constructions feed equally ineluctably and deleteriously into ideas about what should be 'done about' those people.

Thus both 'liberal' and 'reactionary' sedentarisms have posited 'final solutions' to the 'problem of nomads' which actively seek their annihilation. Because of this the historical and contemporary treatment of nomads should not be dichotomised as repressive extermination versus sympathetic assimilation. Rather both approaches have been part of a complex dialectic committed to a 'final solution' to the 'problemof nomads'. (p. 22)

While some observers might find McVeigh's (1997) reference to "a 'final solution" extreme, such a reference emphasises the seriousness and the significance of the kind of analysis undertaken in this thesis. It also reflects the extremely high 'stakes' involved in Traveller education, whereby inappropriate schooling provision can shade all too readily from "sympathetic assimilation" into "repressive extermination" — or from "sympathetic incorporation" into "unsympathetic repression" as McVeigh also named these processes (p. 23). Certainly such inappropriate provision contributes crucially to the ongoing marginalisation of Australian show people, as Chapter Five demonstrates.

The European Federation for the Education of the Children of the Occupational Travellers (EFECOT) was established in 1988(the year before the beginning of the specialised education program for the Queensland show children). Based in Brussels, Belgium, EFECOT is financially supported by various funds provided by the European Commission and uses its connections with other bureaucrats to promote the educational interests of European

occupational Travellers. Its four target groups are fairground children, circus children, bargee children (who live on the barges that ply their trade in Europe's inland waterways) and the children of seasonal fruit pickers (added latterly).

The reason for examining selected EFECOT publications in this literature review is twofold. Firstly, these publications provide a stark contrast to most of the literature on Traveller education that constructs itinerancy as a 'problem' to be 'solved'. EFECOT's approach is generally to insist that the occupational Travellers' itinerant lifestyle is an intrinsically valuable and worthwhile set of experiences, and that educational systems need to change to make specialised and appropriate provision for these people.

Secondly, EFECOT has established a very effective lobbying base to promote the interests of European occupational Travellers. Its congresses are attended by ministers for education of several European countries and high level officials in the European Commission. It has forged close links with leading practitioners of Traveller education in such countries as the Netherlands and the United Kingdom. While Australian show people still experience disruptions to the educational provision for their children when they cross state boundaries, EFECOT has contributed to a reduction in disruption to European occupational Travellers who routinely cross international borders as part of their employment.

In addition to its regular journal *Newsline*, many of the EFECOT publications are administrative, such **as** work programs and annual reports, and developmental, such **as** the strategy to evaluate the TOPILOT project ("To OPtimize the Individual Learning process of Occupational

Travellers"), whereby telematics technology using CD-i were provided to travelling families, initially in four subject areas in Belgium, Germany, the Netherlands and the United Kingdom (Botke & Willems, 1996; Marks & Pullin, 1996). Other publications consist of handbooks for teachers of travelling students, and reports of the outcomes of meetings convened by EFECOT (see for example Bernaert, 1997; Knaepkens, Van Ryckeghem & Verheyen, 1993; Pullin, 1994).

The other major category of EFECOT publications is surveys of particular groups of occupational Travellers, sometimes in specific countries. Representative examples of such publications include *The education of fairground children in the European Community* (Knaepkens, 1989), *Survey: The number, age and geographical distribution of children of show and circusfamilies in the Republic of Ireland* (Magee, 1992) and *The secondary education of circus and fairground children: Addressing discontinuity of learning through information provision and distance learning: Report on the United Kingdom Project A4(B7)* (The European Federation for the Education of the Children of the Occupational Travellers, 1994).

These publications are typically detailed, painstaking and thorough. They are also almost entirely empirically grounded, with no explicit theoretical framework guiding the collection and analysis of the data that they report. Furthermore, the orientation of the research is almost exclusively quantitative, with an abundance of statistics and an absence of voices of individual occupational Travellers. This is perfectly appropriate, given the bureaucratic environment in which EFECOT operates, where the explication of outcomes and the measurement of achievement of those outcomes are integral elements

of EFECOT's interactions with other agencies in the European Commission. This work also makes a valuable contribution to research into Traveller education and itinerancy more generally, by providing information about such matters **as** the number of itinerants and the extent of their itinerancy.

On the basis of the stereotypical and marginalising literature analysed in the previous subsection of this chapter, it would seem that little has changed in the thirty-four years since Morris, Pestaner and Nelson (1967) observed: "Perhapsmobility studies generally have suffered not from deficient but from absent theoretical basis" (p. 78). The exceptional studies canvassed in this subsection, however, indicate that there is a growing recognition of the need to move beyond marginalisation and to conduct research in ways that will resist and hopefully transform the easy assumption that itinerancy is 'deviant' or 'problematic'.

It is certainly that spirit of resistance and hopeful transformation that animates the ongoing and growing contribution to the literature on Traveller education by my colleagues and myself at Central Queensland University. A representative list of our publications appears just before Chapter One of this thesis, and includes the edited book Beyond the ferris wheel: Educating Queensland show children (Danaher, 1998a) and more recently the editing of a theme issue of the International Journal of Educational Research pertaining to Traveller and nomadic education in several different countries (Danaher, 2000b). Although we would certainly not claim to be free at the outset of the research in 1992 from many of the negative assumptions and stereotypes attending the education of itinerant people, since then we have concentrated increasingly on articulating and disrupting those assumptions and stereotypes,

which are central and crucial to the continuing marginalisation of the Australian show people. While this thesis is separate from that research project, it is consistent with, and both extends and makes appropriate use of concepts and theories deployed through, that work.

The main findings of this selective review of research into Traveller education are **as** follows:

- Relatively little has been published about Traveller education, in comparison with other areas of educational research. This highlights the crucial point that, just as itinerant people move 'across' physical spaces, their itinerancy has rendered them largely invisible to academic attention and recognition. This invisibility reinforces the related and equally vital point that where itinerancy occurs, marginalisation follows close behind. This thesis is intended to contribute both to redressing that invisibility and to demonstrating that that attribution of marginalisation is neither appropriate nor inevitable.
- Most studies are written with the assumption that itinerancy differs from the 'norm' of fixed residence and schooling, thereby creating inherent educational 'problems' for travelling students. As I noted above, this might be termed an 'unproblematic othering' of itinerant people, by constructing them as automatically 'other' to 'normal' people and hence as the cause of the educational 'problem' of schooling for Travellers. This thesis is avowedly founded in opposition to this 'unproblematic othering', which lies at the heart of the marginalisation of itinerant people and which stands squarely between them and equitable access to educational services.

- A small number of more 'enlightened' studies either recognises the heterogeneity of itinerancy or values the diversity of experiences and lifestyles that make up itinerancy. As shall become clear, this thesis is conceived as contributing to and extending that valuing, partly by representing itinerancy as the site of potential transformation and positive outcomes for itinerant people.
- At the same time, despite the few excellent studies noted above, there is a possible tendency for some more positive representations of itinerancy to display what might be termed the 'unproblematic celebration' of this distinctive lifestyle. This is the logical opposite of 'unproblematic othering': by highlighting the exotic 'difference' of itinerant people, such studies might suggest that itinerancy is a uniformly easy and enjoyable set of experiences. (In a sense 'unproblematic celebration' parallels the Romantic notion of the 'noble savage' as 'unproblematic othering' parallels the stereotype of the 'ignoble savage': both reflect equally distorted and deleterious images of Indigenous peoples or in this case of occupational Travellers.) This thesis takes issue with that kind of construction, which in certain respects is as suspect and dangerous as 'unproblematic othering', because it diminishes the struggles routinely faced by itinerant people and ignores the vital point that those struggles arise because itinerancy is devalued in comparison with sedentarism.

The literature on Traveller education – with some noteworthy exceptions – therefore has some major deficiencies. In particular, the correlation of itinerancy with 'educational deficit' is likely to replicate the marginalisation of

travelling students that these same studies ostensibly seek to ameliorate. **As** I have written elsewhere (Danaher, 2000c):

Certainly a major issue of concern is the resilience of the deficit model that constructs itinerancy as different, and deviant, from the norm of settled residence, with the corollary assumption that the education of itinerant people is inherently a problem needing "remediation" or a "solution." (p. 224)

This "deficit model" constructs Traveller education as a 'problem' produced from the perspective of those who occupy the official 'place' of education (a point that I elaborate in the next chapter).

By contrast, this thesis constructs the itinerancy of the Queensland show people as creating opportunities for distinctive educational experiences for participants, and for the assertion of a powerful cultural identity in educational domains, thereby promoting resistance and transformation and a counternarrative about itinerancy and Traveller education. I regard this as a highly significant contribution to the literature on Traveller education.

2.4 Review of the chapter

This chapter has presented a critical review of two sets of literature:

- Australian shows
- Traveller education.

In each set, the intention was to evaluate the strengths and limitations of selected existing literature and to suggest entry points whereby this thesis could contribute significantly to knowledge.

With regard to Australian shows, the review demonstrated that these cultural forms have a long history and are associated with various experiences of itinerancy. That same itinerancy was identified **as** the source of considerable and enduring ambivalence in the relations between itinerant and local people: a sense of identification and of having a shared purpose in the activities arranged around a show was offset by mutual suspicion and misunderstanding. The ongoing and deleterious effect of that ambivalence was sustained marginalisation of show people and their lifestyle. This thesis analyses that marginalisation **as** it characterises and influences the interactions between the Queensland show people and others with whom they have regular contact, including the staff members of the Brisbane School of Distance Education.

That same marginalisation was held to characterise much of the literature on Traveller education. On the one hand, many studies characterise Traveller education as a positive response to the specialised educational needs of occupational Travellers such as the Queensland show people. On the other hand, this characterisation often derives from a construction of itinerancy as 'deficit', 'different' and 'disabling'. Notable exceptions to this general trend recognised itinerancy as a valid lifestyle and as a source of resistance and potential transformation of the marginalising dominant discourse. This thesis provides an intensive examination of one site in which itinerancy engages with these three enduring themes of marginalisation, resistance and transformation.

In doing so, it presents a counternarrative to traditional understandings of itinerant people and their education, and it contributes significantly to addressing the gaps and silences identified above in the existing literature.

To elaborate on the significance of that contribution, I return to a point that I made at the end of the previous section of this chapter. There I posited two logical extremes in the literature on Traveller education, which I termed an 'unproblematic othering' and an 'unproblematic celebration' of itinerancy. The former constructs itinerancy as a 'problem' and a 'deviation' from the 'norm' of sedentarism, and accordingly tends to 'blame the victim', by suggesting that itinerant people cause the 'problem' of lack of educational access by their perverse determination on living an 'unnatural' lifestyle. The latter constructs itinerancy as a set of exotic and exciting experiences; it displays a fascination with the 'difference' of an 'unusual' lifestyle. Yet it is equally deficient in understanding itinerancy, which is still conceived through a frame of 'difference' – albeit a more positively valued 'difference' – from the 'norm' of sedentarism. (Clearly the exceptional studies that I identified in the last part of the previous section do not engage in this 'unproblematic celebration', and they certainly eschew 'unproblematic othering'.)

My final point in this chapter is that both these logical extremes are seriously inadequate for achieving a genuine understanding of itinerancy and of Traveller education. Injuxtaposition, they suggest an 'either/or' dichotomy in the field: 'either' itinerant people automatically have negative experiences, 'or' they experience uniform and unalloyed excitement and pleasure, on account of their lifestyle (yet both poles of the dichotomy contribute directly to the ongoing marginalisation of itinerant people). This thesis rejects

completely that dichotomy, and posits instead a 'both/and' approach to understanding itinerancy and Traveller education. Such an approach accepts that itinerancy, like any manifestation of the human condition, entails both positive and negative experiences and possibilities. What is important is an acceptance of the dynamism and fluidity of those experiences and possibilities, in keeping with the physical mobility of itinerancy. This approach has the crucial advantage of moving beyond the twin conceptual perils of fixed marginalisation and superficial exoticisation. Instead, the recognition of the deep and enduring links between itinerancy and marginalisation is leavened by an awareness of the possibility of resistance and transformation of that marginalisation. Even more significantly, this approach allows for a fuller appreciation not only of the education of the specific group considered here but also of 'disadvantaged' groups more generally.

I turn in the next chapter to present and justify this thesis's conceptual framework. Through its application to the show children's education program, that framework demonstrates its viability in significantly extending understandings of itinerancy. In the process, a more dynamic and fluid comprehension of Traveller education, informed by this study of 'learning on the run', can be developed.

CHAPTER THREE

THE CONCEPTUAL FRAMEWORK

"Reading... seems to constitute the maximal development of the passivity assumed to characterize the consumer, who is conceived of as a voyeur (whether troglodytic or itinerant) in a 'showbiz society'."

"In reality, the activity of reading has on the contrary all the characteristics of a silent production: the drift across the page, the metamorphosis of the text effected by the wandering eyes of the reader, the improvisation and expectation of meanings inferred from a few words, leaps over written spaces in an ephemeral dance."

de Certeau, 1984, p. xxi

"In order to understand, it is immensely important for the person who understands to be located outside the object of his or her creative understanding—intime, in space, in culture. For one cannot even really see one's own exterior and comprehend it as a whole, and no mirrors or photographs can help; our real exterior can be seen and understood only by other people, because they are located outside us in space and because they are others."

Bakhtin, 1986a, p. 7

3.1 Overview of the chapter

To this point, I have identified the problem with which this thesis is concerned as being what the Queensland show children's educational provision reveals about the intersection of education and marginalisation, resistance and transformation, as well as about broader issues in Australian Traveller education. The research questions guiding the analysis of the materials gathered to address this problem focus on three dimensions of the show people's lives arising from their itinerancy: their marginalisation, and their resistance and eventual transformation of that marginalised status. The literature reviewed in Chapter Two centred on two topics: Australian shows; and Traveller education. This study was posited as a counternarrative or 'alternative story' to traditional and debilitating stereotypes about itinerant people, particularly Australian show people, and as thereby contributing significantly to, and in some cases departing radically from, the existing literature.

The function of this chapter is to delineate the main elements in the thesis's conceptual framework, and in the process to suggest ways of conceptualising a reinvigorated Traveller education. In doing so, I seek to move beyond the either/or positions identified in the previous chapter: positions within which show people are commonly constructed as passive 'others' or de Certolian exotics. The chapter consists of three sections:

- Michel de Certeau's concept of 'tactics of consumption'
- Mikhail Bakhtin's notions of 'outsidedness' and 'creative understanding'

• a synthesis of the conceptual framework guiding this thesis.

At this point I should emphasise that the selection and synthesis of these conceptual resources is neither accidental nor incidental. On the contrary, I demonstrate throughout this chapter that the conceptual framework has been chosen precisely because of its capacity to frame and inform my responses to the research questions cited above. From this perspective, the conceptual framework is an integral element of a coherent and consistent approach to understanding the educational experiences and opportunities of Queensland show people.

Specifically, I argued at the end of the previous chapter that this study is located directly in the 'middle ground' between two equally unhelpful 'logical extremes' evident in the literature about itinerant people and Traveller education. Those two extremes were identified as an 'unproblematic othering' and an 'unproblematic celebration' of itinerancy, whereby itinerancy is constructed respectively as inherently 'deviant' and 'lacking' and as superficially exciting and 'special'. In eschewing both these constructions, I contended that itinerancy is 'both/and': both a valid and valuable lifestyle and subject to marginalisation that still admits of resistance and possible transformation. To support the study's location in this 'middle ground', I need to deploy conceptual resources that in combination enable me to examine and understand three crucial facets of itinerant people's lives:

- their marginalised status
- their capacity for resistance of that marginalisation

 their potential for transformation of that marginalisation and that resistance.

To elaborate this point, what binds the disparate elements of the conceptual framework together is the proposition that itinerancy involves distinctive relationships between people and space. It is this distinctive set of connections that explains the show people's sense of marginalisation, on account of their appearing to have no 'place' of their own. It explains also their efforts to resist and transform that marginalisation, by working to change the 'spaces' of itinerancy into 'places' that they cancall 'home'.

Specifically, the argument underpinning this chapter – and this thesis – proceeds **as** follows

- The Queensland show people's itinerancy creates opportunities for them to turn the multiple spaces that they enter and leave into the 'spaces' that de Certeau (1984, 1986) envisaged **as** the sites of consumption, and thereby potentially of subversion, of their marginalised status.
- The show people's itinerancy provides opportunities for them to engage the outsidedness and extend the creative understanding (Bakhtin, 1986a) of 'outsiders' to the show circuits, thereby maximising the prospect that 'the rules of the game' that marginalise itinerancy can be subverted and transformed.

It is these bonds among the elements of this conceptual framework that justify the selection and application in a single site of these particular and separate conceptual resources, and that support and strengthen the thesis's goal of presenting a counternarrative about the show people and their education that is theoretically grounded and significant in itself.

3.2 De Certeau's concept of 'tactics of consumption'

Michel de Certeau's contribution to the conceptual framework framing this thesis is his concept of 'tactics of consumption'. This section of the chapter consists of three subsections:

- The practice of everyday life (1984)
- *Heterologies: Discourse on the other* (1986)
- applications and critiques of de Certeau's work.

The section is by no means intended as a comprehensive analysis of de Certeau's ideas, but rather as a selective focus on the 'tactics of consumption' and associated ideas that constitute the first major element of the study's conceptual framework.

Throughout the following discussion, the emphasis will be on how my appropriation of de Certeau's concept of 'tactics of consumption' will enable me to explain and theorise the show people's experiences of marginalisation and their efforts to resist and sometimes to transform that marginalisation. This will help to justify my claim that the thesis constitutes a counternarrative to traditional 'stories' about itinerant people and how they should be

educated. At the same time, I shall point out that an important limitation of de Certeau's approach is the inability to change 'the rules of the game' of marginalisation and subversion, which justifies my drawing also on Bakhtin's notions of 'outsidedness' and 'creative understanding'.

3.2.1 The practice of everyday life (1984)

De Certeau's research included forays into history, literary studies, cultural studies and psychoanalysis. Some of these varied disciplines were represented in de Certeau's book *The practice of everyday life* (1984), which was first published in 1974 and which was the key text for his elaboration of the concept of 'tactics of consumption'. Many elements of the argument of this book were packed densely into de Certeau's dedication, which is accordingly worth citing in full.

To the ordinary man [sic passim].

To a common hero, an ubiquitous character, walking in countless thousands on the streets. In invoking here at the outset of my narratives the absentfigure who provides both their beginning and their necessity, I inquire into the desire whose impossible object he represents. What are we asking this oracle whose voice is almost indistinguishable from the rumble of history to license us, to authorise us to say, when we dedicate to him the writing that one formerly offered in praise of the gods or the inspiring muses?

This anonymous hero is very ancient. He is the murmuring voice of societies. In all ages, he comes before texts. He does not expect representations. He squats now at the centre of **our** scientific stages. The floodlights have moved away from the actors who possess proper names and social blazons, turning first towards the **chorus** of secondary characters, then settling on the mass of the audience. The increasingly sociological and anthropological perspective of inquiry privileges the anonymous and the everyday in which zoom lenses cut out metonymic details - parts taken for the whole. Slowly the representations that formerly symbolised families, groups and orders disappear from the stage they dominated during the epoch of the name. We witness the advent of the number. It comes along with democracy, the large city, administration, cybernetics. It is aflexible and continuous mass, woven tight like a fabric with neither rips nor darned patches, a multitude of *quantified heroes who lose names and faces* **as** *they become the ciphered* river of the streets, a mobile language of computations and rationalities that belong to no one. (p. v)

Despite the excess of masculine pronouns, this passage *can* be read **as** reflecting de Certeau's desire to focus academic attention on 'ordinary people'—the bit players, not the star actors, or those who are continually and routinely positioned at the **margins** rather than in the text itself. In the process, it was the actions and interactions of active social players that concerned him, rather than the operation of artificial forces or deterministic influences. Immediately we have a clear sense of de Certeau's celebration of human

agency on the part of all social participants, regardless of their degree of formal or overt power.¹

Another of de Certeau's contentions, that researchers should concern themselves with the micro more than with the macro level of events, was elaborated in his preface to the English translation of *The practice of everyday life* (1984).

For what I really wish to work out is a science of singularity; that is to say, a science of the relationship that links everyday pursuits to particular circumstances. And only in the local network of labour and recreation can we grasp how, within a grid of socio-economic constraints, these pursuits unfailingly establish relational tactics (a struggle for life), artistic creations (an aesthetic) and autonomous

1

I discuss below my response to **this** kind of critique of de Certeau, arguing that the degree of human agency implicit in de Certeau's analysis is preferable to the less agential analyses against which he reacted, but that his view of **social** life does not readily account for changing 'the rules of the game' confronting such groups **as** the Queensland show people.

Ahearne (1995) expressed "reservation" about this dedication to the 'ordinary man' because "it is easy to forget that he does not exist" (p. 187). In a typically 'double edged' comment that contained both praise and criticism, Ahearne asserted that de Certeau's

^{...}invocation of this fantasmatic, impossible figure (his other), needs to be read with half an eye on other rhetorical invocations of the term (by the Moral Majority, by the tabloid press, by every kind of politician). Only then could Certeau's arresting reemployment of the term be read less as ballast to his writing than as a powerful perspectival displacementup on the operations of learned analysis. (p. 187)

initiatives (an ethic). The characteristically subtle logic of these "ordinary" activities comes to light only in the details. And hence it seems to me that this analysis, as its bond to another culture is rendered more explicit, will only be assisted in leading readers to uncover for themselves, in their own situation, their own tactics, their own creations and their own initiatives. (p. ix)

Here de Certeau linked the idea of human agency that animated his dedication to a single name – "tactics" – that encapsulated and operationalised agency, particularly on the part of the less powerful and strong. That is, although all of us possess and deploy our "owntactics", it is in situations where we are subordinate or vulnerable that we need to call on those specific resources.

Much of the "General introduction" to the book (pp. xi-xxiv) set out in programmatic form the theoretical framework guiding de Certeau's empirical studies. He began by eschewing individuality (the elementary unit of the individual to which social groups can always be reduced). He was more interested in the "modes of operation or schemata of action" than in "their authors or vehicles" (p. xi). He continued:

The purpose of this work is to make explicit the systems of operational combination (les combinatoires d'opérations) which also compose a "culture", and to bring to light the models of action characteristic of users whose status as the dominated element in society (a status that does not mean that they are either passive or docile) is concealed by the euphemistic term 'consumers'. Everyday life invents itself by poaching in

countless ways on the property of others. (pp. xi-xii; emphasis in original)

In this passage, de Certeau introduced the terms "users" and "consumers" in the context of "culture", and his proposition that these groups, despite their status "as the dominated element in society", are in fact neither "passive" nor "docile". Instead, they take part in "poaching in countless ways on the property of others". In certain respects the show people engage in "poaching" on the spaces that they enter as a consequence of their itinerancy, as a prelude to their transformation of those spaces from marginalising to enabling territories for them.

"Consumer production"

Having introduced the argument that human agency is manifested through tactics, particularly by less powerful individuals and groups, and that those tactics are deployed by consumers, who in doing so resist the application of labels such as 'helpless' and 'passive', de Certeau divided his theoretical framework into two categories: "Consumerproduction" (p. xii) and "The tactics of practice" (p. xvii). With regard to the first category, he elaborated three "positive determinations" informing his research into "consumer production". The first was "usage, or consumption" (p. xii): in addition to the representations of a society and its modes of behaviour, he believed that attention should be directed at "theuse to which they are put by groups or individuals" (p. xii). He drew on a favourite example to elaborate this notion:

For instance, the ambiguity that subverted from within the Spanish colonisers' 'success' in imposing their own culture on the indigenous Indians is well known. Submissive, and even consenting to their subjection, the Indians nevertheless often made of the rituals, representations and laws imposed on them something quite different from what their conquerors had in mind; they subverted them not by rejecting or altering them, but by using them with respect to ends and references foreign to the system they had no choice but to accept. They were other within the very colonisation that outwardly assimilated them; their use of the dominant social order deflected its power, which they lacked the mans to challenge; they escaped it without leaving it. The strength of their difference lay inprocedures of 'consumption'.(p. xiii; emphasis in original)²

De Certeau's use of the term "other" in this passage differed crucially from its usual employment to denote a marginalised and generally powerless deviation from 'the norm'. Here "other" contains and nurtures agency, whereby the South American "Indians" were enabled to make use of the Spanish invaders' "rituals, representations and laws" in ways that gave

In view of the 'personalnote' with which I concluded Chapter One, it is appropriate to point out that elements of de Certeau's argument underpinned an earlier thesis in Australian history (Danaher, 1991), without my being aware of this at the time, in my analysis of how the Darnley Islanders in the Torres Strait 'made use of' and 'consumed' the meanings and values of Christianity, 'civilisation' and colonisation brought to the Strait by the London Missionary Society in 1871.

them meaning and purpose in their own terms, rather than in the conquerors' terms. "Consumption" is central to this process by virtue of its association with both subversion of domination and marginalisation and the recognition and enactment of difference. From this perspective, it is clear why de Certeau insisted that consumption is an active, not a passive, performance. For the same reason, I conceive of the show people's consumption of their educational experiences and opportunities as avowedly active and as being directed at resisting their marginalised status.

De Certeau's second determining concept of "consumerproduction" (p. xii) was "theprocedures af everyday creativity". Here he showed both where he drew on and where he departed from Foucault's book Discipline and punish (1979). On the one hand, he concurred that

...the goal is to perceive and analyse the microbe-like operations proliferating within technocratic structures and deflecting their functioning by means of a multitude of 'tactics' articulated in the details of everyday life. ..(p. xiv)

On the other hand:

...the goal is not to make clearer how the violence of order is transmuted into a disciplinary technology, but rather to bring to light the clandestine forms taken by the dispersed, tactical and makeshift creativity of groups or individuals already caught in the nets of 'discipline'. (pp. xiv-xv)

Two important attributes of 'tactics' are revealed in these two passages. Firstly, 'tactics' are pervasive and multifarious, "a multitude" that is "articulated in the details of everyday life". Secondly, 'tactics' are both "clandestine" and immersed in the "creativityof groups of individuals". In other words, a central reason for the success of 'tactics' in subverting marginalised status is that they are chameleon-like, evading attention because of the sheer number of their manifestations and working surreptitiously, 'behind the scenes', away from the spotlight of official notice.

De Certeau called the third organising principle of "consumerproduction" (p. xii) "theformal structure of practice" (p. xv). He contended that, even though they occur in particular contexts and lacked their own "ideologies or institutions" (p. xv), ways of usage and procedures of everydaycreativity "conformto certain rules" based on a particular logic. To discern this logic, he suggested a combination of two techniques: conducting a number of descriptive and empirical investigations; and reading the scientific literature (particularly as it relates to rituals and networks, ordinary language, and formal logics and analytical philosophy). This thesis is conceived as both drawing on a detailed "descriptiveand empirical" study and contributing to "thescientific literature", in order to delineate "theformal structure of practice" in one setting of Traveller education for occupational Travellers.

His elaboration of these three determinants led de Certeau to posit a number of ironical statements about "themarginality of a majority".

Marginality is today no longer limited to minority groups, but is rather massive and pervasive; this cultural activity of the non-producers of culture, an activity that is unsigned, unreadable and unsymbolised,

remains the only one possible **for** all those who nevertheless buy and pay for the showy products through which a productivist economy articulates itself. Marginality is becoming universal. **A** marginal group has now become a silent majority. (p. xvii)

For de Certeau, "a marginal group" was by no means homogeneous. "Socialsituations and power relationships" (p. xvii) come into play to make its members' readings of and responses to ideas and images different and complex. Hence:

...the necessity of differentiating both the 'actions' or 'engagements' (in the military sense) that the system of products effects within the consumer grid, and the various kinds of room to manœuvre left for consumers by the situations in which they exercise their 'art'. (p. xvii)

The relevance of this passage to the argument being advanced in this thesis is twofold. Firstly, the passage reinforces de Certeau's depiction of 'tactics' as taking many and varied forms, in response to "the system of products" operating "within the consumer grid". Secondly, the implied reference to 'tactics' as taking up the "room to manœuvre left for consumers" evokes the proposition that 'tactics' are the means by which variously marginalised individuals and groups are enabled to 'consume' and 'make use of' the spaces in which they are located and/or from which they are denied access. This is signified by de Certeau's statement that "our research has concentrated above all on the uses of space, on the ways offrequenting or dwelling in a place..." (p. xxii).

The culmination of de Certeau's account of "consumerproduction" (p. xii) was its overt politicisation. Given the conflicts, tensions and balances inherent in culture, de Certeau asserted baldly: "The tactics of consumption, the ingenious ways in which the weak make use of the strong, thus lend a political dimension to everyday practices" (p. xvii). This observation was the logical extension of an explanation of social life that highlighted both ways in which power is exercised and – more importantly from the perspective of this thesis – ways in which agency is deployed to resist, subvert and have the potential to transform that power into something more collectively palatable.

To this point, I have examined de Certeau's discussion of "consumer production". Rather than being an oxymoron, this term has emerged as depicting and explaining the exercise of human agency by the less powerful through their deployment of 'tactics' of subversion and particularly of consumption, thereby asserting the validity and value of their difference from 'the mainstream' while minimising the deleterious effects of their divergence from 'the norm'. These 'tactics of consumption' are clandestine and creative, and they demonstrate that labels such as 'marginalised group' and 'minority group' are floating signifiers rather than fixed essences. These 'tactics of consumption' therefore reflect a view of social life that is at once politicised and agential. They are also indispensable to my efforts to conceptualise the marginalisation and the resistance that I argued above are equally central elements of the show people's itinerant lifestyle.

"The tactics of practice"

In addition to "consumerproduction" (p. xii), de Certeau's other major category in his conceptual framework was what he termed "thetactics of practice" (p. xvii). Significantly, he envisaged the main constituents of "the tactics ofpractice" as representing something of a 'corrective' to what he considered "the rather too neatly dichotomized. . .relations between consumers and the mechanisms of production" (p. xvii), so that "theoverly schematic character of the general statement can be somewhat nuanced" (p. xviii). I shall return to this point in a later subsection; here it is sufficient to note that de Certeau himself prefigured some of the criticisms of his conceptual framework as tending to excessive polarisation between more powerful producers and less powerful consumers, and that he sought to counter that tendency in his theorising about "thetactics of practice".

In the process of elaborating the elements of "a problematics that could articulate the material collected" in his research (p. xvii) – "a problematics" that resonates with the 'unproblematic othering' and 'celebration' identified at the end of the previous chapter – de Certeau identified three key concepts: trajectories, tactics (incorporating strategies, place and space) and rhetorics. Firstly, the "trajectories" of consumers, who "moveabout" in "technocratically constructed, written, and functionalized space", "form unforeseeable sentences, partly unreadable paths across a space", and they "traceout the ruses at other interests and desires that are neither determined nor captured by the systems in which they develop" (p. xviii).

So trajectories are an analytical tool that assist researchers to identify and examine the effects of individuals' and groups' 'tactics of consumption'. De Certeau's view of trajectories as performing this function was consistent with his argument that tactics are not easily detected, surveyed or measured. For example, he rejected "statistical investigation" (such as that which characterised some of the literature critiqued in Chapter Two) as being too detached and generalised to be able to portray consumers' trajectories with sufficient flexibility to record "the bricolage (the artisan-like inventiveness) and the discursiveness" (p. xviii) that make up social life. Similarly, de Certeau asserted the importance of avoiding the tendency of trajectories to reduce "acts" to "a tracing", on account of a trajectory's 'planeprojection, a flattening out" (pp. xviii-xix). From this perspective, while I certainly aspire to present "atracing" of the show people's 'tactics of consumption', I equally want to avoid "a flattening out" in my account of their itinerant lifestyle, seeking on the contrary to emphasise continually the dynamism, multiplicity and variability of that lifestyle.

For his second conceptual element of "the tactics of practice", de Certeau elaborated his construction of 'tactics' by proposing a distinction between them and 'strategies' based on the kind of activity involved, who carries it out and for what purposes.

I call a 'strategy' the calculus of force-relationships which becomes possible when a subject of will and power (aproprietor, an enterprise, a city, a scientific institution) can be isolated from an 'environment'. A strategy assumes a place that can be circumscribed as proper. . . and thus serve as the basis for generating relations with an exterior distinct

from it (competitors, adversaries, 'clienteles', 'targets' or 'objects' of research). Political, economic and scientific rationality has been constructed on this strategic model.

I call a 'tactic', on the other hand, a calculus which cannot count on a 'proper' (a spatial or institutional localisation), nor thus on a borderline distinguishing the other as a visible totality. The place of a tactic belongs to the other. A tactic insinuates itself into the other'splace, fragmentarily, without taking it over in its entirety, without being able to keep it at a distance. It has at its disposal no base where it can capitalise on its advantages, prepare its expansions and secure independence with respect to circumstances. The 'proper' is a victory of space over time. On the contrary, because it does not have a place, a tactic depends on time – it is always on the watchfor opportunities that must be seized 'on the wing'. Whatever it wins, it does not keep. It must constantly manipulate events in order to turn them into 'opportunities'. The weak must continually turn to their own endsforces alien to them.

...But these tactics. ...also show the extent to which intelligence is inseparable from the everyday struggles and pleasures that it articulates. Strategies, in contrast, conceal beneath objective calculations their connection with the power that sustains them from within the stronghold of its own 'proper'place or institution. (pp. xix-xx)

This lengthy pair of definitions **skirts** close to the same tendency to polarisation for which de Certeau theorised "the tactics of practice" – although, as I shall discuss later in this subsection, his distinction between 'tactics' and 'strategies' is more dynamic and less rigid than some of his

that he assigns to these two terms. Strategies are associated with "asubject of will and power"; they are enacted in "a place that can be circumscribed as proper" (that is, as officially sanctioned and valued); they are persistent and enduring; they emanate from "the power that sustains them from within the stronghold of its own 'proper' place or institution". By contrast, tactics are associated with "the weak"; they have themselves "no base", so they must insinuate themselves "into the other's place"; they are fleeting and temporary; they reveal the "intelligence" that "is inseparable from the everyday struggles and pleasures that it articulates".

This set of comparisons applies and clarifies de Certeau's earlier account of "consumer production". Thus there is evidence of agency being operationalised by "the weak" through their clandestine and creative tactics as 'users' or 'consumers' of the 'place' of the 'proper', thereby subverting their marginalised positioning by subjects "of will and power". The clarification derives from de Certeau's employment of striking images to depict the respective actions and aspirations of those using 'tactics' and those using 'strategies'. From the perspective of this thesis, the introduction of 'strategies' into de Certeau's theoretical framework provides the basis for my response to the first research question in Chapter Five: that is, it equips me to analyse and explain the show people's experiences of marginalisation arising from the fact of their itinerancy, in a very different way from the 'deficit model' that lies at the heart of the 'unproblematic othering' of itinerant people cited at the end of the previous chapter.

As well as providing a conceptual basis for addressing the first research question, de Certeau's focus on 'strategies' does three other things: it reminds us that the more powerful also have agency (exercised through 'strategies'); it emphasises the integral dynamism of that framework, with the seemingly eternal interactions between 'tactics' and 'strategies' (thereby strengthening de Certeau's position against charges of excessive polarisation, as I discuss in a later subsection); and it introduces another crucial distinction devised by de Certeau – that between 'place' and 'space'.

Afield (lieu) is the order (of whatever kind) in accord with which elements are distributed in relationships of coexistence. It thus excludes the possibility of two things being in the same location (place). The law at the 'proper' rules in the place: the elements taken into consideration are beside one another, each situated in its own 'proper' and distinct location, a location it defines. A place is thus an instantaneous configuration of positions. It implies an indication of stability.

A space exists when one takes into consideration vectors of direction, velocities and time variables. Thus space is composed of intersections of mobile elements. It is in a sense actuated by the ensemble of movements deployed within it. Space occurs as the effect produced by the operations that orient it, situate it, temporalise it and make it function in a polyvalent unity of conflictual programs or contractual proximities. On this view, in relation to place, space is like the word when it is spoken, that is, when it is caught in the ambiguity of an actualisation, transformed into a term dependent upon many different conventions, situated as the act of a present (or of a time), and modified

by the transformations caused by successive contexts. In contradistinction to the place, it has thus none of the univocity or stability of a 'proper'. (p. 117)

Earlier de Certeau asserted that a tactic has "atits disposal **nu** base" and that it therefore "insinuatesitself into the other's place" (p. xix). Here his conceptualisation of 'space' was **as** a physical manifestation of that process of 'insinuation'. That is, the 'space' is the outcome of the process whereby a 'place' has been entered and occupied – however incompletely and temporarily – by a tactic.

De Certeau elaborated this argument by emphasising the contrasts between 'place' and 'space'. On the one hand, 'place' is where "thelaw of the 'proper' rules"; it is "an instantaneous configuration of positions" because of the tendency to clarity and order of "the law of the 'proper"; accordingly it "implies an indication of stability". On the other hand, 'space' is "composed of intersections of mobile elements", rather than being the site of operation of a single "law"; it is "ina sense actuated by the ensemble of movements deployed within it"; and it "has thus none of the univocity or stability of a 'proper' [place]". Laws, in other words, seldom if ever exist in their perfect state.

De Certeau crystallised the distinction between 'place' and 'space' by asserting: "Inshort, space is a practised place " (p. 117). By this he meant that, just as "inrelation to place, space is like the word when it is spoken", "the street geometrically defined by urban planning is transformed into a space by walkers" and "an act of reading is the space produced by the practice of a particular place". All these examples have in common the

proposition that officially designated, sanctioned and valued 'places' are entered, changed and transformed into 'spaces' through tactical consumption of those 'places'.

The significance of de Certeau's proposition about 'place' and 'space' for this thesis's conceptual framework is profound. Firstly, in contradiction to the prevailing assumption in the distance education literature that distance education students and their teachers are located in separate and fixed spaces, taken literally de Certeau's understanding of 'space' is much closer to the situation experienced by itinerant people who travel through different spaces. Thus his references to the "intersections of mobile elements" that constitute space and "the ensemble of movements deployed within it" are almost 'literal metaphors' for the daily lifestyle of itinerants such as the Queensland show people. Given that de Certeau's work operated at the level of poetics and metaphors, it is important to explicate this kind of 'literal metaphor' or semantic connection with this study.

Secondly, de Certeau argued that 'space' is both "polyvalent" and unstable. This is because space is, like a spoken word, ambiguous, "situated" in present acts and "modified...by successive contexts". This means that the interactions between show people and other people that occur when they cross multiple spaces are complex, influenced by multiple contexts and redolent of multiple meanings and values. So space is the site of intersection among competing discourses about such matters as what a 'home' is, the benefits and drawbacks of an itinerant lifestyle and the educational rights and responsibilities of itinerant people—all of which are themes that I explore at length in the data analysis chapters of this thesis.

Thirdly, 'space' is "practised". This point recalls that de Certeau's elaboration of the distinction between 'tactics' and 'strategies', which prompted a discussion of his contrast between 'place' and 'space', was a key element of his articulation of "the tactics of practice". This point is also a timely reminder that the spaces crossed by itinerants are not elements in a hypothetical discussion or an academic thesis – they are sites of meaning making for real, live, thinking and feeling human beings whose lifestyle deviates from 'the norm' of settled residence. This gives purpose and relevance to this study of how one group of itinerants 'uses' and 'consumes' the 'places' that they encounter and thereby transforms them into 'spaces' that have significance for them.

So de Certeau's contribution to my developing conceptualisation of 'space' lies in his insistence that space is mobile, polyvalent, practised and unstable, and the site of the deployment of 'tactics of consumption'. De Certeau's insights support a reading whereby people such as itinerants, through deploying 'tactics of consumption', enter and transform the 'place' of the 'proper' into a 'space' that reverts to being the 'place' of the 'proper' once the tactics are no longer deployed. This conceptualisation is both complex and dynamic — and in the process in keeping with the multiple elements of life on the Queensland show circuits, including marginalisation, resistance and some aspects of transformation.

A brief digression at this point into the contemporary academic and popular obsession of globalisation serves to illustrate my understanding of de Certeau's distinction between 'place' and 'space'. Terry Evans (1997)

captured both sets of experiences underlying these elements of social life when he noted:

...both history and the present show that people view such interchanges [associated with globalisation] across their borders as far from benign, and rather as intrusions, invasions and incursions into their spaces, places and cultures. As I write and you read — events distant in space and time — people will be resisting, protesting and fighting against previous or prospective incursions into their territories and cultures. In this sense, some of the elements which comprise what we call 'globalisation' are the sources of battle, and not of 'development', or especially pleasure. (p. 7)

De Certeau's concepts help me to analyse the "sources" and the effects "of battle" engaged in by the show people as they seek to transform "their spaces" into 'places' of their own.

De Certeau envisaged as the third conceptual element of "thetactics of practice" "the discipline of rhetoric", or "the science of the 'ways of speaking" (p. xx).

The discipline of rhetoric offers models for diferentiating among the types of tactics[;]...rhetoric. ..offers an array of figure-types for the analysis of everyday ways of acting even though such analysis is in theory excluded from scientific discourse. Two logics of action (the one tactical, the other strategic) arise from these two facets of practising language. In the space of a language (sin that of games), a society

makes more explicit the formal rules of action and the operations that differentiate them. (p. xx)

De Certeau's analysis of 'rhetoric' as one of "the tactics of practice" reinforced his argument that an artifact of social life – in this case, a word or statement – is not inherently either 'tactical' or 'strategic' and is in fact an 'empty signifier'. Its deployment by 'real' people with varied intentions and motivations is what brings it into 'practice', and thereby gives it tactical or strategic effect. This point derives from de Certeau's account of the same territory being transformed from a 'place' to a 'space' through the operation of 'tactics'. (It can be argued that a similar point can be made about education, that it too is an artifact and is therefore simultaneously the site of marginalisation and potential transformation.) This is a timely reminder of the need to avoid essentialising 'tactics of consumption' as a unidimensional and stable phenomenon. To do so would be to misrepresent de Certeau's understanding of 'tactics of consumption' as complex, fleeting and unstable. This point has also a strong resonance with the argument advanced in the previous chapter: that itinerancy needs to be understand as 'both/and' enabling and challenging, a resonance that highlights the conceptual cornpatability between that argument and my selection of 'tactics of consumption' as a major theoretical resource in this study.

The bulk of *The practice of everyday life* was devoted to applying de Certeau's conceptual framework, outlined in this and the previous subsection, to various phenomena of everyday life. Examples included "ordinary language" (p. 1), "popular cultures" (p. 15), story telling, pedestrians in city

streets, train travel, writing, reading, persuading others of the correctness of one's viewpoint and conventions in dealing with dying people.

De Certeau's elaboration of "consumerproduction" and "the tactics of practice" in the "General introduction" to The practice of everyday life constitutes the major source for the theorising about 'tactics of consumption' on which this thesis draws. From this elaboration, 'tactics of consumption' emerge as reflecting human agency, subverting domination and asserting difference, demonstrating the fluidity of terms like 'marginalised group' and 'minority group', occurring in 'trajectories', interacting with strategies, transforming 'places' into 'spaces' and being exemplified in the shifting practices of 'rhetoric'. In combination, these conceptual resources suggest that the spaces through which itinerants travel are politicised, temporary and unstable, and the sites of interaction between tactics and strategies and of the transformation of marginalising places into subverting and enabling spaces. This combined conceptual weight is both consistent with, and central to, the 'both/and' approach to understanding itinerancy being advocated in this thesis.

<u>3.2.2</u> <u>Heterologies: Discourse on the other (1986)</u>

Many of the ideas in *Thepractice of everyday life* were taken up in another of de Certeau's influential **books**, *Heterologies: Discourse on the other* (1986). There he applied notions such as tactics, strategies and rhetorics to particular historical and literary cases. Examples included the development of psychoanalysis (in which the repressed past returned surreptitiously to the present);

Freudian ideas applied to literary analysis; selected texts by Montaigne; mysticism in the sixteenth and seventeenth centuries; works by Jules Verne and Alexandre Dumas; a critique of Foucault's books *The order of things* and *Discipline andpunish*; and links among history, science and fiction.

The use to which I wish to put *Heterologies* in this thesis is as a single example of a demonstration of the conceptual resources outlined in *The practice of everyday life*. This example is taken from the final chapter of *Heterologies*, entitled "*The politics of silence: The long march of the Indians*". This chapter depicted the continuing resistance to European colonialism by the "*Indians*" of South and Central America, in countries such as Brazil, Mexico and Panama.

De Certeau began the chapter by pondering the intentions that were likely to motivate the Indians, and consequently the activities in which they were likely to engage in order to achieve those intentions.

The actions the Indians take are directed less toward the construction of a common ideology than toward the "organization" (a word-leitmotif) of tactics and operations. In this context, the political relevance of the geo-graphical distinctions between separate places is echoed. . in the distribution of places of power. . . (p. 227; emphasis in original)

Here de Certeau argued that the Indians' rejection of a common "ideology" among their disparate groups – Tan (19%) referred to "about 200 Indian ethnic groups" making up the subject of de Certeau's discussion (p. 32) – reflected their desire to avoid adopting practices that were similar to those of the conquerors and that were therefore likely to increase, rather than

subvert, their marginalisation. That is, their selection of specific "tactics and operations" was to be based on local contexts and situational analyses, thereby exploiting the transience and flexibility of tactics. Furthermore, the Indians recognised the crucial importance of emphasising, rather than eliding, geographical and political distinctions among "separate places". This reinforced the point that tactics are directed at specific 'places' as a prelude to changing those 'places' into 'spaces'.

The relevance of this passage to this thesis lies in ongoing 'official' efforts to homogenise the Queensland show people by conceiving of them as a single group with certain predictable features and behaviours (in this context, the difficulties of using the label 'show people' in this thesis are considered in Chapter Five). These efforts are opposed by the show people's recognition and celebration of the difference and multiplicity of their identities and their lifestyle, which create opportunities for them to deploy 'tactics of consumption' of such phenomena as official discourses about them and the show children's educational provision to enter official 'places' and make them into 'spaces'. Thus the show people, like the South and Central American Indians, are more interested in exercising their agency through their own 'tactics of consumption' than they are in employing the marginalising strategies of 'the mainstream'.

This point underscores the importance of less powerful groups identifying and being sustained by one or more sources of meaning and even of inspiration. De Certeau related how

because. ...their communities continued to return periodically to the home village, to claim their rights to the land and to maintain, through this collective alliance on a common soil, an anchorage in the particularity of a place. (p. 229)

Furthermore:

The soil. ..enables the resistance to avoid being disseminated in the occupiers' power grid, to avoid being captured by the dominating, interpretive systems of discourse (or by the simple inversion of those discourses, a tactic which remains prisoner to their logic). It "maintains" a difference rooted in an affiliation that is opaque and inaccessible to both violent appropriation and learned coaptation. It is the unspokenfoundation of affirmations that have political meaning to the extent that they are based on a realization of coming from a "different" place (different, not opposite) on the part of those whom the omnipresent conquerors dominate. (p. 229; emphasis in original)

The show people's equivalent of the Indians' "soil" is their proud association with a distinctive contribution to Australian cultural life, centred on their itinerancy. Thus, if 'tactics of consumption' derive from "the particularity of a place", and if resistance to domination involves maintaining "a difference rooted in an affiliation" and "a realization of coming from a 'different' place", that "place" is the physical and symbolic spaces on the show circuits. This is a far more dynamic and fluid understanding of itinerancy than the 'either/or' approach that derives from conceiving of

itinerant people in terms of 'unproblematic othering' or 'unproblematic celebration'.

De Certeau's conviction that 'tactics of consumption' can be very enduring, and that they *can* also contribute to productive social change, was expressed in the following passage:

It is as though the opportunity for a sociopolitical renewal of Western societies were emerging along its fringes, precisely where it has been the most oppressive. Out of what Western societies have held in contempt, combated and believed they had subjugated, there are arising political alternatives and social models which represent, perhaps, the only hope for reversing the massive acceleration and reproduction of totalitarian, homogenising effects generated by the power structures and technology of the West. (p. 231)

Here the less powerful, through exercising agency and deploying tactics to subvert the marginalising strategies, have survived and endured. Moreover, they have developed viable alternatives to "the massive acceleration and reproduction of totalitarian, homogenizing efsects generated by the power structures and technology of the West". That is, the people on 'the margins' have been recognised as having and experiencing something that is superior to 'the mainstream'. Again this demonstrates the conceptual superiority of the 'both/and' to the 'either/or' approaches to understanding itinerancy and how it is enacted in particular sites and contexts.

This capacity of 'tactics of consumption', not merely to endure marginalising strategies, but also to constitute productive alternative forms of social life, relates particularly to the show children's educational provision. Chapter Seven considers in greater detail how that provision encapsulates the possibility of a different, and potentially more enabling, approach to the provision of Traveller education. Here I wish to emphasise de Certeau's insistence that 'tactics of consumption' are not wholly negative or combative, but instead contain the seeds of social transformation.

De Certeau concluded his chapter on "Thepolitics of silence: The long march of the Zndians" by discussing – appropriately for this thesis – the educational dimension of the Indians' interactions with the conquerors. De Certeau argued that "the 'country schools' established thus far" for the Indians were "'a catastrophe", because they had rendered the practices of everyday life "hierarchical, devaluing or crushing difference, and thereby depriving democratic undertakings of cultural landmarks and technical means" (p. 232). De Certeau proposed instead a form of "cultural pluralism" as being "essential to the self-management perspective" of Indians (p. 232), operationalised through such initiatives as teaching traditional medicine and herbalism in schools. The outcome of this "cultural pluralism" would be the establishment of "a space of exchange and sharing" between the Indians and the conquerors (p. 232).

This discussion reinforces two propositions outlined above. Firstly, de Certeau identified education as a site of deployment of 'tactics of consumption' and of the turning of 'places' into 'spaces'. That is, schools function as officially designated centres for publicly approved learning, yet they are also locations for inserting different kinds of knowledge from that that is officially sanctioned, as Chapter Seven will illustrate. Secondly, 'tactics of consumption' are not wholly destructive from the viewpoint of 'the mainstream', but can in fact create opportunities — otherwise not available — for official knowledge to learn from, and to be informed by, alternative understandings of the world.

So this discussion of *Heterologies* has examined de Certeau's application to a single empirical study of the complex conceptual resources outlined in *Thepractice of everyday life*. The discussion has also somewhat extended those resources in three significant respects. Firstly, 'tactics of consumption' are used to resist the elision of difference among and within marginalised groups. Secondly, 'tactics of consumption' derive from a definite sense of association with a different place, outside the realm of 'the proper'. Thirdly, 'tactics of consumption' can be used to sustain difference, and even to extend that difference into 'the place' of 'the proper', where it *can* both subvert the marginalising strategies of the more powerful and enrich their own, **as** well **as** the marginalised groups', social practices.

De Certeau's application of these three points to educational contexts recalls this thesis's focus on the show children's educational provision as the site of the operation of 'tactics of consumption' and 'strategies of marginalisation', the location of the celebration and the elision of difference, the territory of the 'spaces' of itinerancy and 'the place' of 'the proper'. All of this serves to justify the selection of 'tactics of consumption' and associated ideas in helping to explain the 'both/and' approach to understanding itinerancy as simultaneously an arena of limitations imposed from outside and

of the possibility of throwing off those restraints. Those concepts assist in moving beyond the ossified and mutually opposed categories of 'unproblematicothering' and 'unproblematic celebration' towards a far more fluid and shifting account of itinerancy. In the process, those concepts are fundamental in contributing to presenting a counternarrative to constructions of Traveller education **as** either 'deviant' or 'special'.

<u>3.2.3</u> Applications and critiques of de Certeau's work

Thus far in this section I have outlined some key de Certolian ideas. I have argued for the particular utility of those ideas in prosecuting my preferred 'both/and' approach to understanding the multifaceted and shifting nature of itinerancy, thereby avoiding falling into the 'either/or' conceptual 'trap' of constructing itinerancy as inherently disadvantaged or superficially exotic. Now I turn to present a necessarily brief overview of, and engagement with, selected applications and critiques of relevant de Certolian concepts. In doing so, I have two **goals** in mind:

- to justify my distinctive deployment of 'tactics of consumption' and associated ideas, through
 - showing how those same ideas have been used by other commentators
 - demonstrating how some criticisms of those ideas have been misconceived from the perspective of the argument presented in this chapter

• to agree with certain other criticisms of de Certeau's concepts and to use that agreement **as** a 'lead in' to my appropriation of two of Bakhtin's notions.

The emphasis throughout this discussion is on explaining what for me are the crucial triple dimensions of itinerancy: marginalisation, resistance and transformation. Relatedly, I seek to develop the best possible conceptual framework for presenting a counternarrative to traditional understandings of itinerancy and Traveller education. The following disagreement with some criticisms of de Certeau, and agreement with other criticisms, are intended to contribute to and endorse that explanation and that counternarrative.

Within that context, a considerable paradox surrounds the large number of applications and critiques that have been made of de Certeau's work. This paradox is that the 'limitation' for which he has been most consistently criticised – his alleged dependence on reductive binary oppositions – is also the source of the most fruitful interpretations and transpositions to other sites of his ideas. This is because the terms and concepts that appear to be rigid binaries nevertheless generate a great deal of fruitful debate. Thus, while some commentators have agonised over the degrees of conceptual clarity of, and the legitimacy of the distinctions between, such paired categories as 'strategies'—'tactics' and 'place'—'space', other writers have happily 'consumed' those same categories by making them the basis of diverse and generally credible analyses of historical and contemporary social life.

This paradox is directly significant for this thesis in three ways. Firstly, given the degree of enduring controversy that de Certeau's conceptual resources have attracted, I feel justified in adopting an approach to those

resources that emphasises their evident relevance and applicability to this thesis but that also concludes by accusing de Certeau of not facilitating a conceptualisation of changing 'the rules of the game'.

Secondly, my own 'consumption' of de Certeau's work is unashamedly pragmatic: the 'validity' of his concepts – and others' accounts of those concepts – will be asserted to the extent that they appear to advance or retard the research project with which this thesis is concerned. In other words, my method for responding to the divergence between de Certeau's thought and critiques of that thought is to seek to apply one or the other alternative explanation to the actions of the Queensland show people in relation to the educational provision established for their children. This is one important way in which this thesis can and does contribute to theoretical knowledge, by applying and endorsing, and/or contesting and moving beyond, the writings of de Certeau and his critics.

Thirdly, an abiding sense of ambivalence, which characterises relations between itinerants and non-itinerants and with which the Traveller education literature is replete, also courses through the applications and critiques of de Certeau's ideas. This ambivalence is due partly to the sheer range of issues on which he commented, a result of which is that a 'de Certeau position' is sometimes sought on a matter on which he himself might have preferred not to comment. This ambivalence is also due partly to the fact that de Certeau's writing style, while generally concise and lucid, sometimes contained nuances and subtleties that did not necessarily 'survive' the translation from his native French into English. Consequently, some critiques of de Certeau's work derive from an uncertainty about whether to 'read' that work literally or

metaphorically, and if the latter about which alternative metaphorical 'reading' to pursue (hence the notion of 'literal metaphors' identified earlier in the chapter).

Briefly, the link between the argument to this point and this element of the conceptual framework is this. Both itinerants and de Certeau travel across spaces that conventionally remain separate and distinct from one another. For itinerants, this crossing of spaces both marginalises them from, and equips them to interact productively with, non-itinerant people. In the case of de Certeau, his thought traversed a very extensive range of interests and issues, and he deployed his own intellectual "re-employments" (Ahearne, 1995, pp. 29-33) of previous ideas and projects in his supplementary traverses. Thus it is not coincidental that some commentators (Chambers, 1993, p. 193; Jenluns, 1992, p. 223) have identified 'the nomad' as a figure for whom de Certeau's work is particularly germane, and that his biographer (Ahearne, 1995) has referred to de Certeau as having an "intellectual itinerary" and as having engaged in "untiring textual, cultural and interlocutory 'travel'" (p. 2). In the course of his "intellectual itinerary", he both offended some commentators' sensibilities and informed the supplementary thinking of others.

As I explain below, the most crucial area of ambivalence surrounding de Certeau's thought is concentrated on the explanatory power, or alternatively the reductionist meaninglessness, of the various paired categories such as 'strategies'-'tactics' and 'place'-'space' that I introduced at the beginning of this section. The key issue in analysing that ambivalence is whether those

3

categories are fixed binary oppositions or fluid conceptual filters.³ I argue that

In this context, Docker's (1994,p. 163) rhetorical question and subsequent commentary are worth noting as an example of the flavour of the discussion in this subsection:

Don't binary distinctions always strain to homogenise opposing terms, effacing difference and complexity? Terms like dominant and resistant are altogether too polarising, too bloc-like, too unitary and unifying, too unsupple as categories. Dominant discourses, if they exist in a contestatory world, are themselves not monolithic and single in character, but multifarious and contradictory. What happens in culture, 'high', radical or popular, is too ideologically 'mixed', too discursively messy, to be easily cast into the binary either/or boxes of domination and resistance.

While I agree with Docker that the potential for those binaries to be essentialised or naturalised should be resisted and subverted, it is important to remember that binaries such as madwoman and white/non-white are extremely powerful influences on social behaviour. Stuart Hall's reference to "thefashionable postmodernist notion of nomadology – the breakdown of everything into everything" (in a question responding to James Clifford's paper [Clifford, 1992, p. 115]) encapsulated the perhaps equally dangerous potential of following the opposite line of thinking to its logical extreme.

Furthermore, Docker's critique here of "binary distinctions", with its implied disagreement with de Certeau's analysis of social life, was actually phrased in terms of his disapproval of Fiske's account of that analysis in his book *Understanding popular culture* (1989). The quotation at the beginning of this footnote was preceded by the following:

A kind of rampant structuralism also pervades in the way oppositions are created. Throughout Understanding Popular Culture we learn that there is in the world a simple binary opposition between two central forces, the dominant ideology and popular culture, the power-bloc and the people, the system and the subordinate, top-down

in most cases – certainly as they apply to this study of Queensland show people – the categories in question are fluid conceptual filters, but that underlying them is a deep seated binary opposition that for me is encapsulated in the phrase 'the rules of the game'. The point to emphasise here is that the 'binary opposition'—'conceptualfilter' question derives from the ambivalence caused by de Certeau's movement across several intellectual spaces. This is perhaps the least immediately apparent, yet possibly the most significant, point of connection between his work and this thesis's conceptual framework. This reinforces my view that ambivalence, whether in relation to itinerancy or with regard to de Certeau's theoretical position, can be both marginalising and resistant, and that it can ultimately lead to transformation. It also demonstrates the value – once again – of a dynamic 'both/and', rather than a fixed 'either/or', construction of itinerancy.

This subsection consists of the following discussion points:

- 'strategies' and 'tactics'
- 'place' and 'space'
- marginalisation and resistance
- change and transformation.

power and 'bottom-uppower'. Are things so limpidly clear? (p. 163)

I have sought to ensure that **both** my understanding and my appropriation **of** de Certeau's thought are not reducible to **such** crudely conceived binary oppositions.

The first two of these discussion points are the paired categories that have generated so much controversy among de Certeau scholars; the final two are more general issues of concern to both those scholars and myself. In all cases, the discussion points are explicated from the perspective of their contribution to the developing conceptual framework of this thesis. This is achieved by a relatively cursory acknowledgment of the approving applications of de Certeau's ideas and a more extensive engagement with critiques of those ideas. This approach rests on my desire to explain and justify my decision to apply selected de Certolian concepts, based on 'tactics of consumption', and also to explain my use of the Bakhtinian notions of 'outsidedness' and 'creative understanding' to augment those concepts in the thesis.

'Strategies' and 'tactics'

In the two preceding subsections, I outlined de Certeau's concept of 'tactics of consumption', explained it in relation to its categorical pair 'strategies' and provided a rationale for malung it a crucial element of this thesis's conceptual framework. That rationale rested on the assertion that this concept is necessary to explain and justify my interpretation of the Queensland show people's 'consumption' of the educational provision designed for their children. In particular, that interpretation centres on the show people using their 'tactics of consumption' to subvert the marginalised spaces of itinerancy and transform them into more enabling and productive spaces.

From this perspective, it is appropriate to note that several **studies** have made explicit use of de Certeau's 'strategies'-'tactics' paired category, thereby demonstrating the category's utility across a wide range of empirical sites and research agendas. Firstly, Garbutcheon Singh and Miller (1995) applied de Certeau's concept of 'tactics' to extend Homi Bhabha's notion of 'mimicry' in their analysis of the Asia Education Foundation's position statement, "Studies of Asia: A statement for Australian schools". Secondly, Blaber (1989) applied de Certeau's (1984) distinction between 'strategies' and 'tactics' to support his critique of 'the picaresque' as a literary genre. In particular, Blaber's likening of the picaresque to de Certeau's concept of 'tactics' prefigures some of my accounts in the data analysis chapters of the Queensland show people's actions:

If the picaresque is the story of a figure that operates within enemy territory taking advantage of opportunities as they arise but not gaining any real power, and if the picaro poaches and makes use of guileful ruses, then the story of the picaresque is also about the art of the weak. It is a narrative emphasising the tactical. (p. 335)

Below I take up the issue of whether tactics are always and inevitably "the art of the weak". For the moment, I want to emphasise the evident transferability of Blaber's analysis here to the Queensland show people, who might therefore be regarded as operating "within enemy territory taking advantage of opportunities as they arise", and malung "use of guileful ruses". The analogy is by no means exact, but it is nevertheless a vivid and evocative metaphor for the show people's own emphases "on the tactical".

Thirdly, with strong parallels to the empirical site explored in this thesis, Jenkins (1992) linked the image of 'the nomad' directly to de Certeau's concept of 'tactics of consumption'. Thus he referred to "De Certeau's emphasis upon the tactical nature of consumption and the nomadic character of the consumer's culture" (p. 223). Jenluns' description of "fan culture" as being "nomadic, ever-expanding, seeming all-encompassing yet, at the same time, permanent, capable of maintaining strong traditions and creating enduring works" (p. 223) has some parallels with the show people's situation, in the sense that the show people are both literally "nomadic" and "capable of maintaining strong traditions and creating enduring works" (although I dissent from Jenkins' criticism of de Certeau that led him to construct North American television fans as being "ever-expanding, seeming all-encompassing" and "permanent"; certainly I do not assign these terms to the show people). This suggests that being a nomad enables one to cross over into others' territories and to return 'home' at will, borrowing from those territories 'materials' (both physical and intellectual) that can be added to one's store of 'materials' at 'home'. This is similar to the argument that itinerants use 'tactics of consumption' to take from various 'places' what they desire and in the process to transform them into temporary 'spaces'. It also highlights the dynamism of itinerancy and the folly of reducing itinerancy to the 'either/or' fixed binaries identified at the end of the previous chapter.

Fourthly, in another clear parallel with this thesis, Chambers (1993) used the image of 'the nomad' engaging in 'tactics of consumption' in his application of de Certeau's ideas in *The practice at everyday life* (1984) to propose a postmodern geography that envisaged "cities without maps"

(1993, p. 188). Chambers argued in favour of studies that were "attentiveto the different histories, nuances and narratives that combine in making up our present" (pp. 192-193). For Chambers, the part played by 'nomads' lay in responding to the effects of "trajectories of interests and desires that are neither necessarily determined nor captured by the system in which they develop" (p. 193).

In this sense, as de Certeau goes on suggestively to underline, we all become nomads, migrating across a system that is too vast to be our own, but in which we are fully involved, translating and transforming bits and elements into local instances of sense. It is this remaking, this transmutation, that makes such texts and languages – the city, cinema, music, culture and the contemporary world – habitable: as though they were a space borrowed for a moment by a transient, an immigrant, a nomad. (p. 193)

For me, the strength of Chambers' analysis here is twofold. On the one hand, his statement that "we all become nomads" suggests points of identification with 'literal nomads' such as the Queensland show people, thereby reinforcing the recurring argument in this thesis that itinerancy is a floating signifier rather than a fixed essence, and that accordingly it generates considerable ambivalence among both people who are 'literal nomads' and people who have fixed residence. On the other hand, his reference to the agency involved in making the "texts and languages" of everyday life "habitable", and his likening of that process to "a space" being "borrowed for a moment by a transient, an immigrant, a nomad", draw appropriate attention to the proposition that the 'space' of itinerancy is fleeting and

transient. Both these points derive from my earlier discussion of de Certeau's concept of 'tactics of consumption', and from my insistence on constructing itinerancy as occupying the 'middle ground' between the mutually opposed categories of 'unproblematic othering' and 'unproblematic celebration'.

Fifthly, four separate authors have made use of this paired category in diverse and suggestive ways. Weinstein (1994), in his analysis of constructions of and responses to Robot World, a tourist museum in Wisconsin in the United States, stated that "Popularcultures are evasions of authority of the kind described by de Certeau (1984)", whose 'poetic descriptions of the tactics involved in ordinary subterfuges against authority powerfully describe the sensibilities of popular culture. . . "(p. 3). Fiske, in his somewhat controversial book Understandingpopular culture (1989), lauded de Certeau as "one of the most sophisticated theorists of the culture and practices of everyday life" (p. 32), and used de Certeau's concept of 'tactics of consumption' as the basis of his analysis of popular culture in late capitalist countries, particularly the **United** States of America (pp. 32-43 passim), a representative example was his analysis of "two secretaries spending their lunch hour browsing through stores with no intention to buy" as a 'tactic of consumption' of which "boutiqueowners" are aware "butare helpless before them" (p. 39).

Similarly, Buchanan (1996a) championed de Certeau's construction of an active subject, based on the "activecomponents" of "boththe tactical and the strategic", which "are infact principles of action" (p. 114). Geoffrey Danaher (1995) provided an insightful and corresponding defence against the charge that de Certeau's 'strategies'—'tactics' and 'place'—'space' distinc-

tions were *de facto* binary oppositions and therefore conceptually impoverished accounts of social life. Danaher rejected that claim, arguing instead:

Practice can be understood to be a tactic of the oppressed to the extent that the oppressor—oppressed binary is deconstructable; the very act of practice moves the participant away from the oppressed **or** coerced position within which a dominantlreceived discourse positions herlhim. (p. 134)

Considered together, these four authors regarded 'tactics' as "powerfully describ[ing] the sensibilities of popular culture" (Weinstein, 1994, p. 3), as "principles of action" (Buchanan, 1996a, p. 114) and as helping to move "the participant away from the oppressed or coerced position within which a dominantlreceived discourse positions herlhim" (Danaher, 1995, p. 134). For these authors, tactics reflect the exercise of human agency in the face of sometimes seemingly overwhelming dominance or oppression. In this context, it is clearly important to acknowledge that institutional power is experienced all the time by people who do not have access to these institutions. For Danaher in particular, it is this agency that 'deconstructs' "the oppressoroppressed binary", suggesting that this "binary" is in fact an analytically useful paired category, not an immutable dichotomy. If it were otherwise, there would be no possibility for resistance or positive social change. Both these points I elaborate below. This is also yet another justification for rejecting the notion of itinerancy as being encapsulated in either of the marginalising extremes of 'unproblematicothering' or 'unproblematic celebration'.

Sixthly, Ahearne, in Michel de Certeau: Interpretation and its other (1995), "thefirst full-length study of Certeau's thought" (p. 1), made several references to the conceptual versatility of 'tactics' that help to justify their inclusion in this thesis's conceptual framework. For example, he said of tactics that "There is nothing retarded or quaintly outdated about such practices. They constitute instead for Certeau constantly mutating responses to constantly mutating distributions of power in successive socioeconomic and symbolic formations" (p. 161). Thus tactics are both completely 'up to date' and "constantly mutating".

Another dimension of 'tactics' identified by Ahearne was that they "mustoperate in an essentially mobile element. Opportunities and ripostes are not offered up as such but must be seized as they pass, set up as moving targets in transit" (p. 164). The literal mobility of the Queensland show people highlights the particular relevance of this conception of the "essentially mobile element" in which 'tactics' are deployed. These various characterisations of 'tactics', as being 'up to date', "constantlymutating.", "transportable" as a theoretical 'model' and operating "in an essentially mobile element", all enhance the relevance of the concept of 'tactics of consumption' to this thesis's conceptual framework.

Seventhly, Leonie Rowan (1991, 1993, 1994), whose considerable influence on my thinking in this thesis has been noted in the list of acknowledgments, has explored several examples of the intersections between strategies of marginalisation and tactics of subversion, particularly in relation to women and to people of colour. In doing so, she has elaborated an understanding of de Certeau's thinking that I find persuasive and challenging.

For example, in her paper "Strategies of marginalization: An overview" (1991), she outlined a number of ways in which members of marginalised groups are routinely effaced from 'proper' recognition and their voices are elided from official discourses. The methodological challenges of her analysis in "Strategies of marginalization: An overview" for this thesis are profound, particularly as they pertain to the connections between the show people and myself as the author of the thesis, and they are discussed in Chapter Four.

In an article entitled "Thepowers of the marginal: Some tactics for subverting the centre" (1993), Rowan made several points that are germane to this thesis's conceptual framework. Her first point was to distinguish between 'conscious' (which she distinguished from "intentional" [p. 70]) and 'unconscious' tactics on the part of marginalised people, on the basis that "to assume that the 'other' act only from unconscious motivations is both naive and presumptive. Such an assumption implies that the marginal are not capable of formulating and implementing conscious oppositional tactics" (p. 69). Indeed, Rowan's paper concentrated on tactics "for the women and people of colour" with whom she was concerned that were "both conscious and overt" (p. 70). This distinction reinforces the proposition that 'tactics' do not equate with powerlessness, and that marginalised people are able to exercise agency through careful reflection on their current and desired situations. This would certainly seem to be the case with the Queensland show people.

Rowan's second point was to explain how the tactic of 'displacement' can be a particularly effective method of resisting and subverting the marginalising strategies with which women and people of colour, among

others, are routinely confronted. Thus, "displacementtactics highlight the unnaturalness of discrimination against difference and work. . .to deconstruct and displace the notion that any binary can be justified" (p. 70). From this perspective, displacement can be both 'conscious' and 'unconscious', but it is almost always 'intentional', in the sense of deriving from the marginalised group's self-awareness. For Rowan, this was "the primary strength of displacement tactics" (p. 71). That is,

...they originate from the place of the 'other' – a place in which the centre has no interest. Thus the margins become tactical positions in which the oppressed gain strength, formulate conscious moments of opposition and work against the marginalizing strategies of the centre, particularly the notion that any truth can be taken as 'real'.(p. 71)

This discussion will be taken up in the data analysis chapters, where I shall demonstrate that a significant intended outcome of the Queensland show people's 'tactics of consumption' is the displacement of the marginalising strategies that construct their itinerancy as 'abnormal' and the spaces of itinerancy as ephemeral and devalued.

Rowan's third point was to highlight the importance of repetition as "one of the keys" to assisting "the marginal [to] work unceasingly to introduce their stories into the places of the centre" (p. 75). This point helps to explain the show people's persistence, in the interviews cited in the data analysis chapters, in telling me 'their side of the story'. Clearly they had a common message to impart about their distinctive lifestyle and its marginalising stereotype, and they employed the tactic of repetition with me — as they did with educational providers and others — to ensure that I heard their 'voice'.

Rowan's fourth point was to depart from de Certeau's (1984, p. 37) characterisation of 'tactics' as "anart of the weak". Rowan argued, on the contrary, that "I do not see tactics or any group of 'others' as automatically 'weak" (p. 160).

Rather, I would suggest that de Certeau's tactical model is a useful means of acknowledging the ability of all groupslindividuals to fight back against repressive actions or ideologies, regardless of the degree of repression, by using whatever means are available. More than this, however, the marginal, through diverse and repetitious acts of subversion, can undermine dominant discourse to such an extent that the reductive definitions of 'centre' and 'margin'—whichnecessitate an awareness of 'strategy' and 'tactic'—become, in themselves, meaningless. (p. 160)

The analysis advanced in this thesis supports Rowan's careful avoidance of equating 'tactics' with "theweak". Certainly the show people's confident and assertive dealings with educational providers and university researchers evoke considerable strength of identity and purpose, which enables them to communicate their vision of improved educational experiences for their children and of consequent positive social and educational change. These actions are certainly 'tactics of consumption' because they are directed consciously against 'strategies of subversion', but they are also based on strength rather than weakness. Furthermore, it can be argued that it is people who are 'weak' in particular contexts who are most likely to 'need' to operate

tactically. This understanding resonates with the 'both/and' approach to conceptualising itinerancy championed at the end of the previous chapter, whereby itinerancy is understood **as** simultaneously devalued in comparison with sedentarism and equipped to challenge and counteract that devaluing.

The effect of this discussion of other authors' applications and critiques of de Certeau's 'strategies'—'tactics' paired category has been to justify, and in some cases to extend, my appropriation of 'tactics of consumption' as a major component of this thesis's conceptual framework. Generally these authors found this paired category to be analytically and empirically flexible and transferable, rather than disablingly reductionist. Ahearne (1995) expressed this well when he argued:

'Strategies' and 'tactics' cannot necessarily be set against each other as opposing forces in a clearly defined zone & combat. Rather, as Certeau presents them, they enable us as concepts to discern a number &

Ahearne (1995) concurred with the principle underlying Rowan's (1994) critique of de Certeau's (1984) depiction of 'tactics' as "an art of the weak" (p. 37), stating, "This [depiction] seems mistaken" (p. 162). By contrast, Fiske (1989) appeared to endorse de Certeau's depiction when he asserted, "Guerillatactics are the art & the weak; they never challenge the powerful in open warfare for that would be to invite defeat, but maintain their own opposition within and against the social order dominated by the powerful" (p. 19). My following the lead of Rowan (1994) inclines me to agree that 'tactics of consumption' can indeed be likened to "guerilla tactics", but to add that "guerilla tactics" generally reflect strength – of purpose and resolve if not of power – rather than weakness.

heterogeneous movements across different distributions & power. (p. 163)

This is certainly the use to which the concept 'tactics of consumption' will be put in this thesis: to "discern" the "heterogeneousmovements" across both the Queensland show circuits and educational provision for the show children, which from that perspective can be regarded as "different distributions of power". This very fluid and shifting approach accords closely with the antiessentialist approach to understanding itinerancy advocated in this thesis.

'Place' and 'space'

'Place'-'space' is another of de Certeau's paired categories, and its significance for this thesis is twofold. The first point is that this category 'matches' the 'strategies'-'tactics' paired category, in the sense that 'tactics of consumption' are used to change 'the place' of 'the proper', however temporarily, into 'a space'. The second point is that, given that itinerancy consists of movement across conventionally separate and logically distinct spaces, de Certeau's conception of 'space' is salient to understanding ambivalent constructions of the physical and symbolic spaces of itinerancy.

Having reiterated these **two** points, I wish to make **two** additional ones at this juncture. The first is that my reading of *The practice & everyday life* (de Certeau, 1984) and *Heterologies: Discourse on the other* (de Certeau, 1986), discussed in the two previous subsections, evoked several productive resonances with my analysis of the show people's movements through the spaces of the educational provision developed for their children. The second

point is that de Certeau's 'place'—'space' paired category is far more contested and controversial than his distinction between 'strategies' and 'tactics'. I argue here that this is because his conception of 'place' and 'space' generates considerable intellectual ambivalence in a large number of commentators. Again, my purpose here is not to provide a definitive resolution of this ambivalence. Instead, my intention is to relate instances of this ambivalence directly to the conceptual framework underpinning the thesis.

Thus, several authors have both applied and challenged de Certeau's distinction between 'place' and 'space'. Ahearne was largely complimentary, arguing about de Certeau "thathis intellectual strategy consisted precisely in an endeavour to discern and to make ethical and aesthetic space for particular form of interruption" (p. 3). For me, Ahearne's construction of de Certeau in this way highlighted de Certeau's desire to recognise, and therefore validate, specific 'tactics of consumption' by variously marginalised groups. I conceive this thesis as having a similar purpose and function in the particular case of the show people; I consider in Chapter Four the ethical implications of this conception.

Rowan (1994) was similarly complimentary about the 'place'—'space' paired category. She emphasised "the connection between de Certeau's notion of 'proper' places, the corresponding notion of illegitimate spaces, and the marginalising strategy of explanation" (p. 25). She argued, "This distinction [between 'places' and 'spaces'] is connected to the strategy's ability to circulate definitions of and discourses about the 'unlocated other' which emphasise the normality of the strategy and the aberrance of virtually everythingleveryone else" (p. 26). I discuss the ethical implications for this

thesis of "themarginalising strategy of explanation" in Chapter Four. For the moment I wish to emphasise that the show people, on account of their itinerancy, are almost literally "theunlocated other": their regular movement from one 'place' to another renders them as having no 'place' of their own and therefore as lacking location and hence legitimacy and power. Terry Evans (1998) recognised this crucial point when he noted: "Showchildren occupy or traverse a territory rather than a place" (p. xii). This indicates that de Certeau's distinction between 'place' and 'space' is analytically powerful for the argument prosecuted in this thesis.

By contrast, Balides (1993) was far more ambivalent about the intellectual value of that distinction. Certainly, in her analysis of constructions of women in American films in the early 1900s, she differentiated between de Certeau's conception of how 'place' is transformed into 'space' through the deployment of 'tactics of consumption' and the cinema analyst Stephen Heath's (1981; cited in Balides, 1993, pp. 25-26) argument that "space... becomes place in the movement of narrative" (Balides, 1993, p. 26) — that is, "throughthe continual inscription of the spectator in the diegetic world of the film, which is effected through the spectator's identification with the camera and with the looks of characters" (p. 25).

For Balides, the point of drawing on de Certeau's conception was to "displacethis priority given to vision" (p. 26) in Heath's understanding of 'space'. Thus she endorsed the fact that "inde Certeau's analysis space and place are distinct regimes of location", and that "de Certeau delineates a space different from that ofperspectival vision" (p. 26). Furthermore, "Place. . .is bound up with perspective and a panoptic vision" (p. 27), which

contributed to "the perspectival view of de Certeau's 'place'" (p. 28). Balides used this analysis to support her identification of both 'place' and 'space' in A windy day on the roof, a film made in 1904. Balides argued that the plot of the film, in which a house painter gazed voyeuristically at a woman hanging out laundry on the roof of her apartment building, "illustratesa further sense in which space can be understood to take place" (p. 29). On the one hand, "Inthe painter's look at the woman, the practice of everyday life (space) is subordinated to relations of looking (place)" (p. 29). On the other hand, "The film ends with the woman alone in the shot and an assertion of the visibility of this other space – the space of everyday life" (p. 29). This analysis is similar to those of Ahearne (1995) and Rowan (1994) in emphasising the analytical fluidity of the 'place'—'space' paired category.

However, Balides expressed considerable doubt about other **aspects** of de Certeau's distinction between 'place' and 'space'. For example, in a passage that read somewhat oddly beside her able application cited above of this distinction to her own work (and also with the fact that she followed the passage with the rather dismissive comment, "This [accusation of binary oppositions] is not the import of my argument" [p. 27]"), Balides stated:

In some discussions of de Certeau's work there is a problematic tendency to construct a binary opposition between place and space in which place is associated with a deterministic and monolithic conception of use and space is understood to mean – somewhat axiomatically – differentiated uses and resistance. (p. 27)

Balides' major concern about this alleged "problematic tendency to construct a binary opposition" was what she identified as a masculinist bias in de Certeau's work. She argued that for de Certeau "the threat of the city ('the dark space') is identified with the figure of woman" (p. 27). Furthermore, "de Certeau also identifies place with a masculine vision (the distanced view of the voyeur), linking perspective to a sexually implicated looking (the 'lust to be a viewpoint and nothing more')" (p. 27).

Although it is not 'the place' of this thesis to comment on charges of oppositional masculinism levelled at de Certeau, my response, phrased in terms of the theoretical concerns actuating the thesis, is that Balides' analysis belies the fact that feminist scholars (notably Rowan, 1991, 1993, 1994) have effectively used de Certeau's notion of 'space' (despite Balides' claim that this is an excessively feminised concept) to deconstruct masculinist 'places' such as Australian literary texts. Furthermore, McKay (1996) had no doubt that there was a great deal of value for contemporary feminist theorising of de Certeau'swork

The value of de Certeau's workfor feminist analysis is that it reconceptualises the nature of change so that feminist thought on the everyday is edged beyond the binary of oppression and resistance. This insight combined with the polemological foregrounding of a differentiated analysis of power relations sketches out a framework in which the increasingly complex nature of women's daily existence can be addressed. (p. 77)

From this perspective, Balides' critique here seems to **me** to introduce, prompted by her ambivalence about whether de Certeau should be 'read' as 'pro' or 'anti' marginalised groups, an unhelpful additional binary opposition that elides the human agency that de Certeau insisted was exercised in 'spaces' and that she identified in her own championing of de Certeau's conception of 'space' over that of Heath. The parallel with this thesis is clear: I seek to deploy de Certeau's conception of 'space', in alliance with his concept of 'tactics of consumption', to help to explain how the show people are able to move out of the marginalising spaces of itinerancy and to engage in resistance and transformation — an analysis that would not be possible if I 'read' de Certeau **as** being intellectually opposed to such a movement.

Buchanan (1996a) suggested the application of an additional term – "heterophenomenologicalspace" – to de Certeau's distinction between 'place' and 'space', although he intended this additional term to function as extending, rather than correcting, of de Certeau's work. Considering this suggested term provides a partial 'antidote' to the ambivalence towards de Certeau's 'place'—'space' distinction displayed by Balides (1993).

Buchanan's (1996a) account of de Certeau's work concentrated on "his reconceptualisatwn of the problem of theorising space" (p. 111). Buchanan argued that postmodernist constructions of space "havepushed the subject into the steepest decline" (p. 113). By contrast, he sought to demonstrate, through his appropriation of de Certeau's thought, "that subjectivity is not in decline, and that agency is still possible despite the baffling advances of technulogy, and the resulting incomprehensible futurity of space" (pp. 111-112).

"Heterophenomenological space" was Buchanan's own neologism, based on what he argued was the logical extension of de Certeau's theorising of space. By "heterophenomenology. Buchanan meant "aphenomenology predicated by a heterogeneously constituted subject which does not take for granted the unity of the body" (p. 112). This notion, derived directly from de Certeau's work, focussed on the agency of subjectivity in relation to space. Thus, "It is this agency that heterophenomenology seeks to restore against Jameson, and the trend in cultural studies generally, to place the subject at the mercy of space" (p. 122). Furthermore, "Treating space as a perspective cannot put us in touch with space itself, but it can put us in contact with a certain way of constructing space" (p. 128). In this thesis, I am interested in identifying the Queensland show people's constructions of the multiple spaces of their itinerancy, which Buchanan's extension of de Certeau's conceptualisation indicates derive from their agential subjectivities.

This discussion has indicated that de Certeau's 'place'-'space' paired category has both champions and detractors, and that the same author sometimes switches from one view to the other. I have argued that this uncertainty about whether de Certeau's distinction is to be applied or contested derives from ambivalence about his conceptualisation of 'space' — an ambivalence that has prompted the insertion of additional terminology of varying degrees of helpfulness. My own response to this discussion is to reassert the direct relevance of de Certeau's emphasis on 'space' as the site of the deployment of 'tactics of consumption' to understanding how the Queensland show people exercise agency in the spaces of their itinerancy. That emphasis is therefore central to the logic of the study's conceptual framework, because it helps me to move from an 'either/or' approach to

conceptualising itinerancy to the 'middle ground' of the 'both/and' approach to which this thesis aspires.

Marginalisation and resistance

De Certeau did not construct 'marginalisation'—'resistance' as an explicit paired category. Nevertheless, a number of authors has commented on one or other of these terms as they envisaged de Certeau conceptualising those terms. Furthermore, at least some of these authors, in contesting de Certeau's thought on marginalisation and its resistance, demonstrated ambivalence—which sometimes became operationalised as criticism—about where precisely he positioned himself in relation to that issue. The issue, clearly, is crucial to this thesis and the logic of its conceptual framework, because, in combination with transformation, marginalisation and resistance constitute the central dimensions of itinerancy explored in my study.

Rowan's (1994) doctoral thesis made several points that are pertinent to marginalisation as it was theorised by de Certeau. At a general level, her thesis, which explored the constructions of gender and race in six Australian literary texts, revealed the fruitfulness of applying de Certeau's account of marginalisation and resistance to Rowan's selected site of identity construction and contestation. Specifically, I am concerned with three elements of that study that have particular significance for the way that marginalisation is understood in this thesis.

Firstly, Rowan conceptualised both gender and race in terms with strong parallels to my interpretation of the dominant discourse in constructions of itinerancy. In a seemingly inexorable set of logical connections, she posited that "marks of 'difference' determine an individual's position—mainstream or marginal—within society" (p. 22). For the Queensland show people, their itinerancy is a 'mark of difference' that positions them as 'marginal' in relation to the 'mainstream' 'norm' of fixed residence.

Next, Rowan identified "not only difference, but the connection between difference and a perception of 'lack"' (p. 22). Thus the show people are perceived as lacking the physical and symbolic appurtenances of 'home', with its concomitant role in clearly giving people a 'place' of their own and in saying to the world, 'This is what I am like'.

Then Rowan analysed these 'marks of difference' and this perceived 'lack' as "acts of violence which attempt to exclude the voice (andfrequently the bodies) of the 'other' from legitimate discourse, from the public sphere, and from power" (p. 23). This relates to my argument in Chapter One, where I positioned this thesis as seeking to record the voices of the show people about marginalisation, resistance and transformation. The data analysis chapters contain many instances of the show people's consciousness of themselves being cut off by their itinerancy from "legitimatediscourse, from the public sphere, and from power". From that perspective, Rowan's very succinct account of the connections among 'marks of difference', 'lack' and 'acts of violence' encapsulates the marginalising strategies against which the show people deploy their 'tactics of consumption'.

Secondly, Rowan referred to "theabsence" in Jessica Anderson's novel An ordinary lunacy "of a counternarrative that could be seen as an alternative to masculinist hegemony" (p. 31). This notion of 'counternarrative' is a crucial element of the effective employment of 'tactics of consumption'. If the show people are ultimately to subvert the marginalising discourses that render them as different, lacking, having no voice and illegitimate, they need to articulate and prosecute an alternative and more positive construction of itinerancy, one that emphasises the social and educational benefits of this lifestyle and the legitimacy of itinerant people to speak about such issues as Australian cultural traditions and educational provision. I argue in this thesis that the show people's 'consumption' of the educational provision designed for their children is an important site for the elaboration of precisely this kind of 'counternarrative', whereby their previously silenced voices can be heard by those who reside in 'proper' 'places'. Clearly these conceptions of 'counternarrative' and 'voice' are predicated on marginalisation's constituent elements, among them the 'marks of violence', 'lack' and 'acts of violence' discussed above. (The thesis is also considered a 'counternarrative' through its contribution to helping those voices to be heard.)

Thirdly, Rowan emphasised "the disputatious nature of marginalisation; the many different ways of conceiving and responding to otherness;..." (p. 235). My response to "the disputatious nature of marginalisation" is to emphasise the shifting and fluid 'nature' of itinerancy as it is practised by the show people, and to use 'tactics of consumption', 'outsidedness' and 'creative understanding' as conceptual resources that in combination explain both how show people are marginalised and how they are able to resist and

transform that marginalisation. (This highlights anew the need for a 'both /and', rather than an 'either/or', approach to constructing itinerancy.) Furthermore, "the disputatious nature of marginalisation" helps to explain the ambivalence that some authors have displayed with regard to de Certeau's account of marginalisation and resistance.

Some authors have championed de Certeau as giving hope to marginalised groups by assigning to them the possibility of active and effective resistance. From this perspective, Bauman (1991) described de Certeau's analysis in *The practice of everyday life* (1984) of *la perruque*, which de Certeau called "the worker's own work disguised as work for his [sic] employer" (p. 25), as "the tool of defence of the self-regulated sphere of autonomy" (Bauman, 1991, p. 252, n. 19). Similarly, Stavropoulos (1995) relied heavily on de Certeau's analysis in *The practice of everyday life* (1984) of consumption as the exercise of agency to support her call for postcolonialist theories of 'representation' to take proper and sufficient account of resistance, which "has to be central to any approach which is serious about the revising of power relations" (p. 52).

Furthermore, Goodall (1995) drew on de Certeau's work, particularly in *The practice of everyday life* (1984), to support his argument that popular culture is not simply the 'poor relation' of high culture, but rather is the site of considerable and productive resistance:

The work of de Certeauseeks. ...to identify the ways in which the people and their culture are not simply suppressed, or made into unwilling recipients of values and discourses they neither want nor understand, but actively resist and expropriate the dominant culture. (p. 75)

Appropriately for this thesis's conceptual framework, Goodall pointed to significant parallels between the ideas of de Certeau and Mikhail Bakhtin, parallels to which I shall return in the next section of this chapter.

Writing in a similar vein, Jenkins (1992) stated that "De Certeau gives us terms for discussing ways that the subordinate classes elude or escape institutional control, for analyzing locations where popular meanings are produced outside of official interpretive practice" (p. 26). For Jenluns, de Certeau's conceptualisation of marginalisation and resistance was analytically relevant and enabling —just as this thesis finds that same conceptualisation.

Schirato (1993) used his largely approving summary of de Certeau's ideas about popular culture to highlight the centrality to those ideas of marginalisation and resistance:

The thesis de Certeau puts forward. . .is that popular culture is unswervingly subversive of official culture, and the interests official culture serves, including capitalists, the state, power élites — those Bourdieu calls the dominant. Popular culture is subversive, for de Certeau, because it is constituted by a heterogeneity which is oppositional to, and a denial of, homogeneity, community and, eventually, hegemony. Popular culture, however, not only carries the traces of its own difference; even more importantly, it carries the marks of violence of its own exclusion from that hegemonic community; that is to say it tells the stories that hegemonic ideology works to erase. (p. 283)

In contrast to these largely laudatory accounts of de Certeau's conception of resistance in relation to marginalisation, other authors argued that that conception was excessively 'celebratory' of resistance. Thus, although Weinstein (1994) conceded that de Certeau's account of resistance was influential, and that he had used that account "because of the powerful reading that it generates of the Robot World text" (p. 6), he noted that "the oppositional model had traditionally used a series of binary distinctions to describe the relation and difference between the popular and the official" (p. 3), a practice followed by de Certeau in his work. Weinstein considered that "These dichotomous distinctions are often inadequate to capture the overall complexity that is involved in popular cultures" (p. 4), suggesting that there was more resistance than de Certeau's thought was able to conceive.

Precisely the opposite charge – that de Certeau constructed more resistance than was actually possible – was laid by a couple of other authors. McKay (19%) criticised de Certeau's allegedly excessive celebration of popular resistance of marginalisation by claiming that it implied "fetishising the resistant status of everyday practices" (p. 67). For McKay, "The emergence of resistance becomes a cyclical, quasi-natural phenomenon rather than a potentiality whose realisation is contingent upon a particular configuration approver relations" (p. 68).

Similarly, Ahearne (1995) argued that, "inhis concern to uncover a kind of polymorphous flexibility", "Certeau's analyses run the risk of lapsing intermittently into an unqualified apologetics for ordinary practices" (p. 151). He further charged de Certeau with exaggerating the capacity of people to resist marginalising situations, using "anextreme example" to do so:

...he tends to assume that there will always be a given quota of inventiveness to ensure that people can 'get round' whatever mechanisms seek to organize and inform them. One misses in Certeau's account a note of threat, the sense that what he is analysing can never be guaranteed but can undeniably be stamped out in specific cases (an extreme example would be the behaviour of the popular majorities in Nazi or totalitarian states). (p. 185)

So Weinstein (1994) apparently thought that de Certeau downplayed the level of available resistance, while McKay (1996) and Ahearne (1995) accused him of exaggerating that level. Faced with this kind of evident contradiction, my response is to pursue a cautious and self-reflexive middle ground. In particular, I assert that the data analysis chapters of this thesis will demonstrate how the show people's 'tactics of consumption' enable them to engage in resistance of their marginalised itinerancy and in the process to turn the 'place' of their children's educational provision into a 'space' for a more enabling conception of itinerancy to be operationalised. Certainly this level of resistance does not 'go with the territory' of itinerancy, but instead derives from the show people's self-conscious constructions of their and others' identities and their formulation of various 'tactics of consumption' and their effortsto promote 'creative understanding'. The point is that, aided by these resources and techniques, the show people are able to carry out 'just enough' resistance for their purposes.

This suggests that hypothetical debates about 'not enough' or 'too much' resistance in de Certeau's thought need to be replaced by appropriate applications of his ideas to empirical sites of marginalisation and resistance.

Otherwise, the ambivalence about whether he conceived of 'not enough' or 'too much' resistance is likely to preclude productive analyses of 'real life' situations. After all, in the final analysis it is the outcomes of resistance that are most important. This point also articulates with my determination to eschew the twin poles of itinerancy as 'unproblematic othering' and as 'unproblematic celebration' identified at the end of the previous chapter: as with the ultimately futile argument about 'not enough'/'too much' resistance, the 'either/or' approach to understanding itinerancy that they encapsulate must give way to a much more fluid set of conceptual categories contained in the 'bothland' approach championed here.

Justification for my position on this issue was provided by Buchanan's (1996b) review of Ahearne'~(1995) biography of de Certeau. While Buchanan's review was as largely complimentary of the biography as the biography was of de Certeau (despite possible appearances to the contrary in this subsection), Buchanan departed from Ahearne in the latter's conviction that de Certeau emphasised resistance and subversion beyond a point where they could realistically pertain in such extreme situations as Nezi concentration camps during World War Two. On the contrary, Buchanan's citation of literature by survivors from such situations led him to proclaim "that de Certeau is entirely correct" (p. 152). This was because

...de Certeau does not assume 'resilience', he posits it. One does not fall back on 'inventiveness' in order to find an inner resource of 'resilience', rather one is 'resilient' because one is 'inventive'. This 'resilience' is possible because 'inventiveness' is a practice, not a specific outcome. This 'inventiveness', [sic] is, for de Certeau, a matter

of turning existing materials to alternative ends. A matter of 'making with' ('faire avec') not 'making do'. Thus, de Certeau does not 'valorise resistance'. . L' he formalises it. (p. 152)

Buchanan's interpretation of de Certeau's construction of human inventiveness and resilience strongly evokes the show people's own claims about how they negotiate their way through the multiple spaces of their itinerant lifestyle. This interpretation also provides a more agential and enabling view of social life than "existing criticisms of de Certeau" that "conform to a doctrinal view of his work" (p. 154). I seek in this thesis to favour the agential and enabling over the "doctrinal", and thereby to record my interpretation of the show people's itinerant spaces — particularly in situations where the "doctrinal" rests on unconscious ambivalences about de Certeau's thought (or else on the false dichotomy of itinerancy as either 'unproblematic othering' or 'unproblematic celebration').

Not that I seek to champion de Certeau against all criticisms by other authors. As I explicate below, I am particularly concerned by the evident incapacity of 'tactics of consumption' to lead to a change in 'the rules of the game'. With regard to marginalisation and resistance, which are my focus here, I take heed of some timely reminders and warning notes by some authors about the theoretical risk of homogenising resistance. Thus, although I disagree with Weinstein's (1994) comment, in reference to de Certeau, that "he overly celebrates. ..resistance" (p. 3), I concur with his supplementary assertion that "the celebratory attitude tends to overlook the ways that the resistance embodied in popular culture comes cluttered with racist and sexist contradictions" (p. 3).

Similarly, I have noted Jenhns' (1992) cautious warning

. . . against absolute statements of the type that appear all too frequently within the polemical rhetoric of cultural studies. Readers are not always resistant; all resistant readings are not necessarily progressive readings; the 'people' do not always recognize their conditions of alienation and subordination. (p. 34)

Moreover, Ahearne (1995) asserted that de Certeau's "writing is not entirely immune from the dangers inherent in advocating on principle the worth of the 'popular' (despite the displacement effected by the critical dissection of this term)" (p. 185). By this, Ahearne meant that "Certeau tends to valorize resistance as such, whereas this may constitute in some cases a damaging problem which has to be addressed" (p. 186). This contention recalls Frow's (1991) criticism that de Certeau's construction of 'the people' assumes both the homogeneity of such a group and that this group "necessarily operates in a progressive way" (p. 58). In other words, celebrating and championing resistance tends to downplay the potential for resistors to engage in equally repressive and marginalising strategies when they gain power themselves.

My response to these diverse warnings is to identify their **two** separate but related elements. One element is the abjuration to avoid homogenising resistance. This is a crucial point, to which I shall return in Chapter Four and the **data** analysis chapters. I acknowledge the importance of seeking to refrain from eliding differences within the show circuits and among the population that for ethical appropriateness and linguistic convenience I designate **as** 'the show people'. (Such an approach, indeed, would lead me back to the hnd of

literature identified in the previous chapter that rested on the romanticising of the itinerant lifestyle.) This is significant not only because such elisions have the potential to achieve the obverse of my objective in this thesis (by portraying show people **as** a single, passive group), but also because a major thrust of my argument in the second data analysis chapter is that the shifts in meaning and understanding between 'showie' and 'non-showie', and among different hnds of 'showies', are a principal component of the show people's resistance of their marginalising experiences.

The other element of the warnings posed by various commentators on de Certeau's work is the analytical risk of rendering marginalised people as folk heroes who, with 'right on their side', battle against evil and powerful enemies. Again, I seek consciously to avoid this potential tendency in this thesis, both because such a tendency would be patronising, even insulting, to a group of people with considerable pride and articulateness, and also because this portrayal would contradict my understanding of the educational providers from the Brisbane School of Distance Education as well-intentioned and hard worhing professionals seeking to enhance educational access for one of their client groups.

The effect of this discussion has been to confirm that many commentators find – as I do - de Certeau's theorisation of marginalisation and resistance fruitful and compelling. Furthermore, rather than being constituted as binary opposites, these terms emerge as floating signifiers. That is, like the show people whose lives are described in this thesis, 'marginalisation' and 'resistance' are fluid and mobile and need to be analysed in context as contingent and provisional. It is this theoretical fluidity and mobility that

attracts me to de Certeau's ideas, because they are precisely the necessary features of a conceptual framework geared to explaining the show people's interactions among themselves and with 'outsiders'. More specifically, they are vital characteristics of my efforts to explain the marginalisation, resistance and transformation of the show people's lifeworlds, and to prosecute a fluid and shifting understanding of itinerancy that can help to sponsor a counternarrative about it and about Travellereducation.

Change and transformation

At this point it is appropriate to synthesise my response to the various critiques of de Certeau's distinctions between such paired categories as 'strategies'-'tactics' and 'place'-'space' (and to some extent between 'marginalisation' and 'resistance')—a response that goes to the heart of my justification for deploying de Certeau's ideas in this thesis. The charge that de Certeau's thinking is excessively polarised, because it is reducible to a rigid set of binary oppositions, is misplaced for three crucial reasons. Firstly, this criticism downplayed the extent to which individuals and groups are in fact routinely positioned as being relatively powerless or marginalised. In other words, many people perceive themselves as being passive consumers of social life rather than as exercising a decisive influence over the unfolding of events to which they have to respond. To ignore or downplay this fact seeks—consciously or otherwise—toreplace what is to me de Certeau's very accurate and knowledgeable politicised analysis of social life that identifies something of the real scale of marginalisation of many disparate groups at the beginning

of the twenty-first century with a far less satisfactory utopian view of that social life.

Secondly, the criticism of de Certeau's work as being too polarised ignores what to me is a great strength of de Certeau's thought: his explicit recognition, and valuing, of the human agency of the less powerful. *The practice & everyday life* (1984) and *Heterologies* (1986), and the applications of these books by other scholars, are replete with examples of people engaging in tactics of various kinds that make a material and psychological difference to their situations. So the implication that de Certeau's work is reducible to binary oppositions and therefore positions the less powerful as having no control over their lives ignores this evidence of marginalised people having a capacity to change important aspects of their lives. Ironically, this criticism also forgets that a major motivation for de Certeau's work was a similar criticism by him of Foucault's thought, which indicates that he was particularly keen to "redress the balance" in favour of the agency of the marginalised.

Thirdly, this criticism ignores the fact that de Certeau did not portray marginalised people as only engaging in tactics of subversion. On the contrary, he eschewed a homogenised view of 'the powerful' and 'the powerless' (which lies at the heart of the criticism about binary oppositions) in favour of a much more heterogeneous analysis of social life. This prompted him to argue that within marginalised groups relatively powerful individuals sometimes engage in strategies that at least potentially can perpetuate the marginalisation of their peers. This might happen, for example, with tokenism, when 'the centre' coopts, and thereby 'buys the silence', of particularly

articulate and therefore potentially troublesome members of a marginalised group. For these reasons, then, de Certeau's distinctions between 'strategies' and 'tactics', and between 'place' and 'space', far from being excessively polarised and therefore static, are actually dynamic and fluid and therefore very well suited to informing my response to these questions. This suitability in turn underscores the appropriateness of de Certeau's ideas in highlighting the multifaceted dimensions of itinerancy and its incapacity of reduction to the twin poles of 'unproblematic othering' and 'unproblematic celebration'.

My criticism of critiques of de Certeau's alleged 'binary oppositions', then, is prompted by my assertion that the 'litmus test' for such critiques is the extent of change and transformation in the show people's situation. That is, if the show people are able to change their marginalised status by using both overt political action and conscious and unconscious resistance, that will be an indication that the critiques were overstated.

I turn now to another reason for discussing transformation at this juncture (apart from the fact that the latter is one of the three key concepts guiding this thesis). This reason is that such a discussion introduces my own area of greatest ambivalence about de Certeau's thought. This ambivalence is encapsulated in my concern that, relating this argument to the show people, the impact of de Certeau's theory is such that the show people can engage in a number of 'tactics of consumption', and they can fleetingly make 'the place' of 'the proper' into 'a space' of their own, but in the end 'the rules of the games' remain precisely the same as at the beginning of the contest.

To elaborate this argument: demonstrating that the show people 'consume' the educational provision established for their children in ways that resist and subvert the marginalising strategies of 'the centre', and that turn 'the place' of the educational norms associated with fixed residence into 'the space' of itinerancy, although it is significant, is inadequate in one crucial respect. This is that it fails to explain fully how and why the show people are able to contribute positively and substantially to social and educational change and transformation, encapsulated in the greater recognition and valuing of their distinctive lifestyle and educational needs: they are positioned always as 'making do' and never as re-making or reconstructing. As the next section of this chapter explains, this point is my justification for combining the theoretical resources of de Certeau's 'tactics of consumption' and Bakhtin's 'outsidedness' and 'creative understanding', because it is through exercising such practices that show people are enabled to transform 'the rules of the game' and to make more enduring positive changes to the spaces of itinerancy.

My ambivalence about this crucial element of de Certeau's thought is reflected in my endorsement of two seemingly contradictory views of that thought as it pertained to change and transformation. On the one hand, as Aheame (1995) pointed out, "Questions of historical rupture and transition are central to Certeau's work. How do social orders (and the human subjects who live within them) cease to be organized in certain ways and come to be organized in others?" (p. 27).

Similarly, for Rowan (1994), the 'binary oppositions' charge that several commentators levelled at de Certeau's work "canbe met by the constancy of displacement" (p. 76), by which she meant that "to seek the reversal of binaries instead of their displacement is to perpetuate inequalities and injustice" (p. 77). Drawing on the work of Audré Lorde, she used a vivid metaphor to illustrate her argument about the transformative potential of repetition leading to displacement

. . . to constantly and ceaselessly introduce difference into repetition is to weaken the master's control over his house and his tools and to allow for a new understanding of what the house is, how it was built and who[m] it shelters. (p. 77)

From the same perspective, I dissent from the claim by Budd, Entman and Steinman (1990; cited in Jenkins, 1992, p. 27) that nomadic readers "may actually be powerless and dependent" rather than "uncontainable, restless and free", and that "Peoplewho are nomads cannot settle down; they are at the mercy of natural forces they cannot control". This seems to replicate the conceptualisation of itinerancy as a 'deficit' (and therefore as an 'unproblematic othering') that courses through much of the Traveller education literature, a construction that de Certeau's concepts of 'tactics of consumption' and 'space' are well suited to assisting me to resist and change.

However, despite my endorsement of Ahearne's (1995) and Rowan's (1994) comments as encouraging my dissension from Budd et alia (1990), I also concur with Jenkins' (1992) elaboration of the comments by Budd et alia: "As these writers are quick to note, controlling the means of cultural reception, while an important step, does not provide an adequate substitute

for access to the mans of cultural production and distribution" (p. 27). This encapsulates my conviction that 'the rules of the game' do not change sufficiently to transform multiple forms of consumption into privileged sites of production. This is precisely why, in short, I have identified the need to link de Certeau's 'tactics of consumption' with Bakhtin's 'outsidedness' and 'creative understanding'.

I began this subsection by asserting that I would navigate a path through selected applications and critiques of de Certeau's ideas, informed by my commitment to eschewing an 'either/or' approach to conceptualising itinerancy in favour of a 'both/and' approach. At the same time, I posited a paradox in the reception accorded to de Certeau's thought, centred on the fact that many commentators who used his ideas fruitfully in applications to their respective research interests also criticised him for relying on one or more reductive binary oppositions. Furthermore, I argued that that paradox was explicable in terms of the commentators' ambivalence about how to 'read' de Certeau in relation to various paired categories: 'strategies'—'tactics', 'place'—'space' and marginalisation—resistance.In addition, I contended that the ambivalence surrounding de Certeau's work reflected his refusal to remain confined to positions or issues within one or other of the terms of these categories, but rather traversed the boundaries keeping them separate.

This proposition combined ambivalence with the itinerancy of the figure of 'the nomad', and therefore provided an explicit link with my earlier critique of the literature on Traveller education. Moreover, some of these ambivalent accounts of de Certeau's paired categories revealed contradictions, such as

Balides' (1993) and McKay's (19%) efforts to render the 'place'-'space' categories **as** excessively masculinist.

Not that I regard ambivalence as a phenomenon to be avoided at all costs. On the contrary, I acknowledge and celebrate my own ambivalence about de Certeau's work, an ambivalence that I contend can be theoretically and empirically productive. On the one hand, I regard the paired categories discussed in this subsection, not as reductive binary oppositions, but instead as shifting and fluid conceptual filters that *can* assist my interpretation of the dynamism and flexibility of the show people's interactions with educational providers. This is particularly true of de Certeau's 'tactics of consumption' and 'space', which I therefore feel justified in malung integral to this thesis's conceptual framework. This practice accords with my argument that itinerancy *can* be properly understood only from an anti-essentialistapproach that avoids the conceptual traps identified in the literature review in the previous chapter.

On the other hand, I argue that the accumulation of paired categories in de Certeau's work, while not in themselves binary oppositions, leads paradoxically to an incapacity to analyse the possibility of a change to 'the rules of the game'. This incapacity allows the show people to be understood as perennially engaging in 'tactics of consumption' and malung the 'spaces' of itinerancy closer to their vision of 'home' but still not accessing or transforming 'the place' of 'the proper'. In other words, there are vital issues that this framework cannot identify and attend to. It is this ultimate analytical weakness that provides a cue for my move to the next section, where I outline how Mikhail Bakhtin's notions of 'outsidedness' and 'creative understanding' enable just such a transformation to be envisaged. At the same time, I

should emphasise the need to combine both sets of conceptual resources if I am to pursue the goal of a 'both/and' approach to understanding itinerancy and of contributing to a counternarrative of both it and Traveller education. My goal in the next section of the chapter is to locate – by using Bakhtin – analytical resources that allow the identification, not just of marginality and resistance, but also of transformation.

3.3 Bakhtin's notions of 'outsidedness' and 'creative understanding'

As with Michel de Certeau, Mikhail Bakhtin's contribution to this thesis's conceptual framework is intimately connected to my desire to deploy the strongest available theoretical resources to advance the cause of an antiessentialist approach to understanding itinerancy. Furthermore, Bakhtin's contribution is related also to what is effectively a *leitmotif* of the thesis: the ambivalence arising from the spaces of itinerancy. In Bakhtin's case, not only has his thought generated considerable ambivalence among commentators (for example, and again in similarity to de Certeau, in relation to possible feminist appropriations of his ideas), but he also highlighted on many occasions the ambivalence that was fundamental to his conception of social life. This assertion will be demonstrated below, partly in my account of Bakhtin's notions of 'outsidedness' and 'creative understanding'. As I indicated earlier, in the discussion that follows I construct these two concepts as leavening elements that potentially transform 'the rules of the game' so that the

marginalised spaces of itinerancy can become meaningful and productive 'places' on more than a temporary basis.

In discussing de Certeau's thought, I moved from his conceptualisation of 'tactics of consumption' in two of his works (1984, 1986) to an account of various commentators' applications and critiques of his thought. With Bakhtin, I have elected to move in the reverse direction, beginning by putting his thought in the perspective of a comparison with that of de Certeau, then proceeding to focus on two of his principal notions — 'outsidedness' and 'creative understanding' (1986a). The reason for this approach lies in my desire to demonstrate the broad similarities and dissonances between the two theorists' thinking as a means of explaining and justifying my juxtaposition of certain of their ideas in the conceptual framework of this thesis. This demonstration is likely to be most effective by exploring the broader dimensions of Bakhtin's thought before concentrating on the two notions selected for deployment in this study.

Accordingly, I shall discuss the following aspects of Bakhtin's thought in this section of the chapter:

- Bakhtin and de Certeau
- 'outsidedness' and 'creative understanding'.

3.3.1 Bakhtin and de Certeau

There are several points of implicit comparison between Bakhtin and de Certeau, even though to my knowledge they remained unaware of each other's work. In tracing these points of comparison, I demonstrate that Bakhtin and de Certeau are not conceptually incompatible but can logically be linked without misrepresenting either thinker's body of work. In the next subsection of the chapter I outline and justify my focus on 'outsidedness' and 'creative understanding' as conceptual resources capable of explaining how 'the rules of the game', within which de Certeau's 'tactics of consumption' ultimately reside, *can*be challenged and transformed.

Firstly, I draw on Morson and Emerson's extensive biography Mikhail Bakhtin: Creation of a prosaics (1990). In that text, the authors identified three "global concepts", which they argued underpinned all Bakhtin's work and gave an enduring coherence to writings that were published more than fifty years apart. One of those "global concepts" was "prosaics". This was a term that Bakhtin never used and that Morson and Emerson coined as a neologism to encompass a major theme in Bakhtin's work. According to Morson and Emerson, prosaics in its broader sense "is a form of thinking that presumes the importance of the everyday, the ordinary, the 'prosaic'" (p. 16). They argued that in its more restricted sense, as "a theory of literature that privileges prose in general and the novel inparticular over the poetic genres", prosaics "is, so far as we know, Bakhtin's unique and original creation" (p. 15).

The concept of "prosaics" would appear to resonate with de Certeau's continuing interest in daily life, as the title of his book The practice of everyday life (1984) most obviously exemplified. There were also echoes of de Certeau's ideas (although his name was not mentioned) in Morson and Emerson's account of Bakhtin's understanding of prosaics, such as in the sections "Prosaics and everyday language" (pp. 21-23), "Prosaics as a philosophy of the ordinary" (pp. 23-25), "Prosaics and ethics" (pp. 25-27) and "Prosaics and systems" (pp. 27-32). The significance of what at first appears to be a somewhat tenuous connection lies in highlighting the fact that, although both Bakhtin and de Certeau were primarily interested in language and literature, their respective ideas had profoundly material and practical implications. That is, Bakhtin and de Certeau understood that the issues to which their concepts related extended far beyond the intellectual sphere and were inextricably linked with the daily lives of 'ordinary' people. This is another point of justification for applying selected aspects of those ideas in juxtaposition to the empirical site with which this thesis is concerned.

Furthermore, the assertion that Bakhtin, like de Certeau, was deeply concerned with the empirical contexts in which his ideas were grounded was supported by Jones (1993). For example, Jones argued that Bakhtin questioned the capacity of theoretical systems

...to account satisfactorily for the experience of everyday life, or to provide an adequate basis for understanding linguistic communication, interpersonal relations, moral choice, aesthetic judgement, literary texts or indeed any sphere of action or enquiry which takes the experience of everyday life as its chosen field. (p. 244)

Jones' (1993) argument evokes de Certeau's similar interest in the ethical dimensions of 'everyday life'. Furthermore, Jones implicitly approved of Morson and Emerson's (1990) neologism 'prosaics', which he described as "a form & thinking (or style of enquiry) that presumes the overriding importance of the everyday, the prosaic, in language as in experience in general" (p. 250). This summary resonates with de Certeau's account of the importance of everyday life as the site of the exercise of agency, rather than as the setting for wholly meaningless and trivial concerns. Furthermore, it underscores the crucial point that both Bakhtin and de Certeau would be likely to have endorsed an analysis of an empirical site of 'everyday life' from the perspective of marginalisation, resistance and transformation – the foci of this study's research questions. Like de Certeau and Bakhtin, I conceive of everyday life as manifesting the interplay of much deeper and more fundamental forces and processes – in this case, marginalisation, resistance and transformation in relation to itinerancy and Traveller education.

Secondly, Morson and Emerson (1990) articulated what they identified as Bakhtin's aversion to binary oppositions:

As we have seen, Bakhtin rejected the Hobson's choices of modern thought: either there is a system or there is nothing; either there are comprehensive closed structures or there is chaos; either there is in principle an all-encompassing explanatory system or there is total relativism (or perhaps: either God exists, or all is permitted). The assumption that these are the only alternatives has blinded critics to the possibility of radically different kinds of truth, unity, and perspective. (p. 233)

This critique of commentators who have mistakenly constructed Bakhtin's ideas as being excessively polarised, whereas he actually argued strongly against binary oppositions, is very similar to my earlier argument that many of the charges of binary oppositions directed against de Certeau have resulted from ambivalence about, and misunderstandings of, his work. The same kind of ambivalence is evident in several of the following points of comparison between Bakhtin and de Certeau. Certainly the assertion that Bakhtin's thought was consciously anti-binaristic in structure helps to justify my deployment of two of his concepts to explain how it is possible to alter fundamentally the supposedly fixed and immutable 'rules of the game'. This resonates with my earlier, repeated statement of preference to eschew the polarised 'either/or' for the more dynamic and mobile 'both/and' approach to conceptualising itinerancy.

Thirdly, Roderick (1995) contended that Bakhtin, through his articulation of the concept of carnival to denote the reversal of 'normal' social rules, aligned himself explicitly with the 'consumers' rather than the 'producers' of society:

Bakhtin removes the complexity of ..[a] spatial model of class society, making the body politic divisible into only two stratum [sic]. Those lowly topics associated with the lower classes are to be championed by Bakhtin, while higher topics associated with upper classes are to be disparaged. The ascendancy of the higher canons of official life is to be resisted through celebrations or rather unleashings of the material bodily principle within everyday life. (p. 122)

Roderick's interpretation of Bakhtin's approach to theorising carnival is to me far too polarised. On the other hand, that interpretation has the merit of identifying Bakhtin's and de Certeau's common interest in highlighting evidence of resistance of the marginalising strategies directed at less powerful groups, and their delight in portraying the rich heterogeneity of social life. This similarity helps also to justify my focus on consumption **as** an active process of challenging and potentially transforming provision of services that consumers consider insufficient or inappropriate.

Fourthly, a crucial similarity between Bakhtin and de Certeau was their common interest in 'boundaries' or 'margins'. Bakhtin (1975) insisted that cultural entities are effective boundaries:

One must not, however, imagine the realm of culture as some sort of spatial whole, having boundaries but also having internal territory. The realm of culture has no internal territory; it is entirely distributed along the boundaries, boundariespass everywhere, through its every aspect.

. Every cultural act lives essentially on the boundaries: in this is its seriousness and significance; abstracted from boundaries it loses its soil, it becomes empty, arrogant, it degenerates and dies. (Cited in Morson & Emerson, 1990, p. 51)

This resonates with the extract from *The practice of everyday life* (1984) that I cited earlier:

Marginality is today no longer limited to minority groups, but is rather massive and pervasive; this cultural activity of the non-producers of culture, an activity that is unsigned, unreadable and unsymbolised,

remains the only one possible for all those who nevertheless buy and pay for the showy products through which a productivist economy articulates itself. Marginality is becoming universal. A marginal group has now become a silent majority. (p. xvii)

There are three principal points of similarity between these two statements. The first, which follows from my second point above, is that a rigid dichotomy between 'centre' and 'margins' is eschewed in favour of an acknowledgment that the dividing line is fluid and shifting. Additionally, margins are constructed as agential and productive, the sites of creativity for Bakhtin and of consumption for de Certeau. Furthermore, Bakhtin's refusal to depict "the realm of culture as some sort of spatial whole" is in some ways evocative of de Certeau's distinction between 'place' and 'space', in that he rejected a view of space as sequestered from the forces of everyday life. These three points of similarity have a three-fold significance for this thesis. The first significance is that they highlight Bakhtin's and de Certeau's agreement with each other on issues that are crucial for the thesis's conceptual framework. The second significance is that they refocus attention on the ambivalent spaces of itinerancy, by suggesting that such spaces are dynamically and productively marginal. The third significance is to reassert the value of understanding itinerancy as dynamic and shifting, not as fixedly polarised between 'unproblematic othering' and 'unproblematic celebration'.

A fifth point of commonality is highlighted by Holquist (1981), who identified in Bakhtin a tendency that was also characteristic of de Certeau, and that followed their common conviction that marginality can be creative and even powerful. This was the fact that Bakhtin "is preoccupied by centuries"

usually ignored by others; and within these, he has great affectionfor figures who are even more obscure" (p. xvi). Holquist's examples of this tendency in Bakhtin included his study of "a peculiar school of grammarians at Toulouse in the seventh century A.D.", and his recurrent interest in "the Carolingian Revival or the interstitial periods between the Middle Ages and the Renaissance" (p. xvi). In de Certeau's case, equivalent instances included his lifelong enthusiasm for the ideas of Jean-Joseph Sdrin, whom Ahearne (1995) described as "a strange seventeenth-century mystic" (p. 2), and de Certeau's detailed account of a case of multiple diabolic possession in the French city of Loudon in 1632 (Ahearne, 1995, p. 76). This similarity of tendency underscores both Bakhtin's and de Certeau's greater interest in less well known individuals and groups than those with greater financial and cultural capital, a point that highlights the general applicability of their ideas to this study of a 'marginalised' group within Australian society.

Sixthly, Roderick's (1995) depiction of Bakhtin's account of the body contrasted strongly with Fiske's (1989) construction of de Certeau's account of the same topic. Roderick chastised Bakhtin for privileging the collective body of the public space – closely associated with carnival – over the individual body of the private sphere. For Bakhtin (1984), the grotesque public body is enormously productive and resilient, largely because of its direct link to the "ancestralbody of the people" (p. 29).

By contrast, according to Fiske (1989), de Certeau believed "that juridical law can be effective only if people have bodies upon which it can be imposed" (p. 91). In Fiske's analysis of de Certeau's position on the body, "This incarnation of the law and intextuation of the body. ..are also at work

in ordinary everyday practices" such as "clothing, cosmetics, slimming, [and] jogging" (p. 91). What interests me in this analysis is the proposition that Bakhtin presented a more agential 'reading' of the body than that propounded by de Certeau. While 'tactics of consumption' are explicitly agential, my selection of 'outsidedness' and 'creative understanding' as accompanying conceptual resources derives from my argument that they provide an ultimately more productive interpretation of social life than 'tactics of consumption' considered in isolation from efforts to change and transform 'the rules of the game'.

Seventhly and finally, Roderick (1995) levelled a similar feminist critique against Bakhtin to the criticism directed by Balides (1993) and McKay (1996) at de Certeau. Roderick's attack was two-pronged. The first prong was his charge that Bakhtin's study of carnival focussed largely on the construction of space from a masculinist perspective. Roderick summarised these two points in his acerbic comment: "Bringing men into public space where they operate as competent speakers is hardly a radical new project" (p. 132). The second prong was Roderick's contention that Bakhtinian agency was also masculinist. This point related to Roderick's assertion that "Bakhtinis unable to recognise" that the agency that he championed as being exercised by 'the people' was actually based on "asexed subject" (p. 132). Again, I disagree with Roderick's 'reading' of Bakhtin as excessively masculinist, on the basis that several successful feminist appropriations of Bakhtin's ideas have been made (see for example Pearce [1994], who nevertheless also attacked Bakhtin as potentially masculinist).

This admittedly selective comparison of the thought of Bakhtin and de Certeau has revealed three main findings that are relevant to this thesis's conceptual framework. Firstly, a number of commentators revealed considerable ambivalence about certain ideas of both thinkers, an ambivalence that in some cases obscured or misrepresented their work. Secondly, Bakhtin and de Certeau's thought has several points of coincidence, particularly the recognition that they gave to human agency among 'marginalised' groups and the power of that agency in both contributing to, and resisting, that marginalisation. Thirdly, in relation to topics as varied as binary oppositions and the body, Bakhtin showed a willingness to consider how 'the rules of the game' separating groups with varying degrees of financial and cultural capital might be changed and transformed that was not so consistently displayed by de Certeau. In combination, these three findings endorse the 'rightness of fit' between my synthesis of these two thinkers' ideas and the logic of the study's conceptual framework, conceived as it is as simultaneously demonstrating that itinerancy is associated with marginalisation at the same time that it presents the possibilities of resistance and transformation.

3.3.2 'Outsidedness' and 'creative understanding'

Having traced some of the main lines of similarity and divergence between the thinking of Bakhtin and de Certeau, I turn now to examine the two Bakhtinian notions that I apply to the analysis of the study's data. My intention in this subsection is to outline the concepts as conceived by Bakhtin and understood by commentators on his work. I seek also to make a strong case that juxtaposing these concepts with de Certeau's 'tactics of consumption' is

necessary to explain how the show people can go beyond resisting marginalisation to transforming the origins and consequences of that marginalisation.

My inclusion of 'outsidedness' and 'creative understanding' in this study's conceptual framework is neither accidental nor incidental. On the contrary, they have been included because they allow me to answer the third research question framing this thesis, the one relating to transformation, in ways that depending exclusively on de Certeau's 'tactics of consumption' would not enable me to do. Such an exclusive dependence would create an intellectual problem, by not allowing the study to conceive of the possibility of the show people's moving outside the marginalisation—resistance 'loop'. Bakhtin's notions are crucial to solving that intellectual problem, by conceptualising the potential for transformation and the creation of new ways of understanding itinerant people and Traveller education.

To elaborate this argument, in the next chapter I demonstrate how 'outsidedness' and 'creative understanding' are central to my methodological approach to the research design of the study. In this subsection of this chapter my task is to explain how those same notions contribute conceptually as well as methodologically. They do this by allowing me to identify and articulate those actions of the show people that cannot be explained by the broad concept of 'tactics of consumption'. In other words, if I restricted the study's conceptual framework to that broad concept, there are certain political moves and other actions that I would be unable to name and hence analyse, because 'tactics of consumption' cannot conceive of the kind of transformative practice that this thesis posits. Thus Bakhtin's notions are indispensable

because they give me that naming power, and they allow me not only to focus on, highlight and identify, but also to acknowledge and analyse the political significance of, particular practices and moves.

Against the backdrop of this argument about the crucial analytical utility of the two Bakhtinian concepts of 'outsidedness' and 'creative understanding', then, those two concepts are logically related. By this I mean that the processes entailed in 'creative understanding' are predicated on a preceding recognition of the 'outsidedness' of other people. In the analysis that follows in subsequent chapters, I shall demonstrate the ways that the show people display creative understanding as they contribute to the transformation of their educational opportunities. Here I need to explicate precisely what these terms mean.

Morson and Emerson (1990) explained 'outsidedness' in this way: "Whenone personfaces another, his [sic passim] experience is conditioned by his 'outsidedness'. Even in the physical sense, one always sees something in the other that one does not see in oneself. I can see the world behind your back. ..." (p. 53). Although the bases of 'outsidedness' could vary considerably, including "personal, spatial, temporal, national, or any other" (p. 56), "outsidednesscreates the possibility of dialogue, and dialogue helps to understand a culture in a profound way" (p. 55).

Pechey (1997) referred to "the early Bakhtin, who builds the whole house of value on the foundation of our 'outsidedness' to one another, my authorship of you as 'hero' and your answering authorship of me" (p. 37). Pechey used that analysis to assert, "We are each of us a margin and each of us a centre" (p. 37), which recalls my earlier discussion of the similarities

between Bakhtin and de Certeau in their respective views of the 'centre'—'margins' dichotomy. It recalls also Ferguson's (1990) persuasive reference to 'the invisible centre', and his statement that "Whenwe say marginal, we must always ask marginal to, what?" (p. 9). This link between 'outsidedness' and the idea of marginalisation as a floating signifier is important in suggesting 'outsidedness' as part of a process of destabilising the foundations of marginalisation. Again this highlights the analytical power of 'outsidedness': its contribution to that process of destabilisation is vital in identifying and interrogating actions by the show people that help to transport them to a new terrain that moves beyond the marginalisation—resistance interplay so ably theorised by de Certeau.

Morson and Emerson (1990) used a linguistic example to explain the operation of 'outsidedness':

To realize and develop the potential of a language, 'outsidedness'-the outsidedness of another language-is required. That outsidedness may lead to an exchange in which each language reveals to the other what it did not know about itself, and in which new insights are produced that neither wholly contained before. (p. 310)

In addition, Bakhtin (1986a) articulated the crucial role that 'outsidedness' played in enabling and promoting 'creative understanding':

Creative understanding does not renounce itself, its own place in time, its own culture; and it forgets nothing. In order to understand, it is immensely important for the person who understands to be located outside the object of his or her creative understanding—in time, in space,

in culture. For one cannot even really see one's own exterior and comprehend it as a whole, and no mirrors or photographs can help; our real exterior can be seen and understood only by other people, because they are located outside us in space and because they are others. (p. 7; emphasis in original)

Furthermore, "Creative understanding continues creativity, and multiplies the artistic wealth **A** humanity" (Bakhtin, 1986b, p. 142).

Morson and Emerson (1990) traced the elaboration of the Bakhtinian concept of 'creative understanding', which they argued was based on a "profoundly non-Platonic conviction about human thought" (p. 99):

Revising the Kantian triad **t** his earliest period, Bakhtin now claims that understanding is in fact a four-tiered process: first, the physical perception, then its recognition, then a grasping of its significance in context, and finally—and this is the crucial step—'active-dialogic understanding.' This fourth step is more than an acknowledgment **t** existing context; it is implicitly creative, and presumes ever-new, and surprisingly new, contexts. (p. 99)

The concept of 'creative understanding' is useful not only methodologically (as I elaborate in the next chapter) but also analytically, because it allows me to identify instances of the show people engaging in this lund of understanding. This in turn enables me to argue that these are fundamentally transformative acts that move the show people beyond their traditional marginalised status. Creative understanding, understood in this sense, is

something that one does, rather than something that one has or to which one aspires.

My proposition that 'outsidedness' is a prerequisite of 'creative understanding' is made despite my awareness that Morson and Emerson (1990) located 'outsidedness' as a new concept in what they identified as Bakhtin's first period of intellectual development, and 'creative understanding' in his fourth and final period (p. 66). I am aware also that they argued that "Whenhe discovered dialogue, Bakhtin largely abandoned this model" (p. 54) of social life that included 'outsidedness', on the basis that

...the abstractness of the formulation, its sense of self and other as irreducibly counterposed starting points gave way to a richer sense of dialogue as the starting point. He arrived at more profound and integrated conceptions of self and society—two categories that were derivative, reified, and partially misleading when opposed to each other. (p. 54)

Nevertheless, for me a less abstract and rigid conceptualisation of 'outsidedness' is helpful both in constructing my interactions with the Queensland show people and in interpreting their constructions of other people, as a prelude to engaging in dialogue as an expression of 'creative understanding'. This argument is partly justified by the fact that Morson and Emerson asserted that "Bakhtintends to stress the importance & boundaries and & unmerged horizons, which provide the outsidedness that ultimately makes all dialogue and all creativity possible" (p. 166), and by their subsequent statement that "the differences of culture—the outsidedness—... makes [sic] a real dialogue among cultures and periods possible" (p. 429;

emphasis in original). Even more explicitly, they claimed that Bakhtin's "understanding of creative understanding" contained four elements: "outsidedness, live entering, confusion, and active dialogue" (p. 99).

Perhaps the 'solution' to this apparent contradiction – at the very least an example of ambivalence by Morson and Emerson (1990) towards these two Bakhtinian concepts – lies in Emerson's (1984) contention, "That one aspect of Bakhtin's style most inseparable from his personality is the developing idea. Its subtle shifts, redundancies, self-quotations – ultimately, its openendedness – is [sic] the genre in which, and with which, he worked" (p. xxxix).

In other words, rather than seeing 'outsidedness' from Bakhtin's first period and 'creative understanding' from his fourth period as mutually exclusive, it is more appropriate and productive to regard the latter as the considerably elaborated amplification of the former, with important continuities with, as well as crucial divergences from, the earlier concept. This approach echoes Docker's (1994) assertion that

...there is no essential Bakhtin to be appropriated, that Bakhtin himself disliked any attention that tried to seek true or single identity for his writings, that he was embarrassed by agreement, and welcomed disagreement. Inevitably I have used "Bakhtin", my own interpretation of his writing, for my own arguments about cultural history. (p. 183)

In a corresponding way, I am using two of Bakhtin's notions to support my argument about the show people's resistance and transformation of their marginalised situation.

So Morson and Emerson's (1990) ambivalence about the connections between Bakhtin's concepts of 'outsidedness' and 'creative understanding' has given way in my account to the argument that 'outsidedness' is a prerequisite of 'creative understanding', which in turn is a means of moving beyond the fixed conceptual binaries that constitute 'the rules of the game' through its contestation and destabilisation of the basis of those rules and its demonstration of the possibility of new and different rules being established in their place. Certainly if more equitable and less marginalising approaches to Traveller education are to be achieved, creatively and imaginatively comprehending the basis of the other's circumstances and positions is vital to communicating what each group expects such education to accomplish. Furthermore, as I relate in Chapter Four, these two concepts are crucial to justifying how it is possible for a non-itinerant person to write a thesis focussed on the lives of itinerant people, by employing and exploiting the principle of 'outsidedness', and also how such a thesis can itself reflect and promote 'creative understanding' between show people and non-show people.

Both these points – challenging conceptual binaries and helping to frame the study's research design – reinforce the argument pursued in this section of the chapter. That argument has been that a focus on de Certeau's 'tactics of consumption' is helpful in explaining how the show people experience and resist marginalisation, but that such a focus is insufficient to explain how such marginalisation can be and is transformed into more positive experiences and meanings associated with itinerancy. 'Outsidedness' emphasises the contingency and provisionality of the mutually opposed positions attached to marginalisation and resistance. Similarly, 'creative understanding', understood in an active and agential way to denote a conscious process and effort of will,

provides a means of moving beyond those mutually opposed positions to a comprehension of the other's point of view and a realisation that 'things don't always have to be like this'. This is the sense in which I conceive these two Bakhtinian concepts as underpinning a shift from marginalisation and resistance to transformation and a fundamental change to 'the rules of the game' according to which itinerancy is 'played' or enacted. That same shift is crucial to prosecuting a move beyond conceiving itinerancy as either 'unproblematic othering' or 'unproblematic celebration' towards a far more mobile, realistic and respectful understanding of itinerant people.

3.4 Synthesis of the conceptual framework

The purpose of this section of the chapter is to synthesise the main elements of the study's conceptual framework, thereby summarising the argument outlined so far in this thesis and providing a 'bridge' to the research design and data analysis chapters that follow. Figure 3.1 portrays the thesis's conceptual framework in diagrammatic form.

ITEMS	ELEMENTS		
Focus	The spaces of itinerancy		
Key	'Tactics of	'Outsidedness'	'Creative
concepts	consumption'		understanding'
Research	Marginal-	Resist-	Trans-
questions	isation	ance	formation
Literature	Australian	← ►	Traveller
	shows		education

Figure 3.1: The conceptual framework of the study

The starting point of the conceptual framework, which as I noted at the outset of the chapter is integrally associated with the need to pursue an antiessentialist construction of itinerancy, is the study's focus: the spaces of itinerancy. I explained earlier that 'spaces' in this context refers to the physical and symbolic spaces through which itinerants travel, which are therefore recognisably different from the spaces in which non-itinerants conduct their daily lives.

The thesis's focus on the spaces of itinerancy has prompted my selection of the three key concepts informing this study. Firstly, de Certeau's concept of 'tactics of consumption' is crucial to explaining how and why the show people resist and subvert their marginalised status, as a means of maximising

their children's educational access and in the process of turning the often uncomprehending and sometimes hostile 'spaces' through which they travel into 'places' of their own. On the other hand, my ambivalence about de Certeau's concept led me to argue that, although I found many of the binary oppositions ascribed to him by other writers (and about which they expressed their own ambivalence) misconceived, that concept did not enable an analysis of a situation in which 'the rules of the game' that construct people like the 'showies' as marginalised can actually change.

Accordingly, I selected Bakhtin's notions of 'outsidedness' and 'creative understanding' as a means of understanding how 'the rules of the game' can and do change, thereby capitalising on the potential resistance and transformation of the spaces of itinerancy and therefore refusing to privilege marginalisation as the single most influential consequence of itinerancy. 'Outsidedness' as a prerequisite of 'creative understanding', in my 'reading', explains how the show people's interactions with others, including educational providers, can enable a mutually respectful comprehension to develop. This growing comprehension would have the effect of making the 'spaces' of itinerancy 'places' for both groups, rather than a territory that the itinerants would occupy only when the non-itinerants were not looking.

In combination, the focus on the spaces of itinerancy and the key concepts of 'tactics of consumption', 'outsidedness' and 'creative understanding' explain, inform and refine the three research questions underpinning this thesis. Firstly, the show people's experiences of marginalisation result directly from the constructions of the spaces of itinerancy **as** 'unstable' and even 'dangerous' places where the established order is turned upside down.

In consequence, 'strategies of marginalisation' are experienced by the show people, who epitomise these perceived 'threats' to that order. This situation is clearly the antithesis of the exercise of 'outsidedness' in order to reach mutual 'creative understanding'.

Secondly, that same instability and fluidity of the spaces of itinerancy give the show people opportunities to take subversive action and engage in several and varied 'tactics of consumption' in order to resist their marginalisation. Here the focus is on 'outsidedness' rather than 'creative understanding', as the show people draw strength for their resistant tactics from what makes them different from, or 'outside', the established order. So 'outsidedness' becomes transformative and an explicitly political move.

Thirdly, in an attempt to move beyond 'the rules of the game' that position 'tactics of consumption' in opposition to 'strategies of marginalisation', the show people also engage in actions designed to capitalise on their 'outsidedness' in relation to non-itinerant people, and to enhance and maximise mutual 'creative understanding', which thereby and also becomes a political move against those who set 'the rules of the game'. They see these actions as the most likely to enable them to move beyond marginalisation and resistance to transformation, and to make the often hostile 'spaces' of itinerancy into meaningful and welcoming 'places' of their own.

Finally, it is intended that the data analysis chapters of this thesis, having explored marginalisation, resistance and transformation in relation to the Queensland show people, will contribute significantly to extending the existing literature about Australian shows and Traveller education. In particular, it is intended that the thesis function as a counternarrative to the

negative stereotypes still prevalent in that literature, and instead contribute to circulating more accurate and productive understandings of show people and their educational experiences and opportunities.

To reiterate and summarise, then, the synthesised conceptual framework underpinning this thesis uses 'tactics of consumption', 'outsidedness' and 'creative understanding' as analytical lenses to interpret the Queensland show people's experiences of marginalisation, resistance and transformation as they move through the spaces of their itinerancy. Those experiences can either inhibit the 'spaces' of itinerancy from becoming mutually valued 'places' of living and learning, or else they can facilitate that process of transformative becoming. Examples of such facilitation can contribute significantly to making the literatures on Australian shows and Traveller education more rather than less meaningful and productive for educational providers and their varied clients — and of course show people themselves. Equally significantly, these analytical tools are central to the goal of pursuing a counternarrative to the traditional and limited understandings of itinerancy and Traveller education evident in much of the current literature.

3.5 Review of the chapter

In his essay "Travelling cultures" (1992), the anthropologist James Clifford referred to

...the evident fact that travellers move about under strong cultural, political, and economic compulsions and that certain travellers are materially privileged, others oppressed. ...Travel, in this view, denotes a range of material, spatial practices that produce knowledge, stories, traditions, comportments, musics, books, diaries, and other cultural expressions. (p. 108)

Clifford also outlined the concept of "dwelling-in-travel" (p. 102), whereby travellers develop routines and rhythms of tradition and culture that appear to belie the transience stereotypically associated with the status of traveller.

Clifford's identification of the "strongcultural, political, and economic compulsions" motivating travellers and the "range of material, spatial practices" associated with travel, and his reference to "dwelling-in-travel", are a convenient way of reviewing this chapter by returning to its starting point. I began the chapter by proposing, on the basis of the preceding literature review, the need to avoid the twin conceptual 'dead ends' of 'unproblematic othering' and 'unproblematic celebration', and also that itinerancy involves distinctive relationships between people and space, a unifying theme that linked the disparate sections of which this chapter is composed.

I began by outlining my justification for applying de Certeau's (1984, 1986) concept of 'tactics of consumption' to this study. In particular, I asserted the potential value of analysing the show people's actions as examples of their exercise of agency in the form of 'tactics of consumption' intended to make the marginalised 'spaces' of itinerancy into 'places' that they would find meaningful and productive. At the same time, I noted the

considerable ambivalence with which commentators have responded to one or other of his so-called 'binary oppositions', arguing that many of these critiques are misconceived on the basis of appealing to the empirical setting of the show people. On the other hand, I expressed my own ambivalence about de Certeau, particularly my concern about the evident incapacity of 'tactics of consumption' to turn 'spaces' into 'places' on other than a temporary basis.

This apparent inability to change 'the rules of the game' justified my adoption of elements of Bakhtin's thought into the study's conceptual framework. Specifically, I contended that 'outsidedness' and 'creative understanding' (Bakhtin, 1986a) provide a way of theorising how the show people move beyond marginalisation and resistance into transformation of their experiences of itinerancy, making possible more enduring – although no less contingent and provisional – 'places' for them to occupy as their "dwelling-in-travel" (Clifford, 1992, p. 102).

Finally, I essayed a synthesis of this thesis's conceptual framework. I reiterated my focus on the spaces of itinerancy, and my interest in the consequent marginalisation, resistance and transformation of those spaces. I proposed that 'tactics of consumption', 'outsidedness' and 'creative understanding', reflecting different aspects of that marginalisation, resistance and transformation, were appropriate lenses for analysing the show people's educational experiences. Analysing the show people's lives in this way, I argued, would have significant implications for extending, and for functioning as a counternarrative to, the existing literature on Australian shows and Traveller education.

This chapter has therefore used the preceding literature review as the basis for elaborating a conceptual framework to analyse the data collected in this study, a framework whose logic is framed and nurtured by avoiding an essentialist approach to understanding itinerancy. I turn now to consider the research design underpinning the collection and analysis of those data, which are centred on communicating my understanding of 'learning on the run' for the Queensland show children and their families.

CHAPTER FOUR

THE RESEARCH DESIGN

"The ethics and politics of qualitative research. ..[are] a swamp and. ..I have provided no map. Each individual will have to trace his or her own path. This is because there is no consensus **or** unanimity on what is public and private, what constitutes harm, and what the benefits of knowledge are."

Punch, 1994, p. 94

"Everyfocus excludes; there is no politically innocent methodologyfor intercultural interpretation."

Clifford, 1992, p. 97

4.1 Overview of the chapter

In Chapter Two of this thesis, I demonstrated that the literature on Traveller education is the site of interactions among marginalisation, resistance and transformation, and that those interactions can be traced to a general devaluing of the physical and symbolic spaces of itinerancy. I also sought to position the thesis as occupying the 'middle ground' between the two equally unhelpful and disabling constructions of itinerant people as either an 'unproblematic othering' and an 'unproblematiccelebration'. In Chapter Three, I argued that conceptualising itinerancy as involving complex and distinctive relationships between people and spaces provided the foundation of a potentially instructive conceptual framework that combines de Certeau's concept of 'tactics of consumption' and Bakhtin's notions of 'outsidedness' and 'creative understanding'. These terms in combination explain how it is possible for the show people both to resist and ultimately to transform the marginalisation associated with itinerancy, by moving the old 'rules of the game' to the new terrain of a countemarrative about itinerancy and Traveller education. All these concepts have implications for how the study was designed and conducted.

The function of this chapter, therefore, is to outline and justify the study's research design, by making explicit the relationship between theory and methodology. In it I explore my relationship with the participants in the study, viewed through the lenses of my research focus and conceptual framework. The chapter consists of the following sections:

marginalisation, resistance and transformation and researching Traveller education

- 'tactics of consumption' and researching Traveller education
- 'outsidedness' and 'creative understanding' and researching Traveller education
- the ethical and political dimensions of the study
- the research design of the study
- the data gathering techniques of the study
- the delimitations and limitations of the study.

I begin in Sections 4.2, 4.3 and 4.4 by highlighting the central methodological issues that arise when considering the key theoretical concepts employed throughout the thesis. In other words, while I have emphasised in the previous chapter the way that de Certeau and Bakhtin inform my 'head work' or view of the world and research, in this chapter I explore the practical 'field work' implications of this 'head work'. Although theory and practice cannot be separated, while Chapter Three emphasised the kind of analysis to expect, Chapter Four explains how the data for the thesis were collected in a de Certolian and Bakhtinian style. I follow this with an explicit discussion of how the research was designed.

4.2 Marginalisation, resistance and transformation and researching

Traveller education

If show people are routinely marginalised from 'mainstream' society on account of their itinerancy, it follows that a research project about Traveller education – about the education of itinerant people – has the potential to perpetuate that marginalisation. This is because the focus of the research is on what makes the people 'different' from, and therefore often 'inferior' to, others. It is also tied to the unavoidable fact that university research is generally more closely aligned with 'mainstream' than with 'marginal' education. That such an outcome is unlikely to be the researcher's explicit intention does nothing to nullify the point that at several points in the research the itinerant people's difference will be highlighted and emphasised (even though itinerancy is only one of the multiple markers of their identities from which they derive meaning and purpose).

A likely response to this marginalising situation is resistance. Firstly, it is to be expected that show people who are routinely marginalised are liable to approach a research project emanating from an institution that might be construed as complicit in their ongoing discrimination with scepticism, even hostility. Furthermore, it would be understandable if those show people engaged in resistance at the first sign that the project was contributing to their further marginalisation, by 'closing down' the range of their responses and even by refusing to continue their participation in the project. Secondly and conversely, the project might be seen as an opportunity to enlist the researcher's assistance in resisting the show people's marginalisation. If this

occurred, the project might be considered to be discharging a useful social responsibility; on the other hand, some would argue that the researcher would need to ensure that her or his detachment was retained.

The potential mediator between marginalisation and resistance in the context of the 'researcher'-'researched' relationships is transformation. This point recalls Anyanwu's (1998) definition of 'transformative research':

Transformative research is a systematic enquiry into the real conditions which create oppression or hinder self-determination. It produces reflective knowledge which helps people to identify their situation and in doing so, to change such [a] situation for the better. In this regard, transformative research plays the important role of supporting the reflective process that promote positive change. (p. 45)

Just as transformation is conceived in this thesis as breaking down the oppositional binary of marginalisation and resistance, thereby allowing 'the rules of the game' to be challenged and overturned, so transformative research can be seen as moving the 'researcher'—'researched' relationship out of a formalised set of conventional and routinised interactions into something much more positive and unpredictable, whereby people are genuinely understood as individuals outside the enactments of their 'roles'. Furthermore, this is a significant element of what I mean when I state that I seek to make the thesis a 'counternarrative'. By circulating images and understandings of show people gleaned from intensive interactions with them, I hope to contribute to a process of counteracting the negative and inaccurate stereotypes about them already circulating. The transformative dimension of this study therefore assists in the vital process of investigating "the real"

conditions which create oppression or hinder self determination", and "in doing so" helps "to change such [a] situation for the better" (Anyanwu, 1998, p. 45).

As I elaborate in Chapter Eight, Trinh T. Minh-ha (1990) captured something of this 'forward lookingness' of transformation when she observed:

Inevitably, a work is always a form of tangible closure. But closures need not close off; they can be doors opening onto other closures and functioning as ongoing passages to an elsewhere (-within-here). (p. 329)

That is, provided that the researcher is attentive to these possible "doors opening onto other closures", the transformative potential of a research project can promote productive understandings that might otherwise never have come to fruition.

I contend that, regardless of the research design involved, any study of Traveller education will encounter the eddies and flows around marginalisation, resistance and transformation. These eddies and flows move constantly through the spaces of itinerancy, and will always do so as long as an itinerant lifestyle is inherently devalued in contrast to its more socially acceptable 'other'. The challenge for the researcher into Traveller education is to explain and justify how her or his research project engages with the connections among marginalisation, resistance and transformation, how the project communicates in a 'trustworthy' way the selected aspects of the Travellers' lives, and how the project proceeds in an ethically responsible and politically responsive manner. My response to this challenge is inextricably linked with

my combination of 'tactics of consumption', 'outsidedness' and 'creative understanding' in the study's conceptual framework, as I discuss below.

4.3 'Tactics of consumption' and researching Traveller education'

In Chapter Three I noted that commentators have reduced a number of de Certeau's paired concepts to binary categories. I challenged that reduction, which I argued reflected those commentators' ambivalence about de Certeau's conceptualisation of the links between marginalisation and agency. On the other hand, I also asserted that the difficulty of de Certeau's thinking in changing 'the rules of the game' justified my deployment of Bakhtinian 'outsidedness' and 'creative understanding' beside de Certeau's notion of 'tactics of consumption'.

This reminder of de Certeau's paired concepts is important in this account of the implications of 'tactics of consumption' for the design, conduct and reporting of this study. Specifically, de Certeau's juxtaposition of 'writing' and 'reading' as a paired category forms the framework for that account. For de Certeau, reading constitutes a set of tactics of consumption that enables readers to resist the marginalising and totalising strategies of textual production —that is, of writing (1984, p. xxi). I interpret this argument as applying to this study in two respects: the ways in which the show people have 'consumed'/'read' the research project; and the ways in which I have 'produced'/'written' that project.

Turning first to 'consumption'/'reading', and following de Certeau's proposition that "Reading thus introduces an 'art' which is anything but passive" (1984, p. xxii), I assert that the multiple individuals making up my term 'the show people' (itself a textual practice with significant implications for consumption/production) have engaged in several kinds of 'reading' in regard to the research project on which this thesis reports. They have 'read' my initial contact, usually through a 'third party' such as one of the teachers involved in the education program, a member of the Showmen's Guild of Australasia or another person on the show circuits. They have 'read' the information sheet and the request to sign the consent form with which I furnished them. They have 'read' my verbal explanation of why I was conducting the project and why I wished to tape record the requested interviews. They have 'read' the questions that I posed to them during the interviews. They have 'read' such nonverbal indicators as my appearance, my tone of voice, my facial expression and my handshake. They have 'read' the subsequent reports on my research findings, and the associated publications, sent to the Showmen's Guild. They have also 'read' one another's responses to those same activities and events.

I indicated above my understanding of how 'tactics of consumption' (including 'reading') relate to the intersections among marginalisation, resistance and transformation in relation to the spaces of itinerancy. What I wish to emphasise here is that these acts of 'reading' by the show people have in turn consciously and unconsciously influenced my continuing conduct of the research project. That is, the study's 'production'/'writing' has taken account in several ways of how the show people have 'consumed'/'read' it. I elaborate this crucial point below as I seek both to acknowledge the potential

charges against 'production'/'writing' as replicating marginalisation and to demonstrate how I have attempted to avoid those practices and outcomes.

In contrastfopposition to the show people's 'consumption'/'reading' of the research project, the relevance of de Certeau's work to the study is that my 'production'/'writing' is potentially the site of marginalisation of the show people's agency. De Certeau (1988) referred to the writing of history contributing to "replacingthe obscurity of the lived body with the expression of 'will to know' or a 'will to dominate' the body" (p. 6), or to becoming "writing that conquers" because "Itwill transform the space of the other into afield of expansion for a system of production" (de Certeau, 1988, pp. xxv-xxvi). Relatedly, de Certeau (1984) referred to "the scriptural economy" (p. 131) and to the "textual practices" (p. 80) associated with scholarship. Conley (1988), the English translator of de Certeau's work The writing of history (1988), argued that both rhetoric and history "evincea 'strategy' that wills to efface, marginalize, or even repress more complicated and ambivalent designs" than those that they produce (p. x).

Similarly, Rowan (1991) identified universities as being "inextricably bound up with western society's desire to maintain power over those who are different" (p. 11); this is a Qrect warning to a university doctoral candidate and researcher. Furthermore, Rowan (1991) has analysed 'explanation', based as it is "on a distinction between those who explain and those who are explained" (p. 6), as a major 'strategy of marginalisation' in de Certeau's terms, on account of its imbrication in efforts to 'capture' (and thereby potentially to essentialise and naturalise) the 'difference' of the objects of study. This is particularly so if that 'difference' becomes equated with 'lack'

(Rowan, 1991, p. 4), as though the people who are 'different' would be 'all right' and no longer 'deviant' if only they could have whatever it is that separates them from the majority or the 'mainstream'. Correspondingly, Spivak (1990) has highlighted explanation's complicity as a marginalising strategy, such as in her observation, "My explanation cannot remain outside the structure of production of what I criticise" (p. 384).

My response to these potentials for marginalisation is initially to acknowledge that this thesis and the project on which it reports are textual practices, and are certainly not "being written out of any objective/scientific perspective" (Schirato, 1993, p. 283). A corollary of this acknowledgment is that I claim neither to 'speak on behalf' of Queensland show people nor to have created an opportunity for them to 'speak for themselves'. I am fully imbricated in the process of producing this text, in editing the interview transcripts, in selecting quotations to present in this thesis and in presenting my analysis of those quotations. In doing so, however, I have sought to highlight what I have perceived as evidence of the show people's 'tactics of consumption', this intention being the opposite in my view of efforts to 'efface' or 'marginalise' the show people. I have also sought to minimise the prospect of the charge levelled at academic discourses by Rowan (1994) – "...the strategy, in order to better secure its own authority, must explain 'other' groups in terms which emphasise the strength **a** the centre" (p. 26) being 'broughthome' to this study. On the contrary, I have emphasised "the strength" of the show people in contesting and changing the power of "the centre".

A related strategy, according to Rowan (1991, pp. 8, 11), derives from the tendency to 'exoticise' the 'difference' of the objects of study. In the case of the show people, this could happen relatively easily: the stereotypes surrounding occupational Travellers identified in Chapter Two are centred on show people's constructions as 'strange' and 'exotic' as they travel from town to town. My response is to design the research project in such a way that it will facilitate my capacity to discern and document, rather than cover up and ignore, the show people's multiple signifiers of identity. This requires me to interact with the show people in a variety of settings, to meet and speak with as many of them as possible and to design and conduct interviews in ways that will allow these multiple signifiers to be become apparent and explicit.

A crucial issue for the research design of this study is the interpretation of 'tactics of consumption' being metaphorical rather than empirical (Danaher, 1995,p. 138) and therefore as not being amenable to analysis in a thesis such as this. I respond to this issue by stating my conviction of the utility of the deployment to several empirical sites of de Certeau's notion of 'tactics of consumption' by several commentators (Balides, 1993; Rowan, 1994; Weinstein, 1994) to justify my claim that de Certeau's restriction of 'tactics' to the metaphorical realm does not invalidate my application of his concept to the empirical site of the Queensland show people's engagement with the education authorities. Not to do so would be to agree with the critics who say that de Certeau's analysis is ultimately static because it cannot accommodate change. For me, the important aspect of 'tactics' is their effects: how the show people *can* use 'tactics' to resist and subvert marginalisation.

My justification for this approach is twofold. Firstly, I concur with Rowan's (1994) assertion that

...it is problematic for a narrative to criticise various political practices **f** it goes on to suggest the complete and abjectfailure of any attempt by a marginalised individual or group to challenge the system that produces those practices. (p. 155)

In other words, what is the 'point' of a conceptualisation of 'tactics' that is unable to point to productive change in the material circumstances of the marginalised? Secondly, I cite Danaher's (1995) rationale for his appropriation of Foucault's ideas to his account of Australian histories:

...my intention has been to reconstitute Foucault's project in the interests of constructing a theoretical position which might productively engage in a critique of Australian historiography. I have not attempted to produce a 'faithful' account of Foucault's work, but rather one that has sought to emphasise its utility in analysing the making of a modern community out of its original status as a convict colony, and the historiographic treatment of this process. (p. 229)

For me, my extension of 'tactics of consumption' to an empirical site remains "'faithful"' to de Certeau's commitment to understanding how marginalising and totalising strategies can be and are resisted.

A related issue is the research design implications of the sheer number and scale of possible 'tactics'. That is, recognising that 'tactics' can be as diverse as reading 'against the grain' of a text and actively campaigning for change requires me to allow space in the ways in which I plan and conduct the

research to identify the multiple kinds of responses that are potentially 'tactics'. Furthermore, in creating that space I must avoid hierarchising and thereby privileging one response over another, for to do so would be to deny the agency of the research participants and construct them as passive objects of study. A vital benefit of this approach to research design is that it allows me to foreground resistance and transformation in my analysis.

Finally, some commentators have highlighted de Certeau's analysis of 'stories' as "political tactics" and as "an ideological move in a long-running political game" (Schirato, 1993, p. 288), and as "thestones by which we define ourselves, and what it means to be who we are" (Buchanan, 1996b, p. 150; see also Ahearne, 1995, p. 25). In this context, de Certeau asked the highly pertinent question about quoting voices in more recent approaches to writing history: "...who is speaking? to whom?" (1984, p. 157). Rowan (1991) provided a provocative answer to this rhetorical question:

The explanations [contained in academic discourses] and those who produce them work constantly to ensure that the voice of the 'other' is heard by only two groups: those who ... are also marginal and those who will not be affected; and in most cases this second group is composed of the people who formulate the explanations in the first place. (pp. 10-11)

When I extrapolate de Certeau's question (and Rowan's response to that question) to this study, it means that the show people's interview transcripts should be understood, not as deliberate efforts to mislead me, but instead as further examples of 'tactics of consumption' of the research project. My response needs to be, not oppositional 'strategies of marginalisation', but rather efforts to analyse the show people's understandings of the study and

what those understandings reflect about the show people's resistance and transformation of marginalisation in the spaces of itinerancy. My response needs also to use the interview transcripts as the basis for 'outsidedness' with and 'creative understanding' of the show people, as I elaborate in the next section of this chapter.

I argue, therefore, that de Certeau's notion of 'tactics of consumption' has crucial implications for both the research design of this study and the conduct of research into Traveller education more broadly. Firstly, it is important to acknowledge, not deny the possibility and potential of, the binary categories 'consumption'/'reading' and 'production'/'writing'. Secondly, this acknowledgment means that I need to pay conscious attention to both how the show people 'read' the research project and how I 'write' that project, deliberately eschewing, for example, strategies of explanation and exoticisation in favour of less marginalising and more self-conscious textual practices. Thirdly, my claim that 'tactics of consumption' can function at empirical as well as metaphorical levels needs to find its way explicitly into the data analysis chapters of this thesis. Fourthly, I need to understand the show people's interview transcripts as 'moves in the game' and as efforts at 'selfdefinition', and crucially as prompting the question "who is speaking? to whom?" (de Certeau, 1984, p. 157) to which I must respond. My intended contribution to methodological knowledge consists partly of relating these indispensable implications to the broader conduct of Traveller education research, which in turn involves partly an understanding of relations between 'researcher' and 'researched' in terms of, and as framed by, 'tactics of consumption'.

4.4 'Outsidedness' and 'creative understanding' and researching Traveller education

In comparison and in complementarity with de Certeau's concept of 'tactics of consumption', the Bakhtinian notions of 'outsidedness' and 'creative understanding' are intended in the research design of this study to heighten my identification of the show people's consumption of the research project while minimising their marginalisation from it. 'Outsidedness' and 'creative understanding' require that I identify, record and respond to the multiple voices of the research participants, and that I ensure that the study changes as a consequence of what those voices say. This approach requires attending to the show people's resistance and prospective transformation of their marginalisation as itinerant people occupying several spaces. This approach is also in keeping with my argument in the previous chapter that these analytical resources are indispensable both conceptually and methodologically in the study, by highlighting the political significance of the show people's actions that make such transformation possible.

For Bakhtin, the utterance can be considered the basic unit of communication (Jones, 1993, p. 252; Schirato & Yell, 1996, pp. 76-79). Utterances, which can be parts of sentences, complete sentences or much larger pieces of text, are distinguished from sentences by their 'living' social or interactive dimension: "An utterance always evaluates; every utterance has ethical import" (Jones, 1993, p. 252). The methodological significance of the

utterance for this thesis is that it rightly focusses attention on the centrality of what people say and write in the context of with whom, how and why they are communicating. In practical terms, this means that I have not undertaken a formalised discourse analysis of the interview transcripts for this study, but instead I have analysed what the interviewees have said in relation to the study's research questions and focus (marginalisation, resistance and transformation in the spaces of itinerancy) and its organising concepts ('tactics of consumption', 'outsidedness' and 'creative understanding').

Or to put it another way, if de Certeau prompts me to design my study to identify and document, rather than cover up and ignore, the multiple signifiers of identity among Queensland show people, Bakhtin leads me to look for those signifiers specifically in the realm of language – both my own and that of the people whom I interviewed. To that end, the following data analysis chapters report my efforts to explicate the show people's complex and subtle understandings of themselves and others, mediated and revealed through their interactions with me as researcher.

A crucial corollary **of** this emphasis on language as the site of identity construction and contestation is the related concept of 'voice'. According to the writers of the glossary to Bakhtin's book *The dialogic imagination: Four essays* (1981), voice "...is the speaking personality, the speaking consciousness. A voice always has a will or desire behind it" (p. **434)**. Under de Certeau's scheme, the voices that are most commonly heard are those associated with 'strategies of marginalisation'; voices linked with 'tactics of consumption' are almost always silent, lacking as they do **a** normalised 'spealung position' from which to communicate with authority. This definite

and explicit hierarchy is the antithesis of Bakhtin's vision of utterances, of which a deliberate non-hierarchisation of voices is a non-negotiable prerequisite.

Clearly Bakhtin's vision behoves me to use every methodological means at my disposal to achieve a non-hierarchisation of voices in my design and reporting of this study. Yet he – and I – recognise(d) the methodological difficulties in attaining this vision. One example must suffice. According to Bakhtin (1981):

...the speech of another, once enclosed in a context, is — no matter how accurately transmitted — always subject to certain semantic changes. The context embracing another's word is responsible for its dialogizing background, whose influence can be very great. Given the appropriate methods for framing, one may bring aboutfundamental changes even in another's utterance accurately quoted. Any sly and ill-disposed polemicist knows very well which dialogizing backdrop he [sic passim] should bring to bear on the accurately quoted words of his opponent, in order to distort their sense. (p. 340)

While I do not consider myself a "sly and ill-disposed polemicist", nevertheless I concur that striving for communication with the show people at a conceptual level does not necessarily translate automatically into communication at a methodological level. This point helps to justify the 'place' of ambivalence in this study: in this case, my own ambivalence about the methodologies of academic research projects. In practical terms, this requires me to evaluate every utterance – both the interview data and my analytical

statements about those data – to ensure, to the extent that I can do so, that the show people's voices and my voice are heard together.

As part of that process, and as I indicated in Chapter Three and reiterated at the beginning of this chapter, 'outsidedness' is a Bakhtinian concept that has great relevance for this study's conceptual and methodological dimensions. As I noted in Chapter Three, Morson and Emerson (1990), Bakhtin's first biographers, explained outsidedness in this way: "When one person faces another, his [sic passim] experience is conditioned by his 'outsidedness.' Even in the physical sense, one always sees something in the other that one does not see in oneself: I can see the world behind your back..." (p. 53). Although the bases of 'outsidedness' could vary considerably, including "personal, spatial, temporal, national, or any other" (p. 56), "outsidedness creates the possibility & dialogue, and dialogue helps to understand a culture in a profound way" (p. 55).

As a further illustration of outsidedness, a philosopher colleague (Ezra Heymann, personal communication, 23 April 1999) listened carefully to my synthesis of Morson and Emerson's (1990) explanation of this concept as the ability to see the back of someone else's head (something that one cannot do oneself, and that highlights the other person's outsidedness from one), then he responded that a former fiancée used to tell him when he was putting on his 'winter face'. This statement, which referred to how his countenance looked to others (presumably denoting some combination of introspection, discouragement and pessimism), was one of the philosopher's chief grounds for sadness at the end of the relationship: that another person would no longer be there to tell him when he was wearing his 'winter face'.

Thus outsidedness evokes two attributes simultaneously: a certain amount and kind of separation from the other person; and sufficient interest to pay attention to the other person. Methodologically, outsidedness constitutes the basis of my response to a hypothetical objection that a non-show person, who has never himself lived an itinerant lifestyle, cannot possibly attain sufficiently 'trustworthy' understanding of an itinerant people for that understanding to be considered 'reliable' and 'valid'. On the contrary, I argue that – provided that I have the sufficient interest referred to above – my separation from the show people enables me to perceive and identify elements of their multiple signifiers of identity of which they are themselves unaware.

This most emphatically does not denote the voyeuristic look of the omniscient 'expert', gazing with scientific detachment at human specimens, nor lay any claim to objectivity or neutrality. One reason that it does not do so is that my 'interested separateness' is 'a means to an end', not an end to itself. That is, my analysis of the show people's signifiers of identity of which they are unconscious becomes 'meaningful' and 'truthful' only when I use that analysis to augment my growing comprehension of their situation. In other words, outsidedness is a means to the end of creative understanding (both the show people's and my own); in this way, I strive to give equal attention and value to the show people's and my voices.

Similarly, it is helpful to note the evident link between outsidedness and ambivalence, which is a recurring theme in this thesis. I have discerned implicit ambivalence on the show people's part in their perceptions of the research project – or perhaps more accurately in their status **as** 'research subjects'. Our interactions have been irregular and intensive, with months without any contact

being interspersed with data gathering over two or three days. Now that the data gathering has been completed, I have heard indirectly from the show people, such as when someone seeking information about Traveller education contacts me with the words, "The show people suggested that I call you". I interpret this situation as manifesting the show people's ambivalence, in the sense that progressively as the show people got to know and trust me they were prepared to identify me to others as someone who would discuss their experiences and aspirations sympathetically —that is, who would respect their agency without adding to their marginalisation. Given that they see themselves as being accustomed to marginalisation, this ambivalence, or willingness to 'suspendjudgment' until they knew more about my intentions, was a crucial pre-requisite to the conduct of the study. It was also an example of how outsidedness can lead to creative understanding.

In that context, it was helpful that, as I noted in Chapter Three, Bakhtin (1986a) provided the following overview of the links between outsidedness and creative understanding:

Creative understanding does not renounce itself, its own place in time, its own culture; and it forgets nothing. In order to understand, it is immensely important for the person who understands to be located outside the object of his or her creative understanding—intime, in space, in culture. For one cannot even really see one's own exterior and comprehend it as a whole, and no mirrors or photographs can help; our real exterior can be seen and understood only by other people, because they are located outside us in space and because they are others. (p. 7; emphasis in original)

In the previous chapter, I outlined the analytical utility of creative understanding in helping me to identify the ways in which show people engage in such understanding as a basis of the transformation of their marginalised status. Here it is appropriate to emphasise the methodological significance of creative understanding defined in this way for my enactment of the responsibilities and roles of researcher. That methodological significance is the crucial importance of making the study a genuine exchange of information and ideas. By this I mean that, in addition to requesting show people to respond to my developing analysis of their marginalisation, resistance and transformation, I need to listen and respond to their outsidedness and creative understanding in relation to the research project. If I am to participate in practices of mutual comprehension with the show people, I must be open to their explicit and implicit comments on the purposes and conduct of the research project, and I must recognise that they, as much as I, have constructed and carried out the study. This means, for example, that I must seek to hear the show people's voices about topics other than those about which they are ostensibly speaking in the interviews, and strive to relate those voices to my developing answers to the study's research questions.

So the methodological implications of Bakhtinian outsidedness and creative understanding for researching Traveller education can be synthesised as follows. Interested outsidedness and creative understanding can function as an 'antidote' to a reductionist rendering of de Certolian marginalising 'strategies' and resistant 'tactics'. It can operate in this way through productive use of the ambivalence attending the spaces of itinerancy and my own ambivalence about Traveller education research that prompts my openness to the multiple signifiers of the show people's identities manifested

through language, specifically the Bakhtinian notion of the utterance. In these ways, the interview transcripts that are this study's principal data gathering technique can become the basis of ongoing and mutual comprehension between 'the researcher' and 'the researched', and in the process can become the site in which 'strategies' and 'tactics' 'speak' to one another and thereby lead to transformation. A crucial 'sign' whether this possibility is being actualised is the extent to which multiple voices are heard and responded to mutually and non-hierarchically in both this thesis and the study on which it is based. Another, equally vital 'sign' is whether an analysis framed by 'outsidedness' and 'creative understanding' enables a conception of itinerant people that moves beyond the 'unproblematic othering' and the 'unproblematiccelebration' of itinerancy identified in Chapter Two.

4.5 The ethical and political dimensions of the study

The location of the study's research design in the intersections among marginalisation, resistance and transformation in the spaces of itinerancy, and the design's foundation in de Certeau's 'tactics of consumption' and Bakhtin's 'outsidedness' and 'creative understanding', posit several crucial issues pertaining to the study's ethical and political dimensions. Rather than emphasising an artificial separateness between these dimensions, I have elected to consider them as related aspects of the considerations and constraints informing and framing my conscious and unconscious decisions about the collection and analysis of the research data. In particular, I argue that the research involves the exercise of ethical responsibilities for both the

interviewer and the interviewees, and that the research is politically charged rather than value neutral. In that context, manœuvring through and around these potential ethical and political minefields brings to life the three dimensions of itinerancy outlined above, as well as the other participants' and my 'tactics of consumption', 'outsidedness' and 'creative understanding'.

Despite their emphasis on philosophical and textual, rather than empirical and practical, forms of interaction, both de Certeau and Bakhtin were very well aware of the ethical and political dimensions of their respective studies, and were equally conscious of the complexities and subtleties attending those dimensions. For example, Smith (1996) contended that de Certeau's work, "like the thought of Michel Foucault, is of the epistemological/ethical orientation" (p. 19) in philosophical thinking, and that one clear manifestation of that orientation was the "meticulously conscientious attention to difference" (p. 20) characteristic of his writing. Similarly, and again by way of illustration, commentators have generally rejected charges that Bakhtin was amoral or unethical on account of his studies of carnival, and have emphasised the ethical responsibility on which several of his concepts were based. For example, according to Emerson (1997):

(It is worth noting that Bakhtin's vision of outsidedness is wonderfully nonelitist, nonjudgmental, and open to all, whatever our gifts or inclination. He does not stipulate that we do the other party any positive good, only that we assume an outside position towards that party. Even the laziest and most passive outsider can always help me out by letting me know what is happening behind my head; in my laziest, most passive, most testy and unengaged moods I can render outsiders at least that much of a service.) (p. 210)

The foregoing account suggests that the research practices of de Certeau and Bakhtin are compatible with the tradition identified by Jenkins (1992) as the "newer conception of ethnographic authority", encapsulated in "a shift from totalizing accounts of social and cultural processes toward partial, particularized, and contingent accounts of specific encounters within and between cultures" (p. 4). At the same time, I recognise and, given my focus on the ambivalence of itinerant spaces, applaud the ambiguity that this approach to conducting and writing research prompts. In other words, the replacement of an objective, neutral, even omniscient researcher with a human being whose values and cultural capital are fully inscribed in the research process constitutes the beginning, not the end, of the ethical and political issues surrounding research. Thus I can empathise with the feminist poststructuralist writer Patti Lather's (1994) reference to being/feeling "Situated, partial, perplexed" (p. 41). I also endorse Punch's (1994) vivid metaphor:

The ethics and politics of qualitative research...[are] a swamp and...I have provided no map. Each individual will have to trace his or her own path. This is because there is no consensus or unanimity on what is public and private, what constitutes harm, and what the benefits of knowledge are. (p. 94)

The fact that I am engaged in this study reflects my conviction that these methodological difficulties need to be acknowledged but that they should not lead to inaction or paralysis. "Situated, partial, perplexed" I most certainly am, but I am guided and encouraged in the design of this study by my selection of the reference points of my "map": 'tactics of consumption', 'outsidedness' and 'creative understanding' interacting with one another and

with the show people's and my marginalisation, resistance and transformation as I proceed along my "ownpath" through the spaces of itinerancy.

I turn now to illustrate this argument by reference to some of the methodological issues confronting the design of this study, and particularly to their ethical and political dimensions. Several of these issues relate to the risks attendant on planning and conducting research, and demonstrate once more the integral and intricate connections among marginalisation, resistance and transformation in research design. I have referred elsewhere (Danaher, 1998b) to three among several hypothetical interactions between myself and the show people: advocacy (which is also identified by Lather [1992, p. 91]); appropriation; and complicity in perpetuating the show people's purported marginalisation. While I acknowledged the possibility of each interaction and sought to 'defend' myself against each 'charge', the point to emphasise is my recognition of the contingency and fluidity, rather than the finalisability (in Bakhtin's words), of this kind of methodological 'settlement'. That is, the spaces of itinerancy contain the shifting flows and multiple signifiers of the 'researcher'-'researched' relationship, and each new encounter has to be approached anew from this perspective. (Relatedly, Ahearne [1995] explained how, in de Certeau's analysis of the writings of French mediaeval mystics:

Certeau shows how the writings of the mystics both exacerbate and seek to overcome the gap which separates the human subject **from** the language he or she speaks. This gap was not effaced by their texts. They were always compelled to begin again. [p. 115])

Some of these flows and signifiers of the 'researcher'-'researched' relationship are manifested in the practices and procedures commonly associated with obtaining ethical clearance to conduct research. According to

Punch (1994), "In essence, most concern revolves around issues of harm, consent, deception, privacy, and confidentiality of data" (p. 89). Evans and Jakupec (1996) helpfully elaborated these concerns into six "key issues" (p. 79) attending the ethical conduct of research in open and distance learning:

...(i) concern and moral obligation of the researcher to respect privacy and integrity of individuals, (ii) power and empowerment of researcher and researched, (iii) covertness, (iv) using secondary data, (v) resolving the conflict between the right of individuals and the right of society to know and (vi) informed consent. (p. 79)

In relation to these "key issues", I have done my utmost to respect the show people's "privacy and integrity" through my textual practice of not ascribing data to individual respondents, and my avoidance of writing about them in ways that could be construed as denigrating or disrespectful. Given my reservations about the emancipatory connotations of the term "empowerment", I prefer to emphasise my recognition of the agency of the show people, while certainly responding to the need to assist research participants "togive free and informed consent" (p. 82) to the research project. I have not engaged in "covert ana' secret participant observations" (p. 82), the research participants at all times being aware of my status as a university researcher, and I have collected the data reported in this thesis myself, rather than drawing on other researchers' data collections. While I acknowledge that there is an enduring tension between the rights of individuals to privacy and the public's right to know, I draw only on data gleaned from the semi-structured interviews that I conducted and from publicly available documents; if I must state a preference for one or other of these 'rights', it is to endorse the research participants' right for matters that they consider 'private' to remain so.

Finally, while I understand that definitions of "informed consent" can be "complicated, problematic and difficult to implement" (p. 86), I have done my utmost to ensure that the research participants have understood, as comprehensively as possible, the purposes and intended uses of the research project, and that their agreement to participate derives from a tactical and/or creatively understanding perspective.

Evans and Jakupec (1996) summarised their six "key issues" by claiming that

...the two basic principles researchers in open and distance education should follow are: (i) research in open and distance education should not expose individuals to risks of **or** cause unjustified political, personal, economic, physical, emotional, moral or psychological harm; (ii) researchers in open and distance education ought not to undertake research which violates principles of free informed consent. (p. 91)

I applaud the authors' efforts to synthesise a field of such patently shifting sands, and I attest that, to the best of my partial and situated knowledge, I have not infringed either of these "basicprinciples" in this study. Furthermore, although, as I signalled above, I am uneasy about claiming that my research is necessarily 'emancipatory' or 'empowering' of the participants, nevertheless I contend that some more limited benefits, ethically grounded and politically understood, might be regarded as accruing to the participants in the study. Specifically, I hope that the research project will give the participants information about the spaces of itinerancy, and how others perceive the show people, that they might find helpful in transforming their 'tactics of consumption' into more enduringly productive social change. Similarly, I hope that non-itinerant people's understandings of show people

will be considerably enhanced through the information presented in this thesis.

Another element of conducting this study, already referred to in the preceding section of this chapter, needs to be considered here for its ethical and political aspects. This element pertains to my position as a non-itinerant person writing a thesis about itinerant people. Earlier I justified this position by drawing on Bakhtin's notion of 'outsidedness', which I argued was a prerequisite to 'creative understanding'. In the context of this discussion of ethics and politics, I feel obliged to point out that I do not infer from that justification that I have greater understanding of show people than an itinerant researcher would have. That is, I agree with Jenkins' (1992) assertion that the "danger" of potentially conflating one's perceptions with those of the research participants "is not substantially lessened by adopting a more traditionally 'objective'stance" (p. 6). It should be clear that I claim no lund of 'objectivity' in my role as non-itinerant researcher, and that my ambivalent tentativeness about making claims for the 'truthfulness' of my findings is increased rather than decreased by that role. This suggests that being 'reflective' is preferable to claiming to be 'objective', and furthermore that it is appropriate to augment Lather's (1994) reference to researchers feeling "Situated, partial, perplexed" (p. 41) cited earlier to read "Situated, partial, perplexed and self-reflective".

One further issue in conducting the research for this study needs to be mentioned at this stage. This is the distinctive ethical and political aspects of interviewing children. In total I interviewed eighteen children, varying in age from seven to fourteen. All interviews were conducted with the written consent of the children's parents; most interviews were conducted at the local school

where the Brisbane School of Distance Education teachers were worlung with the children during that week. As the data analysis chapters indicate, I was ambivalent about recording and reporting the children's voices except as background noises to their parents' comments. This was not intended to deny or diminish the authenticity of their views or their right to have those views heard (Danaher, Hallinan & Rose, 1998; Kiddle, 1999). On the other hand, the circumstances attending data gathering meant that I had a very limited time to establish rapport with children in order to encourage them to move beyond monosyllabic or brief responses, and in many cases I was unable to achieve this. I was conscious of my ethical responsibilities towards these interviewees, as well as of my political position as an adult and a stranger speaking with them in a formal and potentially threatening situation. However, in this matter I was heartened by the fact that many of their parents were by contrast extremely forthcoming and even voluble.

Having dealt with the ethical and political dimensions of some features of conducting research into itinerancy, I turn now to analyse the same dimensions of selected practices associated with writing about such research. Here I found Denzin's (1994) identification of four "Writing Issues:" "...the interpretation, or sense-making, representation, legitimation, and desiring phases of moving from field to text to reader" (p. 503) helpful. "Sense Making" refers to "making decisions about what will be written about, what will be included, how it will be presented, and so on" (p. 503). I have been guided in this decision making by the deployment of the concepts underpinning the thesis, as well as by my developing observations and understandings underpinning the production of related publications.

In relation to "Representation", I acknowledge Denzin's (1994) points that "Representation, of course, is always self-presentation" and that "The Other who is presented in the text is always a version of the researcher's self" (p. 503). Among other things, I take this to refer to my having a much greater interest in resistance and transformation than in marginalisation (although clearly these phenomena are dialectically related), and to my enduring awareness of the ambivalence underlying my relations with others and theirs with me. More broadly, I also endorse Denzin's awareness that "...even when 'we' allow the Other to speak, where we talk about or for them, we are taking over their voice", and his recommendation, supported by a reference to Bakhtin (1986a), that "A multivoiced as opposed to single-voiced text can partially overcome this issue...." (p. 503).

With regard to "Legitimation", and to the "Traditional Joundationalist topics such as reliability, validity, and generalizability" (p. 503), I prefer the term 'trustworthmess' as a criterion for evaluating the study's contribution to methodological knowledge. That is, given the ethical and political issues to which I have referred, I understand that a different researcher conducting a similar investigation would collect and analyse data in very different ways, with different outcomes and effects from those of this research project. My expectation is that judgments about the 'soundness' and 'appropriateness' of this project will focus on the extent to which my applications of 'tactics of consumption', 'outsidedness' and 'creative understanding' as my key concepts to the data that I have collected and analysed to answer the study's research questions 'make sense' and are 'legitimate' in terms of my research goals, and in terms of whether those research goals afford opportunities to

generate responsive and useful understandings about the show people's educational experiences and opportunities.

Of the four "Writing Issues" identified by Denzin (1994), I am most ambivalent about "Desire", which "refers to the writing practices that field-workers deploy" and that constitute "Thetopic...[of] the pleasure of the text" (p. 504). I agree that "A vital text is not boring", that "It grips the reader (and the writer)" and that "A vital text invites readers to engage the author's subject matter" (p. 504). However, I disagree with the implicit reduction of "Thepostmodem sensibility [that] encourages writers to put themselves into their texts" (p. 504) to whether the writer writes engaging or boring text. Surely "desire" transcends this rather banal indicator to go to the heart of the writer's and the reader's subjectivities as informed and stimulated by the text's subject matter. This is why attending to the ethics and politics of conducting and reporting research is so vitally important: to use the juxtaposition of pronouns that I seek to disrupt, this study is intended to engage with how 'I', 'YOU' and 'they' understand the world, our respective places in it and our aspirations for strengtheningand/or changing those places.

If the primary impressions conveyed by my discussion of the issues associated with writing about my research project are ambivalence and tentativeness, I shall have succeeded in my aim of emphasising the ethical and political dimensions of this crucial element of the study. In this I take heart from Lather's (1992) rejection of a research methodology that aspires to omniscient objectivity in favour of

...the creation of a more humble scholarship capable of helping us to tell better stories about a world marked by the elusiveness with which it greets our efforts to know it. (p. 95)

For me, awareness of and engagement with this "elusiveness", which reflects ambivalence, de Certolian 'tactics of consumption' and Bakhtinian 'outsidedness' and 'creative understanding', are both pre-requisites and consequences of dealing with the ethical and political dimensions of conducting and writing research into the spaces of itinerancy. At the same time, these principles underpinning the design of this study clearly resonate with broader issues in contemporary theorising around the conduct of research, as well as with more traditional debates in educational research.

4.6 The research design of the study

Bearing in mind all the issues explored to this point and the need to ensure a tight articulation between theory and methodology, I turn now to outline explicitly the major elements of the study's research design. That design was consciously focussed on integrating the literature review, conceptual framework, data collection and data analysis explicated in this thesis. The design consisted of the following five phases identified in Figure **4.1**:

IPHASE	ACTIVITY
One	Identification of research problem;
	phrasing of research questions;
	review of literature
Two	Data collection (involving interviews
	with forty-two people as outlined in
	Figure 4.2 below and document
	collection)
Three	Preliminary analysis (identification
	of major themes evident in
	interviews and documents)
Four	Final analysis (guided by the
	study's analytical concepts: 'tactics
	of consumption', 'outsidedness'
	and 'creative understanding')
?ve	Thesis writing

Figure 4.1: Phases in the study's research design

Inevitably presenting the study's research design in this way suggests that it was far more linear, sequential and straightforward than it actually was. Such a presentation also elides much of the messiness, tentativeness and sheer unease that for me is inextricably involved in designing and conducting educational research. To acknowledge that I encountered some 'false starts', or that I did not approach the thesis writing with maximum efficiency, for example, emphasises that like all human endeavours this study was as much

an emotional journey – with attendant exhilaration, frustration and hope in equal measure – **as** it was an intellectual exercise. More importantly, these acknowledgments underscore the crucial dynamic nature of the project: rather than conforming to the scientific paradigm of the application, modification or rejection of a prescribed hypothesis, the study grew and changed shape in response to the participants' responses and suggestions, and **as** a result of the application of the theoretical resources outlined in the previous chapter.

4.7 The data gathering techniques of the study

I used two techniques for gathering data in this study: semi-structured interviews and document collection. In keeping with the study's conceptual framework, those data gathering techniques are conceived **as** sites that the show people can 'consume' and use **as** opportunities for 'outsidedness' and 'creative understanding' with the researcher, in the process revealing their engagements with marginalisation, resistance and transformation in the spaces of itinerancy. That is, like the other elements of the research design explicated in the previous section, the data gathering techniques are fully integrated with the study's analytical concepts in their goal of collecting information whereby the study's research questions can be addressed.

4.7.1 Semi-structured interviews

The principal data gathering technique used in the study was the semistructured interview, whereby I used a fairly general interview schedule (which appears as Appendix A to this thesis) to guide questions but encouraged respondents to talk at length about other issues as they wished to do so (thereby distinguishing the technique from both a fully structured interview and an unstructured interview). The preference was for researcher and respondent to engage in an informal conversation, rather than to follow rigidly a prescribed interview format. In that context, Appendix A reflects my particular assumptions, foci and interests, and is not intended to be applicable beyond the confines of this study. Most interviews were carried out at the local school used by the itinerant teachers and their students during show week; a few took place at the local showgrounds. Ethically appropriate procedures were adopted throughout the research, including obtaining formal cooperation from the Queensland Department of Education (subsequently renamed Education Queensland) and the Brisbane School of Distance Education, which initially administered the program studied by the show children, and distributing explanatory information sheets, as well as consent forms to be signed by interviewees.

Figure **4.2** represents the distribution of interviews conducted for the study. In keeping with the ethical considerations canvassed in this chapter, names of interviewees have not been included.

YEAR	LOC- ATION	INTER- VIEWS WITH CHILD- REN	INTER- VIEWS WITH PAR- ENTS	INTER- VIEWS WITH HOME TUT- ORS	INTER- VIEWS WITH TEACH -ERS	INTER- VIEWS WITH ADMIN -IST -RAT -ORS	TOT-
11992	Mackay	8	2	1	0	0	11
1993	Bun- daberg	5	5	0	1	0	11
1994	Bun- daberg	2	1	2	0	0	5
1995	Emerald	3	6	0	2 .	0	11
1995	Brisbane	0	3	0	0	2	2
1996	Rock-	0	в	1	0	0	1
2000	Yep- poon	0	ß	D	0	1	1
TOT-	-	18	14	4	3	3	<u>42</u>

Figure 4.2: Interviews conducted for the study

The semi-structured interviews were not longitudinal, in that I did not set out deliberately to speak to the same people each year to identify a developmental dimension of their responses to my questions. However, I did note a

definite change in attitude among many parents to the program provided by the Brisbane School of Distance Education, as I elaborate in Chapter Seven.

With regard to linking this data gathering technique with the study's conceptual framework, it was crucial that interviews were conducted in ways that facilitated the exercise of the participants' resistance and transformation rather than replicated their marginalisation through restricting their 'voices' to passive responses to questions in which I was interested but that had no relevance to them. So interviewees had to feel sufficiently comfortable in talking to me that they would discuss issues of genuine concern and moment to them. Those issues in turn 'fed into' my developing focus on the show people's marginalisation, resistance and transformation as research questions of direct significance to their lives, rather than as *apriori* impositions on the study's research design.

4.7.2 Document collection

Although interviews were the major technique of data collection employed in the study, use was made also of a small number of background policy documents and relevant literature. These documents related mainly to the Brisbane School of Distance Education's organisational structure and the particular program investigated in the present study, as well as some information about the proposed separate school for the show children. These documents were collected with the permission of the participants at appropriate times during the study.

Again relating this data gathering technique to the study's conceptual framework, I worked hard to respect the confidentiality and where appropriate preserve the anonymity of sources in cases where show people and educators

provided me with copies of documents. This approach to data gathering enabled me to collect sufficient information while being 'true' to the ethical and political principles outlined in this chapter and being consistent with the study's conceptual framework, the aim being to make the project 'true' to the principle of 'transformative research' (Anyanwu, 1998) rather than the source of further marginalisation of the participants.

4.8 The delimitations and limitations of the study

This study is a number of things; it is <u>not</u> a number of other things. In particular, it is delimited in two crucial respects. Firstly, it is not a formal evaluation of the educational program selected for study. The intention is not to provide policy makers and program planners with detailed or systematic feedback about the initiative that they have created. Rather, the focus is on the show people's 'consumption' of the program and their 'outsidedness' and 'creative understanding' with the individuals responsible for its implementation on the Queensland show circuits. This does not deny, of course, that the research might well have important implications for educators' reflections on the merit or otherwise of particular educational initiatives **as** a consequence of the study's highlighting of significant aspects of the subjectivity of the categories 'student', 'parent' and 'teacher'.

Secondly, the study makes no claims about the replicability of its techniques or the generalisability of its findings. Although the final chapter suggests possible future avenues of research that might confirm, refine or disconfirm emerging theories, the researcher acknowledges the pervasive and

enduring influence of context and circumstance, and cautions against policy makers, program planners or others who would seek to use the findings to support their claims about other, unrelated educational initiatives. Nevertheless, the framework that I have developed, and the kind of analysis that I have undertaken, could potentially be taken into other educational contexts, suggesting that the framework might be transferable even if the research is not replicable.

The study is limited as well **as** delimited in a number of ways. Firstly, as I acknowledged in Chapter Two, the literature review has concentrated on Australian shows but not on British fairs or American carnivals. Similarly, it has dwelt on the Australian and European literature on Traveller education, to the exclusion of North American studies of 'migrant' education (Mexican fruit pickers in southern states) and other occupational Travellers on that continent.

Secondly, the number of interviews able to be conducted during the study is lower than I would otherwise have desired. The mitigating circumstances were the necessity to adjust to the show people's busy schedule during show week, and the fact that the interviews that were conducted yielded a large number of productive data.

Thirdly, I do not presume that any method – including the one used in this study – can ever be totally inclusive and representative of the 'realities' of social life. As the anthropologist James Clifford (1992, p. 97) has pointed out: "Everyfocus excludes; there is no politically innocent methodology for intercultural interpretation". Given the communication strategies operating on the show circuits, it was considerably easier for me to meet and interview people from the 'upper echelons' of the coastal and western Queensland

show circuits. Another undoubted factor was my status as a university researcher; while many of my interviewees were highly articulate, it is probable that I was never introduced to others with less familiarity with formal education. For these reasons, I acknowledge that this study reports the views of members of the Showmen's Guild of Australasia far more readily than those of 'workers', 'itinerants' and 'horsey people'. This does not mean that the findings are invalidated; it does suggest that the range of voices heard in this thesis is smaller than I would like to be the case.

4.9 Review of the chapter

My intention in this chapter has been to outline the research design of the study, and in the process to explicate and justify the link between the study's conceptual framework and that design. This process has served to focus attention on the connections among the marginalising, resistant and transformative dimensions of the spaces of itinerancy and the sequence of propositions that links de Certeau's concept of 'tactics of consumption' and Bakhtin's notions of 'outsidedness' and 'creative understanding'.

The chapter began with an account of the conceptual and methodological links among marginalisation, resistance and transformation in relation to researching Traveller education. I highlighted my own ambivalence about any research project's potential capacity to perpetuate itinerant people's marginalisation and deny their resistance and transformation, even while a researcher might seek to contribute to transforming that marginalisation and celebrating their resistance. I used that ambivalence as a justification for focusing on de Certeau's concept of 'tactics of consumption', which I argued was significant for drawing attention to two binary categories:

'consumption'/'reading' and 'production'/'writing'. I contended that these binary categories have crucial implications for my relations with the show people and for the 'trustworthiness' of my analysis of their words in the interview transcripts. This led to a discussion of Bakhtin's notions of 'outsidedness' and 'creative understanding' as a useful framework for analysing both my interactions with the show people and their relationships with the staff members of the Brisbane School of Distance Education. The point here is that 'outsidedness' and 'creative understanding' help to make the ambivalence of the spaces of itinerancy potentially productive and transformative, rather than automatically replicating the marginalisation associated with itinerancy, by highlighting the political significance of the show people's actions and moves, including their interactions with me.

The next section of the chapter was my consideration of the ethical and political dimensions of the study's research design. I asserted the ethical responsibility and the political situatedness of the study, and portrayed my traversing of the "swamp" that makes up "The ethics and politics of qualitative research" (Punch, 1994, p. 94) as confirming my ongoing ambivalence and my use of the study's conceptual framework as the 'map' guiding my actions in both conducting the interviews and writing about the interview data.

I reported the data gathering techniques of the study, which included semi-structured interviews with forty-two people and an analysis of documents associated with the show children's education program. Finally, I explained how the study is delimited and limited in particular ways beyond my control.

Having explained how the study's research design both derives from and feeds into its conceptual framework, I have set the scene for reporting my understanding of 'learning on the run' for the Queensland show children. That understanding will be communicated through my responses to the study's three research questions, beginning with the show people's experiences of marginalisation.

CHAPTER FIVE

RESEARCH QUESTION ONE: THE SHOW PEOPLE'S EXPERIENCES OF MARGINALISATION

"Well, see, because of our itinerant nature – here today and gone tomorrow

- . . [people are] less likely to trust you."

Y3P1

"...sometimes they don't understand that we're more or less like them but just travel on."

Y1C4

5.1 Overview of the chapter

I indicated in Chapter One that the three key concepts guiding this thesis are marginalisation, resistance and transformation. The interplay among these concepts was highlighted in the literature review and the conceptual framework, and also in the study's research design. These concepts also underpin the three data analysis chapters, beginning with this chapter's focus on marginalisation.

In responding to the study's first research question, "How do the show people experience marginalisation?", I seek in this chapter to demonstrate that the Queensland show people experience marginalisation in complex and diverse forms. Moreover, these forms are frequently invisible to the gaze of non-itinerant people. It is generally only the experience of moving from one town to another every week or so that makes apparent the myriad ways in which show people are routinely and often unconsciously denied access to services that non-itinerants take for granted. At the same time, the elements of the itinerant lifestyle are habitually invested by 'locals' with negative meanings and devalued.

It is vital to trace the multiple dimensions of the show people's marginalisation for two main reasons. Firstly, it is important in itself to establish the reasons for the show people's awareness that their lifestyle and livelihood marginalise them in the eyes of the 'mainstream' community. This awareness stretches back for generations, and is not far below the surface of a show person's interactions with a 'local'. Secondly, marginalisation is far from being the end of the story, but instead forms the precursor and the impetus to the show people's acts of resistance and transformation, which are canvassed

in the next two chapters. If we are to understand how the show people have succeeded in turning distance education into Traveller education – a 'place' of their own – we have first to comprehend their consciousness of being regularly denied access to 'mainstream' educational and other services.

The chapter therefore consists of four sections:

- a review of de Certeau's (1984) concept of 'strategies of marginalisation', which provides the chapter's conceptual framework
- an account of how the show people experience an absence of place as a consequence of their itinerancy
- an analysis of how the show people experience otherness arising from their definition as foreign
- an examination of the show people's past educational experiences as exemplifying the conversion of power over them into forms of seemingly unproblematic knowledge about schooling.

The overarching intention is to establish that these stereotypes and educational experiences combine to mark show people as 'different' and 'deviant' within mainstream strategic contexts and thus as people who cannot take for granted equitable access to social goods.

Although the people I call 'the show people' use different terminology from mine to identify and construct themselves and other people (including myself in the role of researcher), the data below demonstrate both the existence and the strength of those constructions. Moreover, these data go some way towards capturing the heterogeneity of construction and experience

that is implicit in the respondents' tendency to construct themselves as belonging to particular families and occupational groups, and that is belied by the necessarily homogeneous term 'the show people'. The broader issue to which this point relates is that the processes of naming reflect a dual role. On the one hand, naming displays the heterogeneity of identity construction and the resistance of a homogeneous, marginalised identity being imposed on a particular group. On the other hand, naming presents an opportunity for the reinscription of that marginalised identity, and it behoves researchers to avoid as far as possible becoming complicit in that reinscription (which, as I outlined in the preceding chapter, is one of the underlying principles of the study's research design).

Furthermore, although the focus in the data analysis chapters is mostly on the interview responses of 'the show people' (the show children and their families), each chapter, particularly the third, contains some observations by the home tutors; when this is done, it relates to the tutors who come from 'outside' the show circuits, rather than the family members who are already familiar with show life and accordingly might be considered to be 'show people' in any case. In addition, each data analysis chapter, but again particularly the third, draws on comments by the teachers from the Brisbane School of Distance Education. This practice is used as a form of indirect reflection of the show people's effectiveness at transforming their marginalised status, by recording the perceptions of some of those 'others' who have had intensive contact with the show people.

5.2 Strategies of marginalisation

My purpose in this section of the chapter is to present a brief review of de Certeau's (1984) concept of 'strategies of marginalisation'. I have already, in Chapter Three, placed this concept in the broader context of de Certeau's theory of social life and critiqued it from the perspective of several commentaries by other scholars. Here my intention is to remind the reader of the salient features of the concept, as a prelude to applying the concept to the show people's several experiences of marginalisation recorded in the remainder of the chapter.

De Certeau's emphasis in *The practice of everyday life* (1984) was squarely on "users" (p. xi) or consumers of social processes, rather than on the producers of those processes. This emphasis reflected his far greater interest in the multifarious and "ingeniousways in which the weak make use of the strong" (p. xvii) than in the massified forces of marginalisation that construct such a fundamental dichotomy between "the weak" and "the strong". However, he was fully aware of the need to delineate some of the features of the forces of marginalisation, against which the actions of resistance outlined in the rest of his book were directed. **As** I indicated in Chapter Three, he conceptualised two sets of binaries to depict this struggle: 'strategy'/'tactic' and 'place'/'space'. In reviewing these terms in this section, I shall limit my remarks to 'strategy' and 'place'.

With regard to 'strategy', as I noted in Chapter Three, de Certeau provided in the "General Introduction" to the book a definition of the term that he elaborated in a later chapter:

I call a "strategy" the calculus **e** force-relationships which becomes possible when a subject of will and power (aproprietor, an enterprise, a city, a scientific institution) can be isolated from an "environment." A strategy assumes a place that can be circumscribed as proper (propre) and thus serve as the basis **for** generating relations with an exterior distinct from it (competitors, adversaries, "clientèles," "targets," or "objects of research"). Political, economic, and scientific rationality has been constructed on this strategic model. **(1984, p. xix)**

From the outset, then, de Certeau conceptualised strategies in terms of three elements that are integral to the understanding of marginalisation underpinning this chapter. Firstly, marginalisation proceeds from "asubject of will and power", and by definition that same power is denied to the objects of marginalisation. Secondly, marginalisation proceeds from "aplace that can be circumscribed as proper (propre)", underscoring the vital link between 'strategy' and 'place'. Thirdly, marginalisation derives from and depends on the construction of 'otherness', as occurs in the process of "generating relations with an exterior distinct from" the source of the marginalising strategies. I argue in this chapter that power, 'place' and 'otherness' are all clearly manifest in the show people's experiences of marginalisation.

De Certeau elaborated this synthesised overview of strategies in Chapter Three of *The practice & everyday life* (1984):

I call a strategy the calculation (ormanipulation) **E**power relationships that becomes possible as soon as a subject with will and power (a business, an army, a city, a scientific institution) can be isolated. It postulates a place that can be delimited as its own and serve as the base

from which relations with an exteriority composed of targets or threats (customers or competitors, enemies, the country surrounding the city, objectives and objects of research, etc.) can be managed. As in management, every "strategic" rationalization seeks first of all to distinguish its "own" place, that is, the place of its own power and will, from an "environment." A Cartesian attitude, f you wish: it is an effort to delimit one's own place in a world bewitched by the invisible powers of the Other. It is also the typical attitude of modem science, politics, and military strategy. (pp. 35-36)

Again we see de Certeau's emphasis on strategies as being imbricated within the exercise of power; as being located in "a place that can be delimited as its own — and that it seeks to defend from external and internal attack; and as being the more valued half of a binary relationship "with an exteriority" and with "the invisible powers of the Other". The range of examples has been maintained from the earlier to the later conceptualisation, although it has been extended to include "the country surrounding the city" and "military strategy". The repetition of de Certeau's emphasis on power, 'place' and 'otherness' was deliberate and underscores their centrality to his understanding of strategies and to the construction of marginalisation deployed in this chapter.

Having established the central role of these three elements of marginalisation, de Certeau delineated three key attributes of strategies of marginalisation. Firstly: The "proper" is a triumph of place over time. It allows one to capitalize acquired advantages, to prepare future expansions, and thus to give oneself a certain independence with respect to the variability of circumstances. It is a mastery of time through the foundation of an autonomous place. (p. 36)

This means that 'the centre's' strategies of marginalisation exploit the advantages of stable territory that it has seized. These advantages include being accepted by others as occupying 'centre ground' or 'the mainstream', from which derives the power to define itself as 'normal' and others as 'deviant'.

Secondly:

It is also a mastery of places through sight. The division of space makes possible a panoptic practice proceeding from a place whence the eye can transform foreign forces into objects that can be observed and measured, and thus control and "include" them within its scope of vision. To be able to see (far into the distance) is also to be able to predict, to run ahead of time by reading a space. (p. 36)

This capacity to "transformforeign forces into objects. . ., and thus control and 'include' them within its scope of vision" is vital to 'the centre's' maintenance of its power base. It also means that it has the power to define others in its own terms, and furthermore to insist that such a definition is the sole authorised and accepted construction of others.

Thirdly:

It would be legitimate to define the power of knowledge by this ability to transform the uncertainties of history into readable spaces. But it would be more correct to recognize in these "strategies" a specific type of knowledge, one sustained and determined by the power to provide oneself with one's own place. Thus military or scientific strategies have always been inaugurated through the constitution of their "own" areas (autonomous cities, "neutral" or "independent" institutions, laboratories pursuing "disinterested" research, etc.). In other words, a certain power is the precondition of this knowledge and not merely its effect or its attribute. It makes this knowledge possible and at the same time determines its characteristics. It produces itself in and through this knowledge. (p. 36)

Here the emphasis is on a crucial element in the success of the power of the forces of marginalisation – that such power is 'invisible' through being channelled into supposedly "neutral" and objective forms of "knowledge". This can take the appearance of stereotypes – both positive and negative – expressed along the lines that 'Everyone knows that a particular marginalised groups has these particular desirable or undesirable characteristics'. The repetition of these stereotypes, circulated time and again through discursive practices emanating from the 'invisible centre' (Ferguson, 1990), turns them into forms of unproblematic knowledge that cannot easily be questioned but that are indissolubly linked to the exercise of power.

I should point out here that I am well aware that these three attributes of strategies of marginalisation – of being "a triumph of place over time", of permitting the conversion of "foreign forces into objects" and of helping to

turn power into forms of unproblematic knowledge — can be related directly to educational research projects of the lund undertaken in this study. I sought in Chapter Four to communicate my concerns about the ethics and the politics of the study, and my efforts to engage with those concerns. My emphasis in this chapter is on how the same analysis of strategies of marginalisation can be applied to the show people's experiences.

Finally, de Certeau provided a useful summary of his conceptions of strategies and 'place':

In sum, strategies are actions which, thanks to the establishment of a place of power (the property of a proper), elaborate theoretical pieces (systems and totalizing discourses) capable of articulating an ensemble of physical places in which forces are distributed. They combine these three types of places and seek to master each by means of the others. They thus privilege spatial relationships. At the very least they attempt to reduce temporal relations to spatial ones through the analytical attribution of a proper place to each particular element and through the combinatory organization of the movements specific to units or groups of units. The model was military before it became "scientific." (p. 38)

From this perspective, the fact that the show people routinely move through space renders them at a disadvantage in relation to the forces of marginalisation. This is because the space through which they move is actually the 'place' of 'a proper' – it is 'owned' by 'the centre' and their itineraries are tolerated rather than accepted. It is this 'ownership', with its associated capacity to define and enforce where the show people are 'allowed' to go, that goes to the heart of explaining how strategies of marginalisation can exercise

can exercise such a powerful and enduring effect on the experiences and the lives of show people.

5.3 Absence of place

In the previous section, I referred to de Certeau's (1984) identification of three key attributes of strategies of marginalisation: I use those three attributes as the chapter's organising framework for analysing the show people's complex and diverse experiences of marginalisation. In this section, I examine how the show people's itinerancy means that they have an absence of place, with the result that they must find multiple ways of traversing the place(s) owned by others. In the next section, I consider how the show people are defined as foreign, an indispensable ally of the process of constructing and reinforcing their otherness in relation to mainstream society. Finally I identify the show people's past educational experiences as exemplifying the conversion of power over them into forms of unproblematic knowledge, including assumptions about 'normal' schooling. In each section, the focus is on how these respective attributes of marginalising strategies operate – sometimes obviously, far more often invisibly and insidiously - to render the show people as marginal to, and deviant from, the community to which they regard themselves as belonging.

The show people's absence of place underscores the strategic importance of a fixed location. **As** I noted in the previous section of this chapter, this crucial point was expressed succinctly by de Certeau (1984):

The "proper" is a triumph of place over time. It allows one to capitalize acquired advantages, to prepare future expansions, and thus to give oneself a certain independence with respect to the variability of circumstances. It is a mastery of time through the foundation of an autonomous place. (p. 36)

When the show 'comes to town', what does this actually entail? It involves a group of itinerant people moving into a settled community's boundaries for a predetermined period. It involves those people residing and working in an allotted territory in a public place – generally the local showgrounds. It involves their conformity to the local rules and regulations concerning a host of matters from parking vehicles to water supply to ownership of pet animals. In other words, it means an official acceptance of the show people's 'coming to town' provided that they adhere to the local community's policies and *mores*. They are there 'on sufferance', 'on the terms' of the official representatives of the townspeople. These are the various techniques that underpin "thefoundation of an autonomous place".

This fundamental and integral link between place and power, and between the absence of place and powerlessness, was synthesised starkly by historian Richard Broome and show person Alick Jackomos (1998):

Showpeople represented potential disorder. Like wanderers everywhere, they were perceived to be beyond the moral and social controls of the local community, or at least a threat to that control. (p. 29)

1

Hence the practice of local police forces maintaining "a close watch on the Showies", with a Queensland detective being assigned to follow them up the Queensland coast each year until 1987 (p. 29). It is the show people's absence of place, and the concomitant lack of an authorised spealung position, that lead to this lund of marginalising and even discriminatory behaviour against them.

Generally the show people were more interested in discussing the positive dimensions of their lifestyle than in focussing explicitly on their marginalised status (an emphasis that also reflected the tenor of the interview questions). Consequently very few respondents referred directly to themselves as marginalised, and they certainly eschewed the ascription of a 'victim status'. Nevertheless evidence of the deleterious effects of their absence of place can be found in a number of their utterances.

Awareness of absence of place and its consequences for show people was strongly evident in one show parent's extended comments about the concept of 'home'. She identified herself as a member of the Showmen's Guild of Australasia, and she explained that "I was born in the show business, my father was in the show business", and that her grandfather "started in show business by actually importing and selling toys to showmen,. . . and then the family's continued on from there" (Y4P5)\frac{1}{2}. Later in the interview she clarified

For the purposes of preserving the anonymity of respondents, while indicating to the reader which statements were made by particular respondents, I have adopted the procedure, common in qualitative educational research, of identifying interviewees' quotations with codes. In the codes, 'Y' refers to the year of the project in which the interview was conducted, while 'A', 'C', 'HT', 'P' and "T' denotes the number of the interviews conducted respectively with

this situation, saying with considerable pride that "my husband's a six generation showman", and outlining how "...he really has a show, [but] my family are more carnivals, not so much showgrounds" (Y4P5). Subsequently she stated that "we're operating in the itinerant side [of the show], but we live in the Guild side, because we grew up there and that's where we come from" (Y4P5) (a distinction that is elaborated in the next chapter).

This family background is significant for explaining the personal context in which this show person's discussion of 'home' was located. That is, by positioning herself as a member of the Showmen's Guild and as the wife of "a six generation showman", the respondent was constructing herself as someone with considerable cultural capital on the show circuits and as occupying a strong spealung position in relation to 'outsiders' such as the interviewer. This in turn framed her particularly revealing account of 'home', in which she speculated at greater length than any other of my respondents on the relationship between 'home' and show people's ascribed absence of place.

I conducted two interviews with this respondent. The first was a lengthy, audiotaped interview in which she and I were the only people present in a temporarily vacant classroom. The second was a much briefer, videotaped interview in which she and I stood on the school oval with the videographer and the sound recordist standing close to us. The respondent used both these interviews to communicate a confident and articulate construction of show people and other people through the multiple understandings and meanings of

administrators, children, home tutors, parents and teachers.

'home'. In the process, she demonstrated that show people use those understandings and meanings to talk 'otherwise' about their situation, rather than accepting unproblematically their marginalised status.

In response to the explicit question "Where is home?" in the audiotaped interview, the respondent expressed mild irritation at the recurrence of this question being asked by 'outsiders' such as the interviewer: "Everybodyasks where is home" (Y4P5). In her view the question implied an assumption that 'home' necessarily had to refer to a house, whereas for her the concept was more closely related to the extent and the strength of family and community relationships: "I'mat home. Myfamily, my children, my home is in the place [where] I live" (Y4P5). For this show person, a perceived obsession with living in a fixed residence was one of the clearest indications that the interlocutor was not a 'showie'. In the process, she rejected implicitly a perception that the shifts between 'home' as the show circuits, her parents' house and her caravan identified above were necessarily contradictory. Her extended response to the question took up this theme:

Wherever I am, I'm at home. My mum and dad now have a house in Melbourne, and that's my home because my parents live there. But that's not my home because it's in the one place. Do you know what I mean? People have this thing that we 'redisjointed from what we should belong to. Do you know what I mean? If it's got wheels, and it's not earth, we belong to it. And that's where our roots are and that's what we do. (Y4P5)

The repetition of the question "Doyou know what I mean?" reflected the respondent's strong desire to communicate her construction of the show

people as having a very deeply felt attachment to one another and to show life, and of 'home' as encapsulating and incorporating this attachment into practical living arrangements. The respondent also sought to construct other people as failing to understand the show people's enduring attachment to one another and to show life, thereby marking themselves as non-show people by means of this lack of comprehension. Importantly, this same lack of comprehension in turn marks the show people as 'different', 'lacking' and 'deviant' according to the perspective of local people. This last point demonstrates how the show people's absence of place is not simply a minor inconvenience, but instead goes to the centre and the heart of their experiences of marginalisation.

This discussion provided the basis for the respondent's exposition of these ideas in the more public videotaped interview. Again she sought to distinguish between show people and non-show people on the basis of their comprehension of living an itinerant lifestyle.

People don't have a conception of what it is to have a mobile home. People always — the main thing they ask me is, "Where is home?". Like it isn't where I'm at. My home is in my caravan, that's where all my things are, all my everything. My heart is in travelling, and my home is mobile. It has wheels on it, but I have roots in that home. And that's where it is. My parents have a house in Melbourne, but that isn't my home. That's my home because my mum and dad are there, but my home is on the showgrounds travelling, doing what I'm doing. And because it moves and it's not in the one town and it doesn't have a name doesn't mean it's not my home. (Y4P5)

This powerful assertion of identity and cultural pride and of an identification of 'home' as the respondent's caravan was followed by a statement of a perceived benefit of show life not available to most Australian families: "We love what we're doing, we spend time with our family, because our business is a work at home business, more or less" (Y4P5). Here 'home' is the show circuits, with the family living and working together in overall harmony.

Three comments need to be made about this show person's account of the multiple meanings of 'home' for show people. Firstly, the speaker was acutely aware that her itinerant lifestyle constructed her as suffering from an absence of place in the eyes of 'the centre'. This awareness was reflected in her devoting considerable time in both interviews to discussing this issue, which for permanent residents rarely becomes a topic of conversation (although certainly I was keen to pursue the point in both fora). Secondly, the speaker's awareness of this ascribed absence of place prompted her to explain in considerable detail that the show people do in fact experience and benefit from alternative understandings of 'home'. This willingness to explain was evident in her determination to ensure that I – and other locals – accepted the presence of these alternative understandings, through repetition and example. Thirdly, the speaker's focus on the presence of these understandings was a direct response to the presumed absence of place of show people, highlighting the crucial role played by this absence as a strategy of marginalisation.

In addition to negative assumptions about the show people's lack of 'home' as it is conventionally understood, their absence of place was also responsible for the reduction of their multiple signifiers of identity and meaning to a single homogenised label: 'the showies'. Although Broome with Jackomos (1998) traced this appellation to the mid twentieth century, and argued that the term was developed by the people who specialised in operating sideshows to refer to themselves (p. 28), the point to emphasise here is that that same term has been used by locals against show people. The show people's absence of place has also denied them a speaking position and a voice by means of which they can tell non-show people who they are and see themselves as becoming in their own terms. So outsiders' use of the term 'showies' functions much more than as a mere descriptor: it is also a naming device by which negative stereotypes can be uncritically circulated. The ultimate effect of this circulation is the continued marginalisation of the show people, and the closure of opportunities for them to tell their own stories about themselves. (As I demonstrate in the next chapter, show people strive consciously to tell those stories, in the process appropriating and using 'against' non-show people the very descriptors used to marginalise the show people.)

The extent of the show people's consciousness of the negative impact of the term 'showies', when used by others to perpetuate negative stereotypes, was encapsulated in the following interaction with a twelve year old girl on the show circuits:

[You know the local kids — did you talk to them much?]

No. They always say, "Look, they're show kids". (Y2C4)

The bald statement, "Look, they're show kids", is highly evocative of the ease with which show people become objects of the other's gaze, and in the process undergo essentialisation and exoticisation as 'strange' and 'unnatural' beings whose occupations require them to disrupt the norm of fixed residence. This ease illustrates the integral links between naming and power, or its lack, and between the show people's lack of place and their marginalised status.

The combined effects of stereotypically having no 'home' and being subject to homogenised naming practices are to highlight again and again to show people and those who work with them the ongoing and enduring experience of marginalisation that they undergo. A representative example of this consciousness of the negative impact of having an absence of place was a number of statements by a home tutor, who had travelled on the show circuits for six months. A recurring theme in this home tutor's discourse was the assertion that show people have traditionally been poorly treated by the wider society, and that that poor treatment has caused them to retreat into a separate existence on the show circuits.

...no other outsiders would be welcome. ..inside [the show circuits]. They would not. Well,...the world wasn't interested in them,...the communities...didn't know them, all they thought of them was – scum, really. (Y5HT1)

Shortly after this statement, the home tutor reinforced this view.

Everything that the showmen are is a product of the way we treat them – the way their pride **is**, the way they work, the way they are – is a product of years and years, centuries of our treatment of them. And that is the basic line. (Y5HT1)

This thesis focusses on the show people's agency in contesting and transforming their marginalisation, rather than on constructing their status as passive 'victims' as this home tutor appeared to do. Nevertheless, it is important to recognise that her perspective helps to confirm the extent of their marginalisation as seen from her perspective. Later in the interview, the speaker elaborated this view.

For this reason, the speaker claimed, the 'showies' generally confine their interest in the world to the events of show life: "...why should they go out of their way. ...for the outer community, when the outer community have never treated them anything but. ..[poorly]?" (Y5HT1).

The vehemence of the speaker's references to the show people's "years and years" of neglect and mistreatment by "the outer community" derived from the sharp intensity of her observations of those experiences as she lived and worked with the show people 'on the run'. Her position — as an 'informed outsider' or 'participant observer' — provided a distinctive perspective on a situation that many show people acknowledged but that they

were likely to assign to 'the back of their minds' as they engaged in the daily battles of earning a livelihood. All of this derives, I argue, from de Certeau's insight that "The 'proper' is a triumph of place over time", and that "It is a mastery of time through the foundation of an autonomous place" (1984, p. 36). In other words, the strategic importance of a fixed location lies in its inverse ratio to the marginalising impact of the show people's absence of place.

5.4 Constructions of otherness

In the previous section, I demonstrated that a principal element of the show people's marginalisation relates to the first of de Certeau's (1984) three attributes of the strategies of marginalisation: their absence of place. Here I argue that another element of their marginalised status derives from the second of de Certeau's three attributes: their definition as foreign objects, and how that definition operates to construct their otherness.

According to de Certeau (1984):

It [the "proper"] is also a mastery of places through sight. The division of space makes possible a panoptic practice proceeding from a place whence the eye can transform foreign forces into objects that can be observed and measured, and thus control and "include" them within its scope of vision. To be able to see (far into the distance) is also to be able to predict, to run ahead of time by reading a space. (p. 36)

This analysis demonstrates how the strategic importance of a fixed location shades into the treatment of all those 'others' who do not inhabit that fixed location. These 'others' become 'yoreign. . . objects that can be observed and measured", by means of which 'the centre' can "control and 'include' them within its scope of vision". This situation is the opposite of the unpoliticised relativism of the adage that 'beauty is in the eye of the beholder'. On the contrary, 'beauty' or other values such as 'goodness', 'providence', 'responsibility' and 'trustworthiness' are held to exist only in the narrow frames envisaged by 'the centre'. Alternative understandings of these values are silenced and negated – although not completely.

For the show people, this means that their absence of place, arising from their itinerancy, renders them outside 'the centre' and therefore ripe for the operation of the "panoptic practice" that turns a searchlight unblinkingly on them. They become the object of the gaze, unable to speak and communicate their own meanings and values, but instead required to endure their conversion into foreign objects and the construction of their otherness. This process leads to the circulation and repetition of negative stereotypes based not on the show people as active individuals but rather on their status as an objectified and homogenised entity: 'the showies'. This process is an integral element of the show people's marginalisation.

An initial manifestation of the construction of the show people's otherness is perceptions by locals of show people as 'strange'. A show person recalled the local incomprehension of the show people's need to have equipment serviced quickly:

See, I'll come in with a broken TV on Monday and say, "Excuse me, I need it by Friday". They look at you, because people put them in for a month. They can't understand that way. See, I'm moving onto another town, but they don't understand that part. . . . They might think it's a bit weird, but we've only got so long in here, and we've got to do the best we can in thosefew days. (Y4P6)

This reference to local people thinking that "it's a bit weird" resonated with another show person's reflection on the show people's interactions with government officials:

And sometimes when we 'retalking to government bodies and that, they can't understand that it's urgent for us. It's always urgent, because that is the way that our business functions. (Y4P5)

Similarly, another show person stated in a videotaped interview with me, "Because our lives – it's a very sort of strange life to somebody like you" (Y4P1).

These references to perceptions of the show people as "weird" and "strange" reflected their awareness that their itinerancy marks them as 'different'. The result is that they are subject to the gaze of surveillance while they are 'in town'. For each group of local people, this process of objectification rarely occurs more often than once a year. For the show people, however, it takes place every time that they enter a new community. The regularity and the repetition of enduring this kind of interested and objectifying gaze are marginalising in their cumulative effect. This consciousness of the marginalisation that arises from being seen as "weird" and

"strange" certainly underlay a ten year old girl's comment about local children: ". . . sometimes they don't understand that we're more or less like them but just travel on" (Y1C4).

An extension of regarding the show people as "weird" and "strange" is to perceive them as objects of pity, as a consequence of being forced to live a 'second class existence'. A seven year old boy referred to this perception when he identified some of the difficulties in getting to know local children:

No, not easy at all. It's hard to find a friend because [locals think that] all of us [are] so stupid, don't get anything done, but we do get things done. (YIC5)

Similarly, a show parent stated, "So I think we have a better lifestyle than most people when they say, 'How could you do it?', like it's so terrible" (Y4P5). Another show parent expressed this perception as amounting to "a stigma": "Likea lot of people are ignorant to our lifestyle and to the way things work, and they think that things aren't right, and that can cause a stigma" (Y1P2).

It is important not to underestimate the deleterious effects of this construction of show people as objects of pity. This process represents an insidious devaluing of the show people's lifestyle, and a refusal to concede their right to attach their own values to the way that they live and work. It therefore marks them out as 'different' and 'deviant', and accordingly as warranting the intrusive attention and surveillance of 'the centre'. Objectified pity brings with it less, not more, understanding of the show people, and it therefore contributes directly to their experiences of marginalisation.

Another manifestation of the show people being defined as foreign, and of the construction of their otherness, but with a harder edge to it than pity, is animosity. This takes the response to the perception that the show people are 'strange' from ambivalence to hostility. This situation was encapsulated in a show parent's observation that "kids still do tend to serve a little bit of animosity to show children when they come into the school" (Y1P2). The connection between the show people and the feelings of hostility derive from negative stereotypes based on their 'foreignness', rather than from direct interactions between the two groups. In other words, this is a situation in which prejudice feeds marginalisation.

This link between animosity and the construction of the show people's outsidedness was confirmed by two 'outsiders' who had a close knowledge of the show people's working lives. Firstly, a home tutor recalled a set of graphic incidents that for her encapsulated many of the perceived tensions between 'showies' and 'locals':

... I was at school with him [the show child whom she tutored] and all the other kids were staring at us through the window — about thirty children were staring at us, and so naturally they put on this big show at lunch time. We were just sitting eating lunch,...we were just sitting quietly eating lunch, and this teacher walked up, we were sitting on top of these port racks [used for storing students' school bags], didn't even speak to us, just pointed at us, "Get off the port racks." And I thought, "I'm an adult, not a child." And then someone left a paper on top of the port rack and she must have been standing half a metre away from me. Instead of saying, "Excuse me, could you get one of the girls to put it in

the rubbish bin?", she just stood there and put this horrid look on her face and just pointed at me and pointed at the paper. So I just chose to ignore it, like I hadn't seen it. That's totally rude. Just afew things [like] going and ordering your lunch at the tuckshop, sometimes the ladies get a bit funny for the first few days. . . . So you're really well spoken with your best manners so they sort of say to each other, "Oh, she's not so bad afer all." (Y1HT1)

The speaker's references to "about thirty children. ..staring at us through the window", and to "sometimes the [tuckshop] ladies. ..[being] a bit funny for the first few days", underscored her direct experience of how being defined in deficit terms and as 'foreign' shades seemingly inexorably into outright animosity and hostility. The same home tutor noted further:

A lot of the time our kids have learnt to stick **up for** themselves. They're like that – they stick together and they might fight with the other kids, but I don't think that I've ever seen our children go and actually pick it first. (Y1HT1)

Here the clear impression is that physical fights by local children directed at show children are an overt expression of more tightly controlled – but nevertheless real and powerful – feelings of hostility held by older community members towards the show people.

Similarly, one of the Brisbane School of Distance Education teachers confirmed the persistence of this animosity towards show children:

They [show children] tend to stick to themselves when they come into school normally, because the school kids perceive them to be something dijherent. . . (Y2T1)

The teacher identified the show children's responses to this situation:

So they get a bit **a** a rough time at school sometimes, and they're made to feel dijherent, and they react. Some **a** them react; some **a** them are just so used to it that they 'revery nonplussed about the whole thing but don't do anything anyway. (Y2T1)

Here a wealth of attitudes and actions is conveyed by the deceptively simple statement that "they'remade tofeel different". The other noteworthy reference is to the fact that "some of them are just so used to it": being labelled as 'different', and this 'difference' leading to hostility and sometimes to physical conflict, is habitual for many show children — as well as for their parents. This habituation of 'difference' and animosity is a striking illustration of the construction of the show people's otherness.

A highly evocative dimension of the show people's otherness is the stereotypical view of them as 'dirty'. This is a visible sign of a presupposition that their itinerancy reduces their access to running water and washing facilities. More fundamentally, it evokes a notion of 'impurity' and the suspicion that their unwanted presence might defile the 'purity' of the 'normal' people living in the community. This association between being 'dirty' and threatening the host community's 'purity' has strong parallels with two other historically or habitually itinerant groups who are also

subjected to negative stereotypes: Gypsies (Liégeois, 1996, 1998) – whom I discuss below – and Jews (Rose, 1990, 1992).

One of the show parents commented on this stereotypical association with 'dirt':

...we get our reputation, which we're not very pleased with, from the people who work for us. Because they're the louder ones down the street; they're usually the ones that didn't have a [shower] before they went, and people say, "The showies are back in town", and they just recognise the people who work for us as being us. But usually they're the ones that wanted a move around lifestyle. ..but we're here in a business and we 'rehere to stay. So that's not our reputation. (Y4P5)

The workers' claimed habit of not showering before they go into town marks them out in a visible way for inspection by, and disapproval of, the townspeople, thereby reinforcing a negative stereotype based on rendering the show people as objects to be surveyed and thus controlled. This situation reflects the centre's configuration as a site cleansed of the 'impurity' of disorder (see also McVeigh's [1997] assertion that the "veryexistence" of nomadic people "threatens, undermines, 'invades' sedentary identity", and that consequently those same people "receive immediate and oppressive policing by the state in the interest of all sedentary people" [p. 22]).

One of the home tutors also commented on this tendency of local people to attach negative valences to superficial appearances. He estimated that workers constitute

...probably. ..fifty per cent or sixty per cent of what makes up the travelling show. .., and that's the outside that you see. And I suppose you'd see the tattoos and hair and pony tails and think, "Well!". (Y3HT1)

Another home tutor also demonstrated her awareness of the deeper significance of looking 'dirty' and 'unwashed':

...in the morning I'd wake up, have a shower, put makeup on, iron my clothes or anything, and I'd be standing there for — maybe if it was a busy showfrom eight o'clock in the morning and I'd have half an hour for lunch and by the time we'd hit seven o'clock when all the people started to arrive, I'd been standing up off the ground, I was lucky, I didn't have to stand right in the dust, but I looked absolutely filthy. You'd look horrible, you would, your clothes would be justfull of dust, and the thing [the joint that she had been tending] might have broken down and you'd have grease all over you,...and all these local people, when they go to the show, they dress up really well, so they come out all neatly showered in their best clothes and they'd think that you're like a dirty scumbag or something. But you just can't do anything about it, and by the time it hits eleven o'clockyou'rejust —you just look dreadful. It's funny. (Y1HT1)

Again the link between the show people's appearance and deeply seated and irrational beliefs by local people about show people is clear and direct. The show people's appearance is thus an index of their definition as 'foreign', and their construction **as** 'other', to the mainstream community through which they are passing.

The perception that show people are 'dirty' is closely associated with the feeling that they cannot be trusted, and particularly that they are thieves. According to Broome with Jackomos (1998):

The initial ambivalence towards sideshows reflected the mixed feelings many had about show people themselves. Because they travelled from place to place and were not a settled people, they were distrusted. (p. 28)

They noted that "Therewas an old country saying, only half injest, 'lockup your daughters and your chooks [chickens], the showies are coming"' (p. 29). They contrasted this with the more likely cause of theft at this time: "Certainly crimes did occur at show time because of the large number of strangers in town" (p. 29), with most of these crimes being attributable to professional criminals who saw the show as 'good cover' for their activities.

This stereotype of the show people as likely to engage in theft was recalled by Frank Foster, member of one of Australia's oldest show families:

You've got to remember that we show people get a lot of bad publicity. In days gone by when the show came into townpeople used to say, 'Pull your washing in and lock your daughters up'.

I remember on one occasion in this particular town where the showgrounds was close to a lot of houses, this young boy was over at the tap washing out his shirt when he saw a young girl in the garden next door and after they exchanged greetings he asked her if she would like to go to the pictures that night, and she told him she would have to ask her mother and the conversation went something like this, 'Mum there's one of these show blokes out here and he wants to take me to the

pictures', the reply was quick and final, 'Youcome inside and bring the cow with you!'. (Morgan, 1995,pp. 128-129)

This 'untrustworthy criminal' dimension of the show people's constructed otherness was evident in a show parent's claim that local people "see what they want to see", and that "because of our itinerant nature – here today and gone tomorrow – . .[people are] less likely to trust you" (Y3P1). She conveyed her frustration at continuing prejudice against show people, leavened by occasional support from people of good sense:

...there are some people who don't even associate with showmen down in New South Wales and the week we were there their machine got broken into. But this person said that why blame the show just because we're here – it's more likely that they've – certain boys she mentioned – they've done it because they know the showmen are in the town. (Y3P1)

Sometimes the combination of being perceived as 'dirty' and as 'thieves' shaded into a stereotyped association between show people and Gypsies. This association linked the show people's otherness to a much older and more concentrated aversion and prejudice in which Gypsies are typecast as unwashed outcasts, with devastating consequences for their educational access among other aspects of their quality of life (Liégeois, 1998). According to Broome with Jackomos (1998):

While the Showies' subculture was born out **a** positive and shared ideals, it partly depended on the negative feeling **a** being classed as outsiders. Some **a** this stemmed from the public's view that Showies

were the same as Gypsies, although the Showies vehemently denied any connection. (p. 42)

Pursuing this theme, an older show person contrasted "our industry" immediately after World War Two, "when we had a lot of Gypsies in it and so forth", with the current situation, when "it's a pretty big one today in Australia", largely because "Through the Showmen's Guild now we've a very organised body of people" (Y4P3). In other words, being "organised" was intended to counteract public perceptions that show people are like Gypsies, and therefore shiftless, untrustworthy and not contributing to the formal economy. This recalls de Certeau's (1984) insight that "The 'proper'. . .is a mastery of time through the foundation of an autonomous place" (p. 36). That is, the show people's efforts to impose order and routine on their seemingly random and unstable movements through the spaces of their itinerancy constitute an attempt to turn marginalised 'space' into valued 'place'.

A younger show person went further in insisting on a demarcation between show people and the negative images attributed by the public to Gypsies:

And then there's a few people that were **d** English Gypsy orientation, where that was in their blood. So how could it not — they were bad people but that was the way they were, and you can't do anything about that. And that was also part **d** our business because they were showmen from original, so we had to accept them. But they did the wrong thing. We also had to say, "Well, this and this is the rules". So we have had people who haven't represented us in our best interests over the years.

So people may have built up their negative attitudes from one **or** two incidents **or** something. (Y4P5)

Again the implication is clear that local people's "negativeattitudes" are irrational and based on stereotypes of the show people as 'different' and 'foreign', yet also that those "negativeattitudes" have a considerable and enduring regrettable impact on the show people's capacity to live their lives as they wish. Relatedly, as McVeigh (1997) pointed out, "the subtlety of these distinctions", such as between show people and people "of English Gypsy orientation" identified by the person cited in the previous paragraph, "are often lost on sedentary people with the power to define the 'whole vagrant population" (p. 16). It is this very "powerto define", and by implication to devalue and elide the show people's own efforts "todefine", their identities that lies at the heart of the show people's marginalisation.

I have demonstrated in this section the relevance and the accuracy of de Certeau's (1984) insight that strategies of marginalisation involve "amastery of places through sight" (p. 36). From that perspective, the show people are indeed "foreignforces" who have been transformed "intoobjects that can be observed and measured", and who are thus subject to "control" and inclusion within the 'proper's' "scopeof vision" (p. 36). Specifically, I have argued that the show people's itinerancy leads directly to their being perceived as 'strange', objects of pity and of animosity, 'dirty', untrustworthy and likely to engage in theft, and similar to Gypsies. These perceptions recall McVeigh's (1997) assertion:

Travellers are subject to a whole series **a** stereotypes which combine to render them hugely problematic: they are criminals by 'nature', they come from outside the community, they are dirty, they are dishonest, they are immoral and amoral and, most importantly, they are 'nomadic'. Most of these contemporary constructions **a** nomads draw on a long history of establishment fears about the travelling dispossessed and the threat they pose to the moral and political order. (p. S)

In combination, then, and from the perspective outlined by McVeigh (1997), the negative perceptions identified above mark out the show people as 'different' and construct them as 'other' to 'normal' and 'proper' settled society. This process is an extremely powerful instance of the operation of strategies of marginalisation against the show people.

5.5 Forms of unproblematic knowledge

In addition to experiencing, and suffering **as** a result of, absence of place and constructions of otherness, the show people have endured the third attribute of strategies of marginalisation identified by de Certeau (1984): forms of unproblematic knowledge. **As** de Certeau noted:

It would be legitimate to define the power of knowledge by this ability to transform the uncertainties of history into readable spaces. But it would be more correct to recognize in these "strategies" a specific type of knowledge, one sustained and determined by the power to provide oneself with one's own place. Thus military or scientific strategies have

always been inaugurated through the constitution of their "own" areas (autonomous cities, "neutral" or "independent" institutions, laboratories pursuing "disinterested" research, etc.). In other words, a certain power is the precondition of this knowledge and not merely its efect or its attribute. It makes this knowledge possible and at the same time determines its characteristics. It produces itself in and through this knowledge. (p. 36)

This account of forms of unproblematic knowledge signals the direct and immediate link between the show people's marginalised status and their educational experiences. From this perspective, schooling functions, not just to facilitate enlightenment or empowerment, but also as an ally of the strategies of marginalisation that deny show people a place and construct them as other to the rest of the community. This is the real significance of de Certeau's (1984) insight that "a certain power is the precondition of this knowledge and not merely its effect or its attribute", and that "Itproduces itselfin and through this knowledge". In other words, formal education is never innocent or neutral, but rather reflects broader social forces. Whether education works against or for the show people depends in large part on the particular relationship between education and the marginalisation, resistance or transformation of the show people's identities.

How might superficially straightforward conditions of schooling be complicit in the show people's marginalisation? The answer lies in the link between those conditions and the two other attributes of marginalising strategies identified by de Certeau (1984):absence of place and constructions of otherness. Firstly, the show people's itinerancy, which involves their

movement in and out of spaces that are others' places, means that they are without places of their own and hence without the stability of identity, meaning and power that are the preserve of "the 'proper" and that enable and underpin "thefoundation of an autonomous place" (p. 36). Secondly, the show people's absence of place renders them subject to the centre's surveillance, and to their own conversion from "foreignforces into objects that can be observed and measured", and therefore liable to "control" and inclusion "withinits scope of vision" (p. 36). The show people's absence of place thus leads inexorably to the construction of their otherness.

Thirdly, the powerful forces that deny a place to the show people and construct them as other are the same forces that underpin "aspecific type of knowledge, one sustained and determined by the power to provide oneself with one's own place' (p. 36). This knowledge is accompanied by a host of assumptions, which become institutionalised and therefore naturalised, about the 'right' way for education to function in a late capitalist society generally, and in Queensland specifically. Two assumptions are particularly noteworthy in the context of the show people's marginalisation. The first is that schooling is rightly located in specific 'places' called 'schools', which are the officially sanctioned institutions for the dissemination of knowledge. A crucial corollary of this assumption is that the students and teachers who labour in schools themselves live in permanently resident situations – the 'rightness' of which is reflected in the school's equivalent location in a fixed place. The show people's absence of place clearly and dramatically puts them at a fundamental disadvantage in relation to schooling, which thereby contributes to their marginalised status. A striking illustration of this situation is the curriculum of the distance education packages, which are predicated on the assumption of students having a fixed residence (albeit at a distance from 'the centre') and which are constructed in unsuitable ways for the show people's mobile lifestyle. Furthermore, the distance education curriculum generally conceives of the show children as 'blank slates' on which the curriculum needs to be inscribed, rather than as having a myriad of lived experiences and skills into which the distance educational materials should be fitted.

The second noteworthy assumption about schooling is that the understandings of the world held by those who occupy the places of 'the centre' are used as the basis for deciding who and what are 'normal' and who and what, on the other hand, deviate from those social 'norms'. There are no half measures here: you are either 'for us' or 'against us'. This thinking is what underlies the construction of a group's otherness: in some fundamental way 'they' are 'different' and hence are subject to 'our' surveillance and control (or else are overlooked and ignored by 'us'). Certainly for the show people, the stereotypes that position them as 'foreign' are fuelled by their patent inability to 'fit the norm' in terms of educational provision: their otherness creates a 'problem' that educational authorities must seek to resolve.

This analysis suggests strongly that the construction and dissemination of knowledge are tied to how that knowledge is encoded, mediated and delivered. Furthermore, the 'invisibility' of these processes portrays that knowledge as straightforward and unproblematic. From this perspective, it is 'obvious' that schools are located in permanent places, and that groups that differ from 'the norm' are 'other' to 'normal' citizens. The purpose of this section of the chapter is to demonstrate how these forms of unproblematic

knowledge, manifested in particular institutions of schooling, are inextricably imbricated in the show people's marginalisation.

This demonstration is pursued by means of analysing the limited educational options (including location and control of schooling) available to the show people before the establishment of the specialised program provided by the Brisbane School of Distance Education (which is discussed in detail in the next chapter). The unifying theme in the show people's recollections of these previous educational options is that, regardless of their respective 'pros' and 'cons', in combination these options revealed the existence of forms of unproblematic knowledge underpinned by powerful forces of marginalisation.

Prior to the establishment of the specialised program, show people's options for their children's education were restricted to six possibilities:

- sending their children to local schools along the show circuits
- sending their children to boarding schools
- not sending their children to local or boarding schools but instead teaching them correspondence lessons on the show circuits
- coming off the show circuits and finding alternative employment for the duration of their children's education so that the children could attend local schools
- remaining on the show circuits and sending their children to live with relatives and attend local schools
- not sending their children to school at all.

As will be seen below, each option had particular difficulties associated with it and was not considered the foundation of a long term solution to the show people's distinctive educational needs and aspirations. As one parent expressed the situation in relation to show life and Traveller education:

It's good. It's a good life and that **for** the kids. The only thing that really suffers with the kids is the lack **c** education, because you pretty well know before you have children what your opportunities are. It's either boarding school or correspondence or school to school. (Y4P2)

An older show person aptly conveyed the long term consequences of such educational neglect – virtually an entire generation of illiterate show people:

We know what it's like for our children to grow up and come and ask us questions about "Pop, or Dad, what's that word?". Sometimes we couldn't answer it because we never had much education. (Y4P3)

This situation was confirmed by a relative newcomer to the show circuits:

There was a time when a lot of showmen I don't think had any means of teaching their children, and if they couldn't afford to send them to school, they didn't get the schooling at all. (Y2P5)

As with many of the quotations below, this utterance reflected neither 'inverted pride' nor 'special pleading'. It was not easy for show people to talk about their own formal illiteracy, but they considered it necessary to convince educational researchers such as myself of the extent of the educational marginalisation arising from their itinerancy. They also used it as a 'rallying

cry' among themselves to articulate their expectations for more appropriate educational provision and to devise tactics to bring such provision about.

5.5.1 Sending show children to local schools on the show circuits

With regard to the show people's option of sending their children to local schools along the show circuits, their references to that option reflected the operation of the marginalising educational 'norms' associated with that option. For example, one of the key people in lobbying for the new program had a clear recollection of the situation that existed prior to the program's introduction.

[Before the program] the choices that were available to the parents were basically zilch. They [the children] just went from school to school and there was no continuity or gauge on what the kids were doing, so they got to a certain age and lost interest very quickly. (Y1P2)

Here the reference to the absence of a "continuity or gauge" demonstrated the speaker's acknowledgment that such a "continuity or gauge" was a pedagogical device commonly used by teachers to monitor their students' learning. The emphasis on its absence exemplified the educational corollary of the show people's absence of place: that absence meant that the option of show people's sending their children to local schools was educationally unsound.

Another show person described the difficulties of show children attending local schools when the show was in town and trying to join in the lessons of the non-itinerant children.

But sometimes it wasn't evenfor a week. You see, this is a short week here, sometimes it was only for three days. . . . If you arrived late on Sunday night, say one or two [o'clock] in the morning, you wouldn't even get up to send the kids to school the next day. The kids wouldn't get up, so then it might only be Tuesday, Wednesday, Thursday. And then. there was a holiday for the show. And I suppose the teacher thought, "Well, what can I teach this kid in three days? I won't worry about it", so they didn't. And then they did their correspondence at the school. That was a little bit better - but this is a lot better. . . . Before that, the kids just went to boarding school. They just went to school from town to town until they were about ten and then they went to boarding school. (Y1P1)

The speaker's reference to what "the teacher thought" reflected her implicit awareness of a particular form of unproblematic knowledge about schooling: the assumption that learning could take place only with the continuity and routine of permanently resident students going to school in the same place for an extended time. The other aspect of this interview statement to emphasise is the speaker's equal awareness that the show people's own continuity and routines were 'at odds' with the school's operations, and that accordingly their children suffered educationally from the lack of 'fit' between the two systems.

An older show person remembered a time when boarding school was not an option for parents:

None of us could afford boarding school. We couldn't afford it, and again when we came to town, you'd take your children to a school and nine times out of ten they were put down the back of the class. And they were given a little project to do while they were therefor two or three days. They learn very little this way. (Y4P3)

Two young parents remembered their own experiences of (not) learning under those circumstances:

I was only a kid, so you'd be getting the same thing over and over again.

They didn't know what to do with you. They had you for a week, so they would sit you up there and to the best they could. . . . (Y4P6)

I did the same project on New Zealand at five different schools. So there's how you could get away with it. I was familiar with the work. They'd say, "We're doing a project". "Oh,I can do a project on New Zealand". I did the same one. (Y4P5)

But that's what you had to do. (Y4P6)

Because that was what was making the teachers happy, but that wasn't teaching **me** anything. I got to know my project down pat. And that happened often. You'd say, "I know how to read this book", and you'd just read that book. And you'd only do the things that you were familiar with, that you had confidence with. (Y4P5)

Then you'd go to school, and all the kids would be in a school uniform, and you'd come in there, and it was pretty heart breaking. I used to cry all the way. My mother used to get upset. I bet she felt like saying, "Don'tworry about it". But she did it. (Y4P6)

This interchange demonstrated starkly two key components of the show people's past educational experiences. Firstly, the institution of schooling constructed knowledge as unproblematic, a construction that fuelled its representatives' assumptions about the necessary conditions of learning. This process helped to explain the fact that the teachers "didn'tknow what to do with you": the show children deviated from the 'norm' of permanently resident students whom the teachers had been trained to instruct. Because the teachers' professional knowledge was regarded as unproblematic and had not been subjected to scrutiny or critique, the 'problem' was projected onto the itinerant children, and the teachers discharged their responsibility when "they would sit you up there and do the best they could". Another example of this unproblematic knowledge was the assumption that students should wear school uniform, to identify them more closely with the single educational institution that they would attend and mark them as not attending another school.

The second key component of the show people's past educational experiences manifested in this exchange was a corollary of the first: the pervasive influence of forms of seemingly unproblematic knowledge led directly to experiences that were marginalising and harmful for the show people. Both speakers' intense frustration was redolent in their exchange: experiences that occurred more than twenty years earlier were recalled with

vivid clarity. "Getting the same thing over and over again", being permitted to do "only...the things that you were familiar with" and "coming in there" without a school uniform, so that "itwas pretty heart breaking", were graphic illustrations of the sustained and deleterious impact of this option on the show people's educational experiences.

Other parents also recalled the social rather than the educational drawback of this situation. One mother referred to show children "Going in shy, having to walk into a class with thirty other kids and not know any [of them]" (Y1P2). According to another parent:

.. I remember what it was like. It wasn't too bad when there were other kids and that. But when you were the only one by yourself it was the worstfeeling in the world. You're standing up there infront of all the class and you have to say your name and "Hello", and what you do. Youjust stand there and cry. It was just the worstfeeling. ...I think I used to hate that. It was horrible. (Y4P2)

The speaker explained that, largely as a result of this recurring situation, "I just went to sixth class. That was my education" (Y4P2).

This is another example of the conflict between unexamined assumptions about schools and teaching on the one hand and the show people's itinerancy and identities on the other. Unproblematic pedagogical knowledge suggests that introducing new students publicly to the class helps to make them feel welcome and lets other students know something about them. However, for the show people, this practice brought into unhelpful alliance their absence of place and their construction as other in an educational setting. One result of

this alliance was that the local students' stereotypical views of show people were strengthened by this process of literally subjecting the itinerant children to their uncomprehending gaze, because the show children were not in town long enough to develop meaningful relationships that would break down those stereotypes. The other result was the inducement of "justthe worstfeeling" in the show children, who "wouldjust stand there and cry" at having their exotic foreignness publicly displayed in this uncompromising fashion.

Another show parent explained that learning with her daughter, rather than her own educational experiences, had equipped her with formal literacy skills: "I never went into school; I just went from school to school. I never learnt to read and write, so I've been learning since I've been teaching. . [my daughter] - (Y2P1).

A member of the Showmen's Guild of Australasia synthesised these debilitating educational experiences this way:

...Going to school once every couple of weeks, and being told, "Draw that picture there", and the next school, "Draw that picture". "But I drew it at the last school." "Well, draw it again, you'll get good at it." So of course by the time the end of the run came, we had a lot of brilliant artists, and in the top brackets of the teams, we had some very good artists. And that's not a ridiculous statement; it's true, because that's all they did. They painted pictures. All day they drew pictures, all day they traced, all day. They're brilliant signwriters and everything like that, and we've got some wonderful people who can paint out there. But of course, let's be realistic.(Y4P1)

The form of unproblematic knowledge being emphasised here was the pedagogical presumption that 'practice makes perfect'. Being told to "drawit again, you'll get good at it" reflected an uncritical assumption that the best way to occupy the limited time of these itinerant children was to assign to them a generic activity that would not distract the rest of the class and that might give them a skill that they might eventually be able to use in their future occupations. The lack of coherent connection with the show children's learning before or after they entered that school was presumed to be beyond the teacher's capacity to address. This practice also tells us something about the show children's perceived 'right' to knowledge: all that they were being offered was mastery of a task that is rated very lowly in schools and society. (The speaker's concluding injunction, "...let's be realistic", reflected the show people's awareness of the limited value of that task and laid the groundwork for their resistance of such inadequate schooling provision, as the next chapter demonstrates.)

The option of show people keeping their children with them and sending them to local schools along the show circuits was clearly not effective. Limitations included recurring discontinuity in the children's learning and a reinforcement of the show children's sense of marginalisation and alienation from 'mainstream society'. The speakers' recollections of efforts to make this option succeed were filtered through their explanations of how life on the show circuits works. The lack of 'fit' between the two systems reflected and reinforced the show people's absence of place and their construction as other to 'the centre', and it also highlighted the negative impact of the way that schooling institutions represent knowledge in unproblematic ways to which people must conform.

5.5.2 Sending show children to boarding schools

Another previous option available to show people in educating their children was to send them to boarding schools. One parent described her family's experiences and perceptions of exercising this option:

Oh, I don't like boarding school. . . . My dad, he went to boarding school, and he didn't like it. So he never sent me to boarding school for that reason. And. . . [my daughter] nearly went to boarding school, but she said no, she's too young. So I pulled out of that idea. . . . Because I didn't like the idea of sending her away. I thought it's silly to have kids and then send them away for half their life. (Y4P2)

Another parent recalled the pressures on her family when she was growing up in relation to pursuing the boarding school option:

...my mum had been to board and so had myfather, but we. ..[have] a very close family network; your family is your friends and your workmates. So it's difficult with your children being away from you. (Y4P5)

A member of the Showmen's Guild expressed many Guild members' attitudes to boarding school education this way:

"We'regoing to watch...[our children] grow up once every six months if we send them away to a boarding school." And showmen are a very close knit community, and they want their families with them all the time.
..."...and we've got to not send our kids away to boarding school for the first twelve years of their life. We want them with us...". (X4A1)

Regardless of whether show people are more of "a very close knit community" than other groups, and despite the fact that a number of show people have attended prestigious Australian boarding schools, these speakers effectivelyconveyed their resentment that parental concerns to maximise their children's educational success would 'direct' them to pursue an option that physically separated family members. Furthermore, it was implicitly acknowledged that this option was not available to many show people owing to the high costs involved. Again the real point to emphasise is that knowledge that is assumed to be unproblematic about the nature of 'normal' school students and their families is found to exclude and devalue the show people's itinerant lifestyle and the associated restrictions on available options for educating their children.

5.5.3 Teaching show children correspondence lessons on the show circuits

Another educational option previously available to show people was not sending their children to local or boarding schools but instead teaching them correspondence lessons as they travelled along the show circuits. One show person also believed that correspondence schooling, without the children attending local schools in different towns, had several problems.

I don't know how the parents managed with the correspondence. I couldn't have done it. I think you've got to have a lot of patience to do correspondence with your own kids. I mean, to be a mother and a teacher, and the kids just saying, "Well, I'm not doing it". If you're a

teacher, you can say, "Yes, you are", but if you're the mother you just don't seem to be able to do anything about it. I've seen plenty of people nearly fall apart trying to do correspondence. It must be really hard. (Y1P1)

Another parent recalled her own mother's efforts to educate her siblings and herself via correspondencelessons:

Well, my mother used to want to kill us, but I mean, we had to persevere; but we did do it. . . .But no, I have been on correspondence all my life. I haven't got a bright education, but I do know enough to get me by. Which unfortunately for my kids, that wouldn't be enough in this day and age. I'd like them to have a little bit more, (Y4P4)

Her marginalising experiences of correspondence education prompted her to make alternative arrangements for educating her own children:

...I have a nine and a seven year old. And they have done correspondence for a few years on the circuit with me. I am a single mum. I'm bred and born on the showgrounds. And as you can see, I've got another generation coming up in the world. It was hard. I persevered with working and that with them, but only could fit an hour or two in. (Y4P4)

Another parent recalled her own experiences studying via correspondence lessons:

My aunty was a correspondence teacher, my dad's sister. It was really hard. They put us on it when we were kids but we were hopeless. It was too hard. (Y2P1)

Another parent remembered her mother's efforts to provide correspondence lessons for her siblings and herself

I just remember my mum had such a hard time. . . . And she 'd try and do her best, but there were six children in our family. And she could notice that we were missing [out on education], so she took on correspondence for about twelve months, and she had four kids in four different grades, and two toddlers running around underneath her. And it's just horrendous. It's not a way – you know, people can say, "Well, you set aside time". My mum was feeding four men that were out working. So meal times were solid times when she was working. And then she was also selling for us. We didn't have much money; she was trying to get everything as much as we could. So it wasn't easy at all. She was prepared to work the extra hours and do as much as she could, but still she felt her children were missing [out on education]. (Y4P5)

She recalled also her own impatience with some of the content of the correspondence lessons that she had completed:

When they send you correspondence, there's so much junk, like crafts and stuff. We're not lacking in that stuff because we're setting up and we've got vision and we've got scope in what we're doing in our work. So it's not really & value, if you know what I mean. It's the bulk & your school work [that is important]. (Y4P5)

The reference to "so muchjunk, like crafts and stuff" reinforces my earlier point that the show people's mobility positions them 'outside' the assumptions and understandings of educational providers, whose constructions of 'legitimate' knowledge are seemingly unable to encompass clients who routinely move from place to place and who, despite or because of that circumstance, have different but equally valid patterns of knowledge construction.

These recollections construct show parents – almost without exception mothers – as valiantly struggling to educate their children via correspondence lessons that they had to incorporate into a myriad of competing and stressful demands on their time and energies. Implicit in the recollections were the point that many mothers themselves lacked high levels of formal literacy, and the presumption that many show adults and children attached far less priority to formal schooling than to the exigencies of working 'on the run'. Within the spaces of their itinerancy, then, correspondence lessons had an 'alien feel' about them and did not 'fit' into the rhythms and routines of life on the show circuits. This reinforces the proposition that education does not occur in the kind of vacuum – or in the type of discrete place – that curriculum documents often assume. This also demonstrates once again the fundamental point that, rather than being 'natural' and 'universal', existing forms of schooling are designed in ways that privilege some groups over others. In this case, knowledge is encoded, mediated and delivered in forms that work to disadvantage those whose itinerancy creates an absence of place and constructs their otherness – the show people. The 'alien feel' about this educational option is therefore an accurate index of the extent of their marginalisation.

5.5.4 Show people coming: off the circuits and sending: their children to local schools

Although it was theoretically an option for show parents to withdraw from the show circuits for the duration of their children's schooling and send the children to local schools while the parents found alternative employment, very few interviewees referred to this option. One show parent explained partly why this was so:

It breaks your family up if you decide to stay in a house and send your children to school. ..[But] economics push you to that sometimes. If you've got three or four children you can't afford to send all of them to boarding school. And f you haven't got any facilities to do it at home, and if you were trying to work and do correspondence, it's really difficult. (Y4P5)

Another parent concurred

Well, I knew what choices I had, and I didn't want to have to settle down to send. . .[my son) to school, as in leave our business and our home and hisfather. I knew that wasn't on the books. (Y1P2)

Partly her determination not to pursue this option derived from her mother's experiences of pursuing the option for the speaker's siblings and herself: ...once all of us were school age, mum had to leave [the showgrounds] and come home to send us all to school – which was very hard on mum and dad. They stuck it out together; they're still together. But it was really hard on them, and I'm so glad that we don't have to do that. (Y1P2)

This educational option not only had economic and social drawbacks but also constituted a profound if indirect attack on the show people's itinerancy. The logic underlying the option was that it was not possible to work as a show person and at the same time receive an equitable education for one's children. The fact that a few speakers referred to their parents having considered pursuing the option reflected their willingness to 'try anything' that could potentially maximise their children's educational outcomes. More broadly, this option demonstrates with considerable starkness the marginalising impact of umproblematic assumptions about how knowledge should be constructed and disseminated to students. In particular, it reinforces the argument pursued in this section of the chapter that the show people's itinerancy constructs them as unable to conform to the narrow conceptions of how the institution of schooling functions and how children and parents must function in relation to that institution.

5.5.5 Sending show children to live with relatives and attend local schools

Another option previously available to show people in educating their children was for them to remain on the show circuits and to send their children to live with relatives and attend local schools. An older show person explained the situation that caused her to send her children to live with her sister-in-law:

Well, I had them [her children] on correspondence for a while, and then they went away. My husbands's fortunately got a sister who lives on. ..a property. And they went to school at her place before they were old enough to go to boarding school. (Y4P1)

She elaborated on the reasons for taking this action:

...When. ..I first started out, it was very busy for us. We had about seven men who I had to cook for. I had to drive a truck. I had to work on the trailer. And I had three children very close together, thirteen and seventeen months apart. And it was pretty hard. I thought that it was probably better to give them that early schooling by sending them away to. ..[her sister-in-law]. Because I really couldn't cope with the correspondence. (Y4P1).

The speaker also reflected on the family disruption and emotional turmoil attendant on this decision, which were similar to the effects of sending children to boarding schools:

...you'd get yourself so worked up about it, but you sort of had to be cruel to be kind. That was my way of looking at it. And. . . if I had my time over again now, there's no way in the world I would have sent them away. I'd keep them with me, and then persevere some way. But because we were young — and we were just sort of establishing our business in this life, in this business, it was sort of hard for us. You think you're doing all the right things at the time, but now if I had my time over again I wouldn't send my kids away. I'd employ a teacher, or. . . [go] without something to have a teacher travelling with us or something like that. To teach the kids. But I mean, you learn by your mistakes. (Y4P1)

One of the mothers who described above how difficult she had found educating her children via correspondence lessons identified the option of sending her children to live with family members as a possible solution to the situation:

But then my mum and dad bought a property,...and my kids, they said to me, "Bring them down to me and I'll teach them there". So they've been there since one was six and the other was five, and they're doing all right, because it was too hard for me to work and look after my kids and give them a good education. (Y4P4)

Another parent explained the limitations of this option: "... if you've got family, you send them [children] to family, but even there you're still to me missing out on so much of their life" (Y4P2).

Like the other options previously available to show people in educating their children, the option of sending their children to live with relatives and attend local schools had some benefits but more drawbacks. In particular, once again the implicit choice was between giving children a 'good education' by sending them away and keeping them on the show circuits but giving them an 'inferior education'.

Family separations of this kind are outside the comprehension of an education system predicated on students of fixed residence living and attending school in a single location. This is a stark reminder of the negative consequences for groups who lack a place of their own (in the de Certolian sense), and who are construed as 'exotic' and 'foreign'. The result is systemic marginalisation, through these kinds of restrictions on their educational options. This is once again a signal of the power that lies behind and below forms of knowledge — power that is invisible because it is constructed as 'natural' and 'neutral', but whose effects are felt in such ways as through family separations of this lund.

5.5.6 Not sending: show children to school at all

The final option previously available in the education of show children was for their parents not to send them to school at all. I encountered only one direct reference to this option, when one parent recalled:

No correspondence, no follow on with lessons, and it was go to this school because we had to. There was a couple of years there where we had terrible trouble with truant agents and stuff like that, which I

suppose there was [a] needfor. There were children who weren't going at all. (Y4P5)

On the one hand, the reference to having "had terrible trouble with truant agents" evoked an image of further surveillance, with the show people once again being subject to the gaze of 'the centre' on account of their itinerancy, ostensibly in the name of ensuring that their children received equitable educational access. On the other hand, it is likely that these agents of the state lacked the requisite understandings of precisely why some show children and their parents would elect not to attend an institution that they found alien and disempowering. This option, therefore, encapsulates the 'second best' quality of the options previously discussed and represents what for many show people would have seemed the logical culmination of those options: if formal schooling refuses to accord us a legitimate place and constructs us as other, why should we be complicit in that process and thereby deprive our children of the educational opportunities to which they are entitled?

Finally in this account of the show people's previous educational experiences, an important point to emphasise is that a number of show people remembered individual teachers who did what they could to maximise the show children's educational experiences across the range of options selected by parents. For example, **an** older show person stated:

Now and again you'd get a dedicated teacher, who'd take them [show children] in and try and do something for them. But then they'd only be there for two or three days, and [then] off again. (Y4P3)

Another show person acknowledged:

And I will say sometimes you went [to school] and the teacher was enthusiastic to try if you were willing to learn; the teacher was enthusiastic to try and give you something. But they realised it was. . [pointless]; it was only a week and how much could they do? But you did have some. (Y4P5)

She recalled a particular teacher who had taken a special interest in her education:

We did the Queensland run around Brisbane. And so mum found a school where she knew they were really interested in us. So she used to drive us from every show and we went to the same [school]....It was a Catholic school. And there was a Sister Maria, and she really took notice of me. And she spoke to my mother, and she used to keep me for an hour afer school for special remedial [lessons]. And also I used to go in playtimes to the principal's office for special remedial [lessons]. So because they made that effort my mum would drive us back, whether we were at Sandgate, whether we were at all the shows all around Brisbane, to that school. And we went solidly for three months, every year for three or four years. And I think that was a big thing for me, because I was older and they took more time with me. (Y4P5)

In other words, show people consistently articulated their critique of their past educational experiences at the level of systemic failure rather than personal prejudice. The show people's responses to this perceived systemic

failure derived from earlier generations' educational experiences. According to a member of the Showmen's Guild:

And there was no education at the time. It was purely correspondence, but **\(\sigma \)** course the parents before them were very poorly educated as well. ... And **a** course what happened then was when that generation started to have their children, then they realised suddenly that they didn't have an education, and they realised suddenly that they were illiterate and how hard it was for them to survive in the world that was new back in the early '70s,...'80s period. They realised...it's going to be much harder in the '90s and the next century to survive without an education, let alone how hard it is nowfor us. ... So they got their heads together obviously, and **a** course it's all history now, and that is the showmen's education program. And it basically developed from that sense of need from the parents' perspective that they said, "We don't want our kids to grow up without an education, and to have every opportunity possible opportunities that weren't given to us during that '50s, '60s, '70s period". And of course, the Showmen's Guild is a perfect vehicle to establish that program through the government. (Y4A1)

The significance of this statement is its demonstration that the show people's 'tactics of consumption', which are explored at length in the next chapter, have been a direct response to, and engagement with, their experiences of marginalisation. The valiant efforts of individual teachers notwithstanding, the six educational options previously available to the show people worked in alliance with their absence of place and their constructed otherness to locate them outside 'the fold' of educational provision and its associated

unproblematic assumptions about forms of knowledge. The result is that the show people lack "the power to provide oneself with one's own place", that same power that "produces itself in and through" (de Certeau, 1984, p. 36) the same knowledge that operates to marginalise and exclude the show people from the ambit of schooling provision.

5.6 Review of the chapter

This chapter has sought to answer the first research question guiding this study: "Howdo the show people experience marginalisation?". The question was predicated on the assumption that the show people's 'tactics of consumption' in relation to their educational experiences, which are the focus of this thesis, can be understood only against the backdrop of their persistent and pervasive marginalisation arising from their itinerancy. A further assumption was that, while some manifestations of that marginalisation are overt and visible, many are not, thereby necessitating the application of the study's conceptual framework to the research data in order to make clear and transparent what for many show people are unconscious — although no less deleterious for that — experiences of being 'on the margins' of the settled community.

The conceptual lens deployed to identify and critique the show people's marginalising experiences was de Certeau's (1984) notion of 'strategies of marginalisation', considered in combination with the concept of 'place'. As he pointed out:

In sum, strategies are actions which, thanks to the establishment of a place of power (the property of a proper), elaborate theoretical pieces (systems and totalizing discourses) capable of articulating an ensemble of physical places in which forces are distributed. (p. 36)

From this perspective, strategies, like places emanating from and supporting 'the centre', can be seen to place show people immediately at a fundamental disadvantage: their itinerancy renders them outside "a proper" and therefore without power. It is this lack of power as understood by 'the centre' that both leads to, and is reflected in, their marginalised status.

More specifically, the chapter examined the following three attributes of strategies of marginalisation identified by de Certeau (1984) in relation to the show people:

- The show people's absence of place locates them outside a fixed location, whose strategic importance lies in its association with determining who has power and who has not.
- The show people's itinerancy and absence of place render them liable to
 perception as 'exotic' and 'foreign', which leads to the construction of
 their otherness in relation to the settled community.
- from their itinerancy, place them in opposition to forms of unproblematic knowledge about the 'proper' location and provision of schooling, so that the six educational options previously available to them forced them to choose between maintaining their lifestyle and maximising their children's educational opportunities.

In combination, the show people's absence of place, the construction of their otherness and their having to conform to forms of seemingly unproblematic knowledge endorsed by late capitalist schooling are all dimensions and indexes of their marginalisation. The significance of that marginalisation is profound and cannot be overstated: it is at the same time the impetus for their 'tactics of consumption' and their moves towards transformation, and the set of 'levels' or 'marks' against which they judge the extent of their progress at resistance and transformation. The show people's marginalised status is therefore integral to their 'learning on the run'.

CHAPTER SIX

RESEARCH QUESTION TWO: THE SHOW PEOPLE'S PRACTICES OF RESISTANCE

"I've had things happen with different principals and their attitudes have been not very good. But I always **make** the point **cf** setting [them] straight, because unless they learn and they know what we're about, nothing is going to improve. So you tend to try and...[move] in the right direction."

Y1P2

"And there's no boundaries in terms of our education, both in terms of geographical boundaries and in terms of what we can do and can't do."

Y4A1

6.1 Overview of the chapter

The previous chapter answered the first research question 'outlined in the introduction to this thesis. Following a de Certolian analysis, the show people's experiences of marginalisation, arising from their itinerancy, were classified as being focussed on their absence of place, constructions of their otherness and forms of unproblematic knowledge that effectively excluded them from equitable access to conventional schooling.

If that analysis captured the full range of the show people's situation, they would emerge as passive victims, forever destined to respond to circumstances as they arose and incapable of influencing or even shaping the conditions in which they experience education. Yet, as has been contended elsewhere (Danaher, 2000a), acknowledgment of the show people's marginalisation is the beginning, not the end, of the story. The argument underpinning this study is that the show people's successful lobbying for the Brisbane School of Distance Education program and subsequently for a separate school for their children constitutes a counternarrative to the traditional narratives about Traveller education within which show people are constructed as 'different' and 'deviant'. This counternarrative concentrates on the recognition and valuing of multiple forms of residence and hence of educational provision.

Further to my conscious desire to eschew idealising show people and their success as educational innovators (for example, by not suggesting that such a process was 'automatic' or 'easy'), I must emphasise at this point that the move from marginalisation to transformation implicit in the unfolding of this counternarrative is by no means straightforward or unproblematic. On the

contrary, several elements combined to make such a move possible. In particular, resistance – conceived as a positive and agential force, or what McVeigh (1997) termed "the continued possibility of alternatives" (p. 22) – is a necessary but by no means sufficient precondition if marginalisation is to give way to transformation. Exploration of that resistance and its complex nature is the focus of this chapter.

Specifically, the purpose of the chapter is to address the second research question explicated in the introduction: "How do the show people resist their marginalised status?". The chapter follows the same structure as, and articulates with and expands on the points elaborated in, the previous chapter. Thus the chapter consists of the following four sections:

- a review of de Certeau's (1984) concept of 'tactics of resistance', which provides the chapter's conceptual framework
- an account of the show people's resistance of their absence of place,
 illustrated by their multiple understandings of 'home'
- an analysis of the show people's resistance of the constructions of their otherness, demonstrated by their uses of the terms 'showies', 'local's and 'mugs'
- an examination of the show people's resistance of forms of unproblematic knowledge that enormously restricted their previous educational options, exemplified by their lobbying for and refinements of the Brisbane School of Distance Education program.

6.2 Tactics of resistance

In the previous chapter, I synthesised the most salient features of de Certeau's (1984) conceptualisation of strategies of marginalisation as the organising framework for understanding the show people's routine experiences of marginalisation. Here I provide a complementary overview of de Certeau's conceptualisation of tactics of resistance as a means of interrogating the show people's practices in resisting that marginalisation. **As** with the discussion in Chapter Five, this account articulates with the more sustained theoretical analysis of de Certeau's ideas contained in Chapter Three.

I stated in Chapter Five that de Certeau (1984) conceptualised strategies of marginalisation in terms of three elements: that they proceed from "a subject of will and power"; that they proceed from "a place than can be circumscribed as proper (propre)"; and that they engage in "generating relations with an exterior distance from" their source (p. xix). This delineation of marginalising strategies as being intimately connected with power, 'place' and 'otherness' found its counterpoint in de Certeau's corresponding elaboration of what he envisaged as tactics of resistance:

I call a "tactic," on the other hand, a calculus which cannot count on a "proper" (a spatial or institutional localization), nor thus on a borderline distinguishing the other as a visible totality. The place of a tactic belongs to the other. A tactic insinuates itself into the other's place, fragmentarily, without taking it over in its entirety, without being able to keep it at a distance. It has at its disposal no base where it can capitalize on its advantages, prepare its expansions, and secure independence with respect to circumstances. The "proper" is a victory of space over time.

On the contrary, because it does not have a place, a tactic depends on time—it is always on the watch for opportunities that must be seized "on the wing." Whatever it wins, it does not keep. It must constantly manipulate events in order to turn them into "opportunities." The weak must continually turn to their own endsforces alien to them. . . (p. xix)

Thus tactics of resistance are deployed by "Theweak", in contrast to the exercise of marginalising strategies by the powerful. Similarly, this lack of power helps to explain why a tactic "has at its disposal no base" and why "it does not have aplace". That is, tactics are restricted to unstable 'space', with occasional and temporary forays into the 'place' of the powerful. Furthermore, tactics are resorted to by those who are constructed as 'other' by the forces of power with marginalising strategies at their disposal. All three of these characteristics of tactics of resistance therefore position them as diametrically opposed to strategies of marginalisation: instead of having power, 'place' and 'otherness', they lack power, they have only fleeting access to 'place' and they are constructed as 'other' through their marginalising experiences.

As with his elaboration in Chapter Three of *The practice of everyday life* (1984) of his earlier synthesised overview of strategies, so de Certeau extended in the same chapter his initially compressed account of tactics:

...a tactic is a calculated action determined by the absence of a proper locus. No delimitation of an exteriority, then, provides it with the condition necessary for autonomy. The space of a tactic is the space of the other. Thus it must play on and with a terrain imposed on it and organized by the law of a foreign power. ... It operates in isolated

actions, blow by blow. It takes advantage of "opportunities" and depends on them, being without any base where it could stockpile its winnings, build up its own position, and plan raids. What it wins it cannot keep. This nowhere gives a tactic mobility, to be sure, but a mobility that must accept the chance of serings of the moment, and seize on the wing the possibilities that offer themselves at any given moment. It must vigilantly make use of the cracks that particular conjunctions open in the surveillance of the proprietary powers. It poaches in them. It creates surprises in them. It can be where it is least expected. It is a guileful ruse. (pp. 36-37)

I wish to emphasise two elements of this elaborated conceptualisation of tactics. Firstly, the lack of "the condition necessary for autonomy" means that "The space of a tactic is the space of the other". That is, 'tactic' and 'space' are the devalued and marginalised binary pairs of the more powerful and commanding 'strategy' and 'place'. The previous chapter demonstrated the direct and varied consequences of the show people of their absence of place — including the constructions of their otherness and the pervasiveness of forms of knowledge that routinely exclude them. So power and its absence are as much in the centre of the conceptualisation of tactics and 'space' as they are evident in 'the centre's' deployment of marginalising strategies as outlined in Chapter Five.

Secondly, the reference to a tactic's having "a mobility that must accept the chance of serings of the moment" recalls two crucial features of the show people's literal mobility. The first is that that mobility is a visible sign of the show people's assumed absence of place – that they have no 'home' of their

own, and so their itinerancy is at once a cause and a consequence of their absence of de Certolian place. The second is that that same mobility gives show people opportunities to resist the marginalising strategies that they encounter in each new place, by seizing "on the wing the possibilities that offer themselves at any given moment". This suggests that the show people are attentive to such "possibilities" and work "vigilantly" to make the most of them.

I indicated in Chapter Five that de Certeau followed his detailed conceptualisation of strategies of marginalisation with a delineation of three key attributes of such strategies: that they deny a place to those whom they conceive as enemies; that they construct those enemies as 'other'; and that they perpetuate their power through the establishment of forms of seemingly unproblematic knowledge about the world. Correspondingly, de Certeau followed his detailed conceptualisation of tactics of resistance with three further comments (albeit in a slightly different sequence) that provide another counterpoint to his remarks about strategies. Firstly, he noted:

Lucking its own place, lacking a view of the whole, limited by the blindness (which may lead to perspicacity) resulting from combat at close quarters, limited by the possibilities of the moment, a tactic is determined by the absence of power just as a strategy is organized by the postulation of power. (p. 38)

Here is the obverse and the consequence of a strategy's working to deny a place to the less powerful: the absence of place is directly correlated with "the absence of power". This lack is precisely what animates and motivates a tactic.

Secondly, de Certeau argued:

In short, a tactic is an art of the weak. . . . [T]rickery is possible for the weak, and often it is his [sic] only possibility, as a "last resort": "The weaker the forces at the disposition of the strategist, the more the strategist will be able to use deception." I translate: the more the strategy is transformed into tactics. (p. 37)

This refers to the tactic's response to the construction of the less powerful as 'other'. It is important to make it clear here that I interpret "an art of the weak" in this context as referring more to a relative than an absolute lack of power and agency. That is, I envisage tactics as potentially contributing to the 'writing' and 'speaking' of counternarratives in ways that actively resist, and possibly transform, constructions of groups like the show people as 'other'.

Thirdly, de Certeau contended that

...a tactic boldly juxtaposes diverse elements in order suddenly to produce a flash shedding a different light on the language of a place and to strike the hearer. Cross-cuts, fragments, cracks and lucky hits in the framework **a** system, consumers' ways **a** operating are the practical equivalents of wit. (pp. 37-38)

I regard this as an effective counterpoint to the strategy's complicity in concealing the existence of power behind forms of apparently unproblematic knowledge about the world. The reference to "a flash shedding a different light on the language of a place" conjures up a faint but persistent illumination of something previously taken for granted that reveals it in its 'true colours' – that is, as the realm of 'the centre', supposedly invisible and

innocent, but actually the source of the marginalising strategies that work to privilege some groups over others. In this way, little by little, alternative forms of knowledge can be inserted into the 'place' of 'the centre'.

Again as he did in the case of strategies and 'place', de Certeau provided a useful summary of his conceptions of tactics and 'space':

Tactics are procedures that gain validity in relation to the pertinence they lend to time—to the circumstances which the precise instant of an intervention transforms into a favorable situation, to the rapidity of the movements that change the organization of a space, to the relations among successive moments in an action, to the possible intersections of durations and heterogeneous rhythms, etc. . .[T]actics are a clever utilization of time, of the opportunities it presents and also of the play that it introduces into the foundations of power. (pp. 38-39)

From this perspective, if show people are to resist the marginalising strategies that confront them on account of their itinerancy, they need to take full advantage of time, by maximising "theopportunities it presents". That is, they need to use the limited time that they have in each new space through which they pass on their travels and their negotiations to seek to resist traditional narratives, or ways of portraying and understanding show people, and to replace those narratives with counternarratives based on more positive, agential and transformative constructions of themselves. Only in this way will they have any possibility of contesting and transforming "thefoundations of power" that represent them as 'different' and 'deviant'.

6.3 Resisting absence of place: "Where is home to you?"

Following de Certeau (1984), I argued in Chapter Five that a major element of the show people's experiences of marginalisation is their alleged absence of place. Their itinerant lifestyle is held to position them 'outside' the purview and the concern of 'the centre', which, it is possible to argue, finds their constant movement through space a threat to its dominance and which seeks to neutralise that threat by speaking and writing them 'off the page' of official discourse. So absence of official place for the show people functions to prevent their entry into terrain that is crucial to 'the centre's' power and vital to "the foundation of an autonomous place" that is fundamental to the 'proper's' "triumph of place over time" (de Certeau, 1984, p. 36).

Yet, as shall become clear, the show people do not accept meekly or unquestioningly the official status of their absence of place. On the contrary, they use all means at their disposal to construct 'a home away from home' when they are travelling, whereby they strive for similar comforts and routines as those enjoyed by permanent residents. This, of course, is an obvious point: to evince surprise at this situation would be to perpetuate the same stereotypes already critiqued. Less obvious, however, is the fact that they are very well aware of the link between resiliently stereotypical assumptions that they have no home other than their caravans and the perpetuation of their marginalised status. Accordingly they exploit opportunities such as this research project to seek to dispel those assumptions by talking openly about the reality of their residential arrangements. Even more fundamentally, they exhibit a complex and variable set of understandings and experiences of home whose effect is to

give them, on their own terms, a sustaining sense of place. In combination, these tactics effectively resist the absence of place that is ascribed to show people and that is deeply complicit in their marginalisation.

The political significance of this resistance of absence of place cannot be overemphasised. If the show people are to counter traditional and stereotypical images of their otherness, and if they are to achieve a specialised form of educational provision that responds to their difference, they must first be accepted by decision makers and shapers of attitudes that they have a legitimate existence and a legal and moral right to expect equity of access to resources and services. This acceptance in turn depends crucially on their capacity to convince others of their rightful 'speaking position' – as having the knowledge and understanding required to articulate their needs and aspirations. Unless they can replace the marginalising strategy that denies them a 'place' (and hence a respected 'speaking position'), they will not be able to convince anyone of anything, because they will not be heard.

In this context, it is helpful to recall de Certeau's (1984, p. 18) assertion:

Innumerable ways α playing and foiling the other's game. .., that is, the space instituted by others, characterize the subtle, stubborn, resistant activity of groups which, since they lack their own space, have to get along in a network of already established forces and representations. People have to make do with what they have.

The point to emphasise here is that challenging dominant discourses about what is 'home', and all the associated benefits and resources, is a political tactic by show people that helps them "toget along in a network of already

established forces and representations". This tactic of "mak[ing] do with what they have" reflects strength rather than weakness, because it represents an act of political ability and agency by the show people to legitimatise their multiple experiences and understandings of 'home' as a means of resisting their absence of place.

Within the parameters of this argument, then, many show people readily referred to the show people's caravans as their 'homes'. A twelve year old girl explained how she helped her parents to sell food bags from a canteen at the show: "SometimesI do about an hour a day when I come homefrom school, working in there" (Y2C4). A seven year old boy commented, "I'vegot a big stack of books that I like at home" (Y1C7). One boy explained about his seven year old friend, a fellow show child: "Hehas to do his chores at home" (Y1C2). A mother stated, "My children come home from school; they're allowed to go and play", and she was pleased that "the kids come home bouncing and happy. ..." (Y1P2). She described how she supervised her son's homework in the family caravan: "As you would as a parent that was in any school situation, you see what comes home and you ask where he's up to" (Y1P2).

For the show people, their references to their caravans as 'home' was closely connected with their common cultural identity, and hence with resisting their absence of place. The caravan is for show people a functional necessity: it is their means of efficient, reliable movement from one town to the next; it must accommodate themselves and their possessions in a very restricted space; and it is one of the very few 'private places' in a lifestyle in which privacy of thought, word and action is at a premium. The caravan is also a

tangible link with the show people of previous generations, who depended in the same sorts of respects on conveyances similar in function to, but very different in form from, those used by the contemporary 'showies'.

At a deeper level, the initial identification of 'home' as caravan in most interviews functions simultaneously in several different ways. Firstly, it is an obvious and tangible marker of difference between show people and non-show people, thereby illustrating the connection between absence of place and marginalisation (even though the distinction between 'showies' and 'non-showies' is fluid, as the next section of this chapter demonstrates, and even though many Australian who have no connection with show life live in caravans).

Secondly, the show people's identification of their caravans as 'home' reveals a great deal about their sense of their own identity. Although occasionally irate about the work involved in justifying this to others, they are comfortable with publicly identifying a moving residence as their 'home', and one show person was confident enough to conduct a videotaped interview in a caravan (Y4P6) – admittedly her sister's rather than her own, on the grounds that her sister's caravan had recently been extensively remodelled and would therefore convey a more positive image of show people's living conditions. This is significant because it demonstrates how the show people tell other, alternative stories about itinerancy and home and thus wear down the power of traditional narratives about those concepts.

Thirdly, show people vary in the extent to which they conceive of their caravans as their homes on a permanent basis. Apart from being likely to differ in the longevity of their commitment to spending their working lives on

the show circuits, show people understand 'home' in several ways apart from being their caravans. This suggests less a degree of impermanence in the show people's lives than their capacity to engage as creatively with changes to physical residence as with changes to less tangible aspects of their lives.

The point that I am emphasising here is that show people's references to their caravans as 'home', and their use, appropriation and re-definition of that term, reflect the habituated and situated nature of their experiences of living on the show circuits. That is, on a day to day basis such references are literally factually correct: 'showies' are 'at home' when they travel in their caravans. Their differing and complex responses to my explicit question "Where is home to you?", however, demonstrate their awareness that the 'home:caravan' homology is the one uppermost in local people's stereotyped views of 'showies', and furthermore that that homology forms the basis of their marginalised status. This awareness is vital to the show people's resistance of their absence of place through their deployment of multiple meanings of 'home'.

Show children made varying responses to the question "Where is home to you?". An eleven year old boy answered, "Probablywhere I'm like living now. Dayboro" (Y2C1). Many of the nuances in discussions of 'home' were encapsulated in the following exchange with a ten year old boy:

[But if I said to you, "Where is home to you?", what would you say?]

Brisbane.

[Brisbane. You've got a house in Brisbane?]

Oh, it's not really mine, it's my aunty's, but we usually live there.
(Y1C1)

A ten year old girl reported a similar residential situation:

...we have a house in Brisbane, and my nana, sometimes, because she's getting too old now, so she just stays at home, and she tries to help me in school work, so I stay home with her most of the time. (Y1C8)

Another ten year old girl referred to home as

Brisbane. ..because we [stay there] about two or three times a year. It's really our nana's, but, you know. We have a place in Sydney, but I haven't been there since I was about two. (Y1C4)

One other ten year old girl, who claimed not to remember where she was born, identified home as being "in Queensland" (Y2C3).

Adults connected with the show circuits also responded in various ways to the question "Where is home for you?". One parent replied, "Here. Here this week. We just travel all year round" (Y1P1). Another parent said, "Yes. Sort of. Off and on" (Y2P4) in answer to the question, "So is Melbourne home to you?".

The potential contradiction implicit in the conflation of show people's understandings of 'home' as a caravan and a house can be explained partly by the interpolation of the explicit question "Where is home to you?" posed by an interviewer who presumably lived in one place. Under these circumstances, the most likely response to the question was probably going to refer to a specific residence in a particular town. Nevertheless, this conflation of

understandings is significant in this discussion of the show people's resistance of their absence of place. Far from being perpetually 'itinerant' or 'nomadic', these people are fully acquainted with the benefits and drawbacks of both sets of living conditions — of travelling from one town to another once or twice a week, and also of living in one place for a defined period. The latter experience is an important element of the show people's identity. This circumstance points also to an adaptability and a versatility that indicates show people's ongoing engagement with change in their everyday lives and that belies the stereotyped image of a show person seeking refuge from 'the real world' in the glamorous allure of an itinerant lifestyle. Furthermore, this circumstance is a direct riposte to 'the centre's' ascription of absence of place to the show people: measured in 'the centre's' terms, the show people are in fact regularly present in the place officially denied to them.

All of this emphasises the pervasiveness of the stereotyped association between 'home' and a single site of permanent residence. In talking about 'home', show people are aware that for many Australians permanent residence is a mark of stability, reliability and trustworthiness. This awareness has a major influence on their resistance of their absence of place, resulting in their tendency to talk to non-show people (including educational researchers) about 'home' in terms of where their family homes are located. This suggests that it is difficult – but not impossible – for show people to insert alternative forms of residence into the dominant discourse on 'home', forms that undermine the dominant assumption of fixed settlement. It indicates also the enormous impact of that dominant discourse in separating show people from other Australians and in turning their perceived lack of a 'proper home' into the basis of their marginalisation.

Significantly, one of the show parents remarked, in the context of describing previous career changes by other family members and herself, "But as they say, you always come home" (Y4P4). This comment encapsulates many of the multiple meanings identified in this section of the chapter as being ascribed to the term 'home'. Those multiple meanings in turn are vital elements of the show people's complex and subtle sense of place, which this section of the chapter has demonstrated as being a direct and powerful counterpoint to their ascribed absence of place.

This tactic resonates with de Certeau's (1986, p. 227; emphasis in original) insight that "...the political relevance of the geo-graphical distinctions between separate places is echoed...in the distribution of places of power...". As I argued at the beginning of the section, constantly reasserting their legitimate and appropriate experiences and understandings of 'home' is far from being mere pedantry on the show people's part. On the contrary, they are playing for extremely high stakes: nothing less than official and public recognition of their right to live the lifestyle of their choosing, and their concomitant right of access to specialised educational provision. From this perspective, their alternative conceptions of 'home' function to sustain them in their resistance of marginalisation and to underpin their counternarrative to official discourses that construct them as 'other' on the basis of their alleged absence of place.

6.4 Resisting constructions of otherness: 'Showies', 'locals' and 'mugs'

In the previous section of this chapter, I indicated how the show people use their understandings of 'home' to resist, strongly and repetitively, consequences arising from dominant perceptions that they lack not just a static residence but also a political 'place'. Now I turn to examine how they resist, equally strongly and repetitively, their marginalised status in the eyes of 'mainstream society' on account of their perceived 'difference', 'lack' and 'deviance'. In Chapter Five, following de Certeau (1984), I explained how defining the show people as foreign objects led directly and inexorably to the constructions of their otherness. As I cited in that chapter, de Certeau used a dramatic metaphor to convey the power and force of such a construction:

It [the "proper"] is also a mastery of places through sight. The division of space makes possible a panoptic practice proceeding from a place whence the eye can transform foreign forces into objects that can be observed and measured, and thus control and "include" them within its scope of vision. To be able to see (far into the distance) is also to be able to predict, to run ahead of time by reading a space. (p. 36)

In this section, I demonstrate that the show people deploy their own uses of identity associated with the terms 'showies', 'locals' and 'mugs' to resist their constructed otherness. I wish to emphasise at the outset that those alternative uses of names are far from being mere quibbles over nomenclature: they go to the heart of how the show people see themselves 'on their own terms' as well as in relation to others. Their naming practices act in concert

with their multiple experiences and understandings of home, described above, to contest and subvert the debilitating marginalisation to which they have traditionally been subject. Finally, as I explore in the next section of this chapter, the show people's subversive naming practices in turn enable them to resist the forms of seemingly unproblematic knowledge about appropriate schooling for itinerant people outlined at the end of the previous chapter.

What I am arguing, therefore, is that the show people's efforts to recapture and remake in their own image the term 'showies', and use it in combination with the terms 'locals' and 'mugs', are an act of tactical and political resistance. That resistance is directed squarely at the "panoptic practice proceeding from a place whence the eye can transform for eignforces into objects that can be observed and measured" identified by de Certeau (1984, p. 36), and is intended to nullify 'the centre's' power to "control and 'include' them within its scope of vision" (p. 36). Elsewhere de Certeau recognised language's 'double-edged' capacity to function as an ally of both/either marginalisation and./or resistance when he noted that "... language is indeed the privileged terrain on which to discern the formal rules proper to such practices [of resistant or tactical consumption]" (p. 32).

I argue that the show people carry out this resistance by putting forward their own alternative and positive constructions of who they are, both on their own terms and in relation to other Australians. This is based on the assumption that if social agents, which is how I understand the show people to be, seek to resist their marginalised status it is not sufficient for them to say, 'I am not like your view of what I am'. On the contrary, they need to go much further, to be proactive rather than merely reactive, and to say, loudly, clearly

and often, 'This and this and this are what I am – not just *that*'. Resistance understood in this way is crucial to understanding how the show people move from marginalisation, through resistance, towards transformation of their situation.

From the perspective of the preceding conceptual framework, then, the show people resist the use of the term 'showies' by non-show people to mark the show people as 'different', 'lacking' and 'deviant', thereby contributing to the construction of their otherness. That construction is centred on the fact that the term 'showies' is used by non-show people in ways that homogenise the show people. The consequences of this homogenisation include the elision of the diversity of life on the show circuits and its reduction to a single label, and the essentialisation of the show people's 'difference' and 'marginality', by assuming something to the effect of 'once a showie, always a showie'. In combination, this homogenisation and essentialisation work to represent the show people as always and irredeemably 'other' to 'normal', permanently resident citizens.

Evidence of the marginalising stereotypes ascribed to show people, and hence of their constructed otherness, was presented in Chapter Two. As a reminder of those stereotypes, in the view of Broome with Jackomos (1998), Australian "showpeople were viewed by the rest at society with bothfear and wonder, and as outcasts" (p. viii). Examples were provided by two of Bob Morgan's interlocutors in *The showies* (Morgan, 1995). According to Tommy Castles:

...in the old days when the showies came to town, the townspeople used to lock up their daughters and chooks [chickens] – in that order – because these terrible interlopers were all considered gypsies, and occasionally there was a 'wild Card' in the pack. (p. 13)

Similarly, Frank Foster recalled:

You've got to remember that we show people get a load **a** bad publicity. In days gone by when the show came into town people used to say, "Pull your washing in and lock your daughters up". (Cited in Morgan, 1995, p. 128)

These acknowledgments of marginalising stereotypes provided the launchpad for the show people's resistance of the constructions of their otherness.

The show people resist those constructions of otherness in three principal ways, all of which reflect their ability to make of the label 'showies' "something quite different from what their conquerors had in mind" and an associated commitment to using it "with respect to ends and references foreign to the system" in which they are located (de Certeau, 1984, p. xiii). Firstly, the show people assign to the label 'showies' different, and far more positively valenced, connotations than those understood by non-show people. In doing so, they are effectively saying, 'This is the label by which others know us, but we shall use this label in ways that suit us and strengthen rather than weaken us'. Secondly, the show people disrupt the homogenised and essentialist cast of 'showies' as used by non-show people. They do this by using the terms 'showies' and 'locals' in ways that make the distinction between those terms dynamic and fluid rather than a rigid and marginalising

dichotomy. Thirdly, the show people sometimes deploy the term 'mug' as a kind of reverse terminology, whereby 'mugs' are representatives of 'the centre' that thinks that it has safely marginalised and weakened the 'showies', who are in fact stronger and more resilient than they appear and are indeed more competent and successful than those representatives when they venture onto the show people's territory. All of this reinforces the view that the terms 'showies', 'locals' and 'mugs', as deployed by the show people, are the sites of overt and continuing resistance of their constructed otherness.

<u>6.4.1</u> Giving 'showies' a positive valence

Several show people made comments that reflected the first means of resisting their otherness identified above – that is, their attribution of special and positive features to 'showies'. One set of such comments was concerned with identifying certain characteristics or qualities that the speaker asserted as being present among show people and implied as being absent from local people. One man praised the adaptability of show people.

As Tex Morton once said, "Showme **a** showman's son or daughter and there's nofools amongst them". He sang a song about them. They were.

..[such] good children, they adapted to other things. (Y4P3)

One woman explained how this adaptability operated in the practical, everyday conditions of show life.

I mean, it might sound fun, travelling and all that, but there's a lot of hard work involved and it's very hard. . .to teach children and travel and work and that. (Y4P1)

The speaker asserted that, because "they see so much with the travelling", show children's "outlook on life is really great" (Y4P1). Another woman referred to show children's special maturity, which by implication derived from their capacity to adapt to changing circumstances.

...I think they're pretty well very smart kids to start with. They're very grown up. ...[My daughter is] only seven, but you can sit down and have a conversation with her like she's an adult. They know what's going on outside. (Y4P2)

I should point out here that this study was not intended to verify these perceptions articulated by show people. My concern is with their significance as manifestations of the show people's representations of themselves and non-show people, not with the veracity of the show people's claims. From that perspective, these statements are powerful demonstrations of a belief that show people are 'different' and 'special', less with an exclusionary purpose than as an expression of identification with and pride in a particular cultural heritage. Furthermore, if that heritage is perceived as 'normal', 'natural' and 'desirable', show people are more likely to demand access to what others construct as 'normal' resources, including educational provision — in other words, to resist their absence of place and their constructed otherness. At the same time, I need to emphasise that 'difference' is not automatically associated with 'deviance' or 'lack' — that the show people take active agential pride in their multiple signifiers of identity, rather than merely and passively using those

signifiers as 'protective shields' against the hostile strategies of marginalisation.

The belief that show people have special attributes deriving from the exigencies of their distinctive lifestyle prompted a couple of assertions of identity demarcation. In discussing the difficulties faced by newcomers to the show circuits, one woman stated explicitly, "Butit is better, I suppose, if you can marry within your own boundaries" (Y4P1). Another woman, whose nine and seven year old children were currently living with her parents while she travelled on the show circuits, asserted: "If we get a good school or a good education department that will teach our kids in our own environment, I'll bring them back thenfor sure" (Y4P4). Both these statements reflected a direct association between show life and being in one's "ownboundaries" or one's "ownenvironment". The corollary of this association is the necessity of 'the other' as a way of defining what falls within and what lies outside those boundaries and that environment. This is the function of the show people's representations of 'locals', which as I discuss below accordingly and inevitably has a less positive valance than 'showies' for many show people.

This theme of resisting constructed otherness by means of adding a positive charge to the term 'showies' certainly underpinned several statements by the show person whom, for reasons outlined in the previous chapter, I interviewed twice. For example, during the videotaped interview she emphasised the advantages of the show people's lifestyle.

I must be biased, because I think we have the best lifestyle. We have everything. We have Australia's most beautiful places that we see every year. We have travelling, life, colour, movement, all in our world. We do

things, we move, we have activities, we have a little taste of everything. We go to places where they do bungee jumping. We all try. ..whatever's the tourist spot in town, we go and have a look. We get to look at all this beautiful scenery, the countryside. We get to do all that, plus we can earn a living and do it. We don't have the stressful [need to] get up at nine o'clock, but we'll work hard. But we have so much scope in what we can do. ..[W]hat we experience every day is not regular, and it's not routine. So therefore it's exciting to get up each day and not know what may be at the end of the day. . . .But it's so good. You have so many advantages that people can't even conceive. (Y4P5)

This is a powerful and proud articulation of the pleasures and benefits of occupational travelling. It is noteworthy that the speaker argued strongly that show life is full of variety, experience and excitement. Moreover, she represents the show people's lifestyle as being "better" than that of "most people", with "so many advantages that people can't even conceive". This representation suggests that this show person had a clear notion of her own identity and that of her fellow 'showies', and that that identity gave her a secure basis for resisting 'the centre's' efforts to construct her peers and herself as 'other'.

Secondly, the speaker identified certain special characteristics that she ascribed to 'showies', and that by implication were the prerequisite of their enjoying the special advantages that she had already outlined.

We've got the drive, we're not scared to work. See, the thing is with our growing up is our form of doing things is that if it's broken, you fix it. You don't have to ask someone, you have to do it today. We're here for two days, we earn our money now, we do our jobs now. (Y4P5)

This statement functions to position the show people as a young, energetic, dynamic group of people, in contrast to the inbuilt conservatism of some business and government bureaucracies. According to this admittedly simplistic representation, the 'showies' are utilitarian and practical in their approach to life: "if it's broken, you fix it". They are also task oriented and well organised, largely owing to the regularity with which they move from town to town. This regularity means that they must complete tasks here and now: "It's always urgent". Again, the assumption is that these are special characteristics of show people, and that they need to be recognised and celebrated, in place of others' less positive constructions of 'showies' as 'other'.

Thirdly, the speaker's identification of distinctive qualities of 'showies' led her to stake a claim for those qualities underpinning a special identity.

...we need to have our own identity, we need to be separate, and we need to be able to haveflexibility within ourselves to do things that we need. We can't have to waitfor bureaucracy or if it doesn't please one person, they say, "No", and all of a sudden we have to put our hands down. We're not those sort of people. We're the sort of people who get things done and do things. And if we have that independence and freedom, we'll do a lot. (Y4P5)

For the speaker, the basis of the show people's "ownidentity", and of their "need to be separate", is their "flexibility", "independence" and "freedom". These qualities – by implication specific but not necessarily exclusive to 'showies' – make them "the sort of people" that they are: hard working achievers who do not have time to waste on "bureaucracy" and the obstructions of people who do not understand show life and who certainly could not survive on the show circuits. On the other hand, if decision makers have the foresight to give show people what they seek, the 'showies' will be very successful at whatever they set out to achieve: "we'll do a lot".

This is the crucial point about this particular approach to resisting otherness: the show people have something positive to replace the 'deficit' and negative stereotypes ascribed to them by others. This point articulates with one of de Certeau's (1984, p. 30) examples of tactics of consumption in operation:

Thus a North African living in Paris or Roubaix (France) insinuates into the system imposed on him [sic passim] by the construction of a low-income housing development or of the French language the ways of "dwelling" (in a house or a language) peculiar to his native Kabylia. He superimposes them and, by that combination, creates for himself a space in which he canfind ways of using the constraining order of the place or of the language. Without leaving the place where he has no choice but to live and which lays down its law for him, he establishes within it a degree of plurality and creativity. By an art of being in between, he draws unexpected results from his situation.

I argue that the show people's attaching a positive valence to the term 'showies' is a strilung example of how they have created "for [themselves] a space in which [they] canfind ways of using the constraining order of the place or at the language", by loading the term with a far greater range of semantic associations than that envisaged by 'the centre'. This avowedly political tactic resonates with other minority groups' appropriations of derogatory terms such as 'black', 'gay' and 'queer' to disrupt the marginalising strategies connected with their original use. The greater range of meanings assigned by the show people to 'showies' is an example of their establishment of "adegree afplurality and creativity", because it is based on a flexibility and multiplicity of linguistic usage that resist efforts to homogenise and essentialise 'showies'. Furthermore, by 'turning the tables' on 'the centre's' efforts to elide the show people's identity and agency, the show people have indeed drawn "unexpectedresults from [their] situation".

6.4.2 Disrupting.the 'showie'-'non-showie' dichotomy

The second approach to resisting constructions of otherness that I identified earlier in this section was the way in which show people break down the seemingly essentialised, fixed and homogenised 'showie'—'non-showie' dichotomy. This is another way of contesting and subverting the marginalising strategy of representing the show people as naturally and irredeemably 'marginal' and 'other' to 'normal' Australians, by demonstrating that the conceptual barrier on which such a construction is predicated is actually a fluid and shifting set of discourses.

In that context, the show children made several references to the 'showie'-'local' division being fluid and temporary. A twelve year old girl asserted strongly, "No, we travel around, because we're with the showmen" (Y2C4). She explained: "Wesettled down for a while and went to school, and then mum got us correspondence and we started travelling again" (Y2C4). Her reference to having "settled down" suggests that the change of status from 'showie' to 'local' and back to 'showie' was not necessarily an easy or straightforward transition for her. Her description of relations with 'locals' was a somewhat negative one. She recalled that local teachers

...treated us different[ly] from the locals. We weren't allowed to do things that they were allowed to do. ..[W] hen they used to go. . swimming. .., we weren't allowed to go. (Y2C4)

Despite her earlier reference to having "settled down" as a 'local', here 'locals' assumes a different meaning and purpose – to point out how in her view show children have been poorly treated by educational providers. Yet her explanation of her situation also indicates the transitional rather than the separate links between 'showie' and 'local'.

One boy, who stated that "I'vegot a couple offriends who are locals" (Y1C5) but most of whose friends followed the show circuits, described at second hand the reverse transition from the 'showie' to the 'local' categories. He explained that "my brother and 2 had a friend and he went off the show and he's a local. ..[and I don't know where he lives]" (Y1C5). Perhaps his concern at the loss of a friendship underlay his assertion that 'locals' "can come back" to the show circuits, and his statement that "I think a few at them" actually do so (Y1C5). Here this boy's discussion of 'showies' and

'locals' is tempered by his personal knowledge of someone who had traversed the boundary between those categories, and by his evident regret that this process had ended a valued friendship.

Another boy described at second hand the change from 'local' to 'showie' status, a change with a different outcome from the situation quoted in the previous paragraph. He explained how his friendship with another boy endured the latter's changing status in connection with show life that derived from his parents joining the show circuit.

Well,...[he wasn't] actually a local. See, what it was is I said –, which is our friend really, the one on the show, because what it was is I was friends with him about two years back when he was a local, but now he's a full showman.

[...When he's a local, what does that mean?]

That means, like now we can get free rides and all that and he couldn't and all that, and he had to pay for the tickets.

[If he's a local doesn't he travel with the show?]

No. He has to stay in one place. But now he's a showman. (Y1C2)

In both these cases, the show children's references to individuals moving on and off the show circuits in some respects parallel the show people's traversing the physical and symbolic spaces of itinerancy. That is, the emphasis is on fluid and shifting markers of identity and signifiers of meaning, rather than fixed essences. The particular point that I am emphasising here is it is that fluid and shifting character that show people

exploit in resisting the constructions of their otherness – in this case, by emphasising that the difference between them and others is less clearly defined and permanent than other people might realise or like to think. This argument resonates with McVeigh's (1997) contention that historically "the sedentary/nomad distinction was much more ambiguous than it is in contemporary societies", and that

. . . even when this distinction does become unambiguous. . . it bears emphasis that nomadic—sedentary transition is not a one-way process.

Just as people can exchange a nomadic for a sedentary existence, so they can exchange a sedentary for a nomadic existence. (p. 11)

The means of resisting the show people's constructed otherness outlined in this subsection recalls a vivid metaphor provided by de Certeau (1984) and previously discussed in Chapter Two. The preceding discussion has established how the show people's dissolution of the supposedly rigid Qchotomy between 'showies' and 'locals'

...enables the [show people] to avoid being disseminated in the occupiers' power grid, to avoid being captured by the dominating, interpretive systems of discourse (or by the simple inversion of those discourses, a tactic which remains prisoner to their logic). (p. 229)

By this means, show people have succeeded in replacing the ascribed and marginalising ideology of "You are not like me and never can be like me" with the more dynamic representation of "You and I are in many respects not so different after all". This resistance sows the seeds of transformation of the show people's identity constructions, as the next chapter will demonstrate.

<u>6.4.3</u> Giving 'mugs' a negative valence

Finally in this section of the chapter, I turn to the show people's use of the term 'mugs'. I indicated earlier in the section that the third approach to resisting otherness adopted by the show people was to use 'mugs' to describe local people in a similar way to many non-show people's use of the term 'showies' – that is, as a negatively charged label that sets the labelled apart from the labellers. The significance of this approach is less that it is a device of revenge than its reversal role. That is, when 'showies' use the term 'mugs', they are referring to the way that 'locals' operate when they are on the show people's territory – generally incompetently, because they lack the specialised cultural capital that make the show people masters and mistresses of their domain. This version of resisting otherness says that 'These people who marginalise us are not competent in functioning on our territory, so therefore the cultural capital that equips them to construct us as "other" is invalid and weak' – precisely the stereotyped characteristics ascribed to show people.

For me, this is a vivid example of de Certeau's (1984) reference to "the indigenous Indian cultures" (p. 31) conquered by Spanish colonisation: "They metaphorized the dominant order; they made it function in another register" (p. 32). At the same time, it is worthwhile remembering McVeigh's (1997) timely injunction:

We must contrast the capacity of the overwhelmingly dominant settled population to racialise, marginalise and discriminate against the nomad with the incapacity of the nomad to operationalise any anti-sedentary prejudice he or she may hold. (p. 12)

From this perspective, what is surprising is not that show people are routinely and enduringly marginalised in multiple ways, but rather that their tactics of resistance take various forms and are often effective.

In an example of what I mean by this, a twelve year old girl explained the meaning of the term 'mugs' in the context of describing interactions between show children and local children.

...we call them 'mugs'.

[What does 'mugs'mean?]

Oh, it just means that they're locals and we're show kids. So mugs have to pay to get on the rides and we don't because we know all the show kids.

[So I they talk to you like that, you're not going to talk to them about your school work, are you?]

No, we don't talk to them nice[ly] \mathbf{f} they talk to us like that. We call them 'wankers' and that. (Y2C4)

This exchange between interviewer and respondent recalled a show child's explanation, cited earlier in this section, of another child's change of status from 'local' to 'showie'. In both instances 'showies' were represented as not having to pay money to enjoy the rides of 'sideshow alley' – and by implication as experiencing the benefits of an itinerant lifestyle, travelling from town to town. In the earlier exchange, 'local' was taken to refer to a permanent resident in a particular location or locality. In this exchange, 'mug' takes on a certain negative valence, in comparison with the more neutral valence of 'local'. It implies someone who is not very intelligent, or else someone who is being duped or misled. It certainly implies a greater level of hostility, of disdain verging on contempt, than that suggested by the term 'local'. This is a reversal of terminology – and its associated meanings – at work in resisting otherness.

One of the parents also referred to 'mugs'. In the process of explaining why including information about shows in the education program implemented by the Brisbane School of Distance Education was "notreally relevant to us", she stated, "I mean, most of the [show] kids, they take as much money from the mugs as we can" (Y3P1). This statement suggests that 'mugs' has a more restricted usage than 'locals', and perhaps that 'mugs' are the 'locals' who come to the shows and who spend their money without being aware of the ways in which they are encouraged to do so. This statement also indicates that show people attach considerable cultural capital to a person's skill at being able to part 'mugs' from their money. Furthermore, it would seem that the 'showie'-'mug' distinction is more securely fixed than the 'showie'-'local' distinction. For example, while some show people referred to their change of status from 'showie' to 'local' and back again, none of my respondents constructed herself or himself as a 'mug'. This point reinforces the perceived differences between show people and non-show people, and it also emphasises the instability and the contextualised character of those differences. It also underlines the multiple and successful ways in which the show people resist their constructed otherness.

This section of the chapter has outlined the show people's representations of themselves and other people according to the respective attributes of the categories 'showies', 'locals' and 'mugs'. Assigning these labels to people allows show people to say who they are on the basis of identifying who they are not – a powerful and effective reversal of efforts to construct them as 'other' on account of who they are and are not. Attaching a positive valence to 'showies', disrupting the dichotomy between 'showies' and 'locals' and assigning a negative valence to 'mugs' function in combination to contest and subvert the

...panoptic practice proceeding from a place whence the eye can transform foreign forces into objects that can be observed and measured, and thus control and "include" them within its scope of vision. (de Certeau, **1984**, p. 36)

That same resistance of the show people's constructed otherness also serves to equip them to resist the forms of unproblematic knowledge attending their previous educational experiences.

6.5 Resisting forms of unproblematic

knowledge: The Brisbane School of

Distance Education program

To this point in the chapter, I have examined the multiple ways in which the show people successfully resist their absence of place and the constructions of their otherness. That resistance places them in a stronger position to resist the educational dimension of their marginalisation. I identified in the previous chapter the forms of seemingly unproblematic knowledge that had marginalised the show people from equitable educational access for generations. Their previous educational experiences were centred on six options, all of which required them to choose between their traditional lifestyle and their children's formal education. The inadequacy and injustice of those options derived from unquestioned assumptions by educational authorities about how and where 'normal' people live in relation to the authorised sites of schooling.

As de Certeau (1984) noted about the third strategy of marginalisation that he identified:

It would be legitimate to define the power of knowledge by this ability to transform the uncertainties of history into readable spaces. But it would be more correct to recognize in these "strategies" a specific type of knowledge, one sustained and determined by the power to provide oneself with one's own place. Thus military or scientific strategies have always been inaugurated through the constitution of their "own" areas (autonomous cities, "neutral" or "independent" institutions, laboratories pursuing "disinterested" research, etc.). In other words, a certain power is the precondition of this knowledge and not merely its effect or

its attribute. It makes this knowledge possible and at the same time determines its characteristics. It produces itself in and through this knowledge. (p. 36; emphasis added)

From this perspective, unquestioned assumptions that most 'normal' students travel from home to school to receive their education, and that a minority of students stays at a permanently located home to receive distance education, are complicit with, and an integral part of, the means by which "a certain power. . . produces itself in and through this knowledge".

The show people's resistance of forms of seemingly unproblematic knowledge about how and where show people learn occurred in two distinct phases:

- establishing a specialised program with the Brisbane School of Distance
 Education
- consuming the program through a close monitoring of its operations.

While the phases were chronologically distinct, they had in common the show people's determination to turn educational provision for their children from a marginalising and alienating institution into a set of practices directly responsive to their specialised learning needs and aspirations. This constituted a mastery of history, by transforming it into readable spaces. In the process, they sought to attain positive and practical achievements from their resistance of the forms of unproblematic knowledge to which they were traditionally subjected.

6.5.1 Establishing the program

The education program provided for the show people's children by the Brisbane School of Distance Education had certain distinctive features. It was targeted at preschool and primary school children and combined face to face and distance education. Teachers from the school travelled to selected sites along the coastal and western Queensland show circuits and worked with the children in a spare classroom in a local school or in a church hall. They provided enriching activities designed to help the children to complete the written papers prepared by the Queensland Open Access Support Centre for all Queensland distance education students (most of whom live on cattle and sheep properties). When the teachers returned to Brisbane, the children continued worlung on the papers, sending their work regularly to the teachers for marking. They communicated with the teachers via telephone and facsimile machine as the need arose. Some children had access to home tutors employed by their parents to facilitate the completion of their written papers. Even when the show circuits took the children out of Queensland, they continued working on the Queensland program.

Effective learning partnerships among teachers, students and home tutors were clearly vital to the success of the program. In Queensland, teachers apply specifically to join the staff of one of the Schools of Distance Education; successful applicants already have experience in conventional classrooms, and they need to demonstrate a commitment to developing pedagogies and using technologies in ways appropriate to teaching at a distance. Teachers involved in the show children's program applied specifically to join it, and in doing so needed to acquire an understanding of the distinctive learning experiences of those children. This understanding was augmented by formal and informal contacts between teachers and show people, as awareness and appreciation of the other group's circumstances grew. From the teachers' perspective, that awareness and appreciation in turn underpinned their adaptation of the Open

Access Support Centre materials and their preparation of the distance learning packages for the show children. The fact that they had volunteered to work with those children reflected their determination to make the program as successful **as** possible.

The home tutors' role was equally crucial to maximising the show children's learning outcomes. They needed to maintain the children's enthusiasm and motivation when the teachers were not present; to supervise the children's completion of designated work and to send that work to the teachers; and to provide a communication link between teachers and students. **As** I indicated in the previous chapter, some home tutors were relatives of the children who performed their role on a voluntary basis; others were people from outside the show circuits employed especially for the task. (I explore in Chapter Seven the home tutors'—and the teachers'—perceptions of the show people's transformations of their marginalising experiences and resistant practices.)

The change from the six inadequate schooling options outlined in the previous chapter to the program described above, catering specifically for itinerant students (and largely for show children, with a small number of circus children being enrolled as the program progressed) within Queensland's largest school of distance education, demonstrated starkly the efficacy of the show people's resistance of previous assumptions about educating Travellers. The motivation underlying this resistance, and the tactical approach fuelled by that motivation, were clearly articulated by a member of the Showmen's Guild:

...it wasn't just one person. There were a group of people who sat down around the campfire one day and said, "What are we going to do? Our kids are going to be uneducated. They're not going to learn through correspondence" in terms of what correspondence was. (Y4A1)

At one level, the idealised image of decision making "around the camp fire" might be regarded as exemplifying poetic licence. At a more fundamental level, the speaker used a strongly evocative metaphor to (re)present a local event in mythic terms, thereby underlining its significance for the show people's community. The emphasis in this statement was on focussed discussion on the benefits and limitations of available options and the possibility of extending the range of such options. This focussed discussion, derived from a collective experience of an itinerant lifestyle and of the most potentially productive avenues of change, led to the formulation of a plan of equally collective action, while recognising that individual show people's contributions to the resistance would vary according to circumstance.

This notion of collectively motivated and individually varied resistance of forms of unproblematic knowledge about Traveller education was recognised by one of the teachers involved in delivering the program:

...a number of women who were Guild members lobbied hard and strong to get a program like this up and running. Through their hard work they have achieved a lot. ...That's often the only way to get it done, unless you've got a mum who's prepared to take time away from work and do that, because they are very busy, very businesslike in their approach to theirjobs, and very efficient. (Y2T1)

This observation was noteworthy in three respects. Firstly, the references to the show people's being "very busy,...very businesslike..., and very efficient" suggested that their resistance to unproblematic knowledge would be systematic, tactical and targeted. That is, they would deploy their customary skills of organisation and orderliness in a different arena to achieve their goal of extending and maximising their educational options. Secondly, the reference to "the only way to get it done" acknowledged a considerable degree of self-sacrifice on the show people's part: they were prepared to interrupt, perhaps even disrupt, their business routine to devote their energies to operating in a very different arena, that of an educational bureaucracy. Thirdly, operating in that different arena required the show people to take risks of self-disclosure with individuals who were armed with bureaucratic authority. That kind of self-disclosure betokened both courage and determination in challenging and resisting authorised assumptions about the 'place' and form of Traveller education.

These three features of the show people's resistance – organisation, self-sacrifice and self-disclosure – were very much in evidence in a particular tactic in which they engaged. This tactic centred on their interactions with the then Queensland Minister for Education, as one of the participating show people recalled:

 Show,...and through that meeting we got the program going. And that was the big step. (YIP2)

I recognise that, although many members of the general public have sent "somecorrespondence to the Education Department", very few such people have "the opportunity to meet with the Minister" for Education. That the speaker and her colleagues were provided with that opportunity by "afriend of a friend", who was highly connected with "theRoyal Brisbane Show", was testament to the show people's determination and their capacity to enlist the support of people with sufficient capital to bring about an introduction to the state official who was most likely to achieve the outcomes that they desired. In this situation, resistance took the form of refusing to allow themselves to be sidetracked by bureaucratic obfuscation or to be sidelined by circuitous conversations with underlings.

Although it is rarely at the disposal of resistant groups, the tactic of 'going straight to the top' was in this case highly effective at subverting taken for granted assumptions about educating itinerant people. On this occasion, the show people were successful at building on the power associated with the institution of the government, and enlisting that power to garner support for their educational agenda. This calls to mind de Certeau's (1984) lively metaphor about the nature and effect of resistance:

Statistics can tell us virtually nothing about the currents in this sea theoretically governed by the institutional frameworks that it in fact gradually erodes and displaces. Indeed, it is less a matter of a liquid circulating in the interstices of a solid than of different movements making use of the elements of the terrain. (p. 34; emphasis in original)

This analysis of the show people's lobbying for the Brisbane School of Distance Education program demonstrates the literal mobility associated with this kind of resistance of inadequate educational provision,

The success of this tactic of resistance was evident in the speaker's account of the events that followed that meeting:

Before I got to the Minister,...I was starting to get a little bit lost with it, because there're so many different people in charge of different sections that you didn't really know where to tap into to get a response back... And once we got to the Minister and he realised there was such a need for it, there were no problems after that. It followed suit. There were other people appointed to judge our situation, and we worked with it from there. (Y1P2)

The outcome of the initial meeting was therefore that, in contrast to their previous status **as** supplicants 'outside the system', the show people had 'forced an entry' to position themselves 'inside the system' — or at least to position themselves much more strongly with the personal endorsement of "the Minister", which endorsement would undoubtedly change the way that the "other people appointed to judge our situation" exercised that judgment. So identifying the most likely change agent was a very effective antidote to the experience of "starting to get a little bit lost with it" in the bureaucracy of a government department, and was certainly likely to ensure that when the show people contacted that department in the future they would indeed "get a response back". The crucial significance of these actions was that they helped to insert a counternarrative into official educational discourses, one based on the realities of an itinerant lifestyle rather than either replicating inaccurate stereotypes about such a lifestyle or omitting it from consideration at all.

The same speaker conveyed her assumption about how this tactic of resistance works when she stated:

I've always said it's not what you know but **who** you know to a certain point, because obviously **f** we hadn't gotten to see the Minister – he was the only real one to put. ..[the program] into the budget. (Y1P2)

Conversely, she described the result if the directly interpersonal element is absent from the equation:

I was in America when it was all happening [in Victoria] and I missed out on helping. It's hard —I don't really know how to get involved down there, because it's a different group of people. (Y1P2)

Identifying, and forming effective working relationships with, people able to make decisions that lead to change are therefore integral elements of the show people's approach to resisting unproblematic knowledge about who they are and how they should be educated.

This tactic of creating new categories and rules as a corollary of resisting old and marginalising ones was evident in the following statement by the same show person: ". . .[we] were handed over to so many different people. 'That's not really my category', but there was no real category for us. It was a whole new thing" (Y1P2). Again the show people demonstrated unambiguously their determination and their capacity to resist, rather than passively accept, their marginalisation through not belonging to any "real category at all".

In de Certolian (1984) terms, the show people's effective lobbying for a specialised program within the remit of the Brisbane School of Distance Education reflected the utility of "a way of using" (p. 18) what was previous-

ly available to them educationally to create something new and more appropriate. According to de Certeau:

...a way of using imposed systems constitutes the resistance to the historical law of a state of affairs and its dogmatic legitimations. A practice of the order constructed by others redistributes its space; it creates at least a certain play in that order, a space for manœuvers of unequalforces and for utopian points of reference. (p. 18)

From this perspective, the show people's lobbying for the Brisbane School of Distance Education was a stark example of "resistance to the historical law of a state of affairs and its dogmatic legitimations"—in this case, "dogmatic legitimations" through the authorised but inadequate educational options previously available to the show people. Furthermore, the program's establishment constituted "a certain play" in the established order, and it created for the show people "aspace for manœuvers of unequal forces and for utopian points of reference" that enabled them to begin to insert and circulate their counternarrative about what and how Traveller education could and should be.

More broadly, the reference to "a space for maneuvers" recalls de Certeau's (1984) conceptualisation of space elaborated in Chapter Three. Specifically:

In short, space is a practised place. Thus the street geometrically defined by urban planning is transformed into a space by walkers. In the same way, an act of reading is the space produced by the practice of a particular place: a written text, i.e., a place constituted by a system of signs. (p. 117)

"In the same way", the preceding analysis has demonstrated how the show people's lobbying for the Brisbane School of Distance Education clearly indicates their efforts to turn the "place" of formal schooling into "the space" in which they feel comfortable and are able to achieve academically 'on their own terms'. Making this "space" into "a practised place" by means of the program's establishment was a crucial element of the show people's resistance of the forms of seemingly unproblematic knowledge about their educational experiences and opportunities identified in the previous chapter. At the same time, I must reiterate that that establishment was never a foregone conclusion but rather the outcome of a long and continuing struggle. Indeed, the lack of guaranteed permanence associated with the program induced the show people to monitor its implementation very closely.

6.5.2 Consuming the program

The preceding subsection analysed the show people's establishment of the Brisbane School of Distance Education program as resistance of forms of unproblematic knowledge about how itinerant people should be educated. I turn now to argue that the show people's consumption of that program demonstrated a similar resistance. 'Consuming the program' in this context refers to the de Certolian sense of consumption outlined in Chapter Two, whereby consumers are not passive recipients of production but instead make their own uses of what is produced – sometimes uses that are significantly different from the producers' intentions. As de Certeau (1984) noted:

...between the person (who uses them) and these products (indexes of the "order" which is imposed on him [sic]), there is a gap of varying proportions opened by the use that he makes of them. (p. 32)

Furthermore:

...once the images broadcast by television and the time spent in front of the TV set have been analyzed, it remains to be asked what the consumer makes of these images and during these hours. The thousands of people who buy a health magazine, the customers in a supermarket, the practitioners of urban space, the consumers of newspaper stories and legends—what do they make of what they "absorb", receive, and pay for? What do they do with it? (p. 31)

The short response to de Certeau's question "Whatdo they do with it?" is that the show people took every available opportunity to make the program closer to their image of specialised Traveller education. That image was centred on aligning as closely as possible the maintenance of their itinerant lifestyle and the maximisation of their children's educational and employment opportunities. They were aware how superior the program was to the inadequate educational options previously available to them and detailed in Chapter Five. They were also aware how long 'andhow hard they had had to struggle to establish the program. They were therefore determined to consume or use it in ways likely to make it a success.

This indicates that there is a qualitative difference between the forms of consumption identified in this and the preceding subsections of this chapter. When the show people were lobbying for the program's establishment, they were consuming what had previously been available to them, in the sense of 'malung do', or 'malung the best of a bad situation'. They were 'pushing the boundaries' and using 'trial and error' to try to see what could be done to enhance their children's educational prospects against a backdrop of generations of neglect. By contrast, after the program's establishment the

show people consumed that program, in the sense of constantly monitoring its implementation in order to refine it and maximise its alignment with their distinctive learning needs. This suggests that after the program had been established the show people did not automatically assume that it was merely a transition to having their own school. On the contrary, as far as they knew at the time having a separate school was little more than a 'pipe dream', and hence they were determined to make the program as closely 'in their own image' as possible.

From that perspective of consumption as monitoring, it was to be expected that most comments pertaining to the program were complimentary. This situation reflected the show people's conscious awareness of the options that had previously been available to them in educating their children, as well as the fact that a number of show people had been directly involved in lobbying for the program's establishment and subsequently for its extension from the coastal to the western Queensland show circuits. Accordingly they had greater 'ownership' of the program's implementation and refinement. More broadly, the program had been incorporated into the practice of the show people's everyday lives, and consequently was seen as functioning as a 'consumer production' and a 'tactics of practice' of their own, as opposed to an imposition from state authorities. The much greater responsiveness of the program to the show people's distinctive educational needs and aspirations meant that it was helping to resist the show people's marginalised status and particularly the forms of previously unproblematic knowledge about the appropriate 'place' of conventional schooling. In addition, the program was seen to promote the show people's social agency in turning education from a 'place' to a 'space' (a practised place).

Within that context, then, several parents commented on the perceived advantages of the education program in comparison with the options available under their previous educational experiences. According to an older show person, "...we're very proud of what our ladies have been doing in bringing this education program forward. ...[I]t's a very good step for our way of life and our industry. .." (Y4P3). This was a significant endorsement of the program, because it reflected an acknowledgment that, unlike the previous options available to show people in educating their children, the program was at least and at last responsive to the show people's distinctive educational needs and aspirations encapsulated in "ourway of life and our industry".

This crucial point was echoed by a younger show person with a seven year old daughter:

...here you've got the best of both worlds. You can still work, [and] you're not leaving your children. Your kids are still there all the time and that, so it's good. (Y4P2)

The reference to "the best of both worlds" is important: the presumption was that, unlike previous educational options, the program had allowed show people to combine their itinerant lifestyle and their children's educational opportunities, rather than having to choose between them as unproblematic assumptions about educating itinerant people had required. This was a clear example of how the show people closely monitored the program and thereby consumed it to fit as closely as possible with their distinctive learning needs: being "the best of both worlds" functioned as a gauge for evaluating the program's continued effectiveness in meeting those needs.

One parent said that, as a result of the program's establishment, "We have got a very good relationship with the schools in Queensland. They're getting better—every year it's getting better" (Y1P2). This 'special relationship' was often expressed in terms of interactions with individual teachers from the Brisbane School of Distance Education. For example, after the involvement of a particular teacher began, the program "skyrocketed" partly because ". ..she really pushes things along, gets things going" (Y1P2). In other words, the program was a developing situation rather than a static entity: ". . I think the program's been essential for the children and that it has progressed considerably during the last three or four years" (Y1P2). She reaffirmed in relation to the show children's education:

...[I]t's so much better than what they had before the program started.

So they're a thousand times better off than they were four years ago. ..

This is so much better. (Y1P2)

It was instructive to gain this parent's perception of the success of a program in whose establishment she had been so active. She had no doubt that the program was achieving its aims. A principal benefit was that the children now derived some meaning from their schooling.

Now with the program they're not only a part of something that belongs to them, that theyfeel apart of, we have a gauge, a learning system that we can control. I know with. my son who's eight, in Grade Three, he's right **up** to date with the average school, and in some subjects he's also ahead, so you can't be happier with the program. (Y1P2)

Although there had been some problems with the postal service, this parent felt that her son's work had not suffered.

So that's the only gauge you can use. **As** long **as** they're **up** to date and you are getting the work through quickly. And the quality is very, very good, everything's excellent. (Y1P2)

She believed also that the children's social slulls had improved significantly: "Now, as I said, these kids are a part of something and they know it, and their self esteem and confidence is up here" (YIP2). The reference to using a "gauge" to measure the program's effectiveness reflected the show people's close monitoring of its implementation, while identifying the show children as "apart of something" was an implicit acknowledgment that previously show people had been dislocated from, rather than being integrated with, formal educational provision. This close monitoring, as I posited above, represents a particular form of consumption of the program, and hence a resistance of traditional assumptions about how and where itinerant people should be educated.

Another parent, a relative newcomer to the showgrounds, concurred about the value of the program:

I think the program itself is excellent. I mean, it would be nearly. . impossible if the program wasn't made available. It's something that they have where they were having nothing in the years past. (Y2P5)

This person explained that, although her own children were too young to be involved in the program, she had heard good reports of its operations from other parents:

I do know that from my understanding of materials that get sent to them [the teachers], they review it, they grade it and send it back for them to review. They keep them up with their lessons, and I think that's the most important aspect of it. I mean, **f** you didn't have that you couldn't do it.

So it's vital. . . . A lot of the interaction that goes on is between teacher and mother, not teacher and child. So it's good when the teachers come out to see the kids. The kids get a chance to then interact with the actual teachers themselves. (Y2P5)

This statement again reflected the close monitoring to which the show people subjected the program; this parent revealed a detailed worlung knowledge of the program even though she was not involved in it. This suggested not only the considerable degree of dialogue among parents and between parents and teachers about the program's consumption, but also how central that consumption was to the show people's ongoing resistance of their previous educational marginalisation.

Another parent was very enthusiastic about the program: "...it's very successful for us. We love what we've got" (Y2P1). She added: "With this, it's presented so easy. Everything comes over clear and easy. .." (Y2P1). She summed up her overall approval of the program's operation:

Our kids love school. They love school. The teachers are fantastic. If you give them work, it's not like in a classroom, our kids do it all. They might be a little bit backward. ... In a classroom, there's always the kids up the back that you never know, and they get to sixth grade and they still can't read and write. Our kids, when they're given proper first grade work, they don't not learn. We've had a couple [of children] who went onto boarding school, and they are high in their grades. You've got to learn from this; it has to work. We do all right. We're coping. (Y2P1)

Another parent was equally pleased with the outcomes of the program to date. She noted that the "teachersare very good. I mean, everyone's happy with the teachers" (Y3P1). Furthermore, she was particularly enthusiastic about another innovation associated with the program:

...we now have a school camp because of this program. Our kids actually go to school camp, whereas they didn't before. They've done it for the last one, two, three years now. ... It gets them all together away from school, awayfrom home. (Y3P1)

These disparate comments by parents shared a conscious and publicly stated determination to consume the program by subjecting it to ongoing scrutiny and critique, in order to consolidate the educational gains that they had made on behalf of their children.

The show children, the most direct 'tactical consumers' of the education program, were also complimentary about its implementation. They had established strong rapport with the itinerant teachers (whom they addressed by their Christian names, and who in some cases were regarded as members of the extended family), and they largely enjoyed the program. One boy claimed that the show children "havea lot of fun" (Y1C1) with the itinerant teachers. A ten year old girl said that she preferred going to school in Queensland rather than New South Wales because of the program's operation in Queensland and the company of her friends from the show circuit (Y1C8). When asked from whom they seek help with difficult work, students most often identified the Brisbane teachers; otherwise a parent, a grandparent or an older sibling might be called on for assistance.

Children recognised that some subjects were more relevant to the show children's distinctive lifestyle than others. One eight year old boy explained his preference for studying mathematics – which was echoed by many of his contemporaries thus: "Becausel like working at the sums, and you gets lots of [games] in mathematics" (YIC6). This was verified by one of the home tutors, who commented that the show children handled money from an early age as part of the work expected of them within the family business (Y1HT1). The same eight year old boy explained the different phases' of learning involved in the program: content work in the papers when the teachers were in Brisbane, and interactive activities with the teachers when they visited the children along the show circuits:

When we come to a school, . . we don't do much with **our** work. We get activities done on the theme we're on and then we do something, and might just do an afternoon, like a bit **of** maths, a bit of language arts and a bit **of** social studies at the end **of** the day when we come into school. (Y 1C6)

One ten year old girl found the work easy to understand: "...once you know all your stuff, for example, it's real easy to go along, once you know what you're doing" (Y1C4). A twelve year old girl concurred, saying about the program: "...because it's better. It's easier" (Y2C4). A seven year old boy identified the Brisbane School of Distance Education teachers as "our real teachers. ..[who] never get grouchy – never get grumpy like those old teachers" in local schools (Y1C7).

My argument about these laudatory comments about the program by show parents and children is that they reflected the show people's conscious and deliberate consumption of the program that they had lobbied to establish through a close and careful monitoring of its implementation. This underscores the crucial observation that consumption from this perspective is a continuing process rather than an end point. Furthermore, the positive remarks about the program demonstrated the show people's determined resistance of previous assumptions about educating itinerant people that had provided them with inadequate options for their children's schooling. This resistance was evident in the show people's careful identification of the program's positive features, all of which are related to a greater alignment between an itinerant lifestyle and maximum educational opportunities. Whether the comments referred to the quality of the distance education papers, the children's liking for this kind of schooling or the empathetic understandings of the teachers, they encapsulated the show people's active involvement in ensuring the retention and refinement of a program that they had worked hard to establish.

The significance of this argument for relating the show people's consumption of the education program to the study's conceptual framework cannot be overestimated. De Certeau (1984) referred to

. • the models of action characteristic of users whose status as the dominated element in society (a status that does not mean that they are either passive **or** docile) is concealed by the euphemistic term 'consumer' $r \sim (pp. xi-xii)$

The preceding analysis has demonstrated that the show people are indeed far from being "eitherpassive or docile" in consuming the program: on the contrary, they have been actively involved in refining it to suit their distinctive needs and circumstances. Moreover, the analysis has highlighted the validity of de Certeau's (1984) bald assertion: "The tactics of consumption, the ingenious ways in which the weak make use of the strong, thus lend a

political dimension to everyday practices" (p. xvii). The show people's consumption of the problem certainly has "apolitical dimension", largely because resistance is an avowedly political practice.

Finally, the preceding analysis has also confirmed the appropriateness of de Certeau's (1984) identification of

...forces that are the starting point for an intellectual creativity as persistent as it is subtle, tireless, ready for every opportunity, scattered over the terrain of the dominant order and foreign to the rules laid down and imposed by a rationality founded on established rights and property. (p. 38)

These forces, of which the show people's resistant lobbying for and consumption of the education program are a strilung example, point the way forward to creating a new "terrain" by virtue of their "intellectualcreativity". That same "intellectual creativity" enables the resistance and subversion of the "rationality" that underpins the construction of "the dominant order". That "rationality" in turn articulates with de Certeau's (1984) reference to "the power of knowledge. ..[being] to provide oneself with one's ownplace" (p. 36). So "rationality" is actually complicit with forms of seemingly unproblematic knowledge – in this case about the 'place' of schooling and the 'place' of itinerant people – and is therefore suffused with power. This power is, as far as possible, counteracted by the show people's resistance of those forms of seemingly unproblematic knowledge, as reflected in their lobbying for and consumption of the Brisbane School of Distance education program for their children.

6.6 Review of the chapter

I argued in the previous chapter that the show people's experiences of marginalisation, which denied them access to institutional or strategic power as a direct consequence of their itinerancy, were usefully summarised by de Certeau's (1984) identification of three key attributes of strategies of marginalisation:

- the show people's absence of place
- the construction of their otherness
- the imposition of forms of seemingly unproblematic knowledge about how they should be educated.

The purpose of this chapter has been to address the study's second research question: "How do the show people resist their marginalised status?". The posing of this question reflected an assumption that, far from being 'passive victims' of their marginalisation, the show people exhibit enormous agency and ingenuity in subverting that status. Responding to the research question required the application of de Certeau's (1984) concept of 'tactics of consumption', aided by his notion of 'space'. His synthesis of the overlap between these ideas is a timely reminder of their emphasis on agency and ingenuity and of their avowedly politicised dimension:

Tactics are procedures that gain validity in relation to the pertinence they lend to time—to the circumstances which the precise instant of an intervention transforms into a favorable situation, to the rapidity of the movements that change the organization of a space, to the relations among successive moments in an action, to the possible intersections of durations and heterogeneous rhythms, etc. ...[T]actics are a clever

utilization of time, of the opportunities it presents and also of the play that it introduces into the foundations **apower**. (pp. 38-39)

Specifically in relation to the three attributes of strategies of marginalisation identified above, I have made the following key points about the show people's tactics of consumption and their use of space:

- They resist their ascribed absence of place through their multiple experiences and understandings of 'home', which enable them to lay claim to a place of their own and hence to a location of power.
- They resist their constructed otherness by giving the term 'showies' a positive valence to counteract the negative stereotypes ascribed to it by others, disrupting the 'showie'-'non-showie' dichotomy and giving the term 'mugs' a negative valence as a tactic of reversal.
- They resist forms of seemingly unproblematic knowledge about their schooling through their lobbying for, and consumption of, a specialised educational program operated by the Brisbane School of Distance Education, thereby countering the six inadequate and marginalising educational options previously available to them.

It is timely to reiterate that there is nothing inevitable or certain about the effectiveness of the show people's resistance of their marginalised status; their tactics of consumption have had to be deployed across a large range of sites and in a considerable number of contexts. What has unified those disparate endeavours has been their recognition that education holds the key to their efforts to establish a counternarrative to their marginalisation. The show people's resistant practices are therefore as integral as the strategies of marginalisation identified in the previous chapter to their 'learning on the run'.

CHAPTER SEVEN

RESEARCH QUESTION THREE:

THE SHOW PEOPLE'S
TRANSFORMATION OF
THEIR MARGINALISING
EXPERIENCES AND
RESISTANT PRACTICES

"...I teach primary [show] kids...I can take into it my experiences from outside. I can open up their imaginations, I suppose, to things...on a wider scale, which I know is what the parents want."

Y3HT1

"...who knows what the potential is of this program? It's like the potential of all children."

Y4T1

7.1 Overview of the chapter

In Chapter Five of this thesis, I argued that the show people's itinerant lifestyle subjects them to what de Certeau (1984) identified as three key attributes of strategies of marginalisation. These three elements related specifically to the show people's absence of place, the construction of their otherness and forms of seemingly unproblematic knowledge about where and how itinerant people should receive their schooling. In Chapter Six, I asserted that the show people engage in direct and effective resistance of those marginalising strategies, as evidenced through their multiple experiences and understandings of 'home', their own uses of the terms 'showies', 'locals' and 'mugs', and their lobbying for and consumption of a specialised educational program under the auspices of the Brisbane School of Distance Education. These resistant practices were identified as exhibiting de Certeau's (1984) concept of 'tactics of consumption', and as reflecting the show people's determination to turn the alienating 'space' of conventional schooling into a welcoming and enabling 'place' of Traveller education.

In this chapter, I provide answers to the third research question guiding the thesis: "How do the show people transform their marginalising experiences and resistant practices?". In doing so, I demonstrate how the show people's deployment of various 'tactics of consumption' is a prelude to their efforts to change 'the rules of the game' that construct them as marginalised and 'other' to 'mainstream' Australia. That is, my reading of the show people's words and actions focusses on how their politically informed subversion of those 'rules' make alternative and more positive frameworks of Traveller education possible as a counternarrative. It is through these processes of change and transformation that the show people are able to

change the marginalised 'spaces' of itinerancy into valued 'places' of their own. It is important to acknowledge at the outset that these processes occur on a multitude of fronts. For the purposes of this study, however, I have concentrated on the ways in which the various dimensions of the Traveller education program can be read as illustrating resistance and transformation.

The chapter consists of four sections:

- outsidedness, creative understanding and transformation
- transforming absence of place: a place of their own
- transforming constructions of otherness: valuing difference
- transforming forms of unproblematic knowledge: a separate show school.

In the first section, I draw on Bakhtin's (1986a) ideas of outsidedness and creative understanding to provide a conceptual bridge from de Certeau's (1984) focus on marginalisation and resistance to transformation. In the second and third sections, I use the words of the home tutors and the Brisbane School of Distance Education teachers to demonstrate the transformation of the show people's previous absence of place and constructed otherness. Here the argument is that the other groups' growing awareness of the show people's struggles to assert their own notions of 'home' and their positive views of their identities are a testament to the potential for productive change in previously devaluing situations. In the fourth section, I use the show people's words about their idealised hopes for Traveller education to argue that the newly established separate school for show children represents a powerful transformation of previous marginalising and alienating assumptions about the education of itinerant people. In combination, these points illustrate the capacity of the show people, a supposedly disempowered and marginalised group, to work in concert with representatives of the 'mainstream' society

whom they enlist to their 'cause' in order to circulate and replicate counternarratives that are more understanding and valuing of the show people's lifestyle and culture.

7.2 Outsidedness, creative understanding and transformation

Resistant 'tactics of consumption' are a necessary but by no means a sufficient prerequisite for more fundamental and permanent change occurring to the strategies of marginalisation faced by the Queensland show people. In particular, I perceive the transformation of their marginalising experiences and resistant practices as being directly and inextricably linked with their capacity to enlist the understanding and support of other people. As long as the 'battle lines' are drawn between 'them' and 'us', show people are inevitably at a recurring political disadvantage. However, if others take up the battle on their behalf, and work with them to challenge stereotypes and subvert alienating policies, 'the rules of the games' can be altered and a new terrain can be mapped out in which show people can potentially obtain a 'place' of their own.

Theoretically, my concern with transformation begins with de Certeau (1984). I indicated in Chapter Three my dissension from the consensus view that his distinctions between 'strategies of marginalisation' and 'tactics of consumption' and between 'place' and 'space' constitute a static model of social life. On the contrary, I argued that his conceptual framework is dynamic and mobile and fully aware of the political interstices into which subversive tactics can be inserted in order to generate productive change.

It is helpful to be reminded of de Certeau's explicit acknowledgment of the role of transformation in his social theory. **As** I noted in Chapter Three, de Certeau (1984) stated:

It is as though the opportunity for a sociopolitical renewal of Western societies were emerging along its fringes, precisely where it has been the most oppressive. Out of what Western societies have held in contempt, combated and believed they had subjugated, there are arising political alternatives and social models which represent, perhaps, the only hope for reversing the massive acceleration and reproduction of totalitarian, homogenising effects generated by the power structures and technology of the West. (p. 231)

Here is a direct explication of how de Certeau envisaged 'the rules of the games' being changed, centred on the "political alternatives and social models" that have been "emerging along. . [the] fringes" of "Western societies". I regard the show people's actions as a striking example of how groups on the "fringes" of 'mainstream' communities, which the same 'mainstream' communities "have held in contempt, combated and believed they had subjugated", are able to counteract the "totalitarian, homogenizing effects" of those 'mainstream communities'.

Further evidence of de Certeau's (1984) recognition of the displacement potential of 'tactics of consumption' can be found in his use of the term 'transform' in his elaboration of his social theory. For example, he noted:

There is no longer an elsewhere. Because **t** this, the "strategic" model **is** also **transformed**, as if defeated by its own success: it was by definition based on the definition **d** a "proper" distinct **from** everything

else; but now that "proper" has become the whole. (p. 40; emphasis added)

He referred to the purpose of his own inquiry as being "totransform what was represented as a matrix-force of history into a mobile infinity of tactics" (p. 41; emphasis added). He also noted the transformative tendencies of 'space', as when "the street geometrically defined by urban planning is transformed into a space by walkers" (p. 117; emphasis in original).

In this chapter, I analyse interview statements by show people, home tutors and Brisbane School of Distance Education teachers for evidence of the ways in which tactics of consumption become transformative. In doing so, I emphasise what I consider the most significant aspect of that potential: the opportunity to construct a counternarrative about who show people are and how they should receive an education. What I have in mind here is what was conveyed by de Certeau's (1984) reference to "...the tactical trajectories which, according to their own criteria, selectfragments takenfrom the vast ensembles of production in order to compose new stories with them" (p. 35).

It remains the case, however, that de Certeau devoted more attention to the resistance than to the transformation of marginalisation. Consequently he was less suggestive about how transformative actions might be identified and analysed than about the nature of tactics of consumption. Accordingly I have turned to Bakhtin (1986a), and particularly his concepts of outsidedness and creative understanding, to provide a more detailed conceptual framework for understanding how the show people's resistant practices can be transformed into enduring displacements of their marginalising experiences.

As I indicated in Chapters Three and Four, Bakhtin (1986a) provided a helpful overview of the links between outsidedness and creative understanding:

Creative understanding does not renounce itself, its own place in time, its own culture; and it forgets nothing. In order to understand, it is immensely important for the person who understands to be located outside the object of his or her creative understanding—intime, in space, in culture. For one cannot even really see one's own exterior and comprehend it as a whole, and no mirrors or photographs can help; our real exterior can be seen and understood only by other people, because they are located outside us in space and because they are others. (p. 7; emphasis in original)

I have taken Bakhtin's precept of the importance "for the person who understands to be located outside the object of his or her creative understanding—in time, in space, in culture" as a rationale for the organisation of the next two sections of this chapter. As I indicated at the outset, those sections analyse the comments by home tutors and teachers about the show people's supposed absence of place and the constructions of their otherness. This is a crucial element of my understanding of transformation: the show people can and do talk about how they resist those marginalising strategies, but it is necessary for 'outsiders' to do the same if real change in the show people's situation is to occur. (I see a parallel here with debates about Indigenous Australian 'reconciliation': while meanings attached to this term vary widely, the underlying concept is predicated on the active involvement of Indigenous and non-Indigenous Australians alike.) The final major section of the chapter then argues that that evidence of the home tutors' and teachers' creative—and

transformed – understanding of the show people's possession of a place of their own and their difference rather than otherness contributes directly to the show people's transformation of forms of seemingly unproblematic knowledge about schooling into their successful establishment of a separate school for show children.

This approach to analysing transformation resonates with Rowan and Brennan's (1998) overview of some of the crucial elements of fundamental and enduring shifts in cultural life. Firstly, they argued that:

The work of cultural transformation is thus not so much an intended 'rationalproject' in line with some pre-set view of 'progress' as a set of partial and reflexive interventions into constantly changing and reproducing institutions and subjectivities. (p. 5)

This chapter is very much *my* reading of ongoing alterations in the show people's marginalised status, a reading that derives from the particular combination of conceptual resources that I have mobilised in this thesis. Similarly, as I indicated in Chapter Six, I eschew a teleological view of social change in favour of one that constructs the gains made by show people as tentative and needing to be constantly reviewed in order to guard against slippage back into the previous, marginalising situation.

Secondly, and conversely, I concur with Rowan and Brennan's (1998) observation that "Theacknowledgment that marginality is produced rather than permanently and naturally inscribed draws attention to the potential for its resultant material consequences to be displaced..." (p. 9). Herein lie the explanation and the justification for the argument presented in this chapter: through resistance and transformation, the show people are able to use against the strategies of marginalisation the same tools deployed by those strategies to

marginalise them. That is, the show people succeed in changing 'the rules of the games' by arguing that those rules are neither 'natural' nor 'neutral', the same charges levelled against the show people in the process of constructing them as marginalised.

Thirdly, Rowan and Brennan (1998) referred to "counternarratives...[as] stories of cultural change" (p. 10). This chapter demonstrates the effectiveness of the show people's counternarratives in presenting alternative and resistant constructions of their own 'place' and their difference rather than otherness as the basis of, and the justification for, different and more positive forms of Traveller education. Indeed, this chapter and the study of which it forms a part seek to contribute to the creation of those counternarratives about the show people's lives and learning.

7.3 Transforming absence of place: A place of their own

As I indicated above, my task in this section of the chapter is to demonstrate the show people's transformation of their ascribed absence of place into a place of their own through the words and understandings of the home tutors and the Brisbane School of Distance Education teachers. That transformation relies partly on the home tutors and teachers acknowledging the show people's 'right' to pass onto 'their terrain' in their role as representatives of 'the centre' by virtue of their own, more 'normal' residence in fixed locations.

As a consequence of the show people's multiple experiences and understandings of home being communicated to these 'outsiders' to the show circuits, the home tutors and teachers are enlisted to spread a counternarrative

about the show people's forms of residence and group relationships that disrupt the stereotype of no fixed location – and hence of no political 'place'. So the spaces of itinerancy become 'a place of their own' in the hearts and minds of these others, who in turn report those new understandings to those whom they encounter. These shifts in awareness can be likened to moves from outsidedness and creative understanding (Bakhtin, 1986a) that I posited above as the basis of transformation in the context of this study.

Turning first to the home tutors, their 'outsider' status enabled them potentially to combine relatively detached comments about their students' and employers' ideas about 'home' with their own references to this concept (a point that is implicit in the habitual use of 'home' in front of the identification of their role as 'tutors'). One home tutor stated that, at the end of her employment as a home tutor, "I went home, to Mackay. . ." (Y5HT1). Another home tutor explained how she "wenthome" after an initial period of working on a show circuit: "Thatwas in Maryborough, because that's where I come from" (Y1HT1). For a home tutor, this statement was an unusually explicit association between 'home' and "comingfrom" or 'belonging to' somewhere. This statement differed from the show people's initial connection between 'home' and their caravans, but paralleled those same people's references in particular contexts to specific towns as being 'home'.

The home tutors reinforced the show people's correlation between 'home' and the caravans in which they lived. One home tutor stated that "I've been able to spend some time with the kids at home, in their little van things" (Y3HT1). Another home tutor, in the context of explaining how for the show children "Money'sjust an everyday issue. ..", added a reference to the

caravan as home: "...they'll sort of get home and see their parents counting out the change for the next day..." (Y1HT1).

In many ways the 'outsider' home tutors found life on the show circuits exciting and intriguing. In other respects it was stressful for them; they were away from their own families and social networks, and sometimes they found it difficult to become accepted as authentic 'teachers' by their young charges (and occasionally by their employers, the parents). Perhaps some of this tension underlay a home tutor's reference to "anywherethey call home", and his strong opinion (unsubstantiated in the interview) that "if some of the kids were honest they would probably say it would be really nice to live somewhere and not just shift all the time" (Y3HT1). On the other hand, the same home tutor commented perceptively about the centrality of the show circuits in the children's distinctive lifestyle: ". . . their routine is to be at a show – that's their home – to pull up everything – business, home – and shift. . .that's the routine" (Y3HT1). This statement by a home tutor reflected a presumably unconscious incorporation of show people's constructions of the "routine" shifts of the show circuits as a defining characteristic that separates them from other people.

I interpret the ambivalence underlying these home tutors' comments about the show people's conceptions of 'home' as reflecting a crucial and in some ways ongoing struggle between outsidedness and creative understanding. The home tutors' difficulties with constantly moving 'home' are a symptom of the resilience of the marginalising stereotypes about show people's absence of place analysed in Chapter Five. The deleterious effects of generations of lack of understanding and respect do not disappear overnight. On the other hand, the home tutors' comments about 'home' also reflect a

growing comprehension of the centrality of the itinerant lifestyle to the show people's *raison d'être*, and hence of the validity and vitality of that lifestyle. For me, this growing comprehension recalls Bakhtin's (1986b) distinction between creative understanding and empathy and his associated insistence on the need for outsidedness if genuine understanding were to result:

One cannot understand understanding as emotional empathy, as the placement of the self in the other's position (loss of one's own position). This is required only for peripheral aspects of understanding. One [also] cannot understand understanding as a translation from the other's language into one's own. (p. 141)

At least some of the home tutors' perceptions of 'home' cited above were shared by the Brisbane School of Distance Education teachers. After all, they shared with the home tutors — although for shorter periods of time — many of the difficulties attendant on leaving their families and living out of suitcases. One of the teachers, in his first year of involvement with the show people, used his account of how his colleagues and he worked with the show children at local schools to equate their 'home' with their caravan: "We often look at a project for the week. ...so we can finish the documents at the end of the week that they take home. .." (Y2T1). However, his perception of certain limitations in the show children's schooling prompted his reference to a different location of 'home':

Sometimes I'd love to be able to take them home, and bring them into a school for an extended period of time, and really get them up to speed because there's a couple of little gaps, and you stumble across the gaps sometimes. (Y2T1)

For this teacher, 'home' represented the predictable routines of living in one place, a situation in which the distractions of constantly moving gave way to an opportunity to work intensively on perceived learning "gaps".

By contrast, a teacher who had been involved with the program from the outset argued strongly that "part of the success" of that program for the show children

...is the development **a** understandings about the working lifestyle, and the huge time constraints and commitments by our families in maintaining their family, and their working lifestyle, their profession. (Y4T1)

The reference to "ourfamilies" reflected the extent to which this teacher in particular had come to identify personally as well as professionally with the distinctive needs of the show people. This identification helped to underpin her construction of the show people as actively contesting and transforming marginalisation into a way of life that is positive, productive and agential. This sense of the transformation of marginalisation was encapsulated in the speaker's reference to "theirworking lifestyle, their profession", a form of words that placed the show people on the same plane as teachers, whose "profession" remains valuable and vital for all that it is currently subjected to media and political belittling. Implicit in that reference was a recognition of both the difference and the validity of the show people's multiple experiences and understandings of home.

I have sought in this section of the chapter to trace some of the elements of the transformation in others' understandings of the show people's lifestyle and values. That transformation helps to facilitate the show people's success in moving from an absence of place to a place of their own. In doing so, I argue that the home tutors' and teachers' growing comprehension of the

distinctive meanings that show people attach to 'home' articulates with the show people's resistance of the fact, noted in Chapter Five, that "The 'proper' is a triumph of place over time", and that "Ztis a mastery of time through the foundation of an autonomous place" (de Certeau, 1984, p. 36). That resistance, as I observed in Chapter Six, resonates with de Certeau's (1984) insight that "...the political relevance of the geo-graphical distinctions between separate places is echoed. ..in the distribution of places of power. ..." (p. 227). The taking up of that resistance by the home tutors and teachers—albeit tentatively and inconsistently in some cases—helps to turn the disruption of resistance into the displacement of transformation.

I contend that this shift reflects the interplay between Bakhtinian outsidedness and creative understanding. Specifically, I assert that the comments about the show people's meanings of 'home' by outsiders who had been with them for a longer period provided evidence for Bakhtin's later identification of creative understanding as a four stage process, as reported by Morson and Emerson (1990):

...first, the physical perception, then its recognition, then a grasping **d** its significance in context, and finally-and this is the crucial step—"active-dialogic understanding." This fourth step is more than an acknowledgment **d** existing context; it **is** implicitly creative, and presumes evernew, and surprisingly new, contexts. (p. **99**)

For me, a vital dimension of those "ever-new, and surprisingly new, contexts" is the implication that the home tutors' and teachers' increased awareness of the show people's multiple experiences and understandings of home is a crucial element of their transformation from an absence of place to having a place of their own.

7.4 Transforming constructions of otherness: Valuing difference

In Chapter Five, I noted de Certeau's (1984) identification of constructions of otherness as one of the key attributes of strategies of marginalisation of particular social groups. I traced the 'progressive' – in the sense of sequential rather than 'enlightened' – development of those constructions in relation to show people through 'locals" perceptions of show people's 'strangeness' to regarding them as objects of pity to treating them with animosity and hostility to replicating negative stereotypes that they are 'dirty' and likely to be thieves to subliminal associations between show people and Gypsies. I argued that the ever increasing antipathy exhibited by 'locals' towards show people was tied directly, and led inexorably, to their construction as 'other' to the norms and values of the 'mainstream' community and hence as outside that community's protection and unable to gain equitable access to its resources and services.

In Chapter Six, I demonstrated that the show people actively and effectively resist the marginalising constructions of their otherness. They do this through particular 'tactics of consumption': giving the term 'showies' a positive valence; disrupting the 'showie'-'non-showie' dichotomy by emphasising how show people regularly move on and off the show circuits; and giving a negative valence to the term 'mugs', and thereby engaging in some reversal of terminology (in a similar way to other marginalised groups deploying terms such as 'black', 'gay' and 'queer' against their oppressors). This tactic works to turn the 'gaze of surveillance' back onto 'the centre' and allow the show people to present their lifestyle **as** valid and valuable.

My purpose in this section of this chapter is to use the home tutors' and teachers' words and understandings to gauge the degree of transformation in constructions of show people's identities, by moving from a focus on their otherness to an emphasis on valuing their difference. As I have insisted throughout this chapter, 'the rules of the game' that construct show people as 'other' can be changed, and a new terrain for valuing their difference can be mapped, only if they are demonstrably successful in inserting and circulating counternarratives about who they are and want to become. A crucial element of this transformation is the extent of their capacity to displace traditional and marginalising stereotypes about themselves by demonstrating that those stereotypes, far from being 'natural' or 'neutral', are in fact political constructions with tremendously deleterious effects. My aim is to illustrate how dominant, generally negative, attitudes towards the show people can be, and have been, transformed via an exploration of the changing views and growing understandings of the home tutors and teachers.

From that perspective, I turn first to the home tutors' statements about how they perceived the show people. One home tutor noted:

But I found most of them to be really good. Some **t** them aren't, some of them are. It just. . some **t** them will be wonderful to you, and other people won't be that nice to you. (Y1HT1)

This home tutor endorsed the comment by a parent that the show children's lifestyle sometimes approaches "afantasy land" for them (Y1P2). She asserted about the show children: "I think they have a ball, just play [ing] with each other. See, as they get older there's different rides that they're allowed onto. ." (Y1HT1) She went further and stated, "I think a lot at them get a bit spoilt. Not spoilt rotten, but they are used to having money. .." (Y1HT1).

As in the previous section of this chapter, I detect considerable ambivalence in some of the 'outsiders' perceptions of show people. As I noted earlier, it takes a great deal of time and persistence for ingrained assumptions about the show people's identities – the constructions of their otherness identified in Chapter Five and resisted in Chapter Six – to be dispelled and replaced with more positive counternarratives. This home tutor's remarks encapsulated much of that ambivalence, which displayed outsidedness more than creative understanding in Bakhtin's (1986a) terms. Nevertheless there were sparks of valuing the show people's difference evident in comments such as "...some of them will be wonderful to you" and "Ithink they have a ball, just play[ing] with each other".

Another home tutor, who positioned himself explicitly as expressing "an outsider's point of view" (Y3HT1), identified with local people's desire to attend the annual show.

...people seem to be willing to go to the show because it was an event that they remember as kids and wanted to perpetuate it with their own kids. Kids want to go because in a town like this perhaps nothing much happens. (Y3HT1)

This home tutor's distancing of himself from the show people prompted him to speculate on their reaction to him as a home tutor who was also competent at driving a truck.

I think they're wary atfirst. ..perhaps because I'm older – that affords you **some** sort of respect anyway, because they know that you've got qualifications and they know that you 're experienced and [don't] speak like a lot of the people they mix with speak. ..(Y3HT1)

Superficially these comments indicate the speaker's explicit identification with 'locals' and a consequent, even if implicit, dissociation from show people. Yet I analyse his statements in a different vein: rather than replicating the 'showie'-'local' divide, I perceive his remarks as exhibiting both outsidedness and creative understanding. The explicit self-positioning as expressing "anoutsider's point of view" implied two elements: a recognition of the fundamental differences between 'showies' and 'locals'; and an implicit suggestion that his detachment enabled him to observe positive aspects of the show people's lifestyle. This was evident in his identification of reasons why rural parents take their children to the show: that identification indicated that the show people perform a valuable service by providing entertainment "in a town like this". Similarly, the ascribed recognition by show people "thatyou've got qualifications and. ..[don't] speak like a lot of the people they mix with speak. .. " suggested his reciprocal acceptance that many show people also have "qualifications" and speak 'properly' otherwise they would be unable to perceive those 'virtues' in him.

I contend that this reverse recognition exemplifies Bakhtin's (1986a) notions of outsidedness and creative understanding. **As** Morson and Emerson (1990) related Bakhtin's concepts to different languages:

To realize and develop the potential **a** a language, "outsidedness"—the outsidedness **a** another language—is required. That outsidedness may lead to an exchange in which each language reveals to the other what it did not know about itself, and in which new insights are produced that neither wholly contained before. (p. **3**10)

This home tutor's articulation of "anoutsider's point of view" enabled him to express "new insights" into the show people's contributions to local communities and their levels of cultural capital. In doing so, his statements contributed to the critique of the show people's constructed otherness and to the transformation of that otherness into a valued difference.

Another home tutor emphasised what she saw as the fixed character of the boundary between 'showies' and 'locals'. She stated, "Everyone'sthat's outside the show's an outsider" (Y5HT1). Later she reinforced this point when she described people being "on the outside" of the show circuits (Y5HT1). She added, ". . .[there is] nothing like an outside influence that brings them all together. .." (Y5HT1), implying the existence of heterogeneous groups on the show circuits. Given this supposedly fixed division between 'us' and 'them', the home tutor considered that show life provided a safe and secure environment for the expression of group and individual identity. Show life "is their whole world", so that, despite inevitable interpersonal and inter-group tensions, many show people feel comfortable in their relationships on the show circuits, "withouthaving to compete with the rest of the otherpeople in society" (Y5HT1).

In contrast to the previous home tutor cited above, who positioned himself as articulating "anoutsider's point of view" (Y3HT1), this home tutor was at pains to identify herself as being associated in important respects (although not in every respect) with the 'showies'. An indication of her acceptance by the 'showies' was when her employer "showedme where the secret gate [into the showgrounds] was to get in so I didn't have to pay" (Y5HT1). Here the avoidance of paying entrance fees was a badge of identification with show people that the speaker prized highly, indicating in

doing so the considerable cultural capital attached to this relatively modest piece of financial capital. On the other hand, the speaker was aware of the unspoken but understood boundaries around her acceptance as a 'showie': "I'mnot a showman, I'm still an outsider" (Y5HT1). She also stated, "I'm just a person who just lives with the showmen. ..." (a comment that also implied that the show people do have a definite 'home'), rather than describing herself as one of those "showmen" (Y5HT1). These fine distinctions reveal the complexity and dynamism of the 'showie'—'local' divide, despite this speaker's own suggestions that this divide was uncomplicated and unchangeable. This revelation suggested the home tutor's implicit acceptance of the show people's tactic—analysed at length in Chapter Six—of highlighting that complexity and dynamism in order to resist the negative stereotype of 'once a show person, always a show person'.

As I indicated in Chapter Five, a recurring theme in this home tutor's discourse was the assertion that show people have traditionally been poorly treated by the wider society, and that that poor treatment has caused them to retreat into a separate existence on the show circuits.

...no other outsiders would be welcome. ..inside [the show circuits]. They would not. Well,...the world wasn't interested in them,...the communities...didn't know them, all they thought of them was —scum, really. (Y5HT1)

Shortly after this statement, the home tutor reinforced this view.

Everything that the showmen are is a product of the way we treat them — the way their pride is, the way they work, the way they are — is a product of years and years, centuries of our treatment of them. And that is the basic line. (Y5HT1)

Later in the interview, the speaker elaborated this view.

Like I said, the way they are, and the way they act, it's years and years of our treatment of them. That's the basic line. .. because of the way we. . .. have treated them. Their own respect has come from the way they put us down,. .. the way the people outside have put them down. (Y5HT1)

For this reason, the speaker claimed, the 'showies' generally confine their interest in the world to the events of show life: "...why should they go out of their way. ...for the outer community, when the outer community have never treated them anything but. ..[poorly]?"(Y5HT1).

From one perspective, these comments could be considered patronising, with the show people's identity being constructed purely in terms defined by the broader society. From a different viewpoint, however, this was the strongest statement by a home tutor championing the show people as an oppressed and marginalised group. Furthermore, this was consequently the most direct evidence among my interviews with home tutors of the shift from outsidedness to creative understanding. This was a striking example of Morson and Emerson's (1990) observation that, within Bakhtin's (1986a) theoretical framework, people "cantake maximal advantage of the differences and **a**t their outsidedness by an act of creative understanding that is truly dialogic in the best sense" (p. 429). From that perspective, this home tutor's clear articulation of the show people's traditional marginalisation, and her use of the interview with me to posit an alternative view of who they are and want to be, were indeed part of "an act of creative understanding" based on valuing the show people's "differences" and thereby displacing the constructions of their otherness.

A similar process was evident in the same home tutor's reference to 'mugs'. She asserted strongly, "Ohno, not me, I wasn't a mug" (Y5HT1) in a part of the interview where I was seeking to delineate terms that show people used to separate themselves from other people. Yet this tutor also indicated that, although she had travelled with the 'showies' for almost two years, for most of that time she dwelt on the margins of identity, not accepted by show people as a complete 'showie' and yet perceived by 'locals' who attended the shows as being associated with the show circuits. In that context, the speaker's rejection of the label 'mug' as having applied to her at any time reinforces the proposition that it is a more derogatory and specialised term than 'local'. This proposition was supported by the home tutor's subsequent reference to 'mug' in connection with 'rort', as a description of a 'rigged' game: "I mean, how are you going to knock these down? That's a joke!" (Y5HT1). So 'mug' in this sense has a more restricted meaning and use than the more generalised term 'local' - a point that I made in Chapter Six in connection with the show people's use of the term as a reversal tactic to counteract the derogatory terms routinely levelled at them. This home tutor's implicit acceptance of the function of that tactic was evident in her recognition of its negative valence.

Turning the focus from home tutors to the teachers from the Brisbane School of Distance Education, one teacher's admiration for the show children's courage in the face of adversity enlivened his account of a particular action by show children that spanned age differences:

But they 'realso very protective of their own. A couple of the bigger boys, I've got a lot of time for. And they're all very close, and they'll see the younger blokes, and they're not necessarily related to them, little boys in

Grade One, and they'll go and get a drink and the big boys can see some of the locals, they call them 'locals', giving them a bit of stick. And they'll quietly walk over and let them know, not in an unfriendly way, that they're around and like to be seen near their friends. And I'm always very proud of them when they do that because they just handle it. (Y2T1)

The teacher reflected on his own response to a potentially stressful situation that appeared to occur in nearly all the local schools attended by the show children:

Ifeel sorry for them that it would be great f they could just come into a school, enjoy the facilities without being hassled. . . . Sometimes the [show] kids go out and play footy with the [local] kids, and they get involved, but on the whole they tend to stick to themselves. (Y2T1)

This teacher had not been involved with teaching the show children for very long, and he identified elements of their lifestyle that he considered educationally marginalising. In that context, it was suggestive that he complimented the social maturity of particular show children in their interactions with local children. **As** a relative newcomer to the show children's education program, he tended to take at face value show people's constructions of the division between 'showies' and 'locals' as fixed and permanent, whereas teachers with a longer involvement with the show people qualified those constructions with references to specific contexts and individual people.

On the other hand, this teacher acknowledged instances of friendly interaction between show children and local children. He cited **as** evidence the particular approach to the 'buddy system' that was in operation at a school on the western Queensland show circuit.

...we went into a classroom and we were invited to work in the classroom with our kids and we were made to feel part **a** that class. The kids in that class adopted our kids, and they were always coming up to us and saying, "Wouldyou like to play with us?", and being very kind to the kids. That was lovely. (Y2T1)

The selected statements by this teacher encapsulate my argument in this section of the chapter. The teacher's outsidedness was revealed in such comments as "Ifeel sorry for them" which, taken alone, could shade into the marginalising constructions of the show people's otherness identified in Chapter Five. In the context of the teacher's educational discourse, however, his reference to the older show children's maturity in dealing with a potentially hostile situation, and his pleased reaction to instances of show children and local children working and playing together amicably, reflected his growing creative understanding of the pressures on show people and their productive engagements with those pressures.

The home tutor's recognition that show people have to work at disrupting their marginalising and constructed otherness in turn recalled for me Morson and Emerson's (1990) account of Bakhtin's insight – reflected in the works of the German writer Goethe – that

...a real sense of creativity would have to involve human work. . It must involve work growing out of concrete needs but producing something that is also genuinely new and not exhaustively specified by the past. (p.

415)

This is the real and twofold significance of the home tutor's comments about the show children for my argument: the move from outsidedness to creative understanding "must involve work" and a genuine commitment to mutual comprehension on 'both sides'; and such a move can indeed be successful in "producing something that is also genuinely new and not exhaustively specified by the past". Transformative counternarratives that value the show people's difference rather than replicate constructions of their otherness are striking examples of this kind of production.

7.5 Transforming forms of unproblematic knowledge: A separate show school

To this point in the chapter, I have discussed the show people's transformations of their marginalising experiences and resistant practices in relation to changing their absence of place into a place of their own and changing the constructions of their otherness into a valuing of their difference. These transformations are important in their own right, as they go directly to the heart and centre of what it means to be a show person and how that meaning is open to dynamic flows as show people imagine themselves otherwise. These transformations are also vital to the show people's educational experiences and opportunities, as this section of the chapter will demonstrate.

In Chapter Five, I focussed on the third key attribute of strategies of marginalisation identified by de Certeau (1984): the forms of seemingly unproblematic knowledge about how and where show people should receive formal schooling. I explained that, while some of the six options previously available to show people for educating their children were better than others, all of them were fundamentally flawed because they derived from a conceptual opposition between the show people's itinerant lifestyle and maximising their children's educational outcomes. I related that fundamental flaw to de

Certeau's (1984) insight that "thepower to provide oneself with one's own place" is the same power that "producesitself in and through" (p. 36) forms of "knowledge" that are neither innocent nor objective, but that are instead actively complicit with perpetuating the power of 'the centre'. That is, implicitly and unquestioningly assuming that 'normal' children remain in one place to receive their schooling was the basis of the show people's educational marginalisation.

In Chapter Six, I demonstrated that the show people have resisted those seemingly unproblematic forms of knowledge about schooling provision by means of two particular resistant practices: their lobbying for the establishment of the special Brisbane School of Distance Education program for their children; and their tactical consumption of that program to refine and reshape it continuously to fit more closely with their distinctive learning needs. I used the show people's words about the program to reveal their very strong commitment to enhancing their children's educational outcomes and their determination to institutionalise the educational innovation. This commitment and this determination derived from the show people's recognition that gains in soliciting understanding of their situation were by no means guaranteed, and reflected de Certeau's (1984) depiction of tactics as ephemeral and transient in character.

I turn now to examine how the show people, animated and encouraged by their success in transforming their absence of place and their constructed otherness, transform the forms of seemingly unproblematic knowledge about their schooling into a particular manifestation of Traveller education that reflects and suits their distinctive circumstances. Their calls for, and their success in obtaining, a separate school of their own constitutes a counternar-

rative that reverses 'the rules of the game' and a new terrain in educational provision. As I have emphasised throughout this thesis, this transformation is the response to generations of neglect and the result of years of energy and labour; it is therefore less teleological and triumphalist than conceived as the possibly temporary outcome of an ongoing struggle about identities and life chances. As de Certeau (1986) noted about the South American "Indians" in their resistance of the Spanish invasion:

...if the survivors' resistance has found political expression, it is because. ..their communities continued to return periodically to the home village, to claim their rights to the land and to maintain, through this collective alliance on a common soil, an anchorage in the particularity of a place. (p. 229)

The fact that "the survivors. . .continued to return periodically to the home village" is a powerful metaphor for the multiple ways in which the show people have continually revisited their own inadequate educational experiences to 'recharge the batteries' of their efforts to create something new: a genuinely Australian version of Traveller education.

From that perspective, the show people's efforts to transform forms of unproblematic knowledge about their schooling varied according to whether they believed that the Brisbane School of Distance Education program could be amended to accord with their developing expectations or that only fundamental structural change would meet those expectations – in short, whether 'evolution' or 'revolution' should be the basis of a transformative Traveller education for them.

•

In looking ahead, the show people sought to identify ways in which Traveller education could fit more and more easily into the spaces of their itinerancy, and consequently ways in which they could continue to turn those spaces into 'places' consumed and occupied by themselves. In doing so, they reflected their continuing awareness of the crucial importance of harnessing the outsidedness and enhancing the creative understanding of educational providers, whose support they would continue to need if they were to succeed in their progressively more ambitious plans. It is in this respect that my interpretation of Traveller education for Queensland show people is not teleological: each 'settlement' is tentative and temporary, and gains can quickly become losses if the show people do not consciously safeguard their terrain.

In this account of the show people's transformation of forms of unproblematic knowledge I move from the 'evolutionary' to the 'revolutionary'. Some respondents concentrated on the structural basis of the program rather than its components. These speakers identified the program's association with the Queensland government as a major restriction on its future expansion. They speculated about the potential benefits of adding a national government dimension to the Traveller education available to the show children. For example, one parent said of her daughter, "Maybe by the time she's ready to go [to school], it will be right throughout Australia, and that will be even better" (Y1P1). Another parent elaborated this perception:

What I thought was, we tend to work basically in the four states of Australia, which would be Queensland, the Northern Territory, Victoria and New South Wales. And basically, because the setup's been done with the [School of] Distance Education in Brisbane, . . . the kiddies still

do their work, [but] they don't do it as far as the program with the schools, the classrooms and so on in New South Wales and Victoria. If it was funded nationally, federally, it would be a much better idea, I believe. If it was possible to get a grant or something. Because it's very hardfor one government to accept the responsibility, basically. (Y2P3)

As the program developed, however, several stakeholders came to the belief that the show people's future educational experiences lay along a very different path: the establishment of a separate school for show children. A nascent indication of this thinking became evident in the first year of data gathering, in the fourth year of the program's operation, when a show person stated:

When I think this program would go in leaps and bounds, what would help it, is to have a teacher who could travel with us. There was a situation for. . .thirty years on the showground, where a policeman travelled the circuit with us and stayed in motels in each town. And he was the p[ublic] r[elations] through the community for the showmen. And it was really great. Now if they had a teacher who could come and travel with us for this length of time. . .because what we found the problem is, every time the children go into the next classroom, that whole first day is taken upjustfinding out for that new teacher where the kids are at. But if there was a teacher that could keep them in line and organised and know each town where they were at, it would be so [good], and they could work one on one every day with a different child. It would be such a bonus to the program; it would be excellent. (Y1P2)

It is important to point out that the speaker was referring to a situation in which one or more teachers travelled full time or permanently with the show circuits, rather than returning to Brisbane after every second show as occurred at the time of her suggestion. At this early stage, this parent still thought in terms of this permanently travelling teacher or teachers as being part of Education Queensland:

...maybe we could fund the money, maybe use certain parts of the money to have a teacher to travel. And as. ..one of the teachers suggested,...a big 'bus would be great that they could stack up the library books and just drive, and if they wanted to build a room in it, [a] self-contained room, they could camp in our area. ...[Which] would cost them no rent or anything; it would cost them nothing once it was initiated, once it was there. And we might use it for three or four months of the year, and perhaps the Education Department could use it in other situations for the rest of the year. (Y1P2)

As the show people's constructions of their future educational experiences developed over time, however, a distinct shift occurred away from this notion of sharing resources with Education Queensland to calls for specialised resources being allocated for the exclusive use of the show people. Another parent articulated the rationale underpinning this shift in mindset:

We tried to ∞ it where they were integrated, but we're a different system altogether. The distance. . .education. . .kids are there, but they're just there. . . .They're out a hundred k[ilometre]s on a property, and they come to town every so often, and they're contactable, and they're there.

Our kids are completely a different system. So we can't take their rules and make them work for us. And we've done it for a few years, and

we've found it's just inhibiting us too much on what we can strive **for** and do. ...And we are specialised, so therefore we should specialise on us. (Y4P5)

This speaker's mobilisation of a discourse centred on being "different" and "specialised" again reflected the process whereby the show people's resistance of forms of unproblematic knowledge about their schooling shaded into transformation of those forms through the construction of the new terrain of Traveller education. That is, the interviewee's 'spealung position' was located in an awareness of the need to subvert "their rules", which they found "justinhibiting us too much", and create new rules of their own. As in the two previous sections of this chapter, part of that transformation was associated with engaging with educational policy makers' outsidedness from, and developing creative understanding of, the show people's situation. Equally crucial to the process, however, was a confidence in establishing new rules and categories if the existing ones were inappropriate or deleterious. This confidence is central to the show people's construction of the counternarrative of Traveller education.

This assertion of a "different" and "specialised" status for the educational needs and aspirations of show children led, seemingly inexorably, to calls for the establishment of a separate school for such children. The same parent who articulated this status elaborated this call: "We're talking about being a separate school. I think that being a separate school only means that we can specialise and concentrate on the needs of our children" (Y4P5). In making this call, she emphasised that the school would not be restricted to children of members of the Showmen's Guild:

...the itinerant children, there are a few [of them], and our system is open to them, even though it will be called "The Showmen's Guild of Austral[as]ia School" – it won't cut out any people. (Y4P5)

Furthermore, another parent explained that, even though she also wanted a special education program for show children:

...I don't mean we can't mix with other kids. We're not saying that. We don't think we should be all in a little thing with the kids. We need to learn on our own because we're different like that, [but we] can still play with all the other kids. They need that too. Still play with all the kids. (Y4P6)

At about the same time, a member of the Showmen's Guild outlined a variation on this call for a government school for the show children:

. . . . in another twenty years' time, when the showmen's education program has its own school, and I'd like to see it one day become a private school in some sense. Not so much a public or a state school, but certainly one that might be based here [at the Guild's headquarters at Yatala, south of Brisbane], down the back or somewhere like that. It might have its own library, and all this type **a** thing. It's a pipe dream, but certainly it's a possibility one day. (Y4A1)

In the event, after the data gathering phase of this research project had been completed, the show people lobbied the Howard Coalition federal government for the establishment of a separate school for show children. They proved adept at using the government's discourse of 'freedom of choice' in relation to education and at capitalising on the government's 'new schools' policy. At the time of writing, a school for show children has been set up with federal government approval and under the auspices of the Queensland state

government. One of the teachers involved with the Brisbane School of Distance Education program has been appointed as principal, and two teams of teachers travel along the circuits teaching the show children full time (Anonymous, 2000; Special Broadcasting Service, 2000).

De Certeau (1984) identified in strategies of marginalisation "aspecific type of knowledge, one sustained and determined by the power to provide oneself with one's own place" (p. 36) and conversely by the power to deny such a knowledge to others. This section of the chapter has traced the show people's determination "toprovide [themselves] with [their] own place", through the establishment of a separate school for their children. That determination has been fuelled by their resistance of the inadequate schooling options previously available to them and by their lobbying for and consumption of the Brisbane School of Distance Education program. That determination has been strengthened also by the show people's success at having their 'spealung positions' on education recognised and valued by others, notably government officials, whose initial outsidedness has been turned into creative understanding of the show people's situation. These several currents of energy and activity have converged in the new school's opening, which marks a 'sea change' in the institution of Australian Traveller education as a counternarrative to the show people's educational marginalisation and resistance.

7.6 Review of the chapter

This chapter has answered the third research question framing this thesis: "How do the show people transform their marginalising experiences and resistant practices?". The posing of that question reflected an assumption

that, in addition to and beyond the show people's resistance of their marginalised situation arising from their itinerancy, their actions have resulted in fundamental changes to 'the rules of the game' governing their previous educational options and have succeeded in creating a new terrain of Australian Traveller education.

Conceptually the chapter was guided by acknowledging that de Certeau (1984) recognised the potential for resistance to lead to transformation of marginalisation, and by deploying Bakhtin's (1986a) notions of outsidedness and creative understanding. Specifically, I applied Bakhtin's explication of the integral links between outsidedness and creative understanding to my analysis of the words of home tutors and Brisbane School of Distance Education teachers in order to demonstrate the extent of the show people's success at enlisting those others in circulating new stories about the show people's lives and education. In doing so, I made the following specific points:

- The show people, through the home tutors and teachers, have transformed their ascribed absence of place into a place of their own and hence into a location of power by extending to those others their multiple experiences and understandings of 'home'.
- The show people, through the home tutors and teachers, have transformed
 the constructions of their otherness into a valuing of their difference by
 communicating to those others the positive valence of the term 'showies'.
- The show people have transformed forms of seemingly unproblematic knowledge about the appropriate forms and places of educating itinerant people by successfully lobbying for a separate school for show children.

This analysis suggests that, although such an outcome was neither inevitable nor permanent, the show people have succeeded in generating a counternarrative to the traditional marginalisation and resistance of their identities and educational opportunities. From this perspective, the show people's transformative actions, 'spaces' that they turn into 'places' through their exploitation of outsidedness and their facilitation of creative understanding, are a fundamental component of their 'learning on the run'.

CHAPTER EIGHT

SOME ANSWERS TO THE QUESTIONS

"Ido not really wish to conclude and sum up, rounding off the argument so as to dump it in a nutshell on the reader. A lot more could be said about any of the topics I have touched upon. . . . I have meant to ask the questions, to break the frame. . . . The point is not a set of answers, but making possible a different practice."

Kappeler, 1986; cited in Lather, 1992, p. 95

"In research the horizon recedes as we advance, and is no nearer at sixty than it was at twenty. As the power of endurance weakens with age, the urgency of the pursuit grows more intense. ...And research is always incomplete."

Mark Pattison, Isaac Casaubon, chap. 10

8.1 The problem addressed and the research questions answered

In the introduction to this thesis, I enunciated the problem to be explored in the study in this way: what does the Queensland show children's educational provision reveal about the intersection of education and marginalisation, resistance and transformation, as well as about broader issues in Australian Traveller education? The intention was to elaborate some of the dimensions and implications of that highly evocative and redolent phrase, "learningon the run".

Furthermore, I stated the three research questions that would frame the conduct of the investigation carried out to address this research problem:

- "How do the show people experience marginalisation?"
- "How do the show people resist their marginalised status?"
- "How do the show people transform their marginalising experiences and resistant practices?"

This study therefore assigned prominence to the three concepts of marginalisation, resistance and transformation in relation to both the Queensland show people and the physical and symbolic spaces of itinerancy. That is, I perceive itinerancy as simultaneously the site **of** undoubted neglect of and discrimination against travelling people and of the possibilities of a more productive and equitable educational provision for those people.

In Chapter Two of the thesis, I surveyed the current literature in two areas: Australian shows; and Traveller education. I identified varying manifestations of ambivalence, marginalisation, resistance and transformation associated with the literature on Australian shows, particularly in terms of the relations between itinerant and local people. I also discerned considerable ambivalence, marginalisation, resistance and transformation in the literature on Traveller education, with the majority of studies continuing to conform to the features of either an 'unproblematic othering' or an 'unproblematic celebration' of itinerancy, with respectively assumptions of itinerancy being a 'deficit' lifestyle and therefore education having a 'remedial' function, or else an excessive emphasis on the exotic fascination of living an itinerant lifestyle. By contrast, there is also a growing body of literature predicated on itinerant people's rights to pursue the lifestyle that they wished and to have equitable access to appropriate educational provision. The existing literature on Australian shows and Traveller education also creates a space for my intended contribution to that literature arising from this intensive study of Queensland show people's experiences of 'learning on the run'.

A significant part of that intended contribution relates to my application to the empirical site outlined above of the conceptual framework articulated in Chapter Three. That framework was centred on the derivation of a mutually engaged and responsive interplay between Michel de Certeau's concept of 'tactics of consumption', whereby show people seek to change the often uncomprehending and sometimes hostile 'spaces' through which they travel into 'places' of their own, and Mikhail Bakhtin's notions of 'outsidedness' and 'creative understanding', whereby the show people's interactions with others assist them in changing 'the rules of the games' that construct them as

disadvantaged. This conceptual framework has the particular value of throwing into stark relief the marginalised, resistant and transformative aspects of 'learning on the run'.

Chapter Four applied 'tactics of consumption', 'outsidedness' and 'creative understanding', and marginalisation, resistance and transformation, to the study's research design. I explained my own ambivalence about the study's potential for replicating, as well as contesting, the show people's disadvantaged situation, and I argued that de Certeau's concept of 'tactics of consumption', and his two binary categories of 'consumption'/'reading' and 'production'/'writing', provided a way of understanding my relations with the show people and a justification for the asserted trustworthiness of my analysis of their interview data. I further asserted that Bakhtin's notions of 'outsidedness' and 'creative understanding' helped to explain the show people's interactions with both myself and the staff members of the Brisbane School of Distance Education. So ambivalence was posited as a potentially productive approach to both the research design's ethical and political dimensions and the data gathering techniques, which consisted of semi-structuredinterviews with forty-two people and an analysis of relevant documents. As I elaborate below, 'learning on the run' is a descriptor that can be applied as much to me as a researcher as it can to the Queensland show people.

I provided answers to the study's three research questions in Chapters Five, Six and Seven. In Chapter Five, I used de Certeau's (1984) notion of 'strategies of marginalisation' to identify and critique the show people's marginalising experiences. Specifically, the chapter examined three key attributes of strategies of marginalisation directed against the show people:

their absence of place; the construction of their otherness in relation to the settled community; and forms of unproblematic knowledge about the 'proper' location and provision of schooling. In combination, these three elements of marginalisation provided the impetus for the show people's resistance and transformation of their marginalised situation, and is therefore integral to their 'learning on the run'.

I used Chapter Six to explore the show people's ongoing efforts to resist their marginalised status. I identified their resistant practices by means of applying de Certeau's (1984) concept of 'tactics of consumption', aided by his notion of 'space'. Specifically, I argued that the show people resist their ascribed absence of place through their multiple experiences and understandings of 'home'. Furthermore, they resist their constructed otherness by giving the term 'showies' a positive valence to counteract the negative stereotypes ascribed to it by others, disrupting the 'showie'—'non-showie' dichotomy and giving the term 'mugs' a negative valence as a tactic of reversal. They also resist forms of seemingly unproblematic knowledge about their schooling through their lobbying for, and consumption of, a specialised educational program operated by the Brisbane School of Distance Education. The show people's resistant practices are therefore as integral as the strategies of marginalisation identified in Chapter Five to their 'learning on the run'.

In Chapter Seven, I analysed the show people's efforts to move beyond the resistance of marginalisation to the transformation of the spaces of their itinerancy. I applied Bakhtin's (1986a) notions of 'outsidedness' and 'creative understanding' to explain how the show people have succeeded in making fundamental changes to 'the rules of the game' governing their

previous educational options and have succeeded in creating a new terrain of Australian Traveller education. Specifically, I analysed the words of home tutors and teachers from the Brisbane School of Distance Education to demonstrate how the show people have transformed their ascribed absence of place into a place of their own and hence into a location of power by extending to those others their multiple experiences and understandings of 'home'. They have also transformed the constructions of their otherness into a valuing of their difference by communicating to those others the positive valence of the term 'showies'. Moreover, they have transformed forms of seemingly unproblematic knowledge about the appropriate forms and places of educating itinerant people by successfully lobbying for a separate school for their children. From this perspective, the show people's transformative actions, whereby they turn the 'spaces' of itinerancy into 'places' through their exploitation of outsidedness and their facilitation of creative understanding, are a fundamental component of their 'learning on the run'.

8.2 The personal note revisited

At the end of Chapter One of this thesis, I inserted "A personal note" that outlined some autobiographical information. The intention was to emphasise from the outset my awareness of the usually implicit relationship between a research project and the researcher's subjectivity, and moreover of the potential for that subjectivity to be complicit in replicating the existing disadvantage of a group of 'research subjects' or of otherwise not doing them positive good. I have explored this matter elsewhere in somewhat greater

length (Danaher, 1998b), and I consider below the implications of the issue for possible future directions in Traveller education research. Here I shall focus on my continuing and changing relationship with the research project.

Throughout this thesis, my musings on this issue of potential complicity have constituted a set of antiphons in a minor key to the focus – the verses in major chords – on the show people's marginalisation, resistance and transformation. That is, I have made explicit on occasions my ambivalence about particular matters, such as the ethical and political dimensions of the study's research design. Thus I have drawn attention to the resistant and transformative elements of the spaces of the show people's itinerancy while striving to challenge the marginalising elements of those same spaces. At the same time, some feelings of ambivalence about doing so are occasioned by my awareness that getting the balance between these polarities 'right' is always difficult (Kenny, 1997). So an important part of my own 'learning on the run' in writing this thesis has been a growing understanding that such ambivalence is likely to remain with me about the study, and moreover that such ambivalence can have positive effects.

I feel supported in making this statement by the view expressed by Kenway and Willis with Blackmore and Rennie (1998) – to which, as I noted in Chapter Four, I would add the descriptor 'self-reflective':

We know that our preferences **for** action and **our** knowledge are situated, partial and interested. They arise **from** our biographies, our different theoretical, institutional, geographic and time locations. (p. Xii)

In other words, my personal note at the beginning of the thesis, this reflection on that note at the end of the thesis and my musings on my ambivalence about conducting this study are all 'grist to the mill' of making researchers' subjectivities more, not less, visible in investigating Traveller education. This kind of approach helps readers to evaluate the 'truth claims' made in studies such as this, and also keeps 'on the agenda' issues such as researchers' roles in replicating or contesting the marginalisation of itinerant people.

As I stated in Chapter Four, I take heart also from the following observation of Trinh **T.** Minh-ha (1990):

Inevitably, a work is always a form of tangible closure. But closures need not close off; they can be doors opening onto other closures and functioning as ongoing passages to an elsewhere (-within-here)...

What needs to be reconsidered are these widely adopted and imposed forms of closure whose main function is simply to wrap up a product and facilitate consumption. They create neither a space of serenity nor of fecundity for the mind and body to rest and grow; rather, they naturalize the zone of conformity, where freedom consists of filling in to one's taste and monetary capacity, the pre-assigned slots. (p. 329)

My desire is that I have written this thesis in ways that encourage 'tactics of consumption' by the reader, and practices of 'outsidedness' and 'creative understanding' with the reader, that will allow temporary and tentative closures rather than an imposed closure designed "towrap up a product and facilitate consumption". Such temporary and tentative closures are far more in keeping with the various conceptual, methodological and empirical lessons that I have learned 'on the run' about the constituent spaces of itinerancy.

8.3 Possible directions for future research

Focussing on the constituent spaces of itinerancy also suggests some possible directions for future research in Traveller education – or ways whereby 'learning on the run' can be applied to the broader field in which I have positioned this study. A useful starting point in this process is Kiddle's (1999) call for a new "terrain" to be developed in Traveller education in England:

Education and pride in themselves is the power that parents and teachers canjointly give to the children—a power to develop self-esteem, a power for choice, a power to defend against exploitation. . . . I have no illusion about the immense difficulties that presents for both parents and teachers, Traveller and non-Traveller, but I do not care to contemplate the alternative future for the children. (p. 156)

Thus for Kiddle (1999) Traveller educational provision and research have an imperative to transform educational opportunities for Travellers whereby they have "power". Kenny's (1997) "vision statement" (p. 296) for Traveller education also focussed on the political dimension:

Traveller children, in collaboration with emancipatory teachers, will acquire the knowledge and skills needed for emancipatory practice. They will learn to clarify who their enemies are, and direct their resistance with increasing accuracy. Teachers and parents will establish reciprocal relationships in schools and in a developing, politically and culturally

dynamic Traveller society. Thus, the patronage which, in situations of internal colonialism informs relations between the dominant and subordinated ethnic groups, will be confronted and replaced with mutually respectful, dialogic relations. (pp. 296-297)

The reference to "mutually respectful, dialogic relations", and the implication that such relations depend on a transformation of the political dimension of Traveller education, resonate with this study's analysis of the 'tactics of consumption', 'outsidedness' and 'creative understanding' manifested in the Queensland show people's marginalisation, resistance and transformation associated with their itinerancy. As I noted in Chapter Four, that reference and that implication also highlight the enduring importance of Traveller educational research having a "transformative" function as enunciated by Anyanwu (1998):

Transformative research is a systematic inquiry into the real conditions which create oppression or hinder self-determination. It produces reflective knowledge which helps people to identify their situation and in doing so, to change such [a] situation for the better. (p. **45**)

From that perspective, the findings of this thesis are intended to have just such a "transformative" function. My analysis of the interplay among marginalisation, resistance and transformation as the show people use 'tactics of consumption', 'outsidedness' and 'creative understanding' in efforts to make the spaces of itinerancy more congenial and meaningful highlighted aspects of their lives that constitute part of "the real conditions which create oppression or hinder self-determination". At the same time, I pointed to conclusions that potentially might assist the show people "toidentify their

situation and in doing so, to change such [a] situation for the better". I do not mean by this that I have communicated a particular opinion about the 'pros' and 'cons' of having a separate school, for example. I mean instead that this study, and future Traveller education research, can and should contribute to a developing body of knowledge about, and understanding and acceptance of, the specialised educational needs and aspirations of Travellers that will potentially create new and more enabling educational opportunities for them.

On a related matter, I indicated in Chapter One that claims are often made about open and distance learning as a possible panacea for addressing the educational needs of different kinds of learners. The significance of this study is that such claims must be counterbalanced by an intensive examination of the learning experiences of specific groups of learners. In the case of the show people, their resistance and transformation of their marginalisation — derived from and fuelled by their agency — enabled them to consume a particular distance education program on their own terms and in their own image. **As** I have said, this is not always the case, and so a healthy scepticism would seem the most appropriate approach to uncritical and unexamined assertions about the empowering capacity of open and distance learning.

As Kenny (1997) has pointed out, the development of knowledge about Traveller education has some parallels with ongoing research into the links between education and gender:

The shift in mindset is comparable to that involved in addressing issues related to gender equity: what was assumed to be the natural order of things was found to be a male construct, and it was not enough to

simply make room within it for women. We need to realise that there is nothing natural about our sedentary life-style and all that it requires, any more than there is about reading from left to right. (p. 296)

This denaturalisation of sedentarism provides a fitting and welcome riposte to McVeigh's (1997) reference to

. . .a host of other less tangible ideas, actions and structures which construct being sedentary **as** the only possible mode of existence within contemporary society. (p. 9)

Such a denaturalisation would have the associated benefit of promoting theoretical understandings of residence as having multiple forms, each equally valid and worthy of value. In this way, the spaces of itinerancy would become transformed into an infinity of places whose inhabitants would constantly move in and out and who would be connected by means of 'outsidedness' and 'creative understanding'.

As for my own more detailed "visionstatement" (Kenny, 1997, p. 296) of Traveller education research, in 1996 I spoke the following words at an international seminar on open and distance learning for Traveller education organised by the European Federation for the Education of the Children of the Occupational Travellers (EFECOT):

...an artistic analogy. ..is useful in helping me to view the education of the children of occupational Travellers as a number of adjoining panels in a very large landscape painting. The foreground depicts children learning in situ, with their teachers changing their conventional roles as the keyfigure in a permanent classroom. The middle ground shows the

excited faces and the bright colours of a fairground ride **or** a circus tent, and the exhausted but contented demeanours of fruit pickers after a hard day's work. The background consists of the flat land and greenfields of western Europe in some panels and the Australian eucalypti stretching to undulating hills on the horizon in others. My colleagues and I hope that this seminar will bring closer to fruition the joining of the panels and the unveiling of the larger painting, so that the constants and the variables of educating the children of occupational Travellers around the world my be revealed. (Danaher, 1996, pp. **47–48)**

This thesis has been conceived **as** contributing to "thejoining of the panels and the unveiling of the larger painting" that depict the complexities and subtleties of itinerancy and Traveller education. The thesis's contribution to that enterprise lies largely in its delineation of some of the physical and symbolic spaces of itinerancy and Traveller education, particularly the marginalising, resistant and transformative dimensions of those spaces. The thesis has demonstrated how the Queensland show people's 'tactics of consumption', 'outsidedness' and 'creative understanding' enable them to resist and transform the marginalising spaces in which they are located, an achievement that reveals the continuing existence of, and the interplay among, marginalisation, agency and ambivalence for Travellers and Traveller education researchers alike. These, among others, are the important lessons of 'learning on the run'.

REFERENCES

Advisory Council for the Education of Romany and Other Travellers (Ed.) (1993). *The education of Gypsy and Traveller children: Action-research and co-ordination*. Hatfield, UK: University of Hertfordshire Press.

Adams, B. & Smith, D. M. (1967). Gipsies and education. In *Children* and their primary schools: A report of the General Advisory Council for Education (Vol. 2: Research and surveys) (pp. 595-600). London, UK: Her Majesty's Stationery Office.

Ahearne, J. (1995). *Michel de Certeau: Interpretation and the other*. Cambridge, UK: Polity Press.

Anonymous (2000, July-September). Show time for travelling school. *The Outdoor Showman*, 417, 6-8.

Anyanwu, C. N. (1998). Transformative research for the promotion of nomadic education in Nigeria. *Journal & Nomadic Studies*, 1,44-51.

Ashworth, A. (1997, April-June). Letters. The Outdoor Showman, 404, 7.

Bakhtin, M. M. (1981). *The dialogic imagination: Four essays* (translated by *C.* Emerson & M. Holquist). Austin, TX: University of Texas Press.

Bakhtin, M. M. (1984). *Problems & Dostoevsky's poetics* (edited and translated by C. Emerson). Minneapolis, MN: University of Minnesota Press.

Bakhtin, M. M. (1986a). Response to a question from the *Novyi Mir* editorial staff. In "*Speech genres*" and other late essays (edited by C. Emerson & M. Holquist, translated by V. W. McGee) (pp. 1-9). Austin, TX: University of Texas Press.

Bakhtin, M. M. (1986b). From notes made in 1970-71. In "Speech genres" and other late essays (edited by C. Emerson & M. Holquist, translated by V. W. McGee) (pp. 132-158). Austin, TX: University of Texas Press.

Balides, C. (1993, Spring). Scenario of exposure in the practice of everyday life: Women in the cinema of attractions. *Screen*, 34(1), 19-37.

Barrett, C. L. & Noble, H. (1973, May). Mother's anxieties versus the effects of long distance move on children. *Journal & Marriage and the Family*, 35, 181-188.

Bauman, Z. (1991). *Modernity and ambivalence*. Cambridge, UK: Polity Press.

Bernaert, Y. (1997, November 16). *General report: EFECOT conference, Blankenberge, 14-16/11/97: In partnership we progress.* Brussels, Belgium: The European Federation for the Education of the Children of the Occupational Travellers.

Binns, D. (1990, August). History and growth of Traveller education. British Journal & Educational Studies, 38(3), 251-258. Birch, I., Lally, M. & Tomlinson, D. (1986, October). *Itinerancy and early school achievement: A pilot study of children in the eastern goldfields region of Western Australia*. Perth, WA: National Centre for Research on Rural Education, University of Western Australia.

Blaber, R. (1989, November). Encounters with the picaresque: Strategies and tactics. *Southern Review*, 22,331-337.

Blair, J. P., Marchant, K. H. & Medway, F. J. (1984, April). Aiding the relocated family and mobile child. *Elementary School Guidance & Counseling*, 18,251-259.

Botke, J. & Willems, S. (1996, September). 03.1: *Educational needs of occupational Travellers*. Brussels, Belgium: The European Federation for the Education of the Children of the Occupational Travellers.

Brett, J. M. (1982). Job transfer and well-being. *Journal of Applied Psychology*, 67(4), 450-463.

Broome, R. with Jackomos, A. (1998). *Sideshow alley*. St Leonards, NSW: Allen & Unwin.

Brown, A. C. & Orthner, D. K. (1990, August). Relocation and personal well-being among early adolescents. *Journal & Early Adolescents*, 10(3), 366-381.

Brown, F. B. & Brown, Y. (1994). Distance education around the world. In B. D. Willis (Ed.), *Distance education: Strategies and tools* (pp. 3-39). Englewood Cliffs, NJ: Educational Technology Publications.

Buchanan, I. (1996a). Heterophenomenology, or de Certeau's theory of space. *Social Semiotics*, *6*(1), 111-132.

Buchanan, I. (1996b). [Review of the book *Michel de Certeau: Interpretation and its other]. Social Semiotics*, 6(1), 147-155.

Buchanan, I. (1996c). How does one read Deleuze? [Review of the book Gilles Deleuze: An apprenticeship in philosophy]. Southern Review, 29(1), 98-101.

Cannon, J. with St Leon, M. (1997). Take a drum and beat it: The story of the astonishing Ashtons 1848-1990s. Kensington, NSW: Tytherleigh Press.

Chambers, I. (1993). Cities without maps. In J. Bird, B. Curtis, T. Putnam, G. Robertson & L. Tickner (Eds.), *Mapping the futures: Local cultures, global change* (pp, 188-198). London, UK: Routledge.

Clifford, J. (1992). Travelling cultures. In L. Grossberg, C. Nelson & P. A. Treichler (Eds.), *Cultural studies* (pp. 96-116). New York: Routledge.

Codey is circus' fifth generation traveller. (1996, June 6). The Gladstone Observer, 1-2.

Conley, T. (1988). Translators' introduction: For a *literary* historiography. In M. De Certeau, *The writing of history* (translated by T. Conley) (pp. vii-xxiv). New York: Columbia University Press.

Dallas, D. (1971). The travelling people. London, UK: Macmillan.

Danaher, G. R. (1995, June). Discipline and punish *and the discursive* production **a** Australian historiography. Unpublished Doctor of Philosophy dissertation, Central Queensland University, Rockhampton, Qld.

Danaher, P. A. (1991). The Coming of the Light: The response of the Darnley Islanders to the London Missionary Society, 1871-1914. Unpublished Master of Letters thesis, University of New England, Armidale, NSW.

Danaher, P. A. (1993). Conceptualising itinerancy: Lessons from an educational program designed for the children of the Showmen's Guild of Australasia. In *Proceedings of the 1993 Postgraduate Student Association Symposium held at the University* of *Central Queensland 1993* (pp. 71-76). Rockhampton, Qld: Postgraduate Student Association, University of Central Queensland.

Danaher, P. A. (1996). Experiences regarding ODL for occupational Travellers in Australia. In *Report: European seminar open and distance learning* for the children of occupational Travellers Blankenberge November 15th and 16th 1996 (pp. 34-59). Brussels, Belgium: European Federation for the Education of the Children of the Occupational Travellers.

Danaher, P. A. (Ed.) (1998a). Beyond the ferris wheel: Educating Queensland show children (Studies in open and distance learning number I). Rockhampton, Qld: Central Queensland University Press.

Danaher, P. A. (1998b). Ethics and researching educational itinerancy. In P. A. Danaher (Ed.), *Beyond theferns wheel: Educating Queensland show children (Studies in open and distance learning number 1)* (pp. 57-69). Rockhampton, Qld: Central Queensland University Press.

Danaher, P. A. (2000a). What's in a name?: The 'marginalisation' of itinerant people. *Journal of Nomadic Studies*, 3, 67-71.

Danaher, P. A. (Ed.) (2000b). *Mapping international diversity in researching Traveller and nomadic education*, special issue of the *International Journal of Educational Research*, 33(3).

Danaher, P. A. (2000c). Guest editor's introduction. In P. A. Danaher (Ed.), *Mapping international diversity in researching Traveller and nomadic education*, special issue of the *International Journal of Educational Research*, 33(3), 221-230.

Danaher, P. A., Hallinan, P. M. & Rose, C. G. (1998). The children's voices. In P. A. Danaher (Ed.), Beyond the ferris wheel: Educating Queensland show children (Studies in open and distance learning number 1) (pp. 131-148). Rockhampton, Qld: Central Queensland University Press.

de Certeau, M. (1984). *The practice of everyday life* (translated by S. Rendall). Berkeley, CA: University of California Press

de Certeau, M. (1986). *Heterologies: Discourse on the other (Theory and history of literature. Vol.* 17) (translated by B. Massumi). Manchester, UK: Manchester University Press.

de Certeau, M. (1988). *The writing of history* (translated by T. Conley). New York: Columbia University Press.

Denzin, N. K. (1994). The arts and politics of interpretation. In N. K. Denzin & Y. S. Lincoln (Eds.), *Handbook of qualitative research* (pp. 500-515). Thousand Oaks, CA: Sage Publications.

Docker, J. (1994). *Postmodernism and popular culture: A cultural history*. Cambridge, UK: Cambridge University Press.

Duffy, E. P. (1986, November). School change and perceptions of the problem by service parents in Western Australia. Unpublished paper.

Duffy, E. P. (1987, August). Itinerant children in the schools: A neglected social issue. *Australian Journal of Social Issues*, 22(3), 542-552.

Dyer, C. (2000). "Education for All" and the Rabaris of Kachchh, Western India. In P. A. Danaher (Ed.), *Mapping international diversity in researching Traveller and nomadic education*, special issue of the *International Journal of Educational Research*, 33(3), 241-251.

Dyer, C. & Choksi, A. (1998). Education for Indian pastoralists: Strategies for provision. *Journal of Nomadic Studies*, 1, 94-103.

Emerson, C. (1984). Editor's preface. In *Problems of Dostoevsky's poetics* (edited and translated by C. Emerson) (pp. xxix-xliii). Minneapolis, MN: University of Minnesota Press.

Emerson, C. (1997). *The first hundred years of Mikhail Bakhtin*. Princeton, NJ: Princeton University Press.

The European Federation for the Education of the Children of the Occupational Travellers (1994). The secondary education of circus and fairground children: Addressing discontinuity of learning through information provision and distance learning: Report on the United Kingdom Project A4(B7). Brussels, Belgium: Author.

Evans, J. W. (1966, April). The effect of pupil mobility upon academic achievement. *The National Elementary Principal*, 45(5), 18-22.

Evans, T. D. (1997). Section 1: Introduction. In L. O. Rowan, V. L. Bartlett & T. D. Evans (Eds.), *Shifting borders: Globalisation, localisation and open and distance education* (pp. 7-9). Geelong, Vic: Deakin University Press.

Evans, T. D. (1998). Foreword. In P. A. Danaher (Ed.), *Beyond theferris* wheel: Educating Queensland show children (Studies in open and distance learning number 1) (pp. xii-xiii). Rockhampton, Qld: Central Queensland University Press.

Evans, T. D. & Jakupec, V. (1996). Research ethics in open and distance education: Context, principles and issues. *Distance Education*, 17(1), 72-94.

Ezewu, E. E. & Tahir, G. (1997). Ecology and education in Nigeria: Studies on the education of migrant fishermen. Kaduna, Nigeria: National Commission for Nomadic Education.

Ferguson, R. (1990). Introduction: Invisible center. In R. Ferguson, M. Gever, T. T. Minh-ha & C. West (Eds.), *Out there: Marginalisation and contemporary cultures* (pp. 9-14). New York, NY and Cambridge, MA: The New Museum of Contemporary **Art** and the Massachusetts Institute of Technology Press.

Fields, B. A. (1995, Winter). Family mobility. Social and academic effects on young adolescents. *Youth Studies Australia*, 14(2), 27-31.

Fields, B. A. (1997, November). The social and educational effects of student mobility: Implications for teachers and guidance officers. *Australian Journal of Guidance & Counselling*, 7(1), 45-56.

Fiske, J. (1989). *Understanding popular culture*. Boston, MA: Unwin Hyman.

Flores, J. L. (Ed.) (1996). *Children of lafrontera: Binational efforts to serve Mexican migrant and immigrant students*. Charleston, **WV:** Clearinghouse on Rural Education and Small Schools.

Foucault, M. (1979). *Discipline and punish* (translated by A. Sheridan). New York: Random House.

Frow, J. (1991). Michel de Certeau and the practice of representation. *Post-colonial literature and advanced literary theory: English Department study guide.* St Lucia, Qld: University of Queensland.

Garbutcheon Singh, M. J. & Miller, H. (1995). Mimicry as a tactic for engaging cultural policy: Australian studies of Asia. *Southern Review*, 28(3), 303-314.

Giddens, A. (1984). *The constitution & society*. Cambridge, UK: Polity Press.

Goodall, P. (1995). *High culture, popular culture: The long debate.* St Leonards, NSW: Allen & Unwin.

Griffiths, M. (1995, April). Making a difference: Feminism, post-modernism and the methodology of educational research. *British Educational Research Journal*, 21(2), 219-235.

Harrington, S. (1987, November/December). Children of the road. *Instructor*, 97(4), 36-39.

Harry, K. (1991, April). Distance education: A survey of the current literature. *British Book News*, 230-235.

Holquist, M. (1981). Introduction. In M. M. Bakhtin, *The dialogic imagination: Four essays* (translated by *C*. Emerson & M. Holquist) (pp. xv-xxxiv). Austin, TX: University of Texas Press.

Jenkins, H. (1992). *Textual poachers: Television fans & participatory culture*. New York: Routledge.

Jones, M. V. (1993). The creation of a prosaics: Morson and Emerson on Mikhail Bakhtin. *Comparative Criticism*, 15,243-260.

Jordan, E. (2000). The exclusionary comprehensive school system: The experience of Showground families in Scotland. In P. A. Danaher (Ed.), *Mapping international diversity in researching Traveller and nomadic education*, special issue of the *International Journal & Educational Research*, 33(3), 253-263.

Kenny, M. (1997). The routes **a** resistance: Travellers and second-level schooling. Aldershot, UK: Ashgate.

Kenway, J. & Willis, S. with Blackmore, J. & Rennie, L. (1998). Answering back: Girls, boys and feminism in schools. London, UK: Routledge.

Kiddle, C. (1981). *What shall we do with the children?* Barnstaple, UK: Spindlewood.

Kiddle, C. (1999). *Traveller children: A voice for themselves*. London, UK: Jessica Kingsley Publishers.

Kiddle, C. (2000). Partnerships depend on power-sharing: An exploration of the relationships between Fairground and Gypsy Traveller parents and their children's teachers in England. In P. A. Danaher (Ed.), *Mapping international diversity in researching Traveller and nomadic education*, special issue of the *International Journal of Educational Research*, 33(3), 265-274.

King-Stoops, J. B. (1980). *Migrant education: Teaching the wandering ones*. Bloomington, **IN:** Phi Delta Kappa Educational Foundation.

Knaepkens, L. (1989). The education of fairground children in the European Community. Brussels, Belgium: The European Federation for the Education of the Children of the Occupational Travellers.

Knaepkens, L., Van Ryckeghem, K. & Verheyen, I. (1993). *Occupational Travellers a right to education today*. . . and after 1993!?: **Report** 2nd Congress **4-5-6** March 1993 De Haan – Belgium. Brussels, Belgium: The European Federation for the Education of the Children of the Occupational Travellers.

Lacey, C. & Blane, D. C. (1979). Geographic mobility and school attainment – the confounding variables. *Educational Research*, 21(3), 200-206.

Lally, M. (1993). Technology and the provision of specialist educational support services to young children of itinerant families. In C. Boylan & M. Alston (Eds.), *Rural education issues: An Australian perspective* (pp. 201-209). Wagga Wagga, NSW: Society for the Provision of Education in Rural Australia.

Lather, P. (1992, Spring). Critical frames in educational research: Feminist and post-structural perspectives. *Theory Into Practice*, *31*(2), 87-98.

Lather, P. (1994, April). Textual strategies and the politics of interpretation in educational research. *Australian Educational Researcher*, 21(1), 41-62.

Lavender, G. B. & Findlay, M. A. (1984). *Australian thesaurus & education descriptors*. Hawthorn, Vic: Australian Council for Educational Research.

Lee, K. W. & Warren, W. G. (1991, August). Alternative education: Lessons from gypsy thought and practice. British Journal & Educational Studies, 39(3), 311-324.

Levine, M. (1966, Spring). Residential change and school adjustment. Community Mental Health Journal, 2(1), 61-69.

Liégeois, J.-P. (1996, November). Declaration by Mary Robinson, President of Ireland. *Interface*, 24, 21.

Liégeois, J.-P. (1998). School provision for ethnic minorities: The Gypsy paradigm (translated by S. ní Shuineár). Paris, France and Hatfield, UK: Gypsy Research Centre, Université René Descartes and University of Hertfordshire Press.

Long, L. H. (1975, Summer). Does migration interfere with children's progress in school? *Sociology of Education*, 48, 369-381.

Lucassen, L., Willems, W. & Cottar, A. (1998). Introduction. In L. Lucassen, W. Willems & A. Cottar (Eds.), *Gypsies and other itinerant groups:* A socio-historical approach (pp. 1-13). Houndmills, UK: Macmillan.

Lynn, J. & Jay, A. (Eds.) (1986). Yes Prime Minister: The diaries of the Right Hon. James Jacker (Vol. 1). London, UK: BBC Books.

Magee, J. (1992, Spring). Survey: The number, age and geographical distribution **a** children **a** show and circusfamilies in the Republic of Ireland. Brussels, Belgium: The European Federation for the Education of the Children of the Occupational Travellers and the Irish Showman's Guild.

Marks, K. & Pullin, R. T. (1996, May). *D8.1: Evaluation strategy and plan*. Brussels, Belgium: The European Federation' for the Education of the Children of the Occupational Travellers.

Martinez, K. (1994). Postcards from the edge. In K. Smith & P. Marland (Eds.). *Knowledge and competence* for beginning teaching: Report of a policy development initiative (pp. 121-140). Brisbane, Qld: Queensland Board of Teacher Registration.

McKay, L. (1996). Michel de Certeau and the ambivalent everyday. *Social Semiotics*, 6(1), 61-81.

McVeigh, R. (1997). Theorising sedentarism: The roots of antinomadism. In T. Acton (Ed.), *Gypsypolitics and Traveller identity* (pp. 7-25). Hatfield, UK: University of Hertfordshire Press.

Miller, Y. F. & Cherry, J. W. (1991). *On the move: Children in transition*. Silver Spring, MD: National Association of School Psychologists.

Mills, G. E. (1985). Transient children in primary schools. *Education*, 34(2), 7-9.

Mills, G. E. (1986a). Coping with the transient child: A school based approach. *Education*, 35(1), 9-12.

Mills, G. E. (1986b). *The transient child: Some case studies*. Unpublished Master of Education thesis, Western Australian Institute of Technology, Perth, WA.

Mills, G. E. (1986c, August). Changing schools again. *Education News*, 19(9), 12-17.

Minh-ha, T. T. (1990). Cotton and iron. In R. Ferguson, M. Gever, T. T. Minh-ha & C. West (Eds.), *Out there: Marginalisation and contemporary cultures* (pp. 327-336). New York, NY and Cambridge, MA: The New Museum of Contemporary Art and the Massachusetts Institute of Technology Press.

Morgan, B. (1995). *The showies: Revelations of Australian outdoor side-showmen*. Mitcham, Vic: Author.

Morris, J. L., Pestaner, M. & Nelson, A. (1967, Summer). Mobility and achievement. *The Journal of Experimental Education*, *35*(4), 74-80.

Morson, G. S. & Emerson, C. (1990). *Mikhail Bakhtin: Creation of a prosaics*. Stanford, CA: Stanford University Press.

Olszewski, P. (1995, June 3). Carnivals with class. *The Australasian Post*, 2-3.

Osborne, L. (1997, July-September). Behind the loudspeakers, toffee apples and fairy floss. *The Outdoor Showman*, 405, 21.

Panagos, J. L., Holmes, R. L., Thurman, R. L., Yard, *G.* J. & Spaner, S. D. (1981, January). OPERATION SAIL: One effective model for the assimilation of new students into a school district. *Urban Education*, *15*(4), 451-468.

Pearce, L. (1994). Reading dialogics. London, UK: Edward Arnold.

Pechey, G. (1997). Bakhtin and the postcolonial condition. In A. Renfrew (Ed.), *Exploiting Bakhtin (Stratchclyde Modem Language Studies, New series, No. 2)* (pp. 29-37). Strathclyde, UK: Department of Modern Languages, University of Strathclyde.

Plowden, B. (1967). *Children and their primary school.* London, UK: Her Majesty's Stationery Office.

Pullin, R. T. (1985). Swings and roundabouts: The education of travellingfairground children – a survey. Sheffield, UK: The University of Sheffield.

Pullin, R. T. (1994). EFECOT Open/distance Learning Consortium: Project A4/2: Essential considerations in devising an optimum distance learning system for the children of occupational Travellers: A report/discussion paper for the meeting of the EFECOT Distance Learning Consortium, Brussels, 18th July 1994. Brussels, Belgium: The European Federation for the Education of the Children of the Occupational Travellers.

Punch, M. (1994). Politics and ethics in qualitative research. In N. K. Denzin & Y. S. Lincoln (Eds.), *Handbook* of *qualitative research* (pp. 83-97). Thousand *Oaks*, CA: Sage Publications.

Queensland Department of Education Northern Region (1992). *Meeting the needs of mobile students in our school community* (4 booklets). Townsville, Old: Townsville and District Education Centre.

Rahmani, Z. (1985, March). Smoothing out the turbulence. *Education News*, 19(2), 39-41.

Ramsland, J. with St Leon, M. (1993). *Children of the circus: The Australian experience*. Springwood, NSW: Butterfly Books.

Reiss, C. (1975). Education of travelling children. London, UK: Macmillan.

Robinson, B. (1999). Open and distance learning in the Gobi Desert: Non-formal education for nomadic women. *Distance Education*, 20(2), 181-204.

Roderick, I. (1995). The politics of Mikhail Bakhtin's carni-phallic body. *Social Semiotics*, 5(1), 119-142.

Rose, P. L. (1990). Revolutionary antisemitism in Germanyfrom Kant to Wagner. Princeton, NJ: Princeton University Press.

Rose, P. L. (1992). Wagner: Race and revolution. London, UK: Faber and Faber.

Rowan, L. O. (1991, September 21). Strategies of marginalisation: An overview. Paper presented at the "Work in Progress" conference, Brisbane, Qld.

Rowan, L. *O.* (1993, February). The powers of the marginal: Some tactics for subverting the centre. *new literature review*, 68-79.

Rowan, L. *O* (1994, January). *Strategies of marginalisation and tactics of subversion: A study of some recent Australian women's writing*. Unpublished doctoral dissertation, University of Central Queensland, Rockhampton, Qld.

Rowan, L. O., Bartlett, V. L. & Danaher, P. A. (1996). Beyond access and equity: Women's place in Australian open learning. In M. Warry, P. O'Brien, J. Knight & C. Swendson (Eds.), Navigating in a sea of change: Second set of working papers of the Higher Education Policy Project (pp. 38-

49). Rockhampton and Brisbane, Qld: Central Queensland University Press and Graduate School of Education, University of Queensland.

Rowan, L. O. & Brennan, M. T. (1998). **A** the need for transformation. In L. O. Rowan & M. T. Brennan (Eds.), *Cultural transformation: Essays in culture and change* (pp. 5-12). Rockhampton, Qld: Central Queensland University Press.

St Leon, M. (2000). Educational practice in Australian circus, 1847-1930. In P. A. Danaher (Ed.), Mapping international diversity in researching Traveller and nomadic education, special issue of the International Journal of Educational Research, 33(3),285-295.

Saunders, P., Clarke, J., Kendall, S., Lee, A., Lee, S. & Matthews, F. (Eds.) (2000). *Gypsies and Travellers in their own words: Words and pictures of travelling life.* Leeds, UK: Leeds Travellers Education Service.

Schaller, J. (1972). Residential change and emotional maladjustment in childhood and adolescence: A review of research. *Goteborg Psychological Reports*, 2(23), 1-11.

Schaller, J. (1975, May). The relation between geographic mobility and school behavior. *Man–Environment Systems*, *5*(3), 185-187.

Schirato, T. (1993, May). My space or yours?: De Certeau, Frow and the meanings of **popular** culture. *Cultural Studies*, 7(2), 282-291.

Schirato, T. & Yell, S. (1996). *Communication and cultural literacy: An introduction*. St Leonards, NSW: Allen & Unwin.

Scholten, U. (2000) Dutch bargee families: Partners in early childhood education. In P. A. Danaher (Ed.), *Mapping international diversity in researching Traveller and nomadic education*, special issue of the *International Journal of Educational Research*, 33(3), 275-283.

Show offers the best of town and country. (1996, June 12). Capricorn Local News, 6.

Shumaker, S. A. & Stokols, D. (1982). Residential mobility as a social issue and research topic. *Journal & Social Issues*, 38(3), 1-19.

Smith, M. B. (1996). Michel de Certeau's microsubversions. *Social Semiotics*, 6(1), 17-26.

Smith, T. S., Husbands, C. T. & Street, D. (1969). Pupil mobility and IQ scores in the urban slum: A policy perspective. In D. Street (Ed.), *Innovation in mass education* (pp. 256-272). New York: John Wiley.

Special Broadcasting Service (2000, October 26). Show kids. [Transcript of television broadcast]. [Url http://203.15.102.140/insight/test/trans cript.php3?date=2000-10-26&title=Show+kids, accessed 27 October 20001

Spivak, G. C. (1990). Explanation and culture: Marginalia. In R. Ferguson, M. Gever, T. T. Minh-ha & C. West (Eds.), *Out there: Marginalisation and contemporary cultures* (pp. 377-393). New York, and Cambridge, MA: The New Museum of Contemporary Art and the Massachusetts Institute of Technology Press.

Stavropoulos, P. (1995, January). Representation and resistance. *Social Alternatives*, *14*(1), 51-53.

Swann, Lord (1985). Education for all: The report of the Committee of Inquiry into the Education of Children from Ethnic Minority Groups. London, UK: Her Majesty's Stationery Office.

Swendson, E. J. (1958, May). Transplanting children. *NEA Journal*, 332-334.

Tahir, G. (Ed.) (1991). Education and pastoralism in Nigeria. Zaria, Nigeria: Ahmadu Bello University Press.

Tahir, G. (Ed.) (1997). *Nomadic education in Africa: Issues and cases*. Kaduna, Nigeria: National Commission for Nomadic Education.

Tahir, G. & Muhammad, N. D. (Eds.) (1998). Readings on distance education for the pastoral nomads of Nigeria. Kaduna, Nigeria: National Commission for Nomadic Education.

Tan, S. K. (1996). Making space for heterologies: De Certeau's links with post-colonial criticisms. *Social Semiotics*, 6(1),27-44.

Umar, A. & Tahir, G. (2000). Researching nomadic education: A Nigerian perspective. In P. A. Danaher (Ed.), *Mapping international diversity in researching Traveller and nomadic education*, special issue of the *International Journal of Educational Research*, 33(3), 231-240.

United Nations Educational, Scientific and Cultural Organisation (1989, October). Meeting of national specialists on the conditions governing the schooling of the children of mobile populations. Final report. Paris, France: United Nations Educational, Scientific and Cultural Organisation.

Weinstein, M. (1994). Robot World: Crossing the borders between official and popular culture. Paper presented at the American Educational Research Association conference, New Orleans, LA.

Welch, I. (1987, June). As many as 100,000 Australian school children move school each year. *Curriculum Development in Australian Schools*, **4**, 43-44.

Whalen, T. E. & Fried, M. A. (1973, December). Geographic mobility and its effect on student achievement. *The Journal of Educational Research*, 67(4), 163-16.

Winning, A. (1990). Homesickness. *Phenomenology and Pedagogy, 8*, 245-256.

APPENDIX A

SEMI-STRUCTURED INTERVIEW SGHEDULE

NOTE: Individual questions were expressed according to whether the respondent was a student, a parent, a home tutor or an itinerant teacher.

1. Background information: name

age

number and ages of siblings or children

length of time on program to date

previous educational experiences

where school work and homework are done

who helps with school work and homework

favourite and least favourite subjects

career aspirations

participation at the show

recreationalactivities

friendships

where and when holidays are taken

where is home?

2. What are three words that you would use to describe living and learning on the show circuit?

- 3. What are three words that you think locals would use to describe living and learning on the show circuit?
- **4.** Which has a greater influence on the type of people show children are travelling on the show circuit or watching television and videos?
- 5. Are there enough jobs on the show circuit for all the show children when they grow up?
- 6. How easily do show people adapt to living in "the outside world"?
- 7. Tell me about the relationships between show people and locals when the show comes to town.
- 8. What are the main benefits and drawbacks of living and being educated on the show circuit?
- 9. How does being educated on the show circuit compare to going to boarding school or doing correspondence lessons?
- 10. Do show children have special learning needs? If so, how well does this program meet those needs (how well does it break down barriers to learning)? Could the program be improved in any way?
- 11. How much control have the show people been able to exercise over the program? Have they had any say in curriculum content or assessment?
- 12. How much do you think that the writers of the program actually know about show people? How much do they need to know?

LEARNING ON THE RUN

- 13. Would this program be better or worse than boarding school or correspondence lessons for high school students on the show circuit?
- 14. Has your opinion of the program and itinerant teachers/show people changed during your participation in the program?
- 15. What other questions should I have asked you about living and learning on the show circuit?